STUTTGARTER BEITRÄGE ZUR PRODUKTIONSFORSCHUNG

Johannes Full

»Techno-Economic Analysis of Hydrogen Production from Biomass with Carbon Capture and Storage«





Universität Stuttgart

STUTTGARTER BEITRÄGE ZUR PRODUKTIONSFORSCHUNG BAND 164

Johannes Full

»Techno-Economic Analysis of Hydrogen Production from Biomass with Carbon

Capture and Storage«

Herausgeber

Univ.-Prof. Dr.-Ing. Thomas Bauernhansl^{1,2} Univ.-Prof. Dr.-Ing. Dipl.-Kfm. Alexander Sauer^{1,3} Univ.-Prof. Dr.-Ing. Kai Peter Birke⁴ Univ.-Prof. Dr.-Ing. Marco Huber^{1,2}

¹ Fraunhofer-Institut für Produktionstechnik und Automatisierung IPA, Stuttgart
 ² Institut für Industrielle Fertigung und Fabrikbetrieb (IFF) der Universität Stuttgart
 ³ Institut für Energieeffizienz in der Produktion (EEP) der Universität Stuttgart
 ⁴ Institut für Photovoltaik (*ipv*) der Universität Stuttgart

Kontaktadresse:

Fraunhofer-Institut für Produktionstechnik und Automatisierung IPA Nobelstr. 12 70569 Stuttgart Telefon 0711 970-1100 info@ipa.fraunhofer.de www.ipa.fraunhofer.de

Bibliographische Information der Deutschen Nationalbibliothek

Die Deutsche Nationalbibliothek verzeichnet diese Publikation in der Deutschen Nationalbibliographie; detaillierte bibliografische Daten sind im Internet über http://dnb.de abrufbar.

Zugl.: Stuttgart, Univ., Diss., 2023

D 93

2024

Druck und Weiterverarbeitung:

Fraunhofer Verlag Mediendienstleistungen, Stuttgart, Jahr des Drucks Für den Druck des Buches wurde chlor- und säurefreies Papier verwendet.



Dieses Werk steht, soweit nicht gesondert gekennzeichnet, unter folgender Creative-Commons-Lizenz: Namensnennung – Nicht kommerziell – Keine Bearbeitungen International 4.0 (CC BY-NC-ND 4.0).

Techno-Economic Analysis of Hydrogen Production from Biomass with Carbon Capture and Storage

Von der Fakultät Energie-, Verfahrens- und Biotechnik der Universität Stuttgart zur Erlangung der Würde eines Doktor-Ingenieurs (Dr.-Ing.) genehmigte Abhandlung

Vorgelegt von

Johannes Simon Full

aus Ravensburg

Hauptberichter: Mitberichter: Univ.-Prof. Dr.-Ing. Dipl.-Kfm. Alexander Sauer Prof. Dr. Iris Lewandowski

Tag der mündlichen Prüfung:23.10.2023

Institut für Energieeffizienz in der Produktion der Universität Stuttgart 2023

Vorwort des Autors

Die vorliegende Dissertation entstand während meiner wissenschaftlichen Tätigkeit am Fraunhofer-Institut für Produktionstechnik und Automatisierung IPA in Stuttgart und am Institut für Energieeffizienz in der Produktion der Universität Stuttgart (EEP). Das anspruchsvolle, innovative und kollegiale Umfeld dieser Institutionen bildeten den idealen Rahmen für diese Arbeit.

Für die Möglichkeit zur Promotion, das Vertrauen in meine Arbeit, die fachliche und persönliche Förderung sowie die freundliche und konstruktive Hilfe gilt mein Dank in erster Linie meinem Doktorvater, Univ.-Prof. Dr.-Ing. Dipl.-Kfm. Alexander Sauer, dem Leiter des Instituts für Energieeffizienz in der Produktion und des Fraunhofer-Instituts für Produktionstechnik und Automatisierung IPA. Besonderer Dank gilt außerdem meinen Wegbegleiterinnen und Wegbegleitern am Fraunhofer IPA. Besonders hervorheben möchte ich Dr.-Ing. Robert Miehe für die lehrreiche Zusammenarbeit, sein Vertrauen und die Förderung meiner persönlichen und fachlichen Entwicklung im industrienahen Forschungsumfeld. Sonja Ziehn, Marcel Geller, Steffen Merseburg und Tobias Schließ für zahlreiche Diskussionen und die Unterstützung bei der Ausarbeitung der dieser Dissertation zugrundeliegenden Publikationen. Steffen Kiemel, Lara Waltersmann, Alexander Emde, Lennard Sielaff und Anne-Kathrin Nuffer für den kollegialen und freundschaftlichen Zusammenhalt.

Bei Frau Prof. Iris Lewandowski, der Leiterin des Fachgebiets Nachwachsende Rohstoffe in der Bioökonomie der Universität Hohenheim, möchte ich mich herzlich für die Übernahme des Korreferats und der wissenschaftlichen Begleitung der Arbeit bedanken. Für die Übernahme des Vorsitzes der Prüfungskommission danke ich Herrn Prof. Robin Ghosh, dem Leiter der Abteilung Bioenergetik des Instituts für Biomaterialien und biomolekulare Systeme der Universität Stuttgart.

Herzlichst bedanke ich mich bei meiner gesamten Familie und meinem privaten Umfeld für die Unterstützung und den bedingungslosen Rückhalt. Insbesondere gilt meine tiefe Dankbarkeit Alexandra und meinen Eltern Martha und Jürgen.

Inhaltsverzeichnis

Nomenclature						
Li	List of Figures 10					
Κι	Kurzfassung 12					
Al	ostrac	ct14				
1	Ir	ntroduction				
	1.1	Initial Situation				
	1.2	Problem and Objective19				
	1.3	Structure of the Work				
	1.4	Philosophy of Science				
2	Т	he HyBECCS Concept				
	2.1	Summary of Paper A: Conceptualization of the technology approach				
	2.2	Summary of Paper B: Exemplary implementation characteristics				
3	A	ssessment Fundamentals				
	3.1	Summary of Paper C: Techno-economic and environmental assessment				
	3.2	Summary of Paper D: Bioenergy and greenhouse gas mitigation potential				
4	E	xemplary Assessment				
	4.1	Summary of Paper E: Exemplary techno-economic assessment				
5	C	ontributions Overview				
6	5 Critical Reflection and Limitations					
7	Summary and Outlook54					

8	Literature	62
Арре	endix A: Paper A	72
Арре	endix B: Paper B	96
Арре	endix C: Paper C	114
Арре	endix D: Paper D	140
Арре	endix E: Paper E	164

Nomenclature

°C	Degrees celsius
BECCS	Bioenergy with carbon capture and storage
Biohydrogen	Hydrogen obtained from biogenic resources
BW	Baden-Wuerttemberg
CapEx	Capital expenditures
CCS	Carbon dioxide capture and storage
CDR	Carbon dioxide removal
CNH	Carbon-negative hydrogen
CO ₂	Carbon dioxide
CRC	Carbon dioxide removal certificates
CS	Carbon dioxide storage
CU	Carbon dioxide use
DACCS	Direct air carbon dioxide capture and storage
DF	Dark fermentation
DIC	Direct and indirect costs
DP	Dark photosynthesis
FRG	Federal republic of Germany
GHG	Greenhouse gas
H ₂	Hydrogen
НуВЕСС	Hydrogen bioenergy with carbon dioxide capture
HyBECCS	Hydrogen bioenergy with carbon dioxide capture and storage
HyBECCU	Hydrogen bioenergy with carbon capture and use
IPCC	Intergovernmental panel on climate change
КРІ	Key performance indicator
LCA	Life cycle assessment
LCCNH	Levelized costs of carbon-negative hydrogen

LCNE	Levelized costs of negative emissions
LCOE	Levelized Cost of Energy
LCSC	Levelized costs of stored carbon dioxide
LHV	Lower Heating Value
NET	Negative emission technology
OpEx	Operational expenditures
R. rubrum	Rhodospirillum rubrum
SR	Steam reforming
ТРС	Total plant costs
TRL	Technology Readiness Level

List of Figures

Figure 1-1	Science systematics (Ulrich 1976, p. 305)22
Figure 1-2	Seven steps of the research process according to (Ulrich 2001, p.195) and connecting points of this work26
Figure 5-1	Structure and connections of the results of the work
Figure 7-1	Summary of the work: Main structure, paper contributions and connecting points to the research questions

Kurzfassung

Auf dem Weg zu klimaneutralen Systemen muss zukünftig neben der Reduktion von Treibhausgasemissionen auch Kohlenstoffdioxid (CO₂) aus der Atmosphäre entfernt werden. Eine vielversprechende Option dafür bieten Technologien, die Bioenergieerzeugung mit CO₂-Abscheidung und -speicherung koppeln (BECCS). In dieser Arbeit wird ein Technologiekonzept für BECCS vorgestellt, bei dem der Energieträger Wasserstoff aus biogenen Rohstoffen produziert wird: Biowasserstofferzeugung mit Kohlenstoffdioxidabscheidung und speicherung (HyBECCS, engl. hydrogen bioenergy with carbon capture and storage). Für das HyBECCS-Konzept bestehen keine technologiespezifischen Möglichkeiten der technoökonomischen Bewertung. Standardisierte Bewertungsgrundlagen sind jedoch notwendig, damit potenzielle Anwender und Investoren bei ihren unternehmerischen Entscheidungsprozessen im Hinblick auf den Technologieeinsatz in der Praxis unterstützt werden können. Auch politische Rahmenbedingungen und Qualifizierungssysteme für öffentliche Subventionen können damit gezielt verbessert werden. Die in der vorliegenden Arbeit entwickelte Lösung stellt methodische Grundlagen für die techno-ökonomische Bewertung von HyBECCS-Technologien zur Verfügung. Dazu gehören einheitliche Systemgrenzen für ökologische und ökonomische Bilanzierungen und darauf aufbauende Kennzahlen. Diese können für Vergleiche mit anderen Wasserstoffproduktions- und Negativemissionstechnologien eingesetzt werden. Eine Besonderheit dabei ist, dass die konventionellen Kennzahlen neben betriebswirtschaftlichen Kriterien auch gesamtwirtschaftliche Auswirkungen integrieren, die aus den Treibhausgasemissionen und reduktionen der zu bewertenden Technologien resultieren. Anhand eines exemplarischen Anschauungsbeispiels wird eine techno-ökonomische Analyse mit Hilfe dieser Kennzahlen durchgeführt. Dabei wird gezeigt, dass die erarbeiteten theoretischen Grundlagen in der Praxis einsetzbar sind.

Abstract

In addition to reducing greenhouse gas emissions, it is inevitable to remove carbon dioxide (CO₂) from the atmosphere to achieve climate-neutrality. Promising options for this purpose are negative emission technologies (NETs) that couple bioenergy production with carbon capture and storage (BECCS). In this paper, a novel technology concept for BECCS is introduced with the peculiarity that the energy carrier hydrogen is produced from biogenic sources: hydrogen bioenergy with carbon capture and storage (HyBECCS). For the HyBECCS concept, no possibilities for technology-specific assessment exist. However, standardized evaluation fundamentals are necessary to support potential users and investors in their entrepreneurial decision-making processes. They are of great importance with regard to the technical implementation of HyBECCS technologies. Additionally, political framework conditions and qualification systems for public subsidies could also be improved in a more targeted way. The solution developed in this doctoral thesis provides a methodological basis for technoeconomic assessment of HyBECCS technologies. It includes uniform system boundaries for ecological and economic balancing and key performance indicators (KPIs). The KPIs can be applied to compare HyBECCS technologies with other hydrogen production technologies and NETs. In addition to conventional economic criteria, these indicators integrate macroeconomic effects resulting from the GHG emissions and reductions as a special feature. On the basis of an exemplary HyBECCS implementation project, a techno-economic analysis is carried out with the help of these KPIs. It is shown that the theoretical principles developed can be applied in practice.

1 Introduction

1.1 Initial Situation

Human-induced climate change and related rise in weather and climate extremes have already led to irreversible impacts on ecosystems and infrastructure. These impacts are related to socioeconomic consequences like reduced food and water security (IPCC 2022a, p. 9). Near-term actions that limit global warming to close to 1.5°C above pre-industrial levels would substantially reduce further damages related to climate change (IPCC 2022a, p. 13).

Without deep reductions in carbon dioxide and other greenhouse gas emissions in the coming decades, global warming of 1.5°C will very likely be exceeded during the 21st century (IPCC 2021a, p. 14). Considering the reduction of GHG emissions, there are feasible and effective mitigation measures. Within energy system transitions, decentralized renewable energy generation is among the options (IPCC 2022a, p. 25). In this area, hydrogen (H₂) can play a key role. Renewable generation possibilities as well as a wide range of applications in different sectors like heavy-duty transport and heat or power supply make H₂ a promising option (Kovac et al. 2021). It can be generated with low GHG emissions, for example, by water electrolysis using renewable electrical energy or from organic residual and waste materials using thermochemical or biotechnological conversion technologies (Turner 2004, Nikolaidis & Poullikkas 2017). Hydrogen obtained from biogenic resources can be referred to as biohydrogen (Bhatia 2014).

In addition to GHG emission reductions, carbon dioxide removal (CDR) from the atmosphere in the order of 100-1000 GtCO₂ over the 21st century will very likely be unavoidable to limit global warming to 1.5°C (IPCC 2018a, p. 17). CDR aims to compensate for residual emissions to reach net zero CO₂ or net zero GHG emissions (IPCC 2021a, p. 29). This is a requirement for stabilizing GHG-induced global temperature increase (IPCC 2021a, p. 30). Carbon dioxide capture and storage (CCS) technologies make this possible, but only if the CO₂ is of biogenic or atmospheric origin. CCS alone does not remove CO₂ from the atmosphere (IPCC 2021b, p. 2218). Biogenic carbon dioxide can be obtained from biomass conversion, i.e. during bioenergy production (BECCS). Atmospheric CO₂ can be captured directly from ambient air using direct air carbon capture and storage (DACCS) approaches. Besides BECCS and DACCS, there are different CDR measures. Among them are afforestation, reforestation, soil carbon management and biochar (IPCC 2021b, p. 242). Known storage options for CO₂ include geological storage in salt rock formations or depleted natural gas and oil fields. Storage of CO₂ in deep, onshore or offshore geological formations has been proven economically feasible under specific conditions for oil and gas fields and saline formations (IPCC 2005, p. 6).

The concept of BECCS rests on the premise that bioenergy production is carbon neutral. This means that as much CO₂ is captured from the air when growing biomass as is released by combustion or conversion. If this biogenic CO₂ is captured and stored, the net effect is a removal of CO₂ from the atmosphere or negative emissions (IPCC 2021b, p. 763). BECCS is constrained by sustainable biomass availability, but it is estimated to have a worldwide potential to remove 1–85 GtCO₂ from the atmosphere in 2050 (IPCC 2018b, p. 342-343). This range can be narrowed to 0.5–5 GtCO₂ per year (Fuss et al. 2018, p. 31). In Europe, the potential for CDR through BECCS is estimated at 200 MtCO₂ per year (Rosa et al. 2021) and for the Federal Republic of Germany (FRD), a potential of approx. 62 MtCO₂ per year is estimated (Borchers et al. 2022, p. 19). As stated in the Climate Change Act of the FRD, the annual greenhouse gas emissions have to be reduced by 88 percent compared to 1990 until 2040 and reach net neutrality in 2045 (FOJ 2021). Looking at the last step towards GHG neutrality from 2040–2045, an annual amount of about 130 MtCO₂eq of remaining GHG emissions would, therefore, need to be able to be reduced or removed from the atmosphere within five years.

In this work, a new concept for BECCS is introduced with the special characteristic that hydrogen is produced. Biohydrogen production technologies generating biogenic CO₂ as a by-product are suitable for CDR and generate negative GHG emissions. The co-generation of biogenic CO₂ applies to most of the known biohydrogen technologies, e.g. photofermentation (Singh et al. 2017, p. 14, p. 106), dark fermentation (Singh et al. 2017, p. 59, p. 212), pyrolysis (Singh et al. 2017, p. 219-220), microbial electrolysis (Singh et al. 2017, p. 215) and biomethane steam reforming (Singh et al. 2017, p. 272).

The HyBECCS negative emission technology concept introduced in this work combines biohydrogen production from renewable biomass, preferential biogenic residual and waste materials, with CCS. A basis for techno-economic assessment of technologies related to this concept is elaborated. The assessment allows the comparison of HyBECCS technologies among themselves and to other hydrogen production technologies or NETs. It is applied to the example of a specific HyBECCS technology implementation project. This is an important step for the understanding of the HyBECCS concept and its impacts and, therefore, supports its potential transfer into broad implementation.

1.2 Problem and Objective

Individual technology combinations for biohydrogen production with carbon capture and storage have already been described. Examples are based on biomass gasification (Susmozas et al. 2016) or biomethane reforming (Antonini et al. 2020), for instance. However, a comprehensive technology concept that generically describes technical implementation options, peculiarities and prerequisites for HyBECCS has not yet been specified.

Further, because there is no description basis, individual HyBECCS technologies are not yet sufficiently comparable and assessable from a techno-economic point of view. Techno-economic assessments should enable the analysis of monetary costs and benefits of a particular technology or process to determine the economic feasibility and to compare different options. For HyBECCS technologies, there are no specific assessment fundamentals available for this purpose. Abstractions, standardizations and uniform system boundaries have not yet been elaborated. However, private and public investments depend on objective techno-economic comparability. Hence, there is a lack of conceptual techno-economic assessment fundamentals and guidelines for HyBECCS technologies that could serve as an important basis for their transfer to wider application.

Appropriate and standardized key performance indicators (KPIs) are necessary to technoeconomically compare individual technologies with competing ones in the market environment. They are inevitable for analyzing and specifically improving the relative economic performance of upcoming and existing technologies. However, to depict the added value for society, it should further be possible to include societal costs in the assessment. This is particularly important in the case of technology concepts that promise additional socioeconomic benefits. As the HyBECCS concept intends to generate such benefits through avoiding economic damage related to climate change, it is particularly important to assess these technologies. Possibilities to integrate the impacts of HyBECCS technologies through GHG emission reductions or removals into techno-economic KPIs do not yet exist. However, this is an essential precondition for understanding the economic impacts holistically. The realization of the socioeconomic potentials of future technologies can be supported by comprehensive and holistic assessment metrics, which serve as a necessary basis for designing policy frameworks and qualification systems for public funding.

Hence, the objective of this work is to establish a basis for techno-economic assessment of HyBECCS processes. This basis should enable a comparison of HyBECCS technologies with each other and with other hydrogen production technologies and NETs in terms of their technoeconomic performance. It thus should serve technology providers and investors as a basis for decision-making. Moreover, it should be possible to consider aspects of climate protection and resulting economic impacts to serve as a guideline for political and societal decisionmaking. For this purpose, technical abstractions and prerequisites, suitable system boundaries and KPIs are intended to be identified or defined in this scientific work.

According to Kubicek, a central aspect of science is formulating research questions. The legitimacy of research questions is based on the knowledge that can be gained by answering them (Kubicek 1977, p. 14-16). Hence, the following central research question and the associated five sub-questions are derived from the objectives described above:

How can HyBECCS technologies be assessed techno-economically?

- 1. What is the HyBECCS concept?
- 2. Which requirements have to be fulfilled to qualify as HyBECCS technologies?
- 3. Which system boundaries are suitable for HyBECCS technologies to balance monetary, mass and greenhouse gas emission flows?
- 4. How can the economic impact of the greenhouse gas balance be incorporated into the techno-economic evaluation of HyBECCS processes?
- 5. Which key performance indicators are suitable for techno-economic assessment of HyBECCS technologies?

The sub-questions build on each other in the sequence shown. The main research question can be answered conclusively from the results of the answers to the sub-questions.

1.3 Structure of the Work

The research conducted as part of this dissertation can be assigned to the research process according to Ulrich, as shown in Figure 1-2 below (Ulrich 2001, p. 195). The seven single steps of this process are linked to the chapters of this thesis, as illustrated. Since this is a cumulative dissertation, essential contents have already been published in the five scientific papers (A-E) listed below.

- Paper A: Full, Johannes; Merseburg, Steffen; Miehe, Robert; Sauer, Alexander, 2021.
 A new perspective for climate change mitigation introducing carbon-negative hydrogen production from biomass with carbon capture and storage (HyBECCS). In: Sustainability 13.7 (2021): 4026. doi:10.3390/su13074026
- Paper B: Full, Johannes; Hohmann, Silja; Ziehn, Sonja; Gamero, Edgar; Schließ, Tobias; Schmid, Hans-Peter; Miehe, Robert; Sauer, Alexander, 2023. Perspectives of Biogas Plants as BECCS Facilities: A Comparative Analysis of Biomethane vs. Biohydrogen Production with Carbon Capture and Storage or Use (CCS/CCU). In: Energies 2023, 16, 5066. doi: 0.3390/en16135066
- Paper C: Full, Johannes; Ziehn, Sonja; Geller, Marcel, Miehe, Robert; Sauer, Alexander, 2022. Carbon-negative hydrogen production: Fundamentals for a techno-economic and environmental assessment of HyBECCS approaches.
 In: GCB Bioenergy 14.5 (2022): 597-619. doi:10.1111/gcbb.12932
- Paper D: Full, Johannes; Trauner, Mathias; Miehe, Robert; Sauer, Alexander, 2021. Carbon-Negative Hydrogen Production (HyBECCS) from Organic Waste Materials in Germany: How to Estimate Bioenergy and Greenhouse Gas Mitigation Potential. In: Energies 14.22 (2021): 7741. doi: 10.3390/en14227741
- Paper E: Full, Johannes; Geller, Marcel; Ziehn, Sonja; Schließ, Tobias; Miehe, Robert; Sauer, Alexander, 2023. Carbon-Negative Hydrogen Production (HyBECCS): An Exemplary Techno-Economic and Environmental Assessment In: International Journal of Hydrogen Energy, doi: 10.1016/j.ijhydene.2023.09.252

The basic chapter structure and the interconnections to the research process are explained in the following. References to the research questions are made in each case, and the contributions made in the respective research publications (A-E) are described briefly.



Figure 1-1 Seven steps of the research process according to (Ulrich 2001, p.195) and connecting points of this work.

The work begins and ends in practice and is mainly focused on investigating of the application context (Ulrich 1981, p.19). In chapter 1, the practice-relevant problems are identified first. This starts with a description of the initial situation and motivation for the research to be conducted. Problems resulting from human-induced climate change and the novel technological HyBECCS approach to support its mitigation are derived. Moreover, the research gap of insufficient assessability for this approach is explained briefly. This is followed by the elaboration of the philosophy of science assigning the work in a scientific-theoretical context to the applied sciences in general, and to technical sciences in particular. A more detailed specification of the research questions to be answered in this work. The main question of how to assess HyBECCS technologies from a techno-economic perspective is thereby divided into five sub-questions. The respective answers lead to a holistic view of the answer to the superordinate question.

In the second chapter, the HyBECCS concept is introduced, and its associated scientific foundations are defined and described from a technological point of view. Following the first step of the research process according to Ulrich, theories and techniques from the fundamental and the formal sciences are identified (Ulrich 2001). These include, in particular, physical and biological knowledge. Significant generic contributions to this part are covered in Paper A (Full et al. 2021a). It summarizes the state of the art in science and technology for biohydrogen production as well as for NETs focusing on BECCS and carbon dioxide sequestration options. Further, the HyBECCS technology concept is defined generically and subdivided into four basic process steps to be implemented with different technology options. It further provides initial estimates of the technological potential for climate protection and considerations of the economic viability of the HyBECCS concept. An additional contribution on specific application peculiarities is made in Paper B (Full et al. 2023a). Herein, the focus is laid on retrofitting existing biogas plants using biogas steam reforming technology. The technical implementation is described and trade-offs for this approach are discussed. The first and second research sub-questions (cf. Section 1.2) are addressed in chapter 2, introducing and describing the HyBECCS technology concept and defining basic requirements for technological implementation options.

In the third chapter, the fundamentals for techno-economic assessment of HyBECCS technologies are elaborated. Referring to the research process, the relevant application context for this purpose is identified first. As summarized in section 1.2, the results are intended to support investors, technology providers and policy makers in their decision-making processes. Assessment criteria, design rules and models are derived and advices for application in practice are elaborated. The main contributions to this part are covered in Paper C (Full et al. 2022b). Results are uniform system boundaries and KPIs enabling comparison of HyBECCS technologies with other hydrogen production technologies and NETs. Additional contributions are made in Paper D (Full et al. 2021b), outlining terminologies for waste classification and giving methodological advice for substrate selection and bioenergy potential estimation for HyBECCS technologies. Research sub-question 2 is intended to be conclusively answered in this chapter by providing an in-depth understanding of the state of the art for the HyBECCS concept and concretizing the verification of the technical requirements defined. Sub-

questions 3, 4 and 5 are addressed, providing basic fundamentals for techno-economic assessment of HyBECCS technologies. These include uniform system boundaries with standardized material, monetary and GHG emission flows as well as KPIs. Impacts on global climate and their economic correlations are particularly considered.

Chapter 4 is a conclusion chapter summarizing the outcomes of the previous chapters and applying the techno-economic assessment fundamentals. Following the research process according to Ulrich (2001), the rules and models are validated in the context of application. An exemplary HyBECCS implementation project is defined for this purpose. Further practice advice can be derived from the validation example by concretizing the procedures to determine and analyze the relevant KPIs on the basis of the practical application example. The contributions to this part are described in Paper E (Full et al. 2023b). Research sub-questions 2, 3, 4 and 5 are considered to be answered conclusively in this chapter through application validation. The system boundaries with standardized material, monetary and GHG emission flows defined in Paper C are applied for this purpose. The KPIs are derived and analyzed. The example technology can be compared with other hydrogen production technologies and NETs considering the economic effects resulting from GHG emissions and removals.

In chapter 5, the interrelationships between the single contributions (Paper A-E) are outlined, and the common thread of the work is described for a better understanding of the overall context of the results.

In chapter 6, all results elaborated in chapters 2-4 are discussed, and the limitations for their application are explained. This contributes to the seventh step in the research process, providing practice advice. It contributes to answering research sub-question 5, especially in terms of applicability constraints.

Chapter 7 summarizes the results of the work and explains its core findings. The significance of the results for scientific advancement and practice application is outlined. With regard to the research process, concluding advice for practice in terms of holistic application possibilities of the findings is provided. This advice, however, does not represent a temporal end point. It must be understood as a constantly recurring phase, as envisaged by Ulrich (Ulrich 1981, p. 21). In the sense of this iterative character of the overall research process, advice for further

scientific development of the results obtained is given. All substantial results are explained in terms of their contributions to answering the five research sub-questions. The main research question defined for this thesis can finally be answered by recapitulating the essential findings and resuming the answers to the sub-questions in a superordinate overall context.

1.4 Philosophy of Science

In the common understanding, science intends to explain states and events of reality by attributing them to the validity of general laws (Ulrich 2001, p. 169). However, it can also be applied to the systematic derivation of recommendations for human action (Ulrich 2001, p. 170). In general, the science process can be defined as systematic thinking that takes place according to specific rules. As there are no common criteria for those rules, they result in a consensus problem. However, this consensus problem can be addressed by science systematics. Science systematics provide classifications into science categories. These allow guidance to examine the rules attached to individual science categories more specifically. Figure 1-1 shows the essential fundamentals of science systematics are explained and related to the present work to understand the objectives and its position in the context of science philosophy.





As illustrated in Figure 1-2, science can generically be subdivided into formal and real sciences. In formal sciences, constructing languages or character systems with rules for their use is the objective. Thereby, no reference to reality exists and the findings can only be checked for their logical truth. Real sciences describe, explain and shape empirically perceivable sections of reality. The findings must be checked for their truth from a logical and a factual point of view. Real science precisely describes new forms of thinking according to certain rules and generalizes them. Hence, they always need formal sciences as a prerequisite. (Ulrich & Hill 1976, p. 305-306)

Among real sciences, a distinction can be made between fundamental and applied sciences. In fundamental sciences, the explanation is their key focus and explanatory models are formed. The applied sciences analyze human action alternatives for the design of social and technical systems. For this purpose, decision models or processes are developed. An applied science must be able to provide appropriate design models and rules for changing social reality in complex social systems (Ulrich 2021, p. 192). Problems of applied research originate outside of science in practice. The problem to be investigated is not the validity of theories but the applicability of models and rules to support human behavior in practice (Ulrich 2001, p. 2020).

Therefore, the central task of the applied sciences is the study of the application context. It expresses itself in practice, since that is where problems arise, and their solutions are realized. Practice can be defined as human action or, more broadly, their behavior and the effects that result from it. The application of scientific knowledge, as it also takes place in this work, deals with human action or behavior and impacts and provides solutions for changing social reality. (Ulrich 1981, p. 7)

The technical sciences can be located between fundamental and applied sciences. They aim at the research and development of artificial artifacts with the goal of testing a practical functionality (Ropohl 2009, p. 161-162). In a broader sense, the artifact and the actual and possible actions associated with it and their forms of organization must be encompassed (Kornwachs 2011, p. 140).

Due to the practical relevance and the main outcome of an applicable basis for technologyspecific assessment of a novel technology concept, the present work can be assigned to the applied sciences. It aims to provide appropriate design models and rules for changing social reality in a complex social system. A generalizing abstraction of the novel technology concept is derived to test and support the practical implication of individual cases within this concept through the provision of a standardized assessment basis. In particular, the work is also linked to technical sciences as the actual and possible actions associated with the technology concept belong to the application context.

2 The HyBECCS Concept

The HyBECCS (hydrogen bioenergy with carbon capture and storage) technology concept combines biohydrogen production from renewable biomass, preferential waste and residues, with carbon dioxide capture and storage. In this chapter, the technological fundamentals are explained, and implementation options are described.

In Paper A, the basic concept is introduced and the state of the art from a technical point of view is explained. The technology concept is divided into four generic process steps, for each of which different technology options can be selected. Technical requirements for the suitability of the options are established and peculiarities are analyzed. The state of the art on biohydrogen production and NETs is elaborated. Additionally, general perspectives for HyBECCS from economic and ecological points of view are outlined.

Paper B emphasizes a technical implementation option of the HyBECCS concept based on modifying existing biogas plants with biogas steam reforming.

2.1 Summary of Paper A: Conceptualization of the technology approach

Paper A (Full et al. 2021a) introduces the basic approach of the HyBECCS concept. It starts with deriving the role of negative emissions and hydrogen in energy transition and climate change mitigation activities. Thereafter, the state of the art in science and technology for hydrogen technologies and NETs is elaborated. A special focus is placed on biohydrogen production technologies and BECCS approaches. It is made clear that the HyBECCS concept can be implemented in various technological ways.

For better understanding, it is divided into four steps: biomass pretreatment (1), biohydrogen production (2), product gas separation (3), and their processing (4). All steps are interdependent and can be performed by different technology options. For the selection of the substrate biomass, it is suggested to use residual and waste materials. However, their suitability depends on the technology choices for the pretreatment and biohydrogen production steps. Considering biohydrogen production, it is a necessary condition that carbon dioxide is produced as a by-product. This applies to most of the biohydrogen production technologies known. CO₂ storage options include geological storage in salt rock formations or depleted natural gas and oil fields, for instance.

As HyBECCS is introduced as a NET concept, the entire HyBECCS process chain, including biomass supply, the use of hydrogen and storage of CO₂, must have a negative GHG balance. This is an essential basic requirement for HyBECCS technologies, which is explained in more detail in chapter 3. If this prerequisite is fulfilled, the hydrogen produced can be referred to as carbon-negative hydrogen (CNH).

In comparison to other NET approaches, the potential advantages of HyBECCS from a technical point of view are derived. Compared to direct air carbon capture and storage (DACCS), the main advantage of HyBECCS and BECCS in general is the energy-efficient capture of CO₂ from a highly concentrated point source. A disadvantage, however, are ecological and social problems that can arise from biomass recovery. Compared to other BECCS approaches that produce hydrocarbon-based energy carriers, the theoretical relative carbon dioxide removal potential of HyBECCS technologies is higher. For example, in biogas or biomethane production

with carbon capture and storage (CCS), a stoichiometric fraction of the carbon from the biomass remains in the energy carrier methane (CH₄). This carbon is usually distributed, e.g. as fuels for transport applications or heating, and emitted as CO₂ during biogas or biomethane combustion. The distribution complicates the capturing of the GHG. In HyBECCS projects, the carbon-free energy carrier H₂ is produced (and distributed), and CO₂ can be captured centrally at the biohydrogen plant site as a point source. Thus, the main technical advantage of HyBECCS is the thermodynamically efficient capture of a large share of CO₂ from biomass.

Further, the role of HyBECCS in future energy systems is analyzed. Key societal and technological drivers are discussed. At the example of the FRD, ecological impacts on GHG emissions for Germany are estimated briefly. The determination is based on the example of dark fermentation (DF) as the biohydrogen production technology choice. A theoretically possible annual production capacity of hydrogen in the amount of about 9.9–19.9 TWh/a is therefore adopted from Weide et al. (2020, p. 6) for the reference year of 2030. This assumption is based on estimates for theoretically available biogenic residual materials in Germany derived from Brosowski et al. (2016, p. 266-268) and shows how high the theoretical H₂ potential through DP would be if all available waste streams were used. The average value for the theoretical hydrogen yield per ton of residual materials is therefore assumed to be 90 - 180 litres of hydrogen per kilogram of organic dry matter according to Weide et al. (2019, p. 24114-24117). Based on this data, an initial estimation of the theoretical maximum emission reduction potential through substituting natural gas could be made, resulting in an amount of 2.0–4.0 Mio. tCO_2eq/a . However, alternative uses and competition for biomass are not taken into account in this analysis. Since all types of biomass are limited and can have multiple end uses, as analyzed by the European Environment Agency (EEA 2023, pp. 19-21), for example, this leads to resource conflicts and trade-offs. These are not taken into account in this example, thus these results can only be considered and classified with these limitations in mind, as explained further in chapter 6. Refinement of the theoretical potential to a sustainable potential through further work is strongly recommended by the author of this thesis. In addition, GHG emissions from biomass production, transport and pretreatment, the biohydrogen production process and hydrogen distribution and usage, as well as for the carbon dioxide capturing, transport and storage are neglected. Based on stoichiometric conditions, the same amount of biogenic CO₂ (in moles) as hydrogen produced is assumed to be generated through dark fermentation. This number leads to the potential estimation for capturable and storable biogenic CO₂. It also represents the maximum possible potential for negative GHG emissions, which is determined at 6.4-13 Mio. tCO₂/a. The total ecological potential in terms of GHG emissions sums up to 8.49 - 17.06 Mio. tCO₂eq/a of maximum possible CO₂ reduction and capture potential. Additionally, the maximum production costs for CNH to be competitive in Germany are estimated, considering hydrogen market forecasts. They would have to be below 4.30 EUR per kg H₂ in a worst-case scenario and below 10.44 EUR per kg H₂ in a best-case scenario.

In conclusion, an outlook from a technical point of view discusses a concept for biorefineries based on the HyBECCS concept. The main gas fractions of HyBECCS technologies, H_2 and CO_2 , could therefore be used in a combinatorial way. For example, using H_2 as a reducing agent for CO_2 could serve as a basis for the production of hydrocarbons.

2.2 Summary of Paper B: Exemplary implementation characteristics

Paper B (Full et al. 2023a) is a supplementary contribution describing and analyzing a specific technological implementation approach for the HyBECCS concept, retrofitting existing biogas plants: Biohydrogen production via steam reforming of biogas with carbon dioxide capture and storage (HyBECCS process, process option 2). This option is compared to biomethane production from biogas with carbon dioxide capture and storage (process option 1). Technical, economic, and environmental aspects are considered. The aim is to provide insights into the technical configuration of the HyBECCS process and the tradeoffs that must be considered when retrofitting biogas plants.

First, the technical implementation particularities of the options are described. The focus is laid on the biogas steam reforming process step. Relevant reaction equations and a flow diagram for producing biohydrogen and biogenic carbon dioxide are provided and discussed. The exemplary comparisons for the two process options are carried out based on these technological descriptions.

Considering the energy efficiency, the results indicate that the HyBECCS approach in process option 2 leads to an energy loss of approximately 37% compared to process option 1. This is estimated in terms of the absolute lower heating value (LHV) of the products hydrogen and methane per kg of biomass or biogas input. Expanding the observation system to compare usable driving energy for heavy-duty transport applications, using H₂ produced in process option 2 still results in a comparable energy loss of approximately 4.2%.

To determine the theoretical negative emission potential, the maximum amount of storable biogenic CO_2 was calculated for both process options. In comparison, process option 1 results in a maximum amount of 1.65 kg CO_2 per kg of CH_4 produced, while the HyBECCS approach (process option 2) shows a maximum of 4.4 kg CO_2 for the same quantity of biomass input. This corresponds to about 2.7 times the amount of storable biogenic CO_2 , assuming an average composition of the biogas of 37.5 vol% CO_2 and 62.5 vol%.

In the context of this dissertation, the results of Paper B serve to understand technical aspects of the HyBECCS concept better using the example of retrofitting biogas plants with steam

reforming technology aiming to provide an application-oriented basis of understanding for the assessment fundamentals described in the following chapter.
3 Assessment Fundamentals

Fundamentals for techno-economic assessment of HyBECCS processes are elaborated in this chapter. Mathematically verifiable prerequisites for the qualification as HyBECCS technologies, uniform system boundaries, key performance indicators (KPIs) and guidelines for GHG emission reduction, GHG removal and bioenergy potential estimations are the main outcomes of this part.

In Paper C, the main contributions regarding techno-economic assessment fundamentals for HyBECCS technologies are made. A transferable definition of system boundaries is developed with associated material, emission and monetary flows. Economic influences and interrelationships between subsystems are specified, and suitable techno-economic KPIs are derived. These include, for example, the levelized costs of carbon-negative hydrogen (LCCNH) and levelized costs of negative emissions (LCNE). A particular characteristic of these indicators is the possibility to include the environmental impact costs due to GHG emissions and removals. Climate change related damage and damage prevention effects can thus be considered in the techno-economic analysis.

Additionally, a methodological approach to support the evaluation of bioenergy potentials for HyBECCS technologies is described in Paper D. A four-step methodical approach dealing with available waste materials, uncertainties of early-stage processes, and calculation aspects provides information on the estimation of usable energy.

3.1 Summary of Paper C: Techno-economic and environmental assessment

Paper C (Full et al. 2022) provides fundamentals for the techno-economic assessment of HyBECCS technologies. It first summarizes and concretizes the characteristics of the HyBECCS concept outlined in Paper A. For this purpose, the state of the art for CCS, NET, CDR and BECCS, to which the HyBECCS concept can be assigned, is presented. Further, an overview of the regulatory status quo of these concepts is given to understand the techno-economic peculiarities of HyBECCS better.

The main outcome of the paper is to provide assessment fundamentals for HyBECCS technologies. Therefore, uniform system boundaries are elaborated first. An overall system boundary, B3, and two subsystem boundaries, B1 and B2, are defined. System elements are assigned to each of the two subsystem boundaries. For B1, the production of biohydrogen and capturing of CO₂ (HyBECC) and for B2, the transport and storage of CO₂ (CS) are determined. In contrast to the four process steps distinguished on a technical basis in Paper A, the HyBECCS process is subdivided with regard to its ecological and economic balancing. As the two system elements, HyBECC and CS, are likely to be spatially and economically separated, they are considered individually. Both system elements combine to the overall HyBECCS technology. Further, mass flows, monetary flows and GHG emission flows are defined for each of the system boundaries. Connections and interrelations between the flows are described in detail. Mass flows can be linked to monetary flows and also to GHG emission flows in the case of CO₂, for instance. GHG emission flows can also be linked to monetary flows. This occurs in the case of economic systems that provide for internalizing the environmental impacts resulting from the GHG emissions.

With this in mind, interrelationships between ecology and economics are discussed. Through internalization mechanisms, the HyBECCS concepts could benefit economically in two different ways. On the one hand, low process GHG emissions throughout the entire HyBECCS value chain would result in a relative cost reduction compared to other processes with higher process emissions. On the other hand, (net) negative emissions would be a source of revenues through internalization approaches like carbon removal certificates (CRC). Accordingly,

negative emissions can be considered a product of HyBECCS technologies alongside hydrogen. A simplified basic model for regulatory systems in net-zero emission economies is proposed to translate positive and negative GHG emissions into costs and revenues. This model is based on the assumption that the external costs and benefits for the climate are entirely internalized.

As the next step, requirements for HyBECCS technologies are detailed based on the system boundaries and associated flows. As a first prerequisite, the overall GHG emission flow of system boundary B3 must be negative. To ensure this, all GHG emission flows must be calculated following the ISO standards 14040/44. Considering monetary flows, the overall profit must be positive (P>0) for the HyBECCS technology to be economically viable.

To finally assess HyBECCS technologies, eligibility checks are elaborated at first. These could lead to the exclusion of technologies that cannot meet the requirements for HyBECCS. To check the theoretical ability on a technological level, permanent removal of CO₂ must be ensured. It is recommended to review experience values from literature as they are the only available reference points. Further research is proposed, for example, to create a positive list of appropriate CS technologies. By ensuring negative emissions through calculating the total GHG emission flow in B3, eligibility can be checked on a project level. Eligibility checks on both project and technology levels are recommended in the planning phase of HyBECCS technologies (ex-ante). After implementation, they must be validated in reality (ex-post).

Specific KPIs for the assessment of HyBECCS technologies are elaborated further. They are also based on the system boundaries and flows determined. First, the GHG removal efficiency is examined by introducing the emission factor (EF). The EF (in tCO₂eq/tCO₂stored) constitutes the ratio between process GHG emissions emitted within the overall system boundary B3 and the amount of non-fossil CO₂ physically stored. It can be used to quantify the GHG removal efficiency of NETs. For HyBECCS, the EF has to be smaller than one, which is always given if the total emission flow in B3 is negative according to the prerequisite defined. An EF of less than 1 in the case of HyBECCS or BECCS generally means that more biogenic CO₂ is stored than GHG emissions are caused by its extraction and storage. The closer the EF is to zero, the better the GHG removal efficiency of the NET.

To determine techno-economic KPIs, capital expenditures (CapEx), operational expenditures (OpEx), and revenue flows must be obtained for all subsystems. To calculate the CapEx, a decision tree is provided to choose a suitable cost estimation method. The OpEx can be estimated based on the planning status or on experience values. For the internalization costs and revenues, the amounts of the different GHG emission flows are required. They can be adopted using standardized databases for life cycle assessments according to ISO standards 14040/44 and literature values for leakages. Stoichiometric estimation approaches or literature values can be used for estimating biogenic CO₂ production. The translation of positive and negative GHG emission flows into internalization cost and revenue flows can be carried out using the defined basic model for regulatory systems in net-zero emission economies. For this purpose, prices can be assumed that reflect the total societal costs due to GHG emissions.

Both internal and internalized costs and revenues are included in the calculation of two central techno-economic KPIs: The levelized costs of carbon-negative hydrogen (LCCNH) and the levelized costs of negative emissions (LCNE). The costs for the production of the core products (hydrogen and negative emissions) are considered under the assumption of an ideal neutralization of the GHG emissions. Those KPIs allow the comparison of different HyBECCS approaches with other hydrogen production technologies and NETs in terms of their economic competitiveness and environmental performance in terms of GHG emissions. The key points of the KPIs are summarized as follows:

- The LCCNH (in EUR/kWhH₂) constitute the costs incurred to produce 1 kWh (net calorific value) hydrogen with net-negative emissions in a climate-neutralized way.
- The LCNE (in EUR/tCO₂removed) are the costs that occur to achieve one ton of negative CO₂ emissions, i.e. to remove one ton of CO₂ from the atmosphere, in a climate-neutralized way.

Additionally, the levelized costs of stored carbon (LCSC) are defined as follows.

• The LCSC (in EUR/tCO₂stored) are defined as the costs that occur to permanently store one ton of CO₂ with a climate-neutralized process.

Climate-neutralized means, in all cases, that the entire external economic damage or damage avoidance through GHG emissions and removals is internalized according to the basic model for regulatory systems in net-zero emission economies defined.

3.2 Summary of Paper D: Bioenergy and greenhouse gas mitigation potential

In the supplementary contribution Paper D (Full et al. 2021), guidelines to support the determination of bioenergy potential of HyBECCS technologies are presented. It provides information on the amount of usable energy from waste and residue materials after conversion. For this purpose, a four-step methodical conduct is elaborated dealing with available waste materials, uncertainties of early-stage processes, and calculation aspects. Only material flows that do not compete with established, high-value utilization routes such as animal feed are considered. The work focuses on the utilization of waste streams in order to maximize ecology and value creation. As a consequence, particular biomass flows theoretically utilizable for biohydrogen production are discarded, as more sustainable or higher-value utilizations are possible.

At first, the state of the art regarding the legal framework for renewable energy as well as biomass utilization and disposal in Germany is outlined. Moreover, common treatment processes for biomass are presented, briefly touching upon their advantages and disadvantages. The state-of-the-art for biohydrogen production and the HyBECCS technology concept (cf. Paper A) are concretized with regard to the use of the various residues and waste materials. Furthermore, the essential basics of bioenergy potentials are explained. The different levels of utilization pathways, corresponding potential terms and system boundaries for bioenergy potential derivation are summarized.

The four-step methodological guideline elaborated in the main part is suitable to support the determination of the bioenergy potential and negative emission potential of HyBECCS technologies. The first step elaborates a procedure to select relevant biomass categories for hydrogen production. This procedure is divided into six sub-steps and applied through an exemplary biotechnological process. The selected waste types can be translated into technical feedstock potentials using values from further literature. Secondly, suggestions are made for making assumptions and defining framework conditions. Regarding biomass pretreatment, suitable methods or method combinations for the selected waste types are identified, and estimation approaches for their respective energy demands are provided. Further,

suggestions for process energy demands are made, and peculiarities for logistics, substrate storage, and handling of residues and produced gases are discussed. In the third step, the equation for determining the bioenergy potential of HyBECCS processes is derived starting from the fuel potential of hydrogen and considering the energy efficiencies of the conversion steps. The energy demands for substrate pretreatments, as elaborated in the work, and for process operation are included. Both reduce the overall potential bioenergy output. The resulting bioenergy potential represents the utilizable amount of energy after conversion, reduced by the amount of energy consumed throughout the value chain. Finally, potentials for climate change mitigation are discussed in step four. A calculation equation to estimate the maximum potential for GHG reduction through substituting fossil energy carriers is derived. Subsequently, the calculation of the theoretical amount of storable biogenic carbon dioxide representing the theoretically achievable negative emission potential is outlined. Therefore, the product gas flow in the respective biohydrogen process and its CO₂ content must be known or estimated. However, process GHG emissions, leakages and risk deductions must be considered from this maximum potential, as described in detail in Paper C.

Concluding, in Paper D, guidelines for estimating bioenergy potentials specifically tailored for biohydrogen production and HyBECCS processes in the German energy and waste system are elaborated. It helps to establish a generally accepted method to determine bioenergy potentials to make their quantification more accessible and consistent. It can also support technology selection based on the suitability of different substrates for HyBECCS approaches, taking into account the available biomass potential in the respective area.

4 Exemplary Assessment

In this chapter, the assessment fundamentals elaborated in Paper C are summarized and applied by means of an exemplary HyBECCS implementation project. The key performance indicators EF, LCCNH, LCNE and LCSC are determined to compare the HyBECCS process with competing hydrogen production technologies and NETs. The main objective of this chapter is to validate the applicability of the techno-economic assessment fundamentals in practice.

4.1 Summary of Paper E: Exemplary techno-economic assessment

In Paper E (Full et al. 2023b), a techno-economic and ecological assessment is carried out at the example of a theoretical HyBECCS implementation project. The example process created for this purpose combines the microbiological biohydrogen production process "dark photosynthesis" (DP) with the capture and long-term storage of CO₂ in saline formations. The substrate for DP in the example process is fructose-containing waste from fruit processing. The assessment is based on the methodological fundamentals described in Paper C and was conducted primarily to validate them.

At first, the example project is described and the assumptions and framework conditions for the assessment are outlined. These involve the technical plant designs for the process steps, peculiarities for the life cycle assessment (LCA), product yields, economic parameters and surcharge factors for cost estimations. Further, the methodology is described. It strictly follows the guidelines defined in Paper C.

As a first assessment step, eligibility on the technology level is checked for the example project. Waste or residues from fruit processing is assumed as substrate. As fruits are a renewable, fast-growing biomass, the CO₂ in the substrate can be considered to be of biogenic origin. The theoretical permanence of CO₂ storage is given due to the technology choice made for CS. CO₂ storage in saline formations can be considered to be permanent for over 1000 years in the case of a careful site selection (IPCC 2005). Thus, the projects' eligibility on a technology level is given.

The system boundaries described in Paper C are applied to the example project and are used for the calculation of the GHG emission flows. The results of their balancing (carbon footprint) show that the exemplary HyBECCS project would be capable of generating negative GHG emissions under the assumptions made. It would remove 17.72 kgCO₂ from the atmosphere for every kilogram of H₂ produced. Thus, the analyzed exemplary project qualifies to produce carbon-negative hydrogen (CNH) and as a HyBECCS and NET approach. Eligibility on a project level is given. The main share of 74.7 percent of the process GHG emissions within the HyBECC system element is due to the provision of heat for hygienization. Electricity from the German grid accounts for another 15.66 percent. Emissions from the CS system element contribute for a 9.38 percent share. The construction and CO₂ transport remain at 0.24 percent and 0.01 percent, respectively.

In the main section, the KPIs are analyzed and discussed. Compared to other NETs, the GHG removal efficiency represented by the EF (cf. Paper C) of the exemplary HyBECCS project (0.12 tCO₂eq/tCO₂stored) is at the lower end of the EF range of both other BECCS and of DACCS approaches. In relation to the resulting process GHG emissions, this indicates a high CO₂ capture potential of the considered HyBECCS example project.

CapEx estimations result in a total of 9,871,401 EUR for the example project. They can be split into 55 percent on the system element HyBECC and 45 percent on the system element CS. Total plant costs (TPC) and direct and indirect costs (DIC) of system element HyBECC account for 28 percent and 27 percent of the total CapEx respectively. The substrate management has the largest share of the TPC with 43 percent, and third largest share of total CapEx with 12 percent. Total OpEx sum up to 2,188,964 EUR per year. Heat accounts for the largest share of OpEx with 32 percent, followed by personal expenses with 22 percent, process water with 14 percent and CS (incl. transport and monitoring) with 12 percent.

The LCCNH were calculated, based on the assumptions made, to be 0.013 EUR/kWhH₂. Compared to grey hydrogen from natural gas (0.12 EUR/kWhH₂) and green hydrogen from electrolysis using renewable electricity (0.18 EUR/kWhH₂), this indicator points to a clear advantage of the HyBECCS project. Even without the assumed internalization of GHG impacts, the calculations of internal LCOE indicate an economic advantage of the exemplary project (0.12 EUR/kWhH₂) over green hydrogen from electrolysis (0.17 EUR/kWhH₂).

The LCSC of the exemplary project was calculated at -260 EUR/tCO₂stored and the LCNE at -297 EUR/tCO₂removed. This means that the exemplary project generates an income of 297 EUR per ton of CO₂ removed from the atmosphere due to the sales of the co-product biohydrogen. For BECCS, literature shows a LCNE cost range of approx. 39-367 EUR/tCO₂removed and for DACCS of approx. 25-1,127 EUR/tCO₂removed. This indicates a high level of economic viability and competitiveness of the exemplary HyBECCS project with other NETs. A sensitivity analysis is performed to identify the parameters that affect the results most significantly. It shows that the GHG internalization costs and revenues, H₂ and CO₂ yield and the total OpEx have the greatest impact on the LCCNH. The H₂ and CO₂ yields have the greatest effect on the LCCNH, when being reduced. A reduction of the yields causes the LCCNH to converge to infinity. Increasing the total OpEx and decreasing the GHG internalization costs and revenues also have a comparatively high impact driving up the LCCNH. The highest value for LCCNH was calculated at 0.1417 EUR/kWh for reducing the H₂ and CO₂ yield by 50 percent. The lowest value for LCCNH was calculated at -0.0384 EUR/kWh for increasing the GHG internalization costs and revenues by 50 percent. The influences of the parameters on the LCNE show that the H₂ sales price has the greatest impact. While a 50 percent reduction would lead to a cost increase of 91 percent to -25.89 EUR/tCO₂removed, a 50 percent increase would reduce the LCNE by 91 percent to -567.48 EUR/tCO₂removed. Similar to the influences on LCCNH, the H₂ and CO₂ yield and the OpEx have a considerably high impact on the LCNE.

In conclusion, the HyBECCS technology could be evaluated and compared with other hydrogen production methods and NETs. In addition to classical techno-economic evaluation approaches, the economic consequences due to the impact on the global climate were taken into account. Therefore, the validation of the assessment fundamentals at the exemplary project was successful.

5 Contributions Overview

The publications (Paper A-E) on which this work is based are connected, as illustrated in Figure 5-1. The core results developed within those publications are presented in the chapters 2-4. As described in section 1.3, two publications contribute to the HyBECCS concept (chapter 2) and the assessment fundamentals (chapter 3), respectively. In each case, one of the papers constitutes the main contribution and the other one provides additional findings, as depicted in Figure 1-2. The exemplary assessment (chapter 4) is covered by a single publication (Paper E). The connections between the chapters and contributions are discussed in the following.





Chapter 2 mainly provides for a better understanding of the technical basis, characteristics and prerequisites of the HyBECCS concept. The work starts with the introduction and conceptualization of the approach. It comprises a modular-based combinatory system. This consists of four sub-processes with respective technology options, as outlined in Paper A. Additional information is given in Paper B considering opportunities for biogas plant retrofits as a specific, application-oriented implementation area.

The fundamentals for techno-economic assessment are elaborated in chapter 3. With regard to the creation of standardized techno-economic assessment guidelines and KPIs, the HyBECCS concept is further abstracted. The most important contributions to this are derived in paper C. Essential outcomes of this work are uniform system boundaries with respective monetary, mass and GHG emission flows. Based on this framework, the key performance indicators EF, LCCNH, LCNE and LCSC are defined. They serve as a benchmark to compare price scenarios and cost forecasts. The LCCNH can be used to compare HyBECCS technologies to other hydrogen production technologies. The LCNE and LCSC serve to compare NETs in terms of their cost efficiencies. Paper D supplements the assessment fundamentals with a special look at biomass residues and waste materials to be utilized by HyBECCS technologies. Basic instructions for selecting suitable waste materials and estimating the respective bioenergy potentials are laid out. It raises awareness of the high importance of biomass substrate selection for HyBECCS technologies.

The exemplary techno-economic assessment conducted in chapter 4 focuses on applying the results derived in Paper C. The work carried out in Paper E aims to validate the suitability and applicability of the methodological guidelines and KPIs developed. For this purpose, an exemplary process combination is defined and described in detail. Based on this process combination, an exemplary HyBECCS implementation project is defined. All associated assumptions relevant to the techno-economic assessment are derived and described. The system boundaries proposed in Paper C are applied and the related monetary, mass and GHG flows are determined. The KPIs EF, LCCNH, LCNE and LCSC are calculated and discussed. It is shown that the theoretical fundamentals elaborated can be implemented in practice. A comparison of the exemplary HyBECCS project with other NETs and hydrogen production technologies is carried out successfully in this work.

Additionally, the publications on which this thesis is based contain partial results that are not directly related to each other. Paper B and Paper D, in particular, contain additional findings that contribute to a better understanding of the work but are not essential to the overall context. However, they have relevance for the entire work as additional contributions to the overall thread of the thesis. This mainly runs through Paper A, Paper C, and Paper D.

47

6 Critical Reflection and Limitations

In this chapter, various aspects that limit the applicability of the results are discussed, and further issues that significantly influence their practical application are outlined.

First, it has to be clearly stated that this work focuses on providing fundamentals for technoeconomic assessments but does not make specific recommendations for individual technologies, such as dark fermentation (Paper A), biogas steam reforming (Paper B), or dark photosynthesis (Paper E). It is important to acknowledge that HyBECCS technologies are at an early stage of development, which introduces considerable uncertainties in their evaluation. Practical implementations may yield widely varying results, making it challenging to predict the precise ecological impacts and economic outcomes of these technologies. Hence, although the HyBECCS concept's peculiarities and fundamentals for techno-economic assessment have been elaborated in the best possible view of technical realization, they must not be misinterpreted as results for the evaluation of specific technological approaches. All assumptions are solely made to demonstrate the applicability of the assessment fundamentals provided.

Second, deep and critical evaluations of the biomass substrate used for HyBECCS processes are essential for their assessment. In this work, the focus lays on the value chain parts after the biomass supply. However, the choice and production pathways of biomass for use in bioenergy processes profoundly impacts their sustainability. In general, when assessing the role of biomass supply in climate impacts, it is crucial to consider the overall greenhouse gas emissions in agriculture, forestry and other land use (AFOLU) and the food system by examining the links between production, consumption and trade (Blandfort & Hassapoyannes 2018, p. 34). According to the IPCC (2023, p. 44), AFOLU contributed to approximately 22% of total anthropogenic GHG emissions in the reference year of 2019. Within this, biomass generation impacts climate change directly through emissions of N₂O, CO₂ and CH₄. Further, agricultural activities can affect the capacity of soils to absorb and store carbon (soil carbon sequestration). Therefore they additionally contribute indirectly through their impact on land use and land-use change. Significant influences of the biomass production on bioenergy processes have been shown in many cases. For example, Lask et al. (2018) analyzed the environmental impacts of ethanol production from miscanthus grown on marginal land in an average-yield site in Germany. They show that biomass cultivation and harvest accounts for 16.5-18.0 kg CO₂eq per GJ, which corresponds to about 33-62 % of the GHG emissions of the ethanol production pathways considered (Lask et al. 2018, p.276-277). A further aspect is the transportation of biomass to the processing sites. It can contribute significantly to the overall GHG emissions of the value chains. At the example of the production, storage, (un)loading and transportation over 100 km of baled and chopped miscanthus analyzed by Smeets et al. (2009), the transport by truck contributes up to 20% of the total GHG emissions (Smeets et al. 2009, p. 1239). Understanding these impacts is crucial for the overall assessment of all bioenergy processes, including HyBECCS. Against this background, it must also be clearly stated that it is not enough to rely on the use of organic residual and waste materials, as their sustainability is not guaranteed. The varying effects of the biomass substrates used for bioenergy production are outlined by Tonini et al. (2016), for example. They compared greenhouse gas emission factors (EFs) for bioelectricity, biomethane, and bioethanol produced from various biomasses, including manures, wood residues, and food waste. The EFs show wide ranges for the various options. For example, in biomethane production through anaerobic digestion, the EFs differ widely for specific substrates, such as manures or household food wastes (-104 to -44 g CO₂eq / MJ), straw/stover, wild grass or seaweed (20– 51 g CO₂eq / MJ), perennial crops like miscanthus or willow (10–40 g CO₂eq / MJ), annual energy crops like wheat grain (100 g CO₂eq / MJ), and agro-industrial residues like potato pulp (200 g CO₂eq / MJ) (Tonini et al. 2016, p. 129). Further, the availability of biomass sources is also strictly limited and subject to resource conflicts. For example, increased competition for waste and residues may occur in countries with developed economies due to a growing demand for bioenergy (Muscat et al. 2020, p. 4). As highlighted by the European Environment Agency (EEA 2023), many trade-offs must be considered for biomass use in terms of food and energy supply, nature conservation, pollution reduction and climate change mitigation. They conclude that, due to the different roles envisaged for biomass and a potential shortage of biomass supply in the future, decisions are pending on how and for what purposes biomass should be prioritized (EEA 2023, p. 9). Many studies deal with the uncertainties that arise e.g.

from these open decisions by using assumptions or idealizations in their scenarios. For example, Paper A in this work refers to an estimate of the theoretical potential for biohydrogen production from Weide et al. (2020, p. 6) which is based on theoretically available biogenic residual materials in Germany derived from Brosowski et al. (2016, p. 266-268). They assume that all available waste streams were used for biohydrogen production without considering alternative uses. However, the advantages and disadvantages of HyBECCS processes must be compared with other utilization methods for the restricted biomass available. Hereby, the manifold mechanisms that drive competition for biomass use and their interactions should be considered from the perspective of different disciplines like ecology, social science and agro-economy (Lewandowski 2015, p. 39). One prerequisite for the production of biomass for energy use, for example, is that priority is given to the demand for food, industrial roundwood and traditional wood fuel (Smeets et al. 2007, p. 95).

Third, considering the ecological effects of HyBECCS technologies as a part of the technoeconomic assessment, a focus is set exclusively on the impact category of global warming potential. For instance, the EF aims to serve as a basis to compare NETs regarding their GHG removal efficiencies. The LCCNH, LCNE and LCSC also integrate solely GHG footprints. This is due to the main objective of the considered technology concept for climate protection. However, other ecological impact categories remain unconsidered. These include, for example, acidification and eutrophication potential or photochemical oxidant formation, ozone layer and abiotic depletion. In a broader view, since HyBECCS represents a BECCS technology, this novel approach is subject to all sustainability criteria of BECCS technologies. Therefore, all potential issues coming from biomass use should be carefully considered. Secure food supply, avoidance of land degradation and land use conflicts, conservation of water resources, as well as preservation of biodiversity should be taken into account (Fajardy 2019). Moreover, additional ecological risks and side effects related to carbon dioxide storage have to be considered. For instance, overpressure at CS could lead to the pollution of potable water, seismic activity or leaks, which could not only rapidly reverse positive mitigation effects but cause environmental and health damage at the leakage sites, are further potential issues to be addressed (Fuss et al. 2018, p. 14).

Moreover, economic influences from GHG balances are integrated into the key performance indicators LCCNH, LCNE and LCSC, following an idealized exemplary basic model for HyBECCS regulatory systems in carbon-neutral economies. This leads to all of those KPIs only being valid as long as no GHG emission internalizations are part of the internal cost calculations or if they are vanishingly small. Internalizations could be included, for example, with the costs for electricity, heat or fuels for transport. Additionally, internalizations already contained in the values of the LCOE of the comparative technologies must be considered. In this work, the internalizations are negligible in contrast to the assumed internalization prices based on the societal costs of carbon. Those are included in the KPIs separately, as defined. However, GHG emission internalizations would have to be considered and deducted from the internal cost calculations, especially in the case of future increases in internalization costs.

As described by Ulrich and Hill, scientific thinking is characterized by a conscious willingness to constantly test the correctness of the assumptions made (Ulrich & Hill 1976, p. 305). Considering the HyBECCS concept introduced in chapter 2, many of the related technology options for the sub-processes are already known and tested individually. However, a coherent HyBECCS process chain has never been implemented in reality. The concept's technical feasibility has, therefore, only been demonstrated for individual approaches of the respective sub-process steps. The demonstration of a complete process chain is still pending. Further, there is no claim to completeness of the technology options under consideration. The possibilities and combinations are numerous and not defined conclusively. Moreover, further technology options are likely to be developed in the future, expanding the range of technical implementation possibilities. In particular, the exemplary HyBECCS project used as a validation example in chapter 4 has also not been implemented. The results strongly depend on the assumptions made and their respective uncertainties. Although this is demonstrated and discussed in the paper through a sensitivity analysis, further unconsidered characteristics may arise during implementation in reality. Moreover, the exemplary process analyzed in this paper is only one technological HyBECCS approach, amongst many others. It was used to achieve the primary objective of applying and validating the assessment fundamentals described in Paper C. It provides a better understanding of the theoretical basics and thus a better applicability in practice. However, the exemplary HyBECCS project and analysis results are not representative of HyBECCS technologies in general. They only apply to the technology combination considered under the assumptions made.

In conclusion, this work provides techno-economic assessment fundamentals for HyBECCS processes. There is a multitude of external factors that greatly influence the assessment results. These have been considered but cannot be fully covered in a single work. Calculating greenhouse gas footprints for substrate biomass, considering further environmental impact categories, limitations in sustainable biomass availability, and the technical uncertainties surrounding emerging technologies all play pivotal roles. Further research is essential to address these crucial aspects comprehensively and refine the assessment fundamentals of HyBECCS processes provided within this dissertation.

7 Summary and Outlook

Negative emission technologies are very likely to be inevitable to limit global warming to 1.5 °C above pre-industrial levels (IPCC 2018, p. 17). In this doctoral thesis, a concept for a negative emission technology is introduced that combines hydrogen production from renewable biomass, preferential biogenic residues or waste, with carbon capture and storage. This concept is referred to as HyBECCS (hydrogen bioenergy with carbon capture and storage). HyBECCS is not a single technology but can be implemented with many different technological options. The objective of the work is to establish a general and comprehensive basis for techno-economic assessment of these individual technologies. This basis should enable technology comparisons with other hydrogen production technologies and NETs regarding their techno-economic performance. A special focus is placed on the integration of climate protection aspects and resulting impacts on the economy. Figure 6-1 illustrates the basic structure of the work, as outlined in section 1.3. The contributions made to answer the research sub-questions derived in section 1.2 are explained below.



Figure 7-1 Summary of the work: Main structure, paper contributions and connecting points to the research sub-questions.

In the first chapter, the initial situation is described in section 1.1, starting with the problems arising from climate change. The role of carbon dioxide removal (CDR) in limiting global warming to 1.5 °C above pre-industrial levels is explained and technical options for NETs are discussed. They include bioenergy with carbon capture and storage (BECCS) approaches like HyBECCS and direct air carbon capture and storage (DACCS), among others. The important factor for NETs is the carbon dioxide source. It must be either of atmospheric or biogenic origin. CCS alone or based on CO_2 from fossil origin does not remove CO_2 from the atmosphere.

Furthermore, the objective of the work is derived in section 1.2. The research gap of a missing basis for the techno-economic assessment of HyBECCS technologies is outlined. It leads to the derivation of research questions. Hereby, the main question on how HyBECCS technologies can be assessed techno-economically is further subdivided into five sub-questions, as depicted in Figure 6-1.

In section 1.3, the research process and the basic structure of the work are explained. The interconnections between the single contributions are explained and references to the

research questions derived in 1.2 are made. The contributions to the research process according to Ulrich as well as the research questions addressed in the respective chapters and publications (A-E) are described briefly (Ulrich 2001, p. 195).

Moreover, the philosophy of science is explained in section 1.4, assigning the work to the applied sciences. This is due to the practical relevance of the outcomes. They aim to provide appropriate tools for changing reality in a complex social system through methodological advice for private and political decision-making. In particular, the work is also linked to technical sciences as the actual and possible actions are associated with a novel technology concept introduced in this work.

The first research sub-question *"What is the HyBECCS concept?"* is answered in chapter 2 introducing the definition of HyBECCS and outlining the peculiarities of the concept. Preparatory to answering the second sub-question *"Which requirements have to be fulfilled to qualify as HyBECCS technologies?"*, a definition of basic requirements is paved by the technical abstraction and classification of HyBECCS as a NET and CDR option.

For this purpose, the HyBECCS technology concept and the associated scientific foundations are introduced and explained first. Significant contributions to this part are covered in Paper A. It summarizes the state of the art in science and technology for hydrogen technologies and NETs to derive the basic HyBECCS concept. The HyBECCS concept is introduced as a novel option for BECCS. It is based on the physical-technical fact that large shares of highly concentrated biogenic CO₂ are produced as a by-product during most biohydrogen production processes. The combination of biohydrogen production with CCS results to the HyBECCS technology concept as a new perspective for CDR.

The concept is further abstracted into four basic technical process steps: biomass pretreatment, biohydrogen production, product gas separation and its processing and use or storage. Different technological options can be selected for all steps. As a prerequisite for every HyBECCS technology, the GHG balance of the overall process chain must be negative. To calculate the GHG balance, the amounts of physically captured and stored biogenic CO₂ can be subtracted from the GHG emissions resulting from all sub-processes. After the deduction, the total process GHG emissions must be negative for the produced hydrogen to qualify as

carbon-negative hydrogen (CNH). Moreover, technical advantages of HyBECCS technologies in comparison with other NETs are indicated in Paper A. Potentially high capture capabilities for biogenic CO₂ with comparatively low energy inputs are highlighted as the main technical benefits of the approach. Additionally, based on further literature, a theoretical maximum amount of GHG emissions to be avoided through the substitution of natural gas with biohydrogen produced by dark fermentation, as well as the maximum amount of sequestrable biogenic CO₂, is estimated for the geological and political area of the FRG and the reference year 2030.

Additional contributions on more specific application possibilities are made in Paper B, considering the technology option of retrofitting biogas plants to HyBECCS facilities. The technical process of biogas steam reforming is described and discussed considering technical and economic aspects.

In chapter 3, the basis for techno-economic assessment of HyBECCS technologies is introduced. Research sub-question 3 *"Which system boundaries are suitable for HyBECCS technologies to balance monetary, mass and greenhouse gas emission flows?"* is answered by separating the overall concept into two system elements: Biohydrogen production with carbon dioxide capture (HyBECC) and carbon dioxide transport and storage (CS). For both system elements, separate subsystem boundaries B1 and B2 are determined. Monetary, mass and GHG emission flows are defined for each subsystem boundary. They combine to the overall HyBECCS system boundary B3.

These abstractions allow the basic prerequisites for HyBECCS technologies to be substantiated from a mathematically accessible point of view and the definition of specific KPIs. An important value for validating the prerequisite of negative emissions is the emission factor (EF). It relates the process emissions caused over the entire process chain to the physically stored amount of biogenic CO₂. This value must be less than 1 for the HyBECCS approach to qualify as NET and the hydrogen produced as CNH. If the value exceeds 1, more GHG emissions are released over the entire process chain than are removed from the atmosphere. From an economic point of view, it is required to make a profit P. The correlations for the value P are described in the work. It must be above zero for HyBECCS technologies to be economically viable. The mathematical evaluation correlations for both the basic requirements of HyBECCS technologies profitability (P<0) and qualification as a NET (EF<1) are thus completed in this paper, answering research question 2 conclusively.

Sub-question 4 *"How can the economic impact of the greenhouse gas balance be incorporated into the techno-economic evaluation of HyBECCS processes?"* is answered by discussing the correlations between GHG emissions and economic effects of HyBECCS technologies. Internalization of economic damages caused by GHG emissions or benefits that occur through avoiding the damages are identified and described as a link between ecology and economy. Negative emissions, as one of the main products of HyBECCS processes, and emission reductions do not represent an individual, internal benefit but an overall societal, external benefit. Conventional market-based mechanisms and tools are, therefore, not sufficient to make such processes economically viable. The answer to the question lies in the instrument of internalization. Through internalization mechanisms, the HyBECCS concepts could benefit economically in two different ways: Relative cost reductions against competing technologies with higher process emissions and revenues for generating negative emissions. Internalization mechanisms are described specifically for the HyBECCS concept in chapter 3.

Sub-question 5 *"Which key performance indicators are suitable for techno-economic assessment of HyBECCS technologies?"* is answered by defining and applying appropriate KPIs. The LCCNH, LCNE and LCSC are introduced as key KPIs. The LCCNH constitute the costs incurred to produce 1 kWh (net calorific value) hydrogen with negative emissions in a climate-neutralized way, i.e. fully internalize the economic impacts of GHG emissions and removals. The LCNE are the costs that occur to remove one ton of CO₂ from the atmosphere in a climate-neutralized way, and the LCSC are defined as the costs that occur to permanently store one ton of CO₂ in a climate-neutralized way. All of these KPIs assume ideal internalizations of GHG emissions and reductions. This means that the entire external economic damage or damage avoidance through GHG emissions and removals is assumed to be internalized. For this assumption, a basic model for regulatory systems in net-zero emission economies is defined.

In chapter 4, the assessment fundamentals derived in Paper C are applied using an exemplary HyBECCS technology. By successfully performing the techno-economic assessment, research sub-questions 2-5 are answered conclusively, evaluating the applicability of the assessment tools provided. The contributions to this part are described in Paper E. At the example of a

HyBECCS technology approach based on the "dark photosynthesis" process and carbon dioxide storage in saline formations, the EF, LCCNH, LCNE and LCSC are determined and discussed. The exemplary assessment leads to a better understanding of the assessment principles and serves as a guide for application in practice. The results show that the exemplary project would be capable of generating CNH, removing 17.72 kgCO₂ from the atmosphere for every kilogram of H₂ produced. Capital expenditures are estimated at 9,871,401 EUR, and operational expenditures at 2,188,964 EUR per year. The LCCNH are estimated to be 0.013 EUR/kWhH₂. Compared to grey hydrogen from natural gas (0.12)EUR/kWhH₂) and green hydrogen from electrolysis using renewable electricity (0.18 EUR/kWhH₂), the results indicate a high level of economic viability. Even without the assumed internalization of GHG impacts, the calculations point to an economic advantage of the exemplary HyBECCS project (0.12 EUR/kWhH₂) over green hydrogen from electrolysis (0.17 $EUR/kWhH_2$). Converted to EUR/kgH_2 , the internal LCOE of the HyBECCS project amounts to approx. 4 EUR/kgH₂. These costs are below the estimated maximum total costs to be competitive in Germany of 4.30 EUR/kgH₂ in the worst-case scenario of Paper A. Compared to other NETs, the GHG removal efficiency represented by the emission factor (EF) of 0.12 tCO_2eq/tCO_2 stored is at the lower end of the EF range of both other BECCS and DACCS approaches. The LCSC of the exemplary project was estimated at -260 EUR/tCO₂stored, and the LCNE at -297 EUR/tCO₂removed. This can be interpreted for the exemplary project to be capable of generating a respective income per ton of biogenic CO₂ stored (LCSC) or removed from the atmosphere (LCNE). The income is mainly due to the sales of the co-product biohydrogen. For BECCS, the LCNE cost range is estimated at approx. 39-367 EUR EUR/tCO₂removed and for DACCS approx. 25-1,127 EUR/tCO₂removed. This indicates a high level of economic viability and competitiveness of the exemplary project with other NETs.

It is shown that the theoretical assessment fundamentals elaborated in this doctoral thesis can be implemented in practice. A comparability of the exemplary HyBECCS project with other NETs and hydrogen production technologies in terms of their techno-economic performance was achieved. It was also possible to specifically evaluate the economic influences for society coming from GHG emissions for the technologies under consideration. The analysis shows that the findings can support technology providers and investors in decision-making. In addition, especially through the integration of climate protection aspects and resulting impacts on the economy, the assessment fundamentals can also serve as a guideline for political decision-making. The objectives of the work, as stated in section 1.2, are thus fulfilled

However, the results are discussed in chapter 5. Core findings and limitations are derived to give further practice advice for the techno-economic assessment of HyBECCS technologies. For this purpose, parts from all five papers are included in the discussion. Important limitations include the fact that the assumptions made have yet to be validated ex-post based on a real example process. Further, the exclusive consideration of GHG emissions when integrating environmental effects into the assessment is discussed, and the limits of applicability of the defined KPIs are stated. If GHG emission or removal internalization cannot be separated from or neglected within cost and revenue calculations, the KPIs are not applicable.

In conclusion, the central research question on how HyBECCS technologies can be assessed techno-economically has been answered delivering the main outcomes of the work, as summarized in the following bullet points:

- A comprehensive technology concept that describes different process combinations for HyBECCS has been elaborated.
- Requirements and prerequisites for their suitability as NET have been defined, and fundamental assessment tools for their verification have been provided. They make sure that the technologies are economically viable and actually provide negative emissions.
- Standardized rules, system boundaries and key performance indicators for HyBECCS technologies have been derived. They serve to compare HyBECCS technologies with other hydrogen generation technologies (LCCNH) and NET (EF, LCNE, LCSC)
- The added value for society, including economic effects resulting from climate change mitigation, can be depicted analyzing the KPIs. The connection between economic sustainability and the impact on the global climate is thus made tangible.
- A techno-economic assessment of an exemplary HyBECCS implementation project has been conducted to validate the applicability of the elaborated fundamentals in

practice. It has been shown that decision-making for private investments and public subsidies can be supported with the fundamentals provided.

Both the HyBECCS concept and the assessment fundamentals elaborated in this doctoral thesis should be further developed. Considering the HyBECCS concept, in this work, the focus is placed on the use of hydrogen and the storage of biogenic CO₂. However, possibilities for biohydrogen production with CO₂ capture and utilization (HyBECCU) could further increase the economic perspectives. As outlined in Paper A, a coupled use of biohydrogen for the chemical reduction of CO₂ aiming to create higher-value products is a promising approach for HyBECCU. Yet, higher risk of reversals, insecurity concerning the time horizon of storage, and challenges such as monitoring the permanence should be taken into account considering CU (Otto et al. 2017). It has to be clearly stated that CO₂-based products from HyBECCU with short lifetimes before re-emitting the CO₂, such as fuels or disposable plastic products, are not capable of achieving negative GHG emissions.

Further investigations should also be conducted to improve and extend the techno-economic assessment basis. For example, as CS technologies and NETs are associated with different time horizons of storage and risks of reversal, these factors have not been considered yet. They should be taken into account within further research activities to raise the comparability of different NETs and CS technologies. Respective advancements of the KPIs derived in this work are advisable.

Social and political framework conditions should also be adapted. Establishing a certification system for carbon-negative hydrogen, which can ensure that the product has a negative carbon footprint, could help to ensure widespread and long-term acceptance. This could be based on the assessment fundamentals elaborated within this work.

61

8 Literature

Antonini et al. 2020	Antonini, Cristina; Treyer, Karin; Streb, Anne; van der Spek, Mijndert; Bauer, Christian; Mazzotti, Marco, 2020. Hydrogen production from natural gas and biomethane with carbon capture and storage – A techno- environmental analysis. <i>Sustainable Energy &</i> <i>Fuels</i> 4 (5), p. 2967-2986. DOI: 10.1039/D0SE00222D
Bhatia 2014	Bhatia, S. C. (ed.), 2014. <i>Advanced Renewable</i> <i>Energy Systems. Part II</i> . In: Woodhead Publishing India, New Delhi, p. 627-630. ISBN: 978-1-78242-269-3
Blandfort & Hassapoyannes 2018	Blandford, David; Hassapoyannes, Katherine, 2018. <i>The role of agriculture in global GHG</i> <i>mitigation</i> . In: OECD Food, Agriculture and Fisheries Papers (112), OECD Publishing, Paris. DOI: 10.1787/da017ae2-en
Borchers et al. 2022	Borchers, Malgorzata; Thrän, Daniela; Chi, Yaxuan; Dahmen, Nicolaus; Dittmeyer, Roland; Dolch, Tobias; Dold, Christian; Förster, Johannes; Herbst, Michael; Heß, Dominik; Kalhori, Aram; Koop-Jakobsen, Ketil; Li, Zhan; Mengis, Nadine; B. H. Reusch, Thorsten; Rhoden, Imke; Sachs, Torsten; Schmidt-Hattenberger, Cornelia; Stevenson, Angela; Thoni, Terese; Wu, Jiajun; Yeates, Christopher, 2022. Scoping carbon dioxide removal options for Germany - What is their potential contribution to Net-Zero CO ₂ ?. <i>Frontiers in Climate Section Carbon Dioxide</i> <i>Removal</i> 4 , Nr. 81034. DOI: 10.3389/fclim.2022.810343

Brockhoff 2010	Brockhoff, Klaus, 2010. <i>Marktorientierte Technikwissenschaft.</i> In: acatech DISKUTIERT: Technologisches Wissen-Entstehung, Methoden, Strukturen. Berlin: Springer, S.
	183-209. ISBN: 978-3-642-14371-7 DOI: 10.1007/978-3-642-14372-4
Brosowski et al. 2016	Brosowski, André; Thrän, Daniela; Mantau, Udo; Mahro, Bernd, Erdmann, Georgia; Adler, Philipp; Stinner, Walter; Reinhold, Gerd; Hering, Thomas; Blanke, Christian, 2016. A review of biomass potential and current utilisation – Status quo for 93 biogenic wastes and residues in Germany. <i>Biomass and Bioenergy</i> 95, p. 257–272. DOI: 10.1016/j.biombioe.2016.10.017
EEA 2023	Kowalczewska, Katarzyna; van Aardenne, John; Iversen, Peter; Petersen, Jan-Erik, 2023. <i>The European biomass puzzle: Challenges,</i> <i>opportunities and trade-offs around biomass</i> <i>production and use in the EU</i> . In: EU Publications, Luxembourg, European Environment Agency, 170 pp. ISBN: 978-92-9480-604-8 DOI: 10.2800/834565
FOJ 2021	Federal Office of Justice, 2019. Federal Climate Change Act of 12 December 2019 (Federal Law Gazette I, p. 2513), as last amended by Article 1 of the Act of 18 August 2021 (Federal Law Gazette I, p. 3905).
Full et al. 2021b	Full, Johannes; Trauner, Mathias; Miehe, Robert; Sauer, Alexander, 2021. Carbon- Negative Hydrogen Production (HyBECCS) from Organic Waste Materials in Germany: How to Estimate Bioenergy and Greenhouse Gas Mitigation Potential. <i>Energies</i> 14 (22), Nr. 7741.

Energies **14** (22), Nr. 7741. DOI: 10.3390/en14227741

Full et al. 2022b	Full, Johannes; Ziehn, Sonja; Geller, Marcel, Miehe, Robert; Sauer, Alexander, 2022. Carbon-negative hydrogen production: Fundamentals for a techno-economic and environmental assessment of HyBECCS approaches. <i>GCB Bioenergy</i> 14 (5), p. 597-619. DOI: 10.1111/gcbb.12932
Full et al. 2021a	Full, Johannes; Merseburg, Steffen; Miehe, Robert; Sauer, Alexander, 2021. A new perspective for climate change mitigation – introducing carbon-negative hydrogen production from biomass with carbon capture and storage (HyBECCS). <i>Sustainability</i> 13 (7), Nr. 4026. DOI: 10.3390/su13074026
Full et al. 2023a	Full, Johannes; Hohmann, Silja; Ziehn, Sonja; Gamero, Edgar; Schließ, Tobias; Schmid, Hans-Peter; Miehe, Robert; Sauer, Alexander, 2023. Perspectives of Biogas Plants as BECCS Facilities: A Comparative Analysis of Biomethane vs. Biohydrogen Production with Carbon Capture and Storage or Use (CCS/CCU). <i>Energies</i> 16 (13), Nr. 5066. DOI: 10.3390/en16135066
Full et al. 2023b	 Full, Johannes; Geller, Marcel; Ziehn, Sonja; Schließ, Tobias; Miehe, Robert; Sauer, Alexander, 2023. Carbon-Negative Hydrogen Production (HyBECCS): An Exemplary Techno- Economic and Environmental Assessment. International Journal of Hydrogen Energy 52, p. 594-609. DOI: 10.1016/j.ijhydene.2023.09.252

Fuss et al. 2018	Fuss, Sabine; F Lamb, William , Callaghan, Max W , Hilaire, Jérôme; Creutzig, Felix; Amann, Thorben; Beringer, Tim; de Oliveira Garcia, Wagner; Hartmann, Jens; Khanna, Tarun; Luderer, Gunnar; F Nemet, Gregory; Rogelj, Joeri; Smith, Pete; Vicente, José Luis; Wilcox, Jennifer; del Mar Zamora Dominguez, Maria; Minx, Jan C, 2018: Negative emissions – Part 2: Costs, potentials and side effects. <i>Environmental Research Letters</i> 13 (6), Nr. 063002. DOI: 10.1088/1748-9326/aabf9f
IPCC 2005	Metz, Bert; Davidson, Ogunlade; de Coninck, Heleen; Loos, Manuela; Meyer, Leo (Eds.), 2005. <i>A Special Report of Working Group III of</i> <i>the Intergovernmental Panel on Climate</i> <i>Change. (Summary for Policymakers)</i> In: Cambridge University Press, UK. pp 431. The Edinburgh Building Shaftesbury Road, Cambridge CB2 2RU. ISBN: 13 978-0-521-86643-9
IPCC 2018a	Masson-Delmotte, V., P. Zhai, HO. Pörtner, D. Roberts, J. Skea, P.R. Shukla, A. Pirani, W. Moufouma-Okia, C. Péan, R. Pidcock, S. Connors, J.B.R. Matthews, Y. Chen, X. Zhou, M.I. Gomis, E. Lonnoy, T. Maycock, M. Tignor, and T. Waterfield (eds.), 2018. <i>Global</i> <i>Warming of 1.5°C. An IPCC Special Report on</i> <i>the impacts of global warming of 1.5°C above</i> <i>pre-industrial levels and related global</i> <i>greenhouse gas emission pathways, in the</i> <i>context of strengthening the global response</i> <i>to the threat of climate change, sustainable</i> <i>development, and efforts to eradicate</i> <i>poverty. (Summary for Policymakers).</i> In: Cambridge University Press, Cambridge, UK, New York, USA, pp. 3-24. DOI: 10.1017/9781009157940.001

IPCC 2018b	Masson-Delmotte, V., P. Zhai, HO. Pörtner, D. Roberts, J. Skea, P.R. Shukla, A. Pirani, W. Moufouma-Okia, C. Péan, R. Pidcock, S. Connors, J.B.R. Matthews, Y. Chen, X. Zhou, M.I. Gomis, E. Lonnoy, T. Maycock, M. Tignor, and T. Waterfield (eds.), 2018. Global Warming of 1.5°C. <i>An IPCC</i> <i>Special Report on the impacts of global warming</i> <i>of 1.5°C above pre-industrial levels and related</i> <i>global greenhouse gas emission pathways, in the</i> <i>context of strengthening the global response to</i> <i>the threat of climate change, sustainable</i> <i>development, and efforts to eradicate poverty.</i> In: Cambridge University Press, Cambridge, UK, New York, USA, 616. DOI: 10.1017/9781009157940
IPCC 2021a	Masson-Delmotte, V., P. Zhai, A. Pirani, S.L. Connors, C. Péan, S. Berger, N. Caud, Y. Chen, L. Goldfarb, M.I. Gomis, M. Huang, K. Leitzell, E. Lonnoy, J.B.R. Matthews, T.K. Maycock, T. Waterfield, O. Yelekçi, R. Yu, and B. Zhou (eds.), 2021. <i>Climate Change 2021: The Physical Science</i> <i>Basis. Contribution of Working Group I to the Sixth</i> <i>Assessment Report of the Intergovernmental</i> <i>Panel on Climate Change. (Summary for</i> <i>Policymakers).</i> In: Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA, pp. 3–32. DOI: 10.1017/9781009157896.001
IPCC 2021b	Masson-Delmotte, V., P. Zhai, A. Pirani, S.L. Connors, C. Péan, S. Berger, N. Caud, Y. Chen, L. Goldfarb, M.I. Gomis, M. Huang, K. Leitzell, E. Lonnoy, J.B.R. Matthews, T.K. Maycock, T. Waterfield, O. Yelekçi, R. Yu, and B. Zhou (eds.), 2021. <i>Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth</i> <i>Assessment Report of the Intergovernmental</i> <i>Panel on Climate Change.</i> In: Cambridge University Press, Cambridge, UK, New York, NY, USA, 2391. DOI: 10.1017/9781009157896

IPCC 2022a	 HO. Pörtner, D.C. Roberts, M. Tignor, E.S. Poloczanska, K. Mintenbeck, A. Alegría, M. Craig, S. Langsdorf, S. Löschke, V. Möller, A. Okem, B. Rama (eds.), 2022. <i>Climate Change 2022: Impacts,</i> Adaptation and Vulnerability. Contribution of Working Group II to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. (Summary for Policymakers). In: Cambridge University Press, Cambridge, UK, New York, NY, USA, pp. 3–33. DOI: 10.1017/9781009325844.001
IPCC 2022b	 HO. Pörtner, D.C. Roberts, M. Tignor, E.S. Poloczanska, K. Mintenbeck, A. Alegría, M. Craig, S. Langsdorf, S. Löschke, V. Möller, A. Okem, B. Rama (eds.), 2022. <i>Climate Change 2022: Impacts,</i> <i>Adaptation and Vulnerability. Contribution of</i> <i>Working Group II to the Sixth Assessment Report</i> <i>of the Intergovernmental Panel on Climate</i> <i>Change.</i> In: Cambridge University Press. Cambridge University Press, Cambridge, UK, New York, NY, USA, 3056 pp. DOI: 10.1017/9781009325844
IPCC 2023	H. Lee and J. Romero (eds.) 2023. <i>Contribution of</i> <i>Working Groups I, II and III to the Sixth</i> <i>Assessment Report of the Intergovernmental</i> <i>Panel on Climate Change</i> . In: Climate Change 2023: Synthesis Report, Section 4. IPCC, Geneva, Switzerland, p. 35-115 DOI: 10.59327/IPCC/AR6-9789291691647
Kornwachs 2011	Kornwachs, Klaus, 2011. <i>Logische Strukturen technischen Wissens – zur Wissenschaftstheorie der Technikwissenschaften</i> . In: Technologisches Wissen: Entstehung, Methoden, Strukturen. Springer-Verlag. doi: 10.1007/978-3-642-14372-4
Kovac et al. 2021	Kovač, Ankica; Paranos, Matej; Marciuš, Doria, 2021. Hydrogen in energy transition: A review. <i>International Journal of Hydrogen Energy</i> 46 (16), p. 10016-10035. DOI: 10.1016/j.ijhydene.2020.11.256

Kubicek 1977	Kubicek, Herbert, 1977. <i>Heuristische</i> <i>Bezugsrahmen und heuristisch angelegte</i> <i>Forschungsdesigns als Elemente einer</i> <i>Konstruktionsstrategie empirischer Forschung.</i> In: Empirische und handlungstheoretische Forschungskonzeptionen in der Betriebswirtschaftslehre. Stuttgart: Poeschel, p. 3–36. ISBN: 3-7910-0214-7
Lask et al. 2018	Lask, Jan; Wagner, Moritz; Trindade, Luisa M.; Lewandowski, Iris, 2019. Life cycle assessment of ethanol production from miscanthus: A comparison of production pathways at two European sites. <i>GCB Bioenergy</i> 11 (1), p. 269-288. DOI: 10.1111/gcbb.12551
Lewandowski 2015	Lewandowski, Iris, 2015. Securing a sustainable biomass supply in a growing bioeconomy. <i>Global Food Security</i> 6 , p. 34-42. DOI: 10.1016/j.gfs.2015.10.001
Muscat et al. 2020	Muscat, A.; de Olde, E.M.; de Boer, I.J.M.; Ripoll- Bosch, R, 2020. The battle for biomass: A systematic review of food-feed-fuel competition. <i>Global Food Security</i> 25 , Nr. 100330. DOI: 10.1016/j.gfs.2019.100330
Nikolaidis & Poullikkas 2017	Nikolaidis, Pavlos; Poullikkas, Andreas, 2017. A comparative overview of hydrogen production processes. <i>Renewable and sustainable energy</i> <i>reviews</i> 67 , p. 597-611. DOI: 10.1016/j.rser.2016.09.044
Otto et al. 2017	Otto, Alexander; Markewitz, Peter; Robinius, Martin, 2017. <i>Technologiebericht 2.4 CO2-</i> <i>Nutzung.</i> In: Technologien für die Energiewende. Teilbericht 2 an das Bundesministerium für Wirtschaft und Energie (BMWi): 31-32. Wuppertal Institut, ISI, IZES (Hrsg.) Wuppertal, Karlsruhe, Saarbrücken.
Ropohl 2009	Ropohl, Günter, 2009. <i>Signaturen der technischen Welt: Neue Beiträge zur Technikphilosophie.</i> Berlin: Lit. Technikphilosophie 20. ISBN: 978-3-643-10465-6

Rosa et al. 2021	Rosa, L.; Sanchez, D. L.; Mazzotti, M., 2021. Assessment of carbon dioxide removal potential via BECCS in a carbon-neutral Europe. <i>Energy Environmental Science</i> 14 (5), p. 3086– 3097. DOI: 10.1039/D1EE00642H
Singh et al. 2017	Singh, Anoop; Rathore, Dheeraj (eds.), 2017. Biohydrogen Production: Sustainability of Current Technology and Future Perspective. Springer New Delhi XVIII, 320. DOI: 10.1007/978-81-322-3577-4 ISBN: 978-81-322-3577-4
Smeets et al. 2007	Smeets, Edward M.W.; Faaij, André P.C.; Lewandowski, Iris M.; Turkenburg, Wim C., 2007. A bottom-up assessment and review of global bio- energy potentials to 2050. <i>Progress in Energy and</i> <i>combustion science</i> 33 (1), p. 56-106. DOI: 10.1016/j.pecs.2006.08.001
Smeets et al. 2009	Smeets, Edward M.W.; Lewandowski, Iris M.; Faaij, André P.C., 2009. The economical and environmental performance of miscanthus and switchgrass production and supply chains in a European setting. <i>Renewable and sustainable</i> <i>energy reviews</i> 13 (6-7), p. 1230-1245. DOI: 10.1016/j.rser.2008.09.006
Susmozas et al. 2016	Susmozas, Ana; Iribarren, D.; Zapp, P; Linβen, J.; Dufour, J, 2016. Life-cycle performance of hydrogen production via indirect biomass gasification with CO ₂ capture. <i>International</i> <i>journal of hydrogen energy</i> 41 (42), p. 19484- 19491. DOI: 10.1016/j.ijhydene.2016.02.053
Tonini et al. 2016	Tonini, Davide; Hamelin, Lorie; Alvarado-Morales, Merlin; Fruergaard Astrup, Thomas. 2016. GHG emission factors for bioelectricity, biomethane, and bioethanol quantified for 24 biomass substrates with consequential life-cycle assessment. <i>Bioresource Technology</i> 208 , p. 123- 133. DOI: 10.1016/j.biortech.2016.02.052
Turner 2004	Turner, John A, 2004. Sustainable hydrogen production. <i>Science</i> 305 (5686), p. 972-974. DOI: 10.1126/science.1103197
--------------------	--
Ulrich & Hill 1976	Ulrich, Peter; Hill, Wilhelm, 1976. <i>Wissenschaftstheoretische Grundlagen der</i> <i>Betriebswirtschaftslehre (Teil I).</i> In: Wirtschaftswissenschaftliches Studium 5 (7), p. 304–309
Ulrich 1981	Ulrich, Hans, 1981. <i>Die Betriebswirtschaftslehre als anwendungsorientierte Sozialwissenschaft</i> . In: Die Führung des Betriebes: Curt Sandig zu seinem 80. Geburtstag gewidmet Stuttgart: Pöschel. p. 1-25. ISBN: 3-7910-0308-9
Ulrich 2001	Ulrich, Hans, 2001. Systemorientiertes Management. Das Werk von Hans Ulrich. Gesammelte Schriften – Studienausgabe. Bern, Stuttgart, Wien: Haupt Verlag. ISBN: 3-258-06359-1
Weide et al. 2019	Weide, Tobias; Brügging, Elmar; Wetter, Christof; Ierardi, Antonio; Wichern, Marc, 2019. Use of organic waste for biohydrogen production and volatile fatty acids via dark fermentation and further processing to methane. <i>International</i> <i>Journal of Hydrogen Energy</i> 44 , p. 24110–24125. DOI: 10.1016/j.ijhydene.2019.07.140
Weide et al. 2020	 Weide, Tobias; Feil, Frans; Kamphus, Sören; Peitzmeier, Jens; Brügging, Elmar (Hrsg.), 2020. Potentialstudie: Bio-H2 aus organischen Reststoffen mittels dunkler Fermentation in Deutschland und den Niederlanden. Münster, 18.09.2020 URL: http://www.biohydrogen.eu/wp- content/uploads/2020/09/2020-09-18- Wasserstoff-Potential-DE-NL_final.pdf Zugriff am 07.12.2023

Appendix A: Paper A

Contributions to the work

Title: A New Perspective for Climate Change Mitigation - Introducing Carbon-Negative Hydrogen Production from Biomass with Carbon Capture and Storage (Hybeccs) Published in: MDPI Sustainability 13.7 (2021): 4026

DOI: https://doi.org/10.3390/su13074026

CRediT (Contributor Role	Taxonomy)	author	statement:
---------------------------------	-----------	--------	------------

Authors	Johannes Full	Steffen Merseburg	Robert Miehe	Alexander Sauer
Conceptualization	\checkmark			
Methodology	\checkmark			
Formal analysis	\checkmark			
Investigation	\checkmark	\checkmark		
Data Curation	\checkmark	\checkmark		
Writing (Original Draft)	\checkmark	\checkmark		
Writing (Review and Editing)			\checkmark	\checkmark
Visualization	\checkmark			
Project administration	\checkmark			
Funding acquisition	\checkmark		\checkmark	\checkmark

The co-authors hereby agree to the contributions summarized in the table avove.

Johannes Full Name Steffen Merseburg Name Robert Miehe Name

Alexander Sauer Name

Signature S. Mutors

Signature

Signature

Alexander Laur

16.11.2022 Date 17.11.2022 Date 3.5.2023 Date 03.05.2023 Date

Signature





Article A New Perspective for Climate Change Mitigation— Introducing Carbon-Negative Hydrogen Production from Biomass with Carbon Capture and Storage (HyBECCS)

Johannes Full ^{1,2,*}, Steffen Merseburg ¹, Robert Miehe ¹, and Alexander Sauer ^{1,2}

- ¹ Fraunhofer Institute for Manufacturing Engineering and Automation IPA, 70569 Stuttgart, Germany; steffen.merseburg@ipa.fraunhofer.de (S.M.); robert.miehe@ipa.fraunhofer.de (R.M.); alexander.sauer@ipa.fraunhofer.de (A.S.)
- ² Institute for Energy Efficiency in Production (EEP), University of Stuttgart, 70569 Stuttgart, Germany
- * Correspondence: johannes.full@ipa.fraunhofer.de; Tel.: +49-(71)-1970-1434

Abstract: The greatest lever for advancing climate adaptation and mitigation is the defossilization of energy systems. A key opportunity to replace fossil fuels across sectors is the use of renewable hydrogen. In this context, the main political and social push is currently on climate neutral hydrogen (H₂) production through electrolysis using renewable electricity. Another climate neutral possibility that has recently gained importance is biohydrogen production from biogenic residual and waste materials. This paper introduces for the first time a novel concept for the production of hydrogen with net negative emissions. The derived concept combines biohydrogen production using biotechnological or thermochemical processes with carbon dioxide (CO₂) capture and storage. Various process combinations referred to this basic approach are defined as HyBECCS (Hydrogen Bioenergy with Carbon Capture and Storage) and described in this paper. The technical principles and resulting advantages of the novel concept are systematically derived and compared with other Negative Emission Technologies (NET). These include the high concentration and purity of the CO₂ to be captured compared to Direct Air Carbon Capture (DAC) and Post-combustion Carbon Capture (PCC) as well as the emission-free use of hydrogen resulting in a higher possible CO_2 capture rate compared to hydrocarbon-based biofuels generated with Bioenergy with Carbon Capture and Storage (BECCS) technologies. Further, the role of carbon-negative hydrogen in future energy systems is analyzed, taking into account key societal and technological drivers against the background of climate adaptation and mitigation. For this purpose, taking the example of the Federal Republic of Germany, the ecological impacts are estimated, and an economic assessment is made. For the production and use of carbon-negative hydrogen, a saving potential of 8.49–17.06 MtCO₂,eq/a is estimated for the year 2030 in Germany. The production costs for carbon-negative hydrogen would have to be below 4.30 € per kg in a worst-case scenario and below 10.44 € in a best-case scenario in order to be competitive in Germany, taking into account hydrogen market forecasts.

Keywords: negative emission technologies; carbon dioxide removal; HyBECCS; hydrogen bioenergy with carbon capture and storage; biohydrogen; carbon-negative hydrogen; climate positive hydrogen; CCS; BECCS; DACCS; NET; CDR

1. Introduction

The Paris Accord was adopted in 2015 to significantly limit the risks and impacts of climate change. The participating nations committed themselves to keeping the increase in the global mean temperature well below 2 K compared to pre-industrial levels and to making efforts to limit the increase even to 1.5 K. To achieve these agreements, measures are to be taken that will lead to a balance of greenhouse gas (GHG) emissions and sinks, i.e., greenhouse gas neutrality, in the second half of the 21st century [1]. Therefore, laws and policies have been introduced worldwide. For example, the European Union (EU)



Citation: Full, J.; Merseburg, S.; Miehe, R.; Sauer, A. A New Perspective for Climate Change Mitigation—Introducing Carbon-Negative Hydrogen Production from Biomass with Carbon Capture and Storage (HyBECCS). *Sustainability* **2021**, *13*, 4026. https://doi.org/10.3390/ su13074026

Academic Editors: Adriana Del Borghi and Michela Gallo

Received: 15 March 2021 Accepted: 1 April 2021 Published: 5 April 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). resolved the European Green Deal in 2019, which includes several key measures to reach greenhouse gas neutrality by 2050 within the EU [2]. Numerous resolutions and legislations were also passed at national levels. The Federal Republic of Germany, for example, has set the Climate Protection Program 2030 and the Climate Protection Act, which came into force in December 2019 [3,4].

Burning fossil fuels for gaining energy has the greatest impact on climate change in industrialized nations. Energy-related emissions account for about 81% of greenhouse gas emissions across Europe [5] and about 85% in Germany [6]. The steady expansion of renewable energy is therefore considered a key lever for reducing greenhouse gas emissions. These efforts currently take into account mainly climate-neutral electrical energy by wind power and solar plants [2,7]. However, some of the greenhouse gas emissions cannot be avoided through electrification and expansion of renewable electricity generation. For example, sectors such as the steel, chemical, and cement industries are particularly difficult to defossilize, because electrification is not sufficient [8–10]. Moreover, in the transport and mobility sectors, complete electrification is hardly feasible, especially for large commercial vehicles or in aviation and shipping, due to the limited gravimetric energy density of currently existing battery technologies [11–15]. Renewable chemical energy carriers with high energy densities, generated either via electrochemical, so-called power-to-x processes or from renewable biomass, are better suited for this purpose [7,16]. In this context, sustainably produced green hydrogen is seen as a key chemical energy carrier of the future [7,10,17,18]. Green hydrogen can be produced through electrolysis using renewable electricity or from renewable biomass, either through thermochemical processes or by using microorganisms and biotechnological methods. Hydrogen produced from biomass is also referred to as biohydrogen and can lead to a more efficient use of biogenic raw materials exploiting residual and waste materials.

However, even with a complete conversion of the electrical energy supply to renewable power generation and a successful defossilization of the heat and transport sectors, residual emissions remain in areas such as agriculture and waste management. These emissions must be compensated to net zero GHG emissions by 2050 [7]. GHG emissions in the earth's atmosphere are therefore to be actively reduced using negative emission technologies (NET). The latest special report by the Intergovernmental Panel on Climate Change (IPCC) outlines development scenarios that could limit global warming to less than 1.5 K. In all success scenarios, NET are essential components that will play an increasingly important role in climate adaptation and mitigation by 2050 [19]. Known NET include technologies that achieve a negative carbon footprint by capturing the greenhouse gas CO₂ and permanently storing (CCS—Carbon Capture and Storage) or using (CCU—Carbon Capture and Utilization) it [19]. The most relevant NETs are the direct capture of CO₂ from ambient air (DACCS) and Bioenergy with Carbon Capture and Storage (BECCS), where energy sources are extracted from the biomass and the resulting co-product CO₂ is captured [19,20].

This paper introduces a new NET concept combining biohydrogen production with carbon capture and storage, referred to as HyBECCS. HyBECCS couples the sustainable production of hydrogen from renewable biomass, for example from residual and waste streams, with efficient CO_2 capture and subsequent storage. The hydrogen produced has a negative carbon footprint and is therefore referred to as carbon-negative hydrogen or climate-positive hydrogen. The technological advantages of this new concept can support climate adaptation and mitigation in the future by approaching the GHG emission problem from two sides. On the one hand, it contributes to the defossilization, bringing more emission-free hydrogen into use, especially for sectors that are difficult to electrify, and, on the other hand, it contributes to the compensation of the remaining residual emissions through the active CO_2 extraction from the atmosphere. The availability and suitability of organic feedstocks, microorganisms, and process technologies, as well as the economic relations and ecological impacts of the HyBECCS approach need to be investigated to be able to solve societal and environmental problems. This paper gives the basics for

this purpose through defining the new approach and describing related technologies, discussing technological properties, and assessing ecological and economic benefits for climate adaptation and mitigation using the example of Germany.

2. Objectives and Enablers

The HyBECCS concept describes technology combinations that merge the production of hydrogen from renewable biomass with subsequent carbon dioxide capture and storage or use. The main objectives of the HyBECCS concept introduced in this section are given by climate adaptation and mitigation goals. Hydrogen has a key role to play in this context, as it can greatly contribute to the defossilization of critical industrial areas that are difficult to electrify, such as steel production, chemical industry, cement production, or heavy-duty transport industry. Therefore, many countries have already adopted national strategies to establish sustainably produced green hydrogen as a basic energy carrier. These include, for example, the US, Canada, Japan, France, Norway, the Netherlands, Australia, and Chile [21-28]. The use of carbon-negative hydrogen produced with HyBECCS concepts can additionally compensate for residual emissions by achieving a reduction of the GHG concentration in the atmosphere through active CO_2 capture and storage. For companies, the use of carbon-negative hydrogen offers an opportunity to improve their carbon footprint. Its use can have a positive impact on their GHG emission balances and can make a direct and effective contribution to climate change mitigation. Besides, HyBECCS can contribute to a more efficient use of existing raw materials by residual and waste exploitation. However, to provide sustainable value creation, carbon-negative hydrogen must be competitive and have undergone a holistic ecological analysis to ensure positive environmental impacts. The HyBECCS concept must therefore be ecologically as well as economically analyzed and optimized. Therefore, the most important basis is a technical understanding of the underlying processes, the fundamentals of which are explained in the following sections.

Furthermore, societal, political, and scientific developments that could encourage the dissemination of carbon-negative hydrogen and the application of HyBECCS technologies can be observed. One societal factor that has gained momentum in science and politics in recent years is the so-called biological transformation of the industry and society toward a biointelligent value creation which is closely interlinked with the global efforts of building a sustainable bioeconomy [29–31]. The process of biological transformation describes an increasing use of biology and its materials, principles, and organisms to create new and sustainable value creation systems. This development is enabled by the increasing interconnection of biotechnology, production technology, and information technology. The creation of a sustainable energy supply system is a major field of action and objective of this transformation process, which can lead to a sustainable bioeconomy in the form of a technology-based subsistence economy [32,33]. Key characteristics of these activities are efficient and coupled material and energy carrier cycles as well as utilization cascades which result in energy recovery [29,31,34,35]. Accordingly, the introduced HyBECCS concept can be seen as part of the biological transformation and bioeconomy developments and is likely to be strengthened from the momentum they generate.

3. State of the Art

This section introduces hydrogen usage as an energy carrier and raw material and its particular importance for achieving climate protection targets (Section 3.1). A special focus is placed on the sustainable production of biohydrogen from biomass, as it is the key sub-process for the HyBECCS approach presented (Section 3.2). Further, existing technology approaches for NET are explained, and the respective advantages and disadvantages are discussed (Section 3.3).

3.1. The Future of Hydrogen

Hydrogen as an energy carrier has many advantages and will play an important role in the energy infrastructure and circular economy of the future [36,37]. Since hydrogen can be produced in a climate-neutral manner, and only water is released during its use, it is considered a climate-friendly energy carrier [36–39]. Hydrogen that is produced from renewable sources is generally referred to as "green hydrogen". It is contrasted with the so-called "grey hydrogen", which is referred to as hydrogen produced from fossil, nonrenewable sources [18]. Currently, most of the global hydrogen production consists in grey hydrogen produced by steam reforming of fossil natural gas [8,37]. Grey hydrogen can be made carbon-neutral by applying carbon capture and storage (CCS) technology and is then called "blue hydrogen", but it stays a non-renewable source [18,40]. We introduced "carbon-negative hydrogen" to distinguish between carbon-neutral green or blue hydrogen. Due to the high gravimetric energy density and good storage and transportability, there are many possible applications for hydrogen [37]. Fuel cells can be used to generate electric power as well as heat from hydrogen. Hydrogen can also be used as a reducing agent in the steel industry [8]. It is considered an inevitable basic building block in sectors that are difficult to electrify [37].

In addition to electrolysis using renewable electricity, green hydrogen can be produced from biomass, e.g., from organic residual and waste streams referred to as biohydrogen. A distinction is made between thermochemical and various biotechnological generation pathways for biohydrogen, which will be discussed in the following.

3.2. Biohydrogen Production

Biohydrogen production technologies can be divided into thermochemical and biotechnological approaches. On an industrial scale, biohydrogen can be produced from biomass (wood, straw, grass cuttings, etc.) but also from other bioenergy sources (biogas, bioethanol, etc.) using thermochemical processes. Thermochemical processes can be subdivided into gasification and pyrolysis. Depending on the feedstock used, the synthesis gas formed during gasification or pyrolysis gas consists of varying proportions of hydrogen (H₂), carbon dioxide (CO₂), carbon monoxide (CO), methane (CH₄), and other components such as nitrogen, water vapor, light hydrocarbons, hydrochloric acid, alkali chlorides, sulfur compounds, biochar, or tar [41,42]. Using subsequent steam reforming and/or water gas shift reactions, the following chemical reactions take place, resulting in the production of CO_2 and hydrogen [43,44].

$$CH_4 + H_2O \rightarrow CO + 3H_2 \tag{1}$$

$$CO + H_2O \rightarrow CO_2 + H_2 \tag{2}$$

Biotechnological biohydrogen production processes using suitable microorganisms in adequate production facilities can be divided into three main categories: biophotolysis, photofermentation, and dark fermentation [45,46]. Dark fermentation is particularly promising because it is easier to scale up without exposure due to its simpler structure [47–50]. Moreover, technological maturity is the most advanced among the biotechnical processes and thus also relevant knowledge, for example about process parameters such as product gas compositions, is known. Therefore, in this paper, a special focus is given to dark fermentation, knowing that there are alternative processes or process combinations with even higher potential. These include, for example, combinations, e.g., of dark fermentation and photofermentation [51–53]. Sugars and starchy products are particularly suitable feedstocks for dark fermentation [48]. However, with pretreatment, a variety of feedstocks such as food waste, wood, or wastewater can be used for biological hydrogen production [46,49,54,55]. In dark fermentation, the contained glucose is decomposed into hydrogen, carbon dioxide, and the byproducts acetic acid (3), propionic acid (4), or butyric acid (5) [39,48,49,56,57]. The

following reaction equations form the basis of these syntheses and take place enzymatically catalyzed via metabolic processes of the microorganisms used.

$$C_6H_{12}O_6 + 2H_2O \rightarrow 2CH_3COOH + 2CO_2 + 4H_2$$
 (3)

$$C_6H_{12}O_6 \rightarrow CH_3COOH + CH_3CH_2COOH + CO_2 + H_2$$
(4)

$$C_6H_{12}O_6 + 6H_2O \rightarrow 2CH_3CH_2CH_2COOH + 2CO_2 + 2H_2$$
(5)

The net energy ratio (NER) of biohydrogen production of 1.9 is significantly better than conventional hydrogen production with methane steam reforming, which has a NER of 0.64 [58]. Utilizing the resulting byproducts in one- or two-stage processes [48,56,58] can further improve the NER. For example, coupling dark fermentation with photofermentation can further decompose the acetic acid components producing hydrogen and increase the NER to 3.1 [58]. Both single organisms and mixed bacterial cultures can produce other byproducts in a biohydrogen plant such as acetone, ethanol, or butanol [48]. Furthermore, adding nanoparticles of various metal oxides can also result in a better hydrogen yield [56]. Thus, further improvements of the processes can be expected through continuous research [39,49–51,53,56]. The product gas of biohydrogen production consists of 30–65% H₂ and 70–35% CO₂, depending on the process parameters, such as the type of microorganism, temperature and pH [52,59–65]. Hydrogen sulfide, water, and nitrogen are also possible by-products [66,67].

3.3. Negative Emission Technologies

NET create greenhouse gas sinks through anthropogenic activities. Sinks that are not directly influenced by humans are not included [19]. Most prominent NET are also known CCS technologies which include, for example, BECCS and DACCS. They can be classified as NET if CO_2 is removed from the atmosphere and negative emissions are generated. However, if the carbon comes from fossil sources, the net emissions are at best zero. In this case, it cannot be classified as NET. In general, CCS technologies can be divided into two steps: first, the capture and sequestration; and second, the storage of CO_2 . To capture and sequester CO_2 from gas mixtures in the first step, technologies are needed that separate CO_2 from gas mixtures. Four approaches (absorption, adsorption, cryogenic separation, and membrane separation) have already been tested and are used for CCS technologies [68–76]. The differences between those options in their effectiveness, robustness, energy requirements, and costs regarding their potential use for HyBECCS concepts are discussed below.

Absorption uses a solvent such as water or polyethylene glycol in which CO_2 is first dissolved and then released by changing the conditions [68]. Pressurized water scrubbing, for example, is used in biogas plants to separate CO_2 and H_2S from methane. In this process, water serves as an absorber for CO₂ through increased pressure. This absorber technology is considered comparatively robust and cost-effective [69,70]. Adsorption, in contrast, occurs at the surface of solids [76]. Again, CO_2 remains on the solid sorption material under certain conditions and can be regenerated by a change in temperature, pressure, or humidity [74]. A technically mature adsorption process is pressure swing adsorption on activated carbon. At a low temperature, of about 5 °C, and increased pressure, of about 2-7 bar, CO₂ accumulates on the surface of the activated carbon. The gas must first be free from impurities such as sulfur or dust for the activated carbon to achieve long lifetimes of up to 20 years [72]. Cryogenic separation takes advantage of the different condensation temperatures of the gases. For example, CO_2 condenses at -78.5 °C and hydrogen at -252 °C [77]. At very low temperatures the liquified CO₂ can easily be separated from other components still in a gaseous state. A very high purity of the recovered CO₂ fraction of 99.9% can be achieved, but this requires a comparatively high energy input. Membrane technology, in which the selective permeability of gases through certain material is exploited, is particularly suitable for gas mixtures that already contain a very high CO_2 content, of more than 90% [72–74]. Membrane technologies

are considered to be more environmentally friendly and cost-effective than the other options described [67,78]. This is due to the simpler modular design and lower energy requirements [76,78]. Challenges include selectivity, permeability, and sensitivity to sulfur compounds [68,74,79]. Further, the comparatively low purity of the CO_2 fraction after separation is to be mentioned. The difference in purity, however, moves in a wide range, from 80–99.9999%, and depends on the membrane technology chosen and the materials used [78,80,81].

In general, gas mixtures with a relatively low CO_2 concentration require technologies with higher selectivity to effectively separate CO₂ [82]. It is thermodynamically advantageous if the gas mixture to be separated already contains a large amount of CO_2 [83]. Thus, the CO_2 source has a strong influence on the choice of the separation process and the costs per captured tonnes of CO_2 . In this context, the initial situations are differentiated between the separation of CO_2 from ambient air or from so-called point sources, which already have a significantly higher CO_2 concentration than the ambient air. The possibility to capture CO_2 from ambient air is known as Direct Air Capture (DAC). The obvious advantage of DAC is that the CO_2 concentration in the atmosphere can directly be lowered and does not require point sources [84,85]. However, because the ambient air has a relatively low CO_2 concentration, of about 400 ppm, on average, the capture of CO_2 is thermodynamically less favorable than using point sources [83]. This is a reason why BECCS outperform, e.g., DACCS in terms of the primary energy required per tonne of carbon sequestered [86]. Chemical absorbers or adsorbers that undergo a chemical reaction with CO_2 are often used for DAC [71,72]. Because of the strong binding and selectivity, even the small amount of CO_2 in the air mixture can be filtered out. However, more energy must be expanded for the regeneration of the CO₂, and a large volume of air must be passed through the equipment to capture CO_2 in appreciable quantities. This results in high energy requirements and high costs for DAC [20,82,87]. If the high energy requirement for DAC is not carried out exclusively with renewable excess energy that does not compete for other uses, this can even lead to a positive emission balance instead of net negative emissions [20]. The second option is to capture CO_2 at point sources that already contain a higher concentration of CO_2 . These include, for example, exhaust gases from coal or gas-fired power plants, bioenergy plants, or other industrial facilities. The higher CO_2 concentration of the gas mixture allows a higher energy efficiency of gas separation [87]. However, if the gas mixture has a complex and pollutant composition, the separation can be very costly and technologically complex [69]. Therefore, post-combustion carbon capture (PCC), in which combustion exhaust gases are filtered, is in many cases cost-ineffective, because the heterogenous gas composition is harmful to the separation technology [72]. Additionally, it cannot be classified as NET if the carbon originates from fossil sources, as in the case of PCC at coal-fired power plants.

The second step of CCS technology is the permanent storage of the captured CO_2 . This can be done, for example, in geological formations such as aquifers [88]. Furthermore, there is the possibility to upgrade and use CO_2 for new products like plastics or lithium carbonate for accumulators. If the captured CO_2 is used, this is referred to as CCU technologies [89]. However, there is no consensus on whether this is a NET because of the limited lifetime of the products. It may only be a delay in emission [90]. But not only CO_2 can be used to achieve negative emissions. Residues from processed biomass can increase the carbon content of soil in so-called soil carbon sequestration and therefore lead to negative emissions [91,92]. Beyond the described CDR technologies, naturally occurring processes that can be influenced by humans, such as reforestation or enhanced weathering, also fall under the term NET. However, these are not relevant to the derivation of the introduced HyBECCS approach and are therefore not considered in this paper.

BECCS is highlighted in the IPCC Special Report as one of the key NET on achieving the 1.5 K target [19]. Additionally, the presented HyBECCS concept can be classified as a new BECCS technology approach. Therefore, BECCS technologies are discussed in more detail in the following paragraph.

With BECCS technologies, plant-based biomass is used for energy generation, and the CO_2 that is released can be captured and permanently removed from the atmosphere [93,94]. Regarding BECCS, three basic ways of generating energy from plant biomass are distinguished: the thermochemical conversion, the physicochemical conversion, and the biological conversion of biomass to solid, liquid, or gaseous energy sources [95]. One example of thermochemical conversion is incomplete combustion or pyrolysis. In this process, wood and stalk-type biomass are converted into coal, pyrolysis oil, and various gases [69]. The relative proportions of solids, liquid, and gas depend on the combustion parameters and residence time in the combustor [96]. Depending on the feedstock and process conditions, gas mixtures are produced that contain H_2 , CO, CH₄, and CO₂ as well as water vapor, nitrogen, particles, tar, hydrochloric acid, alkali chlorides, and sulfur compounds, as described in Section 3.2 [97,98]. The gas mixture must then be separated and processed for use as fuel [69,95–100]. Physicochemical conversion can be used for energy recovery from plantbased oils. They can be mechanically pressed or extracted from plants and then be chemically upgraded, for example by transesterification [101]. The resulting so-called biodiesel is a well-known product from physicochemical conversion processes, but research is also being conducted on the production of kerosene substitutes for aviation from algae oil, for example [102,103]. In biological conversion, biomass is decomposed by microorganisms and converted to methane, hydrogen, or alcohol, for example. A well-known biological conversion process is agricultural biogas production [94]. In conventional biogas plants, microorganisms produce a mixture consisting of 40–75% methane and 15–60% CO₂. Impurities such as H₂S are other possible components that can occur [57,75]. To use it as biogas, this mixture is separated. Biogas with high methane contents can be transported better, and its calorific value increases [68,75]. The CO₂ can be captured and permanently removed from the atmosphere in storage facilities, thus causing negative emissions. Moreover, the digested residue can be used as fertilizer, potentially leading to a further increase of the negative emission effect [92]. However, when the methane biogas is used, CO_2 is released into the atmosphere unless another CCS measure is implemented.

4. Basic Considerations for the HyBECCS Concept

In this section, the HyBECCS concept is described in detail, deriving main technological advantages that lead to lower costs and energy efficiency compared to competing technologies, such as DACCS and other BECCS processes.

The HyBECCS concept summarizes modular process combinations that can be split into four basic process steps (cf. Figure 1): the biomass pretreatment, the biohydrogen production, the product gas separation, and its processing. The authors suggest drawing the system border for economic and ecological classification and balancing around this four steps. Using a consistent system border will make it easier to compare different HyBECCS process combinations. For each of the four steps there are variable sub-processes which have either already reached market maturity or whose functionality has been demonstrated and tested at laboratory scale or prototype scale. Many of them are described in Section 3. However, a HyBECCS process combination has never been experimentally implemented as a whole process linking all of the four steps. Respective technological peculiarities, challenges, and opportunities in all of the four steps are discussed below. The subsequent processes of transportation and storage or utilization of CO₂ and options of hydrogen usage are not in the focus of consideration in this paper, since they have already been discussed and compared in other publications [104–107]. In addition, biomass production and transportation is not included in this work. The following diagram outlines the modularity of the HyBECCS approach described in the following.

Sustainability 2021, 13, 4026

81



Figure 1. Process diagram for carbon-negative hydrogen production using the Hydrogen Bioenergy with Carbon Capture and Storage (HyBECCS) approach.

For the key step of all HyBECCS approaches to biohydrogen production, the possible feedstocks are very diverse, as described in Section 3.2. However, upstream pretreatment processes (step 1) are required in most cases. For example, to be able to apply biotechnological fermentation processes, residual and waste materials containing sugar are usually pre-filtered and woody biomasses have to undergo a saccharification process. Substrates containing starch, fats, or proteins are subjected to hydrolysis pretreatment releasing single sugars, amino acids, and fatty acids for further microbial metabolism, as is common, for example, in most existing biogas production plants [108–113]. In addition, mechanical comminution or physical digestion methods are necessary. They are also needed for many thermochemical biohydrogen production processes. Moreover, thermochemical processes often require preparatory drying. One challenge of the pretreatment step is therefore the efficient processing of the widest possible range of residual and waste materials for use in biohydrogen processes. For the biohydrogen production (step 2), there is a choice of different thermochemical or biotechnological process options or combinations, as described in Section 3.2, that must be selected according to the specific characteristics of the available biomass streams and pretreatment options. Relevant for subsequent steps and the overall efficiency of the HyBECCS approach is the product gas composition resulting from the biohydrogen production process. In most cases, hydrogen and carbon dioxide are the main outputs [52,59–65]. The proportions of these two components should be maximized to exploit the ecological and economic potential and prepare for efficient gas separation (step 3). The lower the heterogeneity of the gas mixture to be separated and the higher the proportions of the main fractions H_2 and CO_2 , the better the conditions for efficient gas separation. Using biotechnological biohydrogen production processes, for example, the mixture usually does not contain combustion residues or other harmful gases that can negatively affect the downstream gas separation, as is the case with PCC or some thermochemical biohydrogen production processes [87]. To fractionate the product hydrogen for use and the carbon dioxide for downstream storage or use, a suitable gas separation process which can be carried out in various options is to be chosen, as described in Section 3.1. To separate H_2 and CO_2 , the physical and chemical prerequisites of an efficient separation are given, for example, due to the different polarities and molecular sizes. Membrane technologies, for example, are an attractive alternative to conventional processes, such as pressure swing adsorption or cryogenic distillation, due to their comparatively high separation quality and energy efficiency [67]. As described in Section 3.1, the disadvantage of DAC regarding the gas separation step is the low content of CO_2 in the ambient air, leading to higher energy consumption and high plant investment compared to the point source resulting from, e.g., biohydrogen production processes [83]. In the final step of processing

(step 4), the separated CO_2 has to be compressed or otherwise prepared for storage or use, and hydrogen has to be compressed or liquefied and purified so that it can either be fed into the natural gas network or transported to users, e.g., in composite pressure tanks. The carbon-negative hydrogen produced can be used materially for chemical processes or by means of fuel cells as energy carriers to generate heat and electricity or to power vehicles, as described in Section 3.1. Using fuel cells, the reaction with oxygen from the ambient air produces only water. In contrast to hydrocarbon-based e-fuels or bio-fuels such as biogas, no greenhouse gas emissions are emitted during its use. This results in an inherently higher CO_2 capture potential for HyBECCS, leading to a decisive advantage over BECCS technologies that aim to produce hydrocarbons as energy carriers. However, to use hydrogen in fuel cells, purity requirements must be met, which are specified in European standards, for example. The purity of hydrogen for its use in hydrogen fuel cells is specified therein at the value 3.7, which corresponds to a purity of 99.97% [114]. The following three points summarize the key advantages of HyBECCS concerning their environmental and economic potential:

- 1. The product gas composition of biohydrogen processes consisting mainly of H₂ as the main product and CO₂ as a co-product makes the capturing of CO₂ energy- and cost-effective compared to low concentrated CO₂ sources like ambient air [52,59–65,83,87,115].
- 2. The possibility to use biotechnological methods for biohydrogen production within HyBECCS approaches leads to carbon sources that are not polluted with combustion residues. In comparison to PCC, there is no risk to impair the subsequent separation or further processing [66,67,69,87].
- 3. Hydrogen is a carbon-free energy carrier. Since there is no carbon remaining in the product, HyBECCS approaches have a comparatively high cumulative CO₂ capture rate and thus a larger specific GHG emission reduction potential compared to, e.g., BECCS technologies producing hydrocarbon-based biofuels [36–39].

Due to these advantages, the HyBECCS concept can be seen as a new promising perspective within BECCS methods. However, to be able to make conclusive statements on process ecology and economics, more in-depth techno-economic analyses and ecological assessments must be carried out. The following section gives an insight into these assessments using the example of Germany.

5. General Perspectives for HyBECCS Approaches

Due to the advantages described in Section 4, HyBECCS offers a perspective for sustainable and efficient production of hydrogen as well as climate adaption and mitigation. Associated political regulatory measures such as CO₂ pricing and emissions trading could result in monetary credits from which the HyBECCS concept could benefit. Further, the global political and societal push toward establishing green hydrogen as a sustainable basic energy carrier across several sectors will have a positive effect for carbon-negative hydrogen production by better meeting the resulting demand. Furthermore, the increasing efforts of companies to leverage its externalities through carbon footprint improvements for business purposes in so-called social or green marketing concepts are also potential drivers of the HyBECCS concept [7,10,18,116,117]. An important basis for further development and industrial application of HyBECCS is a fundamental understanding of their economic and ecological impacts. In the following, from the example of Germany, possible impacts for climate adaptation and mitigation (Section 5.1) and the economic viability of HyBECCS concepts are discussed (Section 5.2).

5.1. Ecological Classification

The amount of sustainably available biomass is limited and should therefore be used as efficiently as possible. One promising way to do so is carbon-negative hydrogen production. The main ecological advantage of this possibility in terms of climate adaption and mitigation is the removal of CO_2 from the atmosphere. Compared to other NET, this can be done particularly efficiently by HyBECCS, as described in Section 4. In the

following, an estimate of the GHG emission reduction potential of HyBECCS in Germany is carried out, starting from bioenergy potentials already investigated for biohydrogen production through dark fermentation processes. Biohydrogen production as a standalone technology is already considered as a potentially climate-neutral process with a wide substrate spectrum. Among the options described in Section 3.2, biotechnological pathways allow for hydrogen production with less energy-consumption than thermochemical and electrochemical processes, which is due to lower process temperatures and pressures [46]. Besides, biotechnological hydrogen production has the advantage of offering a carbon source with high purity and no combustion residues, that could impair the subsequent separation or purification of CO₂, as described in Section 4. Thermochemical processes like pyrolysis or gasification are disadvantaged in these points. In the following, a special focus will therefore be placed on a biotechnological hydrogen production process as the basis for HyBECCS. In this investigation, the best-known and most widespread biotechnological approach, dark fermentation, is considered for the following ecological assessment. Further, the analysis is reduced to the environmental impact category global warming and to Germany as the spatial and systemic frame of reference.

The future potential of biohydrogen production from residuals through dark fermentation has been investigated in a recent study for Germany [118]. It should be noted that only biomass potentials from residues were taken into account in this consideration, which serves as a basis for the subsequent analysis. Social conflicts concerning land-use, for example through competitive situations with the production of food and feed, are thus not to be expected. For Germany, it was estimated that a theoretically achievable annual energy supply of 9.9 TWh/a up to 19.9 TWh/a will be possible in 2030 through dark fermentation of residual and waste materials. This is the same order of magnitude as that of Germany's targets for the production of hydrogen by electrolysis using renewable electricity. Within the national borders, these are set up to 14 TWh/a until 2030 [18]. Assuming that this energy amount of green hydrogen from dark fermentation substitutes an equal amount of energy provided by natural gas, this results in a saving of 2.0–4.0 MtCO₂,eq/a. In addition to natural gas, coal, and lignite, grey hydrogen or heating oil could also be substituted, which would lead to even greater savings due to higher specific GHG emissions related to the amount of energy produced [119]. Besides, the substitution of electricity from fossil sources is possible using hydrogen fuel cells. However, the replacement of natural gas serves as a plausible assumption, since hydrogen can be used directly in the mixture with natural gas in many cases without technological adjustments and can also be fed directly into natural gas distribution networks [18]. HyBECCS approaches further increase the GHG savings potential by creating negative emissions, as described in Section 4. Assuming that the entire amount of CO₂ produced as a by-product during dark fermentation processes were to be separated and stored, 6.4-13.0 MtCO2, eq/a of negative GHG emissions could be achieved in Germany by 2030. This assumption was based on a realistic stoichiometric-based product gas composition of fermentative biohydrogen production through the dark fermentation of CO_2 and H_2 in equal quantities, as can also be seen in Section 3.2 and Equations (4) and (5) [64–70]. In total, the use of carbon-negative hydrogen, under the assumptions made, results in a saving potential of 8.49-17.06 MtCO₂ eq/a due to CO₂ avoidance through fossil energy source substitution and its active removal from the atmosphere. In Figure 2, the estimated emission reduction potentials using HyBECCS are summarized.



Total emission reduction potential

Figure 2. Emission reduction potential through HyBECCS in Germany based on carbon-negative hydrogen production from residual biomass using dark fermentation.

The most important challenge in tackling climate change at a global level is the reduction of GHG emissions. However, a sole consideration of the GHG balance is not sufficient to explore the sustainability of HyBECCS approaches [115,120]. Further impact categories within life cycle assessments (LCA) according to DIN EN ISO 14040/44 are to be considered. Based on EEA classifications 6 and 10 for the chemical industry and agricultural products according to Grosse-Sommer et al., this also includes resource depletion, acidification, consumptive water use, eutrophication, human toxicity potential, photochemical ozone formation, ozone depletion, and land use [121]. Furthermore, the impact categories biodiversity loss and ecosystem services should also be considered [122]. However, this paper cannot make a final assessment taking into account all relevant impact categories. For a holistic view and final assessment, further investigations are still needed.

5.2. Economic Classification

The use of carbon-negative hydrogen can be economically attractive for companies, as it can provide a way to improve their corporate image, reducing the company's carbon footprints, and contribute to social or green marketing concepts. However, to be broadly applied, carbon-negative hydrogen must also be price-competitive, and its production must be profitable. The profitability depends on many internal and external impact factors, some of which are summarized in Figure 3. The internal impact factors describe influences that emanate directly from the technical setup and equipment used for the HyBECCS process combination as described in Section 4, leading to differences in CO₂ separation rates, hydrogen yield, energy efficiency, or plant investment costs, for example. Internal factors are necessary information for comprehensive techno-economic analyses. However, they are very difficult to estimate for HyBECCS due to the low level of technological maturity of the central process step of biohydrogen production and the lack of corresponding industrial-scale production technologies as data sources. Although further research in this regard has already been initiated by the authors, the information situation is not yet sufficient for a comprehensive analysis.



Figure 3. Impact factors on the profitability of carbon-negative hydrogen and HyBECCS.

External factors, in contrast, include forces that affect the competitiveness of carbonnegative hydrogen in the marketplace and are not directly dependent on the technical setup of the chosen process combination and technical setup. These include, for example, market or policy conditions affecting the price of hydrogen or the profitability of carbon-negative hydrogen production through financial incentives. A key external factor in this context are political regulatory measures from environmental and climate policies. Thereby, for the political regulation of GHG emissions via financial incentive systems, a distinction can be made between the taxation of CO_2 emissions and the incentivation of negative emissions establishing credits for NET. Both approaches have a fundamentally high impact on the economics of all NET [20] and carbon-negative hydrogen production. To provide a basis for initial quantitative assessments on economic prerequisites for HyBECCS, the maximum production costs required for competitive carbon-negative hydrogen are estimated in the following considering external factors. Therefore, the existing estimates of production costs for competing green hydrogen and expected incentives for the active removal of CO_2 from the atmosphere are carried out in the following, using the example of Germany.

The market price for hydrogen is a major external impact factor on the economics of carbon-negative hydrogen. HyBECCS plant operators must be able to produce at costs that are below this market price, which depends, among other things, on the general hydrogen demand and its supply situation. Forecasts predict a sharp increase in hydrogen demand in the coming decades [18,37]. Despite the extensive expansion of electrolysis capacities, there will be a large gap between demand and generation capacity in Germany that is to be closed by imports of green hydrogen according to their national hydrogen strategy, as described in Section 3.1. Therefore, a distinction regarding hydrogen prices in Germany can be made between imported green hydrogen and green hydrogen that is produced in the internal market. Since internal demand is very unlikely to be met by the production of green hydrogen by electrolysis within national borders [18], there will be no direct competition between green hydrogen and carbon-negative hydrogen within Germany. The production costs for carbon-negative hydrogen should therefore be measured against the price of imported green hydrogen. Accomplished studies have estimated the price for hydrogen produced in North Africa and transported to Germany either via pipelines or by ship at 3.10 € per kg in 2050. Despite the lower costs for renewable energy, this is more expensive compared to hydrogen produced in the domestic market (2.45 €). Average

transport costs are estimated at 1.70 per kg H₂ [123]. Other studies come up with supply costs of up to 119 \notin /MWh with the example of importing green hydrogen from North Africa to Central Europe, which equals $3.97 \notin$ per kg H₂ [124]. Further studies predict production costs between $3.85 \notin$ and $4.81 \notin$ per kg H₂ from solar energy in Morocco [125], amounting to supply costs of $5.55 \notin$ and $6.51 \notin$ including transport costs, according to [123].

Additionally, credits for CCS should be considered to assess the economics of NET like carbon-negative hydrogen production. The introduction of policy instruments that incentivize NET deployment by generating revenues linked to the amount of CO₂ captured and safely stored is likely to be seen [20]. The expected potential credits per tonne of CO_2 taken from the atmosphere through HyBECCS are estimated in the following. Worldwide, the so-called social costs of carbon (SCC) serve as benchmarks for pricing or tax models. These correspond to the damage caused by an additional tonne of CO_2 to the economy and society and were first defined by Nordhaus [126]. In recent decades, numerous quantitative estimates of SCC have been made. Based on several available models, the National Academies of Sciences, Engineering, and Medicine quantify the SCC for one tonne of CO₂ at an average of 46 \in in 2020 and 55 \in in 2030 [127]. These figures are based on well-documented empirical assessments of climate damage and can be considered conservative [128]. However, there is uncertainty about the extent of the expected warming, damage, and risks. An expert survey on expected climate damage with 386 respondents from different disciplines placed the average estimate of SCC at 70–90 \notin per ton CO₂. Other investigations from the German Environment Agency (GEA) consider an SCC of more than 180 € per tonne of CO₂ realistic [129,130]. Governments around the world are basing their statutory CO₂ prices on these calculations. In Germany, for example, since January 2021, the emissions of carbon dioxide in the building and transport sectors have been affected by a fixed price of 25 \in per tonne of CO₂, thus covering areas that are not yet affected by the European emissions trading system. This uniform price per tonne of CO_2 emitted is to rise gradually, to 30 \in per tonne in January 2022, 35 \in in 2023, 45 \in in 2024, and 55 \notin in 2025 [131]. Switzerland, on the other hand, has been charging 96 Swiss francs per tonne of CO_2 since 2018, equivalent to 86.50 \in , and has passed legislation that could raise the price to 210 Swiss francs per tonne of CO₂, equivalent to 189.22 € [132].

An estimate of the impact on the production of one kilogram of carbon-negative hydrogen in Germany is based in the following on the assumption that a credit per tonne of CO₂ actively extracted from the atmosphere is made at the level of SCC or existing CO₂ taxes. For this, a worst-case and best-case distinction is made. In the worst-case path considered, we use the agreed price in Germany of $55 \notin$ per tonne of CO₂, applicable from 2025 [131], and for the best-case scenario the proposed price of 180 \notin per tonne of CO₂ of the GEA [130] is assumed. Provided that the use of 1 mol (0.002 kg) of carbon-negative hydrogen is associated with the capture and storage of 1 mol of CO₂ (0.044 kg), the expected credits that could be paid per kg of carbon-negative hydrogen can be estimated. For the worst-case scenario, this results in an incentive of $1.20 \notin$, and for the supply price of competitive green hydrogen, a distinction is made between a minimum case of $3.10 \notin$ and a maximum case of $6.51 \notin$ [123,125]. In the figure below, the CCS incentive forecasts, together with the estimated production costs for imported green hydrogen, are shown.

As shown in Figure 4, the incentives for CO_2 capture and storage can have a major impact on the economics of the HyBECCS concept. The sum of the estimated target prize per kg of carbon-negative hydrogen and CCS incentives gives a basic value from which to draw initial conclusions about the maximum production costs of carbon-negative hydrogen to be competitive in the future hydrogen market in Germany. Under the assumptions made, the production costs would have to be below $4.30 \in$ per kg of carbon-negative hydrogen in the worst case and below $10.44 \in$ in the best case to be able to compete with imported green hydrogen. To date, no comparative estimates of the production costs of carbon-negative hydrogen have been available, but this first classification sets a target for cost-effectiveness analyses, which should consider both external and internal impact factors.



Reference price per kg imported green hydrogen

Figure 4. Pricing fundamentals for carbon-negative hydrogen considering the supply costs of competing imported green hydrogen and the expected incentives for CO₂ capture and storage in Germany.

6. Limitations

The HyBECCS concept described in this paper comprises a modular-based system consisting of variable sub-processes. Many of them are already known and tested. However, a complete HyBECCS process combination has never been implemented in a contiguous facility. Their testing in pilot and experimental facilities is still pending and could lead to new findings or process challenges that are not described in this paper. The concept's classifications of ecology (Section 5.1) and economics (Section 5.2) described in this paper are based on simplifications and assumptions made by the authors in an effort to make a conservative estimate. For the ecological assessment, only single-stage dark fermentation, the exclusive use of biogenic residues, a focus on the densely populated country (Germany), and the use of H_2 for the substitution of natural gas were assumed to give a realistic estimate and in order not to create unrealistic expectations. On the other hand, possible emissions connected to farming and the pretreatment of biomass, as well as during carbon storage, which would have an effect on the emission balance, were not taken into account in this paper. In the economic estimation, possible financial benefits through the use of CO_2 as raw material were excluded, as well as possible financial benefits from soil carbon sequestration. Furthermore, carbon-negative hydrogen was compared with imported green hydrogen, although a better carbon footprint of carbon-negative hydrogen is likely to lead to a market advantage over green hydrogen. Nevertheless, process-related peculiarities that are still unknown due to the low technology maturity can lead to negative effects on the economic viability of the HyBECCS approach. The assumptions made need to be reviewed and validated in further research. Additionally, a detailed techno-economic analysis of the process concepts, testing in pilot and experimental plants as well as holistic life-cycle assessments are needed. Then, more precise statements about the potential of carbon-negative hydrogen and the HyBECCS approach will be possible.

7. Summary

In this paper, the HyBECCS (hydrogen bioenergy with carbon capture and storage) concept was introduced as a new approach for NET. This concept combines hydrogen

production from renewable biomass with downstream processes for CO₂ capture and storage or use. The hydrogen produced has a negative carbon footprint and is referred to as carbon-negative hydrogen. HyBECCS processes are described as modular process combination options that can be assigned to four basic steps: the pre-treatment of the substrate biomass, the biohydrogen production process, the separation of hydrogen and CO_2 from the product gas, and their treatment. All steps are interdependent and can be performed by different technology options, some of which are described in this paper. Although HyBECCS has never been technically implemented as a physically connected process combination, there are technology options for each of the four steps that have either already reached market maturity or whose functionality has been proven and tested on a laboratory or prototype scale. The main advantage of HyBECCS over competing NETs can be seen in the efficient capture of CO_2 with high purity produced as point source in a comparatively high concentration during the second step, biohydrogen production. Therefore, technological disadvantages arising either from the contamination of CO_2 with combustion residues or from the energy- and cost-intensive concentration of CO₂ from heterogeneous gas mixtures can be avoided. Further, hydrogen can either be used in the chemical industry, for steel production, or to generate emission-free heat and electricity, in contrast to GHG-emitting hydrocarbon-based energy carriers like biogas or other biofuels produced, for example, from biomass within BECCS approaches. Due to these advantages, the ecological and economic benefits of the HyBECCS concept can be concluded. These benefits were analyzed using the example of Germany. An ecological assessment was made estimating the GHG emission reduction potential through HyBECCS approaches. For the production and use of carbon-negative hydrogen, a theoretical saving potential of 8.49-17.06 MtCO₂,eq/a was estimated for the year 2030 in Germany based on existing estimations of biohydron production through dark fermentation processes. This results from the potential combination of CO_2 avoidance by substitution of natural gas with hydrogen as an energy source (2.00–4.02 MtCO₂ eq/a) and the active extraction of CO₂ from the atmosphere (6.49-13.04 MtCO₂, eq/a). In addition, to provide a basis for initial estimates of the economics of the HyBECCS concept, the maximum costs for competitive carbon-negative hydrogen production were estimated for Germany in two scenarios. For this purpose, existing estimations for the production costs of the competing product green hydrogen were added to potential monetary incentives for the extraction of CO_2 from the atmosphere. According to these estimations, the production costs of carbon-negative hydrogen would have to be below 4.30 € per kg of carbon-negative hydrogen in the worstcase and below 10.44 € in the best-case scenario to be able to compete with imported green hydrogen in Germany. In this paper, the definition and differentiation of the HyBECCS concept for sustainable energy solutions as well as initial potential assessments have set the basis for a more in-depth consideration of this novel approach.

8. Outlook

In further investigations, the HyBECCS concept presented in this paper should be analyzed in more detail so as to identify particularly promising process combinations and prepare for experimental research and technological implementation. A holistic techno-economic and ecological understanding of the HyBECCS process's combination possibilities must therefore be gradually developed in order to drive their development and optimization in a targeted manner. To ensure the ecology of the processes, relevant environmental impacts must be identified and analyzed on this basis. A holistic life cycle assessment following DIN EN ISO 14040/44 for the production of carbon-negative hydrogen through HyBECCS should be established. In addition, cross-impact analyses and model calculations on the overall economic implementation in the energy system and industry should be carried out with a focus on intersectoral system developments, e.g., by means of material and energy flow system model analyses. However, to bring carbon-negative hydrogen into widespread use, not only economic and ecological optimization is needed. Social and political framework conditions must also be adapted. A robust

quantification of negative emissions under international supervision is required. Further, social and environmental conflicts, in particular land and water use conflicts, must be prevented. Establishing a certification system for carbon-negative hydrogen, which can ensure that the product has a negative carbon footprint, could help to ensure widespread and long-term acceptance.

From a technical point of view, HyBECCS process combinations can be combined with bioeconomic biorefinery concepts in perspective. An expanded range of products and substrates could lead to higher value creation and improve economic efficiency. The main product gas fractions H_2 and CO_2 could therefore be used in a combinatorial way, for example by using H_2 as a reducing agent for CO_2 , as outlined in Figure 5. Ecoefficient carbon cycles with high added value can thus be created based on the presented HyBECCS approach.



Figure 5. Exemplary scheme for the expansion of the product range to create a biorefinery concept.

However, the most important basis for both HyBECCS and bioeconomy approaches is a deeper techno-economic understanding as well as a comprehensive life-cycle assessment of the basic process step of biohydrogen production from biomass.

Author Contributions: Writing, J.F. and S.M.; supervision, R.M. and A.S. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the German Federal Ministry for Economic Affairs and Energy and the Ministry of Rural Affairs and Consumer Protection of Baden-Wuerttemberg, Germany.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Acknowledgments: The research shown in this paper was conducted within the projects "RhoTech" and "HyBECCS-BW". The authors gratefully acknowledge the financial support of the German Federal Ministry for Economic Affairs and Energy and the Ministry of Rural Affairs and Consumer Protection of Baden-Wuerttemberg, Germany.

Conflicts of Interest: The authors declare no conflict of interest

References

- 1. United Nations. *Paris Agreement*. 2015. Available online: https://unfccc.int/sites/default/files/english_paris_agreement.pdf (accessed on 2 March 2021).
- 2. Commission to the Euopean Parliament, The European Council, The European Economic and Social Committee, Committee of the Regions. *The European Green Deal*; European Commission: Brussels, Belgium, 2019.
- Bundesministerium f
 ür Umwelt, Naturschutz und Nukleare Sicherheit. Klimaschutzprogramm 2030 der Bundesregierung zur Umsetzung des Klimaschutzplans 2050. Available online: https://www.bundesregierung.de/breg-de/themen/klimaschutz/ massnahmenprogramm-klima-1679498 (accessed on 31 January 2021).
- 4. Bundesministerium für Umwelt, Naturschutz und Nukleare Sicherheit. Fact Sheet Klimaschutzgesetz. Available online: https://www.bmu.de/gesetz/bundes-klimaschutzgesetz/ (accessed on 31 January 2021).
- 5. World Resources Institute. Climate Watch Historical GHG Emissions. Available online: https://www.climatewatchdata.org/ghgemissions (accessed on 13 January 2021).
- 6. Umweltbundesamt. Energiebedingte Emissionen. Available online: https://www.umweltbundesamt.de/daten/energie/ energiebedingte-emissionen#energiebedingte-treibhausgas-emissionen (accessed on 13 January 2021).
- Dambeck, H.; Ess, F.; Falkenberg, H.; Kemmler, A.; Kirchner, A.; Kreidelmeyer, S.; Lübbers, S.; Piégsa, A.; Scheffer, S.; Spillmann, T.; et al. Klimaneutrales Deutschland: Studie im Auftrag von Agora Energiewende, Agora Verkehrswende und Stiftung Klimaneutralität. 2020. Available online: https://static.agora-energiewende.de/fileadmin/Projekte/2020/2020_10_KNDE/A-EW_192_KNDE_Zusammenfassung_DE_WEB.pdf (accessed on 13 January 2021).
- Wyns, T.; Khandekar, G.; Robson, I. Industrial Value Chain-A Bridge towards a Carbon Neutral Europe. Available online: https://static.agora-energiewende.de/fileadmin/Projekte/2020/2020_10_KNDE/A-EW_192_KNDE_Zusammenfassung_ DE_WEB.pdf (accessed on 3 April 2021).
- 9. Schneider, C.; Lechtenböhmer, S.; Bauer, T.; Nitz, P.; Hettesheimer, T.; Wietschel, M.; Meulenberg, W.; Gurtner, R. Low-Carbon-Industrie: Elektrifizierung und geschlossene Kohlenstoffkreisläufe. In *Innovationen für die Energiewende: Beiträge zur FVEE-Jahrestagung 2017*; Forschungsverbund Erneuerbare Energien: Berlin, Germany, 2017; pp. 38–48.
- Witecka, W.; Hauser, P.; Sartor, O. Breakthrough Strategies for Climate-Neutral Industry in Europe (Summary): Policy and Technology Pathways for Raising EU Climate Ambition. Available online: https://static.agora-energiewende.de/fileadmin/Projekte/2020/2020_10 _Clean_Industry_Package/A-EW_197_Strategies-Climate-Neutral-Industry-EU_Summary_WEB.pdf. (accessed on 3 April 2021).
- 11. Bills, A.; Sripad, S.; Fredericks, W.L.; Singh, M.; Viswanathan, V. Performance Metrics Required of Next-Generation Batteries to Electrify Commercial Aircraft. ACS Energy Lett. 2020, 5, 663–668. [CrossRef]
- 12. Sutter, D.; van der Spek, M.; Mazzotti, M. 110th Anniversary: Evaluation of CO₂-Based and CO₂-Free Synthetic Fuel Systems Using a Net-Zero-CO 2 -Emission Framework. *Ind. Eng. Chem. Res.* **2019**, *58*, 19958–19972. [CrossRef]
- 13. Lindstad, E.; Eskeland, G.S.; Rialland, A.; Valland, A. Decarbonizing Maritime Transport: The Importance of Engine Technology and Regulations for LNG to Serve as a Transition Fuel. *Sustainability* **2020**, *12*, 8793. [CrossRef]
- 14. Hepperle, M. Electric Flight-Potential and Limitations. In Proceedings of the Energy Efficient Technologies and Concepts of Operation, Lisbon, Portugal, 22–24 October 2012.
- Hall, D.; Pavlenko, N.; Lutsey, N. Beyond Road Vehicles: Survey of Zero-Emission Technology Options across the Transport Sector. Available online: https://theicct.org/sites/default/files/publications/Beyond_Road_ZEV_Working_Paper_20180718.pdf (accessed on 22 January 2021).
- 16. Bajpai, R.P.; Chandrasekhar, U. (Eds.) *Innovative Design and Development Practices in Aerospace and Automotive Engineering*; Springer: Singapore, 2017; ISBN 978-981-10-1770-4.
- 17. Staiger, R.; Tanțău, A. (Eds.) *Geschäftsmodellkonzepte Mit Grünem Wasserstoff*; Springer Fachmedien Wiesbaden: Wiesbaden, Germany, 2020; ISBN 978-3-658-30575-8.
- Bundesministerium f
 ür Wirtschaft und Energie. Die Nationale Wasserstoffstrategie. 2020. Available online: https://www.bmwi.de/ Redaktion/DE/Publikationen/Energie/die-nationale-wasserstoffstrategie.pdf?__blob=publicationFile&v=16 (accessed on 30 July 2020).
- 19. Masson-Delmotte, V.; Pörtner, H.O.; Skea, J. (Eds.) *Global Warming of* 1.5 °*C: An IPCC Special Report on the Impacts of Global Warming of* 1 °*C above Pre-Industrial Levels and Related Global Greenhouse Gas Emission Pathways, in the Context of Strengthening the Global Response to the Threat of Climate Change, Sustainable Development, and Efforts to Eradicate Poverty; Summary for Policymakers;* IPCC: Geneva, Switzerland, 2018; ISBN 978-92-9169-151-7.
- 20. Honegger, M.; Reiner, D. The political economy of negative emissions technologies: Consequences for international policy design. *Clim. Policy* **2018**, *18*, 306–321. [CrossRef]
- 21. Department of Energy. *Hydrogen Program Plan;* 2020. Available online: https://www.hydrogen.energy.gov/pdfs/hydrogen-program-plan-2020.pdf (accessed on 11 February 2021).
- 22. Government of Canada. *Hydrogen Strategy for Canada: Seizing the Opportunities for Hydrogen. A Call to Action;* 2020. Available online: https://www.nrcan.gc.ca/sites/nrcan/files/environment/hydrogen/NRCan_Hydrogen%20Strategy%20for%20 Canada%20Dec%2015%202200%20clean_low_accessible.pdf (accessed on 11 February 2021).

- 23. Government of France. *Stratégie Nationale Pour le Développement de L'hydrogène Décarboné en France;* 2020. Available online: https://minefi.hosting.augure.com/Augure_Minefi/r/ContenuEnLigne/Download?id=5C30E7B2-2092-4339-8B92-FE249 84E8E42&filename=DP%20-%20Strat%C3%A9gie%20nationale%20pour%20le%20d%C3%A9veloppement%20de%20l%27 hydrog%C3%A8ne%20d%C3%A9carbon%C3%A9%20en%20France.pdf (accessed on 11 February 2021).
- 24. Ministry of Economy, Trade and Industry. *Japan's Basic Hydrogen Strategy Determined*; 2017. Available online: https://www.meti. go.jp/english/press/2017/1226_003.html (accessed on 11 February 2021).
- 25. Minister for Climate and Environment; Minister for Petroleum and Energy. *The Norwegian Government's Hydrogen Strategy: Towards a Low Emission Society*; 2020. Available online: https://www.government.nl/documents/publications/2020/04/06 /government-strategy-on-hydrogen (accessed on 11 February 2021).
- 26. Government of The Netherlands. *Government Strategy on Hydrogen;* 2020. Available online: https://www.government.nl/documents/publications/2020/04/06/government-strategy-on-hydrogen (accessed on 11 February 2021).
- 27. COAG Energy Council. *Australia's National Hydrogen Strategy*; 2019. Available online: https://www.industry.gov.au/data-and-publications/australias-national-hydrogen-strategy (accessed on 11 February 2021).
- 28. Ministry of Energy, Government of Chile. *National Green Hydrogen Strategy: Chile, a Clean Energy Provider for a Carbon Neutral Planet;* 2020. Available online: https://energia.gob.cl/h2 (accessed on 11 February 2021).
- 29. A Sustainable Bioeconomy for Europe: Strengthening the Connection between Economy, Society and the Environment: Updated Bioeconomy Strategy; Publications Office of the European Union: Luxembourg, 2018; ISBN 978-92-79-94144-3.
- 30. Kuckertz, A. Bioeconomy Transformation Strategies Worldwide Require Stronger Focus on Entrepreneurship. *Sustainability* **2020**, *12*, 2911. [CrossRef]
- Bundesministeriumfür Bildung und Forschung (BMBF) Bundesministerium für Ernährung und Landwirtschaft. Nationale Bioökonomiestrategie. March 2020. Available online: https://www.bmbf.de/upload_filestore/pub/BMBF_Nationale_ Biooekonomiestrategie_Langfassung_deutsch.pdf (accessed on 30 July 2020).
- Miehe, R.; Bauernhansl, T.; Beckett, M.; Brecher, C.; Demmer, A.; Drossel, W.-G.; Elfert, P.; Full, J.; Hellmich, A.; Hinxlage, J.; et al. The biological transformation of industrial manufacturing—Technologies, status and scenarios for a sustainable future of the German manufacturing industry. J. Manuf. Syst. 2020, 54, 50–61. [CrossRef]
- 33. Bauernhansl, T.; Brecher, C.; Drossel, W.-G.; Hompel, M.T.; Wolperdinger, M. *Biointelligenz: Eine Neue Perspektive für Nachhaltige Industrielle Wertschöpfung: Ergebnisse der Voruntersuchung zur Biologischen Transformation der Industriellen Wertschöpfung (BIOTRAIN);* Fraunhofer Verlag: Stuttgart, Germany, 2019; ISBN 978-3-8396-1433-4.
- 34. Dietz, T.; Börner, J.; Förster, J.; von Braun, J. Governance of the Bioeconomy: A Global Comparative Study of National Bioeconomy Strategies. *Sustainability* **2018**, *10*, 3190. [CrossRef]
- 35. Full, J.; Miehe, R.; Kiemel, S.; Bauernhansl, T.; Sauer, A. The Biological Transformation of Energy Supply and Storage— Technologies and Scenarios for Biointelligent Value Creation. *Procedia Manuf.* **2019**, *39*, 1204–1214. [CrossRef]
- Staiger, R.; Tanțău, A. Analyse von Umwelt- und ökonomischen Merkmalen von grünem Wasserstoff. In Geschäftsmodellkonzepte mit grünem Wasserstoff; Springer: Cham, Switzerland, 2020; pp. 47–90.
- 37. Hebling, M.C.; Ragwitz, T.; Fleiter, U.; Groos, D.; Härle, A.; Held, M.; Jahn, N.; Müller, T.; Pfeifer, P.; Plötz, O.; et al. Wietschel. In *Eine Wasserstoff-Roadmap für Deutschland*; Fraunhofer-Institut für System- und Innovationsforschung ISI: Freiburg, Germany; Karlsruhe and Fraunhofer-Institut für Solare Energiesysteme ISE: Freiburg, Germany, 2019.
- Prognos; Öko-Institut; Wuppertal-Institut. Towards a Climate-Neutral Germany. Available online: https://static.agoraenergiewende.de/fileadmin/Projekte/2020/2020_10_KNDE/A-EW_193_KNDE_Executive-Summary_EN_WEB_V111.pdf. (accessed on 3 April 2021).
- 39. Osman, A.I.; Deka, T.J.; Baruah, D.C.; Rooney, D.W. Critical challenges in biohydrogen production processes from the organic feedstocks. *Biomass Conv. Bioref.* **2020**. [CrossRef]
- 40. Bundesministerium für Wirtschaft und Energie. Was ist Eigentlich Grüner Wasserstoff? Available online: https://www.bmwienergiewende.de/EWD/Redaktion/Newsletter/2020/07/Meldung/direkt-erklaert.html (accessed on 31 January 2021).
- 41. Tian, H.; Li, J.; Yan, M.; Tong, Y.W.; Wang, C.-H.; Wang, X. Organic waste to biohydrogen: A critical review from technological development and environmental impact analysis perspective. *Appl. Energy* **2019**, 256, 113961. [CrossRef]
- 42. Abdoulmoumine, N.; Adhikari, S.; Kulkarni, A.; Chattanathan, S. A review on biomass gasification syngas cleanup. *Appl. Energy* **2015**, *155*, 294–307. [CrossRef]
- 43. Arregi, A.; Amutio, M.; Lopez, G.; Bilbao, J.; Olazar, M. Evaluation of thermochemical routes for hydrogen production from biomass: A review. *Energy Convers. Manag.* 2018, *165*, 696–719. [CrossRef]
- 44. Inayat, A.; Tariq, R.; Khan, Z.; Ghenai, C.; Kamil, M.; Jamil, F.; Shanableh, A. A comprehensive review on advanced thermochemical processes for bio-hydrogen production via microwave and plasma technologies. *Biomass Conv. Bioref.* **2020**. [CrossRef]
- 45. Chang, P.-L.; Hsu, C.-W.; Lin, C.-Y.; Hsiung, C.-M. Constructing a new business model for fermentative hydrogen production from wastewater treatment. *Int. J. Hydrogen Energy* **2011**, *36*, 13914–13921. [CrossRef]
- 46. Levin, D.B.; Chahine, R. Challenges for renewable hydrogen production from biomass. *Int. J. Hydrogen Energy* **2009**, *35*, 4962–4969. [CrossRef]
- 47. Ntaikou, I.; Antonopoulou, G.; Lyberatos, G. Biohydrogen Production from Biomass and Wastes via Dark Fermentation: A Review. *Waste Biomass Valor* **2010**, *1*, 21–39. [CrossRef]

- 48. Wang, S.; Zhang, T.; Bao, M.; Su, H.; Xu, P. Microbial Production of Hydrogen by Mixed Culture Technologies: A Review. *Biotechnol. J.* **2020**, *15*, e1900297. [CrossRef]
- 49. Łukajtis, R.; Hołowacz, I.; Kucharska, K.; Glinka, M.; Rybarczyk, P.; Przyjazny, A.; Kamiński, M. Hydrogen production from biomass using dark fermentation. *Renew. Sustain. Energy Rev.* **2018**, *91*, 665–694. [CrossRef]
- 50. Jabbari, B.; Jalilnejad, E.; Ghasemzadeh, K.; Iulianelli, A. Recent Progresses in Application of Membrane Bioreactors in Production of Biohydrogen. *Membranes* **2019**, *9*, 100. [CrossRef]
- 51. Lin, C.-Y.; Nguyen, T.M.-L.; Chu, C.; Leu, H.-J.; Lay, C.-H. Fermentative biohydrogen production and its byproducts: A mini review of current technology developments. *Renew. Sustain. Energy Rev.* **2018**, *82*, 4215–4220. [CrossRef]
- 52. Lu, C.; Zhang, H.; Zhang, Q.; Chu, C.; Tahir, N.; Ge, X.; Jing, Y.; Hu, J.; Li, Y.; Zhang, Y.; et al. An automated control system for pilot-scale biohydrogen production: Design, operation and validation. *Int. J. Hydrogen Energy* **2020**, *45*, 3795–3806. [CrossRef]
- 53. Singh, A.; Rathore, D. *Biohydrogen Production: Sustainability of Current Technology and Future Perspective*; Springer India: New Delhi, India, 2017; ISBN 978-81-322-3575-0.
- 54. Liu, Y.; Min, J.; Feng, X.; He, Y.; Liu, J.; Wang, Y.; He, J.; Do, H.; Sage, V.; Yang, G.; et al. A Review of Biohydrogen Productions from Lignocellulosic Precursor via Dark Fermentation: Perspective on Hydrolysate Composition and Electron-Equivalent Balance. *Energies* **2020**, *13*, 2451. [CrossRef]
- 55. Dimitrellos, G.; Lyberatos, G.; Antonopoulou, G. Does Acid Addition Improve Liquid Hot Water Pretreatment of Lignocellulosic Biomass towards Biohydrogen and Biogas Production? *Sustainability* **2020**, *12*, 8935. [CrossRef]
- Mishra, P.; Krishnan, S.; Rana, S.; Singh, L.; Sakinah, M.; Ab Wahid, Z. Outlook of fermentative hydrogen production techniques: An overview of dark, photo and integrated dark-photo fermentative approach to biomass. *Energy Strategy Rev.* 2019, 24, 27–37. [CrossRef]
- 57. Bharathiraja, B.; Sudharsanaa, T.; Bharghavi, A.; Jayamuthunagai, J.; Praveenkumar, R. Biohydrogen and Biogas—An overview on feedstocks and enhancement process. *Fuel* **2016**, *185*, 810–828. [CrossRef]
- 58. Mansih, S.; Banerjee, R. Comparison of biohydrogen production processes. Int. J. Hydrogen Energy 2008, 33, 279–286. [CrossRef]
- 59. Bakonyi, P.; Kumar, G.; Nemestóthy, N.; Lin, C.Y.; Bélafi-Bakó, K. Biohydrogen purification using a commercial polyimide membrane module: Studying the effects of some process variables. *Int. J. Hydrogen Energy* **2013**, *38*, 15092–15099. [CrossRef]
- Bakonyi, P.; Nemestóthy, N.; Bélafi-Bakó, K. Biohydrogen purification by membranes: An overview on the operational conditions affecting the performance of non-porous, polymeric and ionic liquid based gas separation membranes. *Int. J. Hydrogen Energy* 2013, *38*, 9673–9687. [CrossRef]
- 61. Chen, C.; Wang, L.; Xiao, G.; Liu, Y.; Xiao, Z.; Deng, Q.; Yao, P. Continuous acetone-butanol-ethanol (ABE) fermentation and gas production under slight pressure in a membrane bioreactor. *Bioresour. Technol.* **2014**, *163*, 6–11. [CrossRef]
- 62. Delgado, J.A.; Águeda, V.I.; Uguina, M.A.; Brea, P.; Grande, C.A. Comparison and evaluation of agglomerated MOFs in biohydrogen purification by means of pressure swing adsorption (PSA). *Chem. Eng. J.* **2017**, *326*, 117–129. [CrossRef]
- 63. La Licata, B.; Sagnelli, F.; Boulanger, A.; Lanzini, A.; Leone, P.; Zitella, P.; Santarelli, M. Bio-hydrogen production from organic wastes in a pilot plant reactor and its use in a SOFC. *Int. J. Hydrogen Energy* **2011**, *36*, 7861–7865. [CrossRef]
- 64. Wang, Y.; Joshee, N.; Cao, W.; Wu, Q.; Tahir, N. Continuous hydrogen production by dark and photo co-fermentation using a tubular multi-cycle bio-reactor with Paulownia biomass. *Cellulose* **2019**, *26*, 8429–8438. [CrossRef]
- 65. Krupp, M.; Widmann, R. Biohydrogen production by dark fermentation: Experiences of continuous operation in large lab scale. *Int. J. Hydrogen Energy* **2009**, *34*, 4509–4516. [CrossRef]
- 66. Bakonyi, P.; Bogdán, F.; Kocsi, V.; Nemestóthy, N.; Bélafi-Bakó, K.; Buitrón, G. Investigating the effect of hydrogen sulfide impurities on the separation of fermentatively produced hydrogen by PDMS membrane. *Sep. Purif. Technol.* **2016**, *157*, 222–228. [CrossRef]
- 67. Gouveia, A.S.L.; Ventaja, L.; Tomé, L.C.; Marrucho, I.M. Towards Biohydrogen Separation Using Poly(Ionic Liquid)/Ionic Liquid Composite Membranes. *Membranes* **2018**, *8*, 124. [CrossRef] [PubMed]
- 68. Basu, S.; Khan, A.L.; Cano-Odena, A.; Liu, C.; Vankelecom, I.F.J. Membrane-based technologies for biogas separations. *Chem. Soc. Rev.* 2010, *39*, 750–768. [CrossRef] [PubMed]
- 69. Kaltschmitt, M.; Hartmann, H.; Hofbauer, H. *Energie aus Biomasse: Grundlagen, Techniken und Verfahren*; 2., Neu Bearbeitete und Erw. Aufl.; Springer: Heidelberg, Germany; New York, NY, USA, 2009; ISBN 978-3-540-85095-3.
- 70. Ofori-Boateng, C.; Kwofie, E. Water Scrubbing: A Better Option for Biogas Purification for Effective Storage. *World Appl. Sci. J.* **2009**, *5*, 122–125.
- 71. Fasihi, M.; Efimova, O.; Breyer, C. Techno-economic assessment of CO2 direct air capture plants. *J. Clean. Prod.* **2019**, 224, 957–980. [CrossRef]
- Viebahn, P.; Horst, J.; Scholz, A.; Zelt, O. *Technologiebericht 4.4 Verfahren der CO₂-Abtrennung aus Faulgasen und Umgebungsluft*; Wuppertal Institut f
 ür Klima, Umwelt, Energie: Wuppertal, Germany, 2018.
- 73. Xu, G.; Liang, F.; Yang, Y.; Hu, Y.; Zhang, K.; Liu, W. An Improved CO2 Separation and Purification System Based on Cryogenic Separation and Distillation Theory. *Energies* **2014**, *7*, 3484–3502. [CrossRef]
- 74. Olajire, A.A. CO₂ capture and separation technologies for end-of-pipe applications—A review. *Energy* **2010**, *35*, 2610–2628. [CrossRef]
- 75. Chen, X.Y.; Vinh-Thang, H.; Ramirez, A.A.; Rodrigue, D.; Kaliaguine, S. Membrane gas separation technologies for biogas upgrading. *RSC Adv.* **2015**, *5*, 24399–24448. [CrossRef]

- 76. Zhang, Z.; Zhang, W.; Lichtfouse, E. *Membranes for Environmental Applications*; Springer International Publishing: Cham, Switzerland, 2020; ISBN 978-3-030-33977-7.
- 77. Zhang, Y.; Evans, J.R.G.; Yang, S. Corrected Values for Boiling Points and Enthalpies of Vaporization of Elements in Handbooks. J. Chem. Eng. Data 2011, 56, 328–337. [CrossRef]
- 78. Burkhanov, B.G.S.; Gorina, N.B.; Kolchugina, N.B.; Roshan, N.R.; Slovetsky, D.I.; Chistov, E.M. Palladium-Based Alloy Membranes for Separation of High Purity Hydrogen from Hydrogen-Containing Gas Mixtures. *Platin. Metals Rev.* 2011, 55, 3–12. [CrossRef]
- 79. Rezakazemi, M.; Ebadi Amooghin, A.; Montazer-Rahmati, M.M.; Ismail, A.F.; Matsuura, T. State-of-the-art membrane based CO2 separation using mixed matrix membranes (MMMs): An overview on current status and future directions. *Prog. Polym. Sci.* 2014, 39, 817–861. [CrossRef]
- 80. Brunetti, A.; Scura, F.; Barbieri, G.; Drioli, E. Membrane technologies for CO₂ separation. *J. Membr. Sci.* **2010**, 359, 115–125. [CrossRef]
- 81. Li, W.; Cao, Z.; Cai, L.; Zhang, L.; Zhu, X.; Yang, W. H₂S-tolerant oxygen-permeable ceramic membranes for hydrogen separation with a performance comparable to those of palladium-based membranes. *Energy Environ. Sci.* **2017**, *10*, 101–106. [CrossRef]
- 82. Al-Mamoori, A.; Krishnamurthy, A.; Rownaghi, A.A.; Rezaei, F. Carbon Capture and Utilization Update. *Energy Technol.* 2017, *5*, 834–849. [CrossRef]
- 83. Lackner, K.S. The thermodynamics of direct air capture of carbon dioxide. Energy 2013, 50, 38–46. [CrossRef]
- Chalmin, A. Direct Air Capture: Recent Developments and Future Plans. Available online: http://www.geoengineeringmonitor. org/2019/07/direct-air-capture-recent-developments-and-future-plans/ (accessed on 30 July 2020).
- 85. Service, R. Cost plunges for capturing carbon dioxide from the air. *Science* **2018**. [CrossRef]
- 86. Creutzig, F.; Breyer, C.; Hilaire, J.; Minx, J.; Peters, G.P.; Socolow, R. The mutual dependence of negative emission technologies and energy systems. *Energy Environ. Sci.* **2019**, *12*, 1805–1817. [CrossRef]
- 87. Azarabadi, H.; Lackner, K.S. Postcombustion Capture or Direct Air Capture in Decarbonizing US Natural Gas Power? *Environ. Sci. Technol.* **2020**, *54*, 5102–5111. [CrossRef] [PubMed]
- Sanna, A.; Uibu, M.; Caramanna, G.; Kuusik, R.; Maroto-Valer, M.M. A review of mineral carbonation technologies to sequester CO₂. *Chem. Soc. Rev.* 2014, 43, 8049–8080. [CrossRef] [PubMed]
- 89. Langanke, J.; Wolf, A.; Hofmann, J.; Böhm, K.; Subhani, M.A.; Müller, T.E.; Leitner, W.; Gürtler, C. Carbon dioxide (CO₂) as sustainable feedstock for polyurethane production. *Green Chem.* **2014**, *16*, 1865–1870. [CrossRef]
- 90. Tcvetkov, P.; Cherepovitsyn, A.; Fedoseev, S. The Changing Role of CO₂ in the Transition to a Circular Economy: Review of Carbon Sequestration Projects. *Sustainability* **2019**, *11*, 5834. [CrossRef]
- 91. Béghin-Tanneau, R.; Guérin, F.; Guiresse, M.; Kleiber, D.; Scheiner, J. Carbon sequestration in soil amended with anaerobic digested matter. *Soil Tillage Res.* **2019**, *192*, 87–94. [CrossRef]
- 92. Tsachidou, B.; Hissler, C.; Delfosse, P. Biogas Residues in the Battle for Terrestrial Carbon Sequestration. *ERCIM News* **2020**, *121*, 19–20.
- 93. Fajardy, M.; Mac Dowell, N. Can BECCS deliver sustainable and resource efficient negative emissions? *Energy Environ. Sci.* 2017, 10, 1389–1426. [CrossRef]
- 94. Klepper, G.; Thrän, D. (Eds.) *Biomasse im Spannungsfeld Zwischen Energie- und Klimapolitik: Potenziale-Technologien-Zielkonflikte: Analyse*; Acatech-Deutsche Akademie der Technikwissenschaften: München, Germany, 2019; ISBN 978-3-9820053-0-0.
- 95. Kaltschmitt, M.; Lenz, V.; Thrän, D. Zur Energetischen Nutzung von Biomasse in Deutschland-Potenziale, Stand und Perspektiven; Leibnitz Institut für Interdisiplinäre Studien: Berlin, Germany, 2008; Volume 12.
- 96. Edenhofer, O.; Pichs Madruga, R.; Sokona, Y. *Renewable Energy Sources and Climate Change Mitigation: Special Report of the Intergovernmental Panel on Climate Change*; Cambridge University Press: New York, NY, USA, 2012; ISBN 978-1-139-15115-3.
- 97. Bhattacharya, S.; Mizanur Rahman Siddique, A.; Pham, H.-L. A study on wood gasification for low-tar gas production. *Energy* **1999**, *24*, 285–296. [CrossRef]
- 98. Chan, W.P.; Veksha, A.; Lei, J.; Oh, W.-D.; Dou, X.; Giannis, A.; Lisak, G.; Lim, T.-T. A hot syngas purification system integrated with downdraft gasification of municipal solid waste. *Appl. Energy* **2019**, *237*, 227–240. [CrossRef]
- 99. Sheth, P.N.; Babu, B.V. Production of hydrogen energy through biomass (waste wood) gasification. *Int. J. Hydrogen Energy* **2010**, 35, 10803–10810. [CrossRef]
- 100. Pandey, B.; Prajapati, Y.K.; Sheth, P.N. Recent progress in thermochemical techniques to produce hydrogen gas from biomass: A state of the art review. *Int. J. Hydrogen Energy* **2019**, *44*, 25384–25415. [CrossRef]
- 101. Ma, F.; Hanna, M.A. Biodiesel production: A review1Journal Series #12109, Agricultural Research Division, Institute of Agriculture and Natural Resources, University of Nebraska–Lincoln.1. *Bioresour. Technol.* **1999**, *70*, 1–15. [CrossRef]
- 102. Ou, X.; Yan, X.; Zhang, X.; Zhang, X. Life-Cycle Energy Use and Greenhouse Gas Emissions Analysis for Bio-Liquid Jet Fuel from Open Pond-Based Micro-Algae under China Conditions. *Energies* **2013**, *6*, 4897–4923. [CrossRef]
- 103. Zhou, Y.; Hu, C. Catalytic Thermochemical Conversion of Algae and Upgrading of Algal Oil for the Production of High-Grade Liquid Fuel: A Review. *Catalysts* **2020**, *10*, 145. [CrossRef]
- 104. Leung, D.Y.; Caramanna, G.; Maroto-Valer, M.M. An overview of current status of carbon dioxide capture and storage technologies. *Renew. Sustain. Energy Rev.* 2014, *39*, 426–443. [CrossRef]
- 105. Hendriks, C.; Crijns-Graus, W.; Bergen, F. Global Carbon Dioxide Storage Potential and Costs. 2004. Available online: https://www.researchgate.net/publication/260095614_Global_Carbon_Dioxide_Storage_Potential_and_Costs (accessed on 3 April 2021).

- 106. Cuéllar-Franca, R.M.; Azapagic, A. Carbon capture, storage and utilisation technologies: A critical analysis and comparison of their life cycle environmental impacts. *J. CO*₂ *Util.* **2015**, *9*, 82–102. [CrossRef]
- Mikhelkis, L.; Govindarajan, V. Techno-Economic and Partial Environmental Analysis of Carbon Capture and Storage (CCS) and Carbon Capture, Utilization, and Storage (CCU/S): Case Study from Proposed Waste-Fed District-Heating Incinerator in Sweden. Sustainability 2020, 12, 5922. [CrossRef]
- 108. Wang, Q.; Gong, Y.; Liu, S.; Wang, D.; Liu, R.; Zhou, X.; Nghiem, L.D.; Zhao, Y. Free Ammonia Pretreatment To Improve Bio-hydrogen Production from Anaerobic Dark Fermentation of Microalgae. ACS Sustain. Chem. Eng. 2019, 7, 1642–1647. [CrossRef]
- 109. Amin, F.R.; Khalid, H.; Zhang, H.; Rahman, S.U.; Zhang, R.; Liu, G.; Chen, C. Pretreatment methods of lignocellulosic biomass for anaerobic digestion. *AMB Express* 2017, 7, 72. [CrossRef]
- 110. Balan, V. Current challenges in commercially producing biofuels from lignocellulosic biomass. *ISRN Biotechnol.* **2014**, 463074. [CrossRef]
- 111. Agbor, V.B.; Cicek, N.; Sparling, R.; Berlin, A.; Levin, D.B. Biomass pretreatment: Fundamentals toward application. *Biotechnol. Adv.* **2011**, 675–685. [CrossRef]
- 112. Taherzadeh, M.J.; Karimi, K. Pretreatment of lignocellulosic wastes to improve ethanol and biogas production: A review. *Int. J. Mol. Sci.* 2008, *9*, 1621–1651. [CrossRef] [PubMed]
- 113. Aruwajoye, G.S.; Kassim, A.; Saha, A.K.; Gueguim Kana, E.B. Prospects for the Improvement of Bioethanol and Biohydrogen Production from Mixed Starch-Based Agricultural Wastes. *Energies* **2020**, *13*, 6609. [CrossRef]
- 114. DIN EN 17124:2019-07, Wasserstoff als Kraftstoff_-Produktfestlegung und Qualitätssicherung_-Protonenaustauschmembran(PEM)-Brennstoffzellenanwendungen für Straßenfahrzeuge; Deutsche Fassung EN_17124:2018; Beuth Verlag GmbH: Berlin, Germany, 2019.
- 115. Marx, J.; Schreiber, A.; Zapp, P.; Haines, M.; Hake, J.-F.; Gale, J. Environmental evaluation of CCS using Life Cycle Assessment—A synthesis report. *Energy Procedia* 2011, *4*, 2448–2456. [CrossRef]
- 116. Papadas, K.-K.; Avlonitis, G.J.; Carrigan, M. Green marketing orientation: Conceptualization, scale development and validation. *J. Bus. Res.* **2017**, *80*, 236–246. [CrossRef]
- 117. Dangelico, R.M.; Vocalelli, D. "Green Marketing": An analysis of definitions, strategy steps, and tools through a systematic review of the literature. *J. Clean. Prod.* 2017, *165*, 1263–1279. [CrossRef]
- 118. Weide, T.; Feil, F.; Kamphus, S.; Peitzmeier, J.; Brügging, E. Bio-H2 aus organischen Reststoffen mittels dunkler Fermentation in Deutschland und den Niederlanden. Available online: http://www.biohydrogen.eu/2020/09/18/bio-h2-aus-organischen-reststoffen-mittels-dunkler-fermentation-in-deutschland-und-den-niederlanden/ (accessed on 3 April 2021).
- 119. Garg, A.; Kazunari, K.; Pulles, T. 2006 IPCC Guidelines for National Greenhouse Gas Inventories; Institute for Global Environmental Strategies (IGES): Kanawaga, Japan, 2006.
- 120. Strazza, C.; Del Borghi, A.; Gallo, M. Development of Specific Rules for the Application of Life Cycle Assessment to Carbon Capture and Storage. *Energies* **2013**, *6*, 1250–1265. [CrossRef]
- 121. Grosse-Sommer, A.P.; Grünenwald, T.H.; Paczkowski, N.S.; van Gelder, R.N.; Saling, P.R. Applied sustainability in industry: The BASF eco-eEfficiency toolbox. *J. Clean. Prod.* 2020, 258, 120792. [CrossRef]
- 122. Klöpffer, W.; Grahl, B. Life Cycle Assessment (LCA): A Guide to Best Practice/by Walter Klopffer and Birgit Grahl; Wiley-VCH: Weinheim, Germany, 2014; ISBN 978-3-527-32986-1.
- 123. Statista. Prognostizierte Kosten für die Wasserstoffherstellung in Deutschland im Jahr 2050. 2020. Available online: https://de.statista.com/statistik/daten/studie/1193246/umfrage/prognostizierte-kosten-zur-herstellung-von-wasserstoffin-deutschland/ (accessed on 11 February 2021).
- 124. Timmerberg, S.; Kaltschmitt, M. Hydrogen from renewables: Supply from North Africa to Central Europe as blend in existing pipelines—Potentials and costs. *Appl. Energy* 2019, 237, 795–809. [CrossRef]
- 125. Touili, S.; Alami Merrouni, A.; Azouzoute, A.; El Hassouani, Y.; Amrani, A. A technical and economical assessment of hydrogen production potential from solar energy in Morocco. *Int. J. Hydrogen Energy* **2018**, *43*, 22777–22796. [CrossRef]
- 126. Nordhaus, W.D. To Slow or Not to Slow: The Economics of The Greenhouse Effect. Econ. J. 1991, 101, 920. [CrossRef]
- 127. Valuing Climate Changes; National Academies Press: Washington, DC, USA, 2017; ISBN 978-0-309-45420-9.
- 128. Edenhofer, O.; Flachsland, C.; Kalkuhl, M.; Knopf, B.; Pahle, M. Optionen für Eine CO2-Preisreform: MCC-PIK-Expertise für den Sachverständigenrat zur Begutachtung der Gesamtwirtschaftlichen Entwicklung. 2019. Available online: https://www.google.com/url?sa=t&rct=j&q=&esrc=s&source=web&cd=&cad=rja&uact=8&ved=2ahUKEwiGuN6 _tdruAhVKrxoKHXVADpYQFjAAegQIARAC&url=https%3A%2F%2Fwww.mcc-berlin.net%2Ffileadmin%2Fdata%2FB2.3 _Publications%2FWorking%2520Paper%2F2019_MCC_Optionen_f%25C3%25BCr_eine_CO2-Preisreform_final.pdf&usg= AOvVaw2Ky9e4nNe_LuN-9uT7jevA (accessed on 8 February 2021).
- 129. Pindyck, R.S. The social cost of carbon revisited. J. Environ. Econ. Manag. 2019, 94, 140–160. [CrossRef]
- 130. Umweltbundesamt. Hohe Kosten Durch Unterlassenen Umweltschutz: Eine Tonne CO2 Verursacht Schäden von 180 Euro—Umweltbundesamt Legt Aktualisierte Kostensätze Vor. Available online: https://www.umweltbundesamt.de/presse/pressemitteilungen/hohe-kosten-durch-unterlassenen-umweltschutz (accessed on 3 April 2021).

- 131. Die Bundesregierung. Grundlage für CO2-Preis Steht. Available online: https://www.bundesregierung.de/breg-de/themen/klizmaschutz/nationaler-emissionshandel-1684508 (accessed on 8 February 2021).
- 132. Bundesamt für Umwelt BAFU Schweiz. Erhebung der CO₂-Abgabe auf Brennstoffe. Available online: https://www.bafu.admin. ch/bafu/de/home/themen/klima/fachinformationen/verminderungsmassnahmen/co2-abgabe/erhebung.html (accessed on 12 March 2021).

Appendix B: Paper B



Article



Perspectives of Biogas Plants as BECCS Facilities: A Comparative Analysis of Biomethane vs. Biohydrogen Production with Carbon Capture and Storage or Use (CCS/CCU)

Johannes Full ^{1,2,*}, Silja Hohmann ^{1,2}, Sonja Ziehn ¹, Edgar Gamero ^{1,2}, Tobias Schließ ¹, Hans-Peter Schmid ³, Robert Miehe ^{1,4} and Alexander Sauer ^{1,2}

- ¹ Fraunhofer Institute for Manufacturing Engineering and Automation IPA, Nobelstraße 12, 70569 Stuttgart, Germany
- ² Institute for Energy Efficiency in Production (EEP), University of Stuttgart, Nobelstraße 12, 70569 Stuttgart, Germany
- ³ WS Reformer GmbH, Allmandring 35, Dornierstraße 14, 71272 Renningen, Germany
- ⁴ Institute of Industrial Manufacturing and Management (IFF), University of Stuttgart, Allmandring 35, 70569 Stuttgart, Germany
- * Correspondence: johannes.full@ipa.fraunhofer.de

Abstract: The transition to a carbon-neutral economy requires innovative solutions that reduce greenhouse gas emissions (GHG) and promote sustainable energy production. Additionally, carbon dioxide removal technologies are urgently needed. The production of biomethane or biohydrogen with carbon dioxide capture and storage are two promising BECCS approaches to achieve these goals. In this study, we compare the advantages and disadvantages of these two approaches regarding their technical, economic, and environmental performance. Our analysis shows that while both approaches have the potential to reduce GHG emissions and increase energy security, the hydrogen-production approach has several advantages, including up to five times higher carbon dioxide removal potential. However, the hydrogen bioenergy with carbon capture and storage (HyBECCS) approach also faces some challenges, such as higher capital costs, the need for additional infrastructure, and lower energy efficiency. Our results give valuable insights into the trade-offs between these two approaches. They can inform decision-makers regarding the most suitable method for reducing GHG emissions and provide renewable energy in different settings.

Keywords: HyBECCS; hydrogen; BECCS; biogas; biohydrogen; biomethane; CDR; NET; carbon-negative hydrogen; steam methane reforming; SMR

1. Introduction

The urgent need to mitigate climate change has led to the development of innovative technologies that can reduce greenhouse gas (GHG) emissions and promote sustainable energy production. Additionally, negative emission technologies (NETs), also known as carbon dioxide removal (CDR) technologies, are crucial for limiting human-induced global warming to less than 2 °C above pre-industrial times [1,2].

Biogas, a renewable energy source produced by the anaerobic digestion of organic waste, has emerged as a promising alternative to fossil fuels [3]. During the process of biogas production, microorganisms break down organic matter in an oxygen-free environment. The result is a gas mixture that is mainly composed of methane (CH₄) and carbon dioxide (CO₂) [4]. The CH₄ contained in biogas can be purified and is then referred to as biomethane [5]. Once the biogas has been purified to methane, it can be used in various ways, e.g., as a fuel for transportation or for industrial applications. It can also be injected into and transported through the natural gas grid. In biomethane production with carbon dioxide capture and storage (CCS), the biogenic CO₂ released during the anaerobic digestion is captured and permanently stored. However, biomethane is a greenhouse gas itself



Citation: Full, J.; Hohmann, S.; Ziehn, S.; Gamero, E.; Schließ, T.; Schmid, H.-P.; Miehe, R.; Sauer, A. Perspectives of Biogas Plants as BECCS Facilities: A Comparative Analysis of Biomethane vs. Biohydrogen Production with Carbon Capture and Storage or Use (CCS/CCU). *Energies* **2023**, *16*, 5066. https://doi.org/10.3390/ en16135066

Academic Editors: Hamidreza Gohari Darabkhani and Abdel-Hamid Soliman

Received: 7 June 2023 Revised: 22 June 2023 Accepted: 28 June 2023 Published: 30 June 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

when emitted into the atmosphere, and its use in combustion engines or combined heat and power (CHP) units results in CO₂ emissions.

Alternatively, biogas can be processed by steam reforming to produce hydrogen (H_2) [6,7]. H_2 can also be used as a fuel for transportation or injected into the natural gas grid [8,9]. An advantage of H_2 over biomethane is that it can be used decentrally without emitting CO₂ or other greenhouse gases, as it is a carbon-free energy carrier. Furthermore, during the steam-methane reforming (SMR) process, in addition to the CO₂ in the biogas, biogenic CO₂ is generated as a product and can be captured and stored. This leads to a higher CO₂ removal potential compared to biomethane production with CCS. However, energy losses resulting from the conversion processes must be considered. Technologies that combine biohydrogen production with CCS are referred to as Hydrogen Bioenergy with Carbon dioxide Capture and Storage (HyBECCS) approaches [10].

Biogas reforming for hydrogen production is a promising area of research and development. Significant progress has been made in the efficiency and scalability of the technology [11–13]. Furthermore, several companies and research institutions are actively engaged in advancing biogas reforming processes, optimizing catalysts, improving reaction kinetics, and exploring their integration with CCS technologies. Some examples include the Canadian company Xebec Adsorption [14], the French Air Liquide [15], and WS Reformer in Germany [16]. Similarly, pilot projects and demonstration plants are being developed, indicating a growing interest in harnessing the potential of biogas reforming as a sustainable pathway for hydrogen production (see for example [17–19]).

For CO₂ storage, there are several options. One common method is injecting the CO₂ into underground geological formations, such as depleted oil and gas reservoirs, saline aquifers, or coal seams [2,20]. Furthermore, there are also several options for the long-term use of captured carbon dioxide (CCU) [21]. One example for CCU is the production of building materials, such as cement, concrete, and mortar, which can help reduce greenhouse gas emissions by replacing GHG-intensive materials in the building sector. As the captured and stored CO₂ is from biogenic sources, its storage or long-term use can lead to negative emissions, and the process can be considered a CDR approach or NET [22,23].

This paper compares biomethane production from biogas with CCS with biohydrogen production from biogas with CCS. The analysis considers the technical, economic, and environmental aspects of each approach. The aim is to provide insights into the trade-offs to be considered in terms of energy efficiency, their potential for negative emission generation and economic potentials. First, a general energy efficiency analysis is presented. For this, a use case is defined that involves the use of biomethane or hydrogen as a fuel in heavy-duty vehicles. Next, the negative emission potential (NEP) for each process option is calculated and compared. Finally, a discussion about the associated costs and profitability of both processes is presented.

2. Basics

2.1. Biomethane Production through Biogas Purification

Biogas is produced through the anaerobic digestion of organic waste [3,24]. It consists mainly of methane (CH₄) and carbon dioxide (CO₂). Typically, the amount of CH₄ in biogas is between 55 vol% and 70 vol% [4]. Other gaseous components like hydrogen sulfide (H₂S), hydrogen (H₂), and nitrogen (N₂) can be present in small quantities. Biogas can be purified to biomethane using various technologies such as pressure swing adsorption (PSA), membrane separation, or cryogenic distillation [5,25]. The purification process removes CO₂ and impurities such as water and trace gases to increase the CH₄ content of the gas [26]. In terms of physical properties, biomethane and methane are identical, as they are the same substance. The different name only refers to their origin and production process. While "methane" refers to any type of methane, including methane derived from fossil resources, biomethane specifically designates methane obtained from biogas [27].

2.2. Biohydrogen Production through Biogas Steam Reforming

Biogas steam reforming is a process used to produce H_2 from biogas. The process involves the conversion of CH_4 into gaseous H_2 and CO_2 through a reaction with steam and the parallel occurring carbon monoxide (CO) shift. According to Rostrup-Nielsen and Christiansen, the process is described by the following two reactions [28]:

$$n \operatorname{CO}_2 + \operatorname{CH}_4 + \operatorname{H}_2\operatorname{O}_{(g)} \leftrightarrow 3 \operatorname{H}_2 + \operatorname{CO} + n \operatorname{CO}_2; \Delta H = +206.15 \text{ kJ/mol}$$
(1)

$$CO + H_2O_{(g)} \leftrightarrow H_2 + CO_2; \Delta H = -41.2 \text{ kJ/mol}$$
(2)

where n describes the molar ratio of CO_2 to CH_4 in the feed gas. This ratio is between 0.43 and 0.82, as the molar amount of CH_4 is between 55% and 70% for typical biogas plants, assuming ideal gas conditions [4]. Assuming the complete conversion of the C feedstock into CO_2 , the overall stoichiometric reaction, obtained by combining Equations (1) and (2), is [29]

$$n CO_2 + CH_4 + 2 H_2O_{(\sigma)} \leftrightarrow 4 H_2 + (n+1) CO_2; \Delta H = +164.95 \text{ kJ/mol}$$
 (3)

2.3. Carbon Dioxide Removal

NETs are essential to counterbalance exceedingly high GHG emissions and to limit global warming to 2 °C above pre-industrial levels [1]. NETs consist in removing CO₂ from the atmosphere and storing it for long periods, effectively reducing the overall concentration of CO₂ in the atmosphere. Negative emissions correspond to the amount of removed and stored CO₂ that exceeds the GHG emissions caused during the entire process of generating such carbon removals. This means that these GHG emissions [30]. For process comparisons based on similar technologies, the maximum possible negative emissions can be estimated. For this purpose, all GHG emissions are neglected, and only the maximum captured amount of biogenic (or atmospheric) CO₂ is considered. This is a comparative value referred to as negative emission potential (NEP) [23].

CDR involves the capture and storage of atmospheric CO₂ from direct air carbon capture approaches (DACCS, direct air carbon capture and storage) and of biogenic CO₂, e.g., from the energetic use of biogenic waste (BECCS, bioenergy with carbon capture and storage). The most common method of CO₂ storage is geological storage, which involves injecting CO₂ into deep geological formations such as depleted oil and gas reservoirs, saline aquifers, and unmineable coal seams [2,31]. Another option for CO₂ storage is mineralization, which involves the reaction of CO₂ with naturally occurring minerals to form stable carbonates [32]. This process can occur naturally over long periods. However, it can be accelerated through various methods like mineral carbonation. This involves the reaction of CO₂ with magnesium and calcium silicate minerals to form stable carbonates. In addition to the permanent storage for CO₂, some carbon capture and utilization (CCU) approaches can be considered CDR, as long as the utilized carbon is durably stored in the respective product. Examples are the production of long-term materials like high-tech plastics or construction materials [33,34].

2.4. The HyBECCS Concept

HyBECCS is an umbrella term for processes in which H_2 is produced from biogenic residual and waste materials and the resulting CO_2 is captured and permanently stored. It can also be seen as a subsidiary branch of bioenergy with carbon capture and storage (BECCS) with the particularity that H_2 is produced as an energy source. The technical approach can be split into four basic process steps: 1. substrate biomass pretreatment, 2. the production of biohydrogen and biogenic CO_2 , 3. the separation of the product gases CO_2 and H_2 , and 4. their processing for the use of H_2 and long-term storage or the use of biogenic CO_2 , including the permanent CO_2 storage or use itself [10]. For each of the four

steps, there are different technology options. A main technological advantage of BECCS and HyBECCS approaches is the energy-efficient capture of biogenic CO_2 as it generally occurs as a point source. It provides a double effect on climate mitigation by providing GHG-free energy and the storage of biogenic CO_2 with the potential to provide negative emissions. The basic requirements are, however, the deduction of all GHG emissions occurring over the entire process chain to determine the amount of negative emissions, economic viability, and a reliable long-term CO_2 storage [30]. Considering biogas plants, there are several possibilities to retrofit existing plants to HyBECCS facilities [35]. The technical approach considered in this work combines biogas-steam reforming with CCS, as described in the following section.

3. Process Descriptions and Main Assumptions

This section defines and describes the two process options to be compared in this work: biomethane production with CCS (process option 1) and biohydrogen production through biogas-steam reforming with CCS (process option 2). The various stages of each process option with a focus on the differences are described and the main assumptions for their comparison are outlined.

An overview of the two process options is provided in Figure 1. The first stage of both process options involves the preparation of the feedstock, followed by biogas production through the anaerobic digestion of organic waste in large vessels that provide an oxygen-free environment for microorganisms to break down the organic matter [3]. Different types of organic waste can be used, including agricultural waste, food waste, and sewage [36,37].



Figure 1. Process options and system boundaries (A and B) for biogas usage (own representation).

In process option 1, the second stage is separating the CH_4 from the other gas components in the biogas. This is typically performed by PSA that uses adsorbent materials to selectively adsorb CO_2 and impurities, leaving behind purified biomethane, as described in Section 2.1 [25,27,38]. Alternatively, membrane separation or water scrubbing can be used.

For process option 2, the biogas is fed into a steam reforming process, as described in Section 2.2. The resultant gas mixture is subjected to further treatment, including a water-gas shift to increase the H_2 and CO_2 concentration, and gas separation through PSA to gain the product fractions CO_2 and H_2 . Relevant reaction equations for this process option are derived below.

The next step for both process options is CO_2 capture and storage. In process option 1, the biogenic CO_2 from the biogas supply is captured after the separation of the biomethane. In process option 2, biogenic CO_2 is present in the biogas fuel and feed and enriched during biogas burning, reforming and CO-shift steps. After the H₂ separation, the tail gas is redirected to the burner to be oxidized, resulting in further CO_2 . The aggregated CO_2 can be captured from a single exhaust gas stream. The essential difference between the two process options is that in process option 1, only the CO_2 present in the biogas can be captured. Part of the carbon content of the original biomass is bound in the CH_4 and is released during the decentralized use of the methane. In contrast, in process option 2, the entire carbon content of the biogas is released during the SMR step and can be captured from the "Exhaust Gas" stream (cf. Figure 2) as CO_2 .



Figure 2. Flow diagram of the steam methane reforming process step within process option 2 (own representation).

After its separation, the CO_2 is compressed for transportation either via pipelines or in containers by truck. Storage sites are often located offshore, which may require additional transport by ship. Which option is more economically and environmentally viable must be decided for each case specifically. Transport over short distances can be carried out by truck and ship, while transport via pipeline involves higher expenditures, but becomes viable in the long term if CO_2 is transported over long distances and when the amounts of CO_2 transported are higher [39,40]. Finally, the CO_2 is stored to prevent its emission into the atmosphere, e.g., in geological formations, such as depleted oil and gas reservoirs or saline aquifers [2].

It is assumed that the use of both chemical energy carriers H_2 and CH_4 provided in process options 1 and 2 is decentralized and that no carbon capture occurs during their use. For example, vehicles can be fueled, or heat can be supplied. Two cases are defined for exemplification: Case A (see system boundary A in Figure 1) compares the biomethane from process option 1 to the biohydrogen produced via process option 2 in terms of their chemical energy content as secondary energy carriers, expressed by the lower heating value (LHV). As depicted in Figure 1, the system boundaries for case A range from the biomass supply to the biomethane and H_2 production.

Case B (see system boundary B in Figure 1) compares the useful energy generation potentials of both process options for heavy-duty transport applications. In process option 1, the useful energy from biomethane is generated by bio-liquefied natural gas (LNG) combustion engines. In process option 2, the energy from H₂ is generated with fuel cell power trains. Hence, the system boundary B is set from the biomass supply to the use of biomethane and H₂, respectively, for heavy-duty transportation (system boundary B in Figure 1). Burning liquefied biomethane in gas engines can reach efficiencies up to 44% [41]. State-of-the-art fuel cell systems for electric truck powertrains peak at efficiencies

5 of 16

of 63% [42–44]. In the following, with regard to further technological development and optimization, high efficiencies of 44% (for bio LNG engines) and 63% (for H₂ fuel cell engines) are assumed for both process options.

Relevant reaction equations for Process option 2 are derived in the following: An average biogas composition of 37.5 vol% CO_2 and 62.5 vol% CH_4 is assumed [4]. By converting the units and referring to the product of 1 kg H₂, reaction Equation (3) can be written as follows:

3.3 kg CO₂ + 2.0 kg CH₄ + 4.5 kg H₂O_(g)
$$\leftrightarrow$$
 1 kg H₂ + 8.8 kg CO₂; Δ H = +5.73 kWh/kgH₂ (4)

This reaction is endothermic and, thus, energy must be provided to drive the reaction towards H_2 and CO_2 production Typically, this energy is supplied in the form of heat, generated by the oxidation of a part of the biogas and of the tail gas after H_2 separation. Figure 2 shows the schematic structure of the SMR process with the respective material and energy flows.

As depicted in Figure 2, the Δ H in reaction (3) representing the endothermic heat of the reaction has to be provided by the burner being part of the SMR unit. Additionally, there is heat demand for steam generation and preheating the feedstock to typical operation temperatures in the reformer of around 800 °C, as well as for compensating wall heat losses and latent heat of the products. The total heat energy supplied amounts to approx. 25 kWh per kg of H₂ produced. This value is based on empirical data from existing biogas SMR units and is taken as an assumption in this paper. The combustion of biogas with air is described by the following equation:

$$n CO_2 + CH_4 + 2 (O_2 + m N_2) \rightarrow 2 H_2O_{(g)} + (n+1) CO_2 + m N_2; \Delta H = -802.25 \text{ kJ/mol}$$
(5)

where n describes the molar ratio of CO_2 to CH_4 in the feed gas and m describes the ratio of N_2 to O_2 in air. Usually air contains around 79 vol% N_2 and 21 vol% O_2 [45]. By converting the units and referring to 1 kg CH₄, Equation (5) can be rewritten as follows:

$$1.65 \text{ kg CO}_2 + 1 \text{ kg CH}_4 + 4 \text{ kg O}_2 + 13.17 \text{ kg N}_2 \leftrightarrow 2.25 \text{ kg H}_2\text{O} + 4.4 \text{ kg CO}_2 + 13.17 \text{ kg N}_2 \tag{6}$$

It can be seen that 62.5% of the CO₂ in the exhaust gas originates from the combustion of methane, while 37.5% comes from the biogas.

Summing up, the two process options consist of several identical process steps but also show some differences: The preparation of the feedstock, the biogas production in the biogas plant and the product gas treatment are identical for both processes. To compare both processes from an ecological point of view, these identical process steps can be excluded from the comparison according to the so-called "black-box" approach [46,47]. Also, for the economic comparison, identical plant components can be neglected.

The main differences between the two process options are: (i) The additional biogas SMR step in process option 2. (ii) The higher amount of captured CO_2 leading to more CO_2 to be compressed, transported, and stored in process option 2. Both (i) and (ii) increase the investment and operational costs of process option 2. (iii) The different products with their respective technology options for heavy-duty transportation: biomethane and biogenic CO_2 in process option 1 versus biohydrogen and more biogenic CO_2 in process option 2. Further differences are the technological and cost differences between H_2 and biomethane in compression and transportation.

4. Technology Comparison

4.1. Considerations on Product Application

In this section, the two products, H_2 and CH_4 , are compared regarding their product properties. This comparison is independent of the two use cases described in Section 3 and intends to give a broader view on possible applications.

 CH_4 has a higher volumetric energy density than H_2 , which means it contains more energy per unit volume [48]. This makes CH_4 generally more suitable for applications where space is limited. However, H_2 has a higher energy content per unit mass, which makes it more efficient for applications where weight is a concern [49]. CH_4 and H_2 are both flammable and explosive above certain concentrations [50]. However, H_2 has a wider flammability range and can ignite at lower concentrations. With proper handling and storage, both gases can be used safely [51].

Both biomethane and H_2 are considered to play key roles in future energy provision, e.g., as substitutes for natural gas [52,53]. They can be used for transportation, heating, power generation, and in industrial processes, as described briefly in the following.

CH₄ is commonly used as a fuel for vehicles, especially in the form of compressed natural gas (CNG) or LNG [54,55]. CNG is used in light-duty vehicles such as cars, while LNG is used in heavy-duty vehicles like trucks and buses [56,57]. H₂ can be used with fuel-cell vehicles (FCVs) as well as modified internal combustion engine vehicles [58]. Furthermore, CH₄ as well as H₂ can be used for heating and power generation in residential, commercial and industrial applications. H₂, however, generally requires a specific environment for combustion due to its lower volumetric energy density, faster laminar flame speed, and higher combustion temperatures (1600 °C) compared to CH₄ (1300 °C) [59]. Subsequently, conventional gas engines must first be either retrofitted or replaced with gas turbines that are H₂-ready and thus adequate for the use of 100% H₂.

4.2. Energy Efficiency

Comparing process options 1 and 2, as described in Section 3, the chemical energy contents of the produced gases are a critical factor for energy efficiency comparison. Therefore, the available thermal energy produced by a combustion of the gases, expressed by the mass-specific lower heating value (LHV), was considered. As described in Section 2.2, the reactions of biomethane to CO_2 and H_2 proceed under energy input. The thermal energy for the endothermic reaction is obtained via the oxidation of biogas. Since CO_2 is completely oxidized, the thermal energy is obtained via full oxidation of the CH_4 in the biogas.

For the following calculation, it is further assumed that biogas consists of 62.5 vol% CH₄ and 37.5 vol% CO₂. This is an average value based on existing biogas processes, as described in Section 2.2. Equation (7) is derived from Equation (4) under consideration of the 25 kWh thermal energy input required for the overall SMR process, as described in Section 3.

$$2.0 \text{ kg CH}_4 + 4.5 \text{ kg H}_2\text{O}_{(g)} + 25 \text{ kWh}_{th} \rightarrow 1.0 \text{ kg H}_2 + 5.5 \text{ kg CO}_2$$
(7)

The chemical energy content of CH_4 and H_2 was calculated with Equation (8) using the LHV for the quantities given. The LHV of CH_4 is 13.9 kWh/kg and of H_2 33.33 kWh/kg [60].

$$E_i = LHV \times m_i \tag{8}$$

 E_i is the total chemical energy content for the considered mass of the species i [in kWh] and m_i the mass of the species i [in kg]. In this case, 2.0 kg of CH₄ has an energy content of 27.8 kWh. Thus, 52.8 kWh of energy input, divided into 2.0 kg CH₄ (LHV) and thermal energy of 25 kWh, leads to 33.33 kWh of chemical energy produced in the form of 1 kg H₂ (LHV), as shown in Table 1. Therefore, the efficiency of the considered steam-reforming process is 63% in terms of the chemically stored energy content of the product H₂ in relation to the overall energy input.

Table 1. Energy balance for the production of 1.0 kg H_2 by biogas steam reforming (process option 2, system boundary A).

	Energy Input	Energy Output
Chemical energy content CH ₄	27.8 kWh	0.0 kWh
Chemical energy content H ₂	0.0 kWh	33.33 kWh
Thermal energy	25.0 kWh	0.0 kWh
Total	52.8 kWh	33.33 kWh

To compare the energy efficiency of CH_4 and H_2 in the heavy-duty transportation sector (case B), the efficiencies of the corresponding technologies must be included. As listed in Section 3, gas engines for liquefied biomethane can reach efficiencies up to 44%, while fuel cell systems with batteries for H_2 peak up to 63% [42–44]. The efficiencies can be included using the following equation.

$$E_{\rm in} = E_{\rm out} / \eta \tag{9}$$

where E_{in} is the energy in kWh, E_{out} is the usable energy in kWh and η is the efficiency of the corresponding technology.

For biomethane, providing 1 kWh of usable energy for transport requires an energy input of 2.27 kWh at an efficiency of 44%. For H₂, the calculation must also take into account the energy efficiency of its production via steam reforming with a value of 63.1%, as shown above and in Table 2. Thus, for a 63% efficiency of the fuel cell system with a battery, the total efficiency of process option 2 for case B is 39.8%. This means that to supply 1 kWh of usable energy in the form of H₂, an energy input of 2.52 kWh is needed. Hence, using H₂ produced in process option 2 means a loss of approx. 4.2% of the usable energy in the form of propulsion energy for heavy-duty transport compared to process option 1.

Table 2. Required energy input for the production of 1.0 kWh usable energy (process options 1 and 2, system boundary B).

	Process Option 1	Process Option 2
Usable Energy	1.0 kWh	1.0 kWh
Efficiency (heavy-duty transport)	44.0%	63.0%
Relative Efficiency (production)	100%	63.1%
Total Efficiency	44.0%	39.8%
Energy Input	2.27 kWh	2.52 kWh

However, this is within the range of several uncertainties due to the assumptions made, e.g., assumed powertrain efficiencies and different driving behaviors. Furthermore, the compression of the two gases, CH_4 and H_2 , are not considered. To convert CH_4 gas into LNG, an energy loss of 10–20% can be assumed (1.39–2.78 kWh/kg CH_4) [61]. For H_2 compression to 400 bar, an energy loss of about 10.8% (3.61 kWh/kg H_2) can be assumed [62]. Since the energy consumptions for both applications are associated with uncertainties and are in a similar order of magnitude, they can be neglected in the calculations.

In conclusion, however, it can be said that for Use Case B, the efficiency losses that occur in process option 2 due to the reforming process step (Use Case A) can be counterbalanced by better efficiency in the FCEV power train, leading to the useful energy for heavy-duty transport being in a similar order of magnitude for both process options.

4.3. Negative Emission Potential

The NEP represents the amount of negative emissions that are possible if all GHG emissions caused along the entire value chain up to the use or storage of the product gases are neglected, as described in Section 2.3. It can be used to compare the two process options described in Section 3 following the "black box" approach, where identical process steps are excluded from the comparison [46,47]. To determine the NEP, the theoretical maximum amount of storable biogenic CO₂ is calculated for the process options. Therefore, the amount of CO₂ generated via biogas production in the biogas plant is considered first. It is assumed, as described in Section 4.2, that biogas consists of CH₄ (62.5 vol%) and CO₂ (37.5 vol%). Thus, 1 Nm³ biogas consists of 0.375 m³ CO₂ and 0.625 m³ CH₄. Furthermore, the biogas was considered an ideal gas. According to the ideal gas law [63], this leads to the following equation being applicable:

$$pV_k = n_k RT, \tag{10}$$
where p is the pressure in Pascal, V_k is the volume of the species k in m³, n_k is the amount of the species k in mol, R is the general gas constant (in J/mol·K), and T is the temperature in K. From the ideal gas law, a molar ratio of 0.6 mol CO₂/mol CH₄ is obtained for the assumed average composition of biogas. This corresponds to a mass ratio of 1.65 kg CO₂/kg CH₄ and means that 1.65 kg of CO₂ is produced via the biogas plant per kilogram of CH₄ produced in process option 1. In process option 2, an additional 2.75 kg of CO₂ is produced from heat supply through biogas combustion and reforming, when the same amount of biogas containing 1 kg of CH₄ is supplied as input stream. Adding this amount to the biogenic CO₂ in the biogas leads to a total of 4.4 kg CO₂ to be captured in process option 2.

In summary, the biomethane production (process option 1) results in an NEP of 1.65 kg CO_2 per kg CH_4 produced, while the HyBECCS approach (process option 2) shows a NEP of 4.4 kg CO_2 for the same biomass input. This corresponds to about 2.7 times the amount of storable biogenic CO_2 of process option 2 compared to process option 1. Furthermore, depending on the biogas quality (CH_4 content), this ratio deviates, as shown in Figure 3. The NEP of process option 2 increases to up to 3.3 times the amount of process option 1 for a CH_4 content in the biogas of 70 vol%. For a CH_4 content in biogas of 55 vol%, the NEP for process option 1 is still 2.2 times higher than that of process option 1.



Figure 3. Ratio of biogenic CO_2 production of process option 2 vs. process option 1 for different CH_4 volume fractions in biogas (own representation).

4.4. Economic Comparison

The costs required and revenues to be expected for the two process options can vary significantly depending on the specific technology and scale of the project. However, some general considerations can be made.

Costs that arise in both process options can be neglected in the comparison, e.g., the main technologies used for upgrading biogas to biomethane are PSA and membrane separation [64]. This is also required for the purification of the biohydrogen and can therefore be neglected for the comparison [65]. Additionally, there may be costs associated with obtaining permits, certifications, and approvals for the biomethane as well as the

107

biohydrogen production, depending on the regulations in the specific country or region. These costs can also vary depending on the project location and regulatory environment and are neglected in this work.

The costs for biogas steam reforming to produce H_2 include the capital costs of the reforming unit, which is the main equipment required for the process. Therefore, the capital costs for process option 1 are generally lower than for process option 2. This is primarily because additional plant components for the reforming process step are needed in process option 2. The costs for the SMR upgrade have to be analyzed specifically for each case. However, the additional costs for the SMR upgrade have to be compensated by a higher income from the sale of the products in order to be economically feasible. The income side is described in the following.

Furthermore, the different sources of income with their respective market prices, today and in the future, can be compared. In process option 1, the marketable products are CH_4 and negative emissions, whereas, in process option 2, the marketable products are H_2 and negative emissions. The total income for each process option depends on the market prices of the respective products today and in the future during the lifetime of the plant as well as the generated amount of each product. A total of 3.6 kg of CH_4 is necessary for the production of 1 kg of H_2 , as described in Section 3. Process option 2 therefore requires 3.6 times the amount of biogas input per kg of produced energy carrier. Considering negative emissions, however, process option 2 has the potential to produce about 2.7 times the amount of storable biogenic CO_2 to be captured and stored, as shown in Section 4.3.

The current market prices of CH₄ and H₂ vary depending on the region and quality. However, some general price comparisons can be made based on available data. In the United States, the retail price of CNG for transportation ranges from USD 2-4 per gasoline gallon equivalent (GGE) (EUR 0.72-1.45 per kg natural gas [66]), whereas the retail price of LNG for transportation ranges from USD 2–5 per diesel gallon equivalent (DGE) (EUR 0.68–1.69 per kg LNG [66]), both depending on the location and the supplier. The retail price of H₂ for transportation ranges from USD 9.5–13.2 per kg (EUR 8.8–12.3 per kg), depending on the location and the supplier [67]. In Europe, the fuel prices for transportation vary from country to country. In the first quarter of 2022, the average fuel price range for CNG in Europe was EUR 5.23 to EUR 12.51 per 100 km travelled (average EUR 7.24 per 100 km), based on information submitted by nine member states [68]. Assuming an average consumption of 3.64 kg of CNG per 100 km [69], this corresponds to a price range of EUR 1.44 to EUR 3.43 per kg of CNG (Average: EUR 1.99 per kg). For LNG, the average retail price range for the same period was from EUR 0.45 to EUR 2.81 per kg of LNG, with an average price for Europe (EU-27) of EUR 1.86 per kg LNG [70]. With an average consumption of 2.14 kg LNG per 100 km for heavy trucks [56], this corresponds to a price range of EUR 0.96 to EUR 6.01 per 100 km (average: EUR 3.98 per 100 km). Average H_2 prices in the first quarter of 2022 ranged from EUR 9.00 to EUR 15.06 per 100 km (average: EUR 11.22 per 100 km), based on information submitted by six member states. Assuming an average consumption of 0.8 kg H_2 per 100 km [71], an average price range of EUR 11.25 to EUR 18.83 per kg of H_2 (average: EUR 14,01 per kg H_2) is obtained. An overview of the pump price ranges for different fuels in the European Union is shown in Table 3.

Table 3. Overview of price ranges for different fuels in the European Union (average for Q1 of 2022) [68,70].

Fuel	Price (EUR/kg)	Price (EUR/100 km)	Reference
CNG	1.44–3.43 Average: 1.99	5.23–12.51 Average: 7.24	[68]
LNG	0.45–2.81 Average (EU-27): 1.86	0.96–6.01 Average (EU-27): 3.98	[70]
H ₂	11.25–18.83 Average: 14.01	9.00–15.06 Average: 11.22	[68,71]

Summarizing, the current market prices of CH_4 are clearly lower than those of H_2 . This is due to H_2 being a downstream product predominantly produced from CH_4 obtained from natural gas [72]. However, the long-term price forecasts for H_2 indicate potential for lower prices due to cost reductions through technological alternatives like electrolysis as well as economies of scale [73].

Both process options generate negative emissions, when the biogenic CO_2 is not released into the atmosphere but captured, transported to a storage site, and permanently stored. However, generating negative emissions from biogenic CO₂ creates further costs and will, therefore, only happen when they are overcompensated by an income. One option is the sale of so-called carbon removal units (CRUs) issued for verified negative emissions (in tCO₂e) and can be sold to buyers who wish to claim them. Already today, several private entities offer CRUs [74]. They stem from different carbon-removal approaches in different countries with different underlying quality standards and show broad price ranges. Examples are EUR 20/CRU for bio-based construction materials in France to EUR 535/CRU for biochar in Sweden [75]. In November 2022, the European Commission proposed an EU carbon-removal certification framework (CRCF) to establish common rules for the monitoring, verification, and reporting of negative emissions for the voluntary carbon removal market [76]. This EU initiative is still ongoing. However, this standardization process within the geographical boundaries of one of the biggest carbon trading schemes worldwide shows the awareness of the political bodies for the urgent need to create a secure and credible playing field for the commercialization of CRUs. Summing up, negative emissions are already today a marketable product, and, in the future, ongoing standardization processes will increase the visibility, reliability, and, thus, the market volume for negative emissions in Europe and worldwide.

As outlined in Section 4.3, process option 1 results in an NEP of 1.65 kg CO_2 per kg of CH_4 produced, while process option 2 shows an NEP of 4.4 kg CO_2 for the same amount of biomass input. The NEP represents the maximum amount of negative emissions to be produced and qualified for CRUs in each option. Process option 1 generates 1.0 kg of bio-LNG, while process option 2 generates 0.5 kg H_2 for the same amount of biomass input, as outlined in Section 3. The following price ranges, also described above, give the following price averages for the products: For H₂, the price range of 0.93–3.72 EUR/kg H₂ results in an average price of 2.325 EUR/kg H_2 . For LNG, the price range of 0.68–1.69 EUR/kg LNG results in an average price of 1.185 EUR/kg LNG. For CRU, the price range of 20–535 EUR/CRU results in an average price of 277.5 EUR/CRU. For process option 1, this means that the products (LNG and CO_2) can generate an income of 459.06 EUR/kg of CH_4 produced. For process option 2, the products (H_2 and CO_2) can generate an income of 1222.16 EUR for the same amount of biomass input. In this simplified exemplary calculation, process option 2 would generate more income. However, cost consideration is necessary for final economic comparison and CRU pricing is still not fixed or legislated. The calculation shows, however, that depending on the market value of the different products, yields determine the superiority in terms of income of the process options and gives first indications and orientation values.

5. Limitations

This paper focuses on comparing the biomethane vs. biohydrogen with CCS options in terms of their energy efficiency, economic aspects, and negative emission potential. Other environmental impacts are not considered. However, both approaches' impacts on the environment must be analyzed for a holistic assessment. Impacts on social sustainability are neglected in this paper.

The energy efficiency for the steam reforming of biogas was calculated under the assumption that an additional 25 kWh of thermal energy is required for the process. This includes, on the one hand, the reformation energy but also, for example, the energy required to heat the gases or to evaporate the liquid water. Depending on the ambient parameters or energetic optimizations, the value for the required thermal energy can be higher or lower,

Energies 2023, 16, 5066

109

resulting in a change in energy efficiency. This assumption is based on experience values and should be validated. This also applies to all other assumptions, such as the efficiency of the LNG or fuel-cell propulsion and the compression/cooling of the gases in Use Case B or the average biogas composition.

Considering the economic comparison and the wide price ranges, as well as the different cost structures of both process options, requires a deeper economic analysis, taking into consideration more influencing factors such as investment and operational costs, location, and the time of operational initiation of the plant.

6. Summary and Conclusions

Climate change constitutes one of the most pressing challenges of our times. Innovative technologies are needed that can reduce GHG emissions in all sectors. Action is especially necessary in the energy sector, which is currently responsible for the largest amount of emissions. A transition to cleaner energy sources like renewable energies and carbon-free energy carriers like H₂ is thus urgently needed. Additionally, to limit human-induced global warming to less than 2 °C compared to pre-industrial levels, NETs or CDR technologies are indispensable.

In this paper, biomethane production from biogas with CCS (process option 1) was compared to biohydrogen production via the SMR of biogas with CCS (process option 2), i.e., a HyBECCS approach. The analysis considers technical, economic, and environmental aspects of the two options. The aim was to provide insights into the trade-offs to be considered.

Considering energy efficiency, the production of biohydrogen in process option 2 results in an energy loss of approximately 37% in terms of the absolute LHV of the products per kg of biomass or biogas input. Looking at usable driving energy for heavy-duty transport applications, using H_2 produced in process option 2 still results in a comparable loss of approx. 4.2% in terms of usable driving energy for the heavy-duty transport sector. However, this is within the range of several uncertainties due to the assumptions made, e.g., assumed powertrain efficiencies, LNG liquefaction (10–20% energy loss), and different driving behaviors. In conclusion, the useful energy generation for heavy-duty transport can be expected to be in a similar order of magnitude for both process options.

To determine the negative emission potential, the theoretical maximum amount of storable biogenic CO₂ (NEP) was calculated for both process options. In comparison, process option 1 results in an NEP of 1.65 kg CO₂ per kg of CH₄ produced, while the HyBECCS approach (process option 2) shows an NEP of 4.4 kg CO₂ for the same amount of biomass input. This corresponds to about 2.7 times the amount of storable biogenic CO₂. Depending on the quality of the biogas process step (CH₄ content in the biogas), even more biogenic CO₂ can be captured and stored with the HyBECCS approach compared to process option 1. The NEP of process option 2 increases to up to 5 times the amount of process option 1 for a CH₄ content in biogas of 70 vol%. For a CH₄ content of biogas of 55 vol%, the NEP for process option 2 is still 2.2 times higher than that of process option 1.

The costs required for the two process options can vary significantly depending on the specific technology and scale of the reforming plant considered. Biogas SMR requires additional equipment leading to higher investment costs. Further to this, the current market prices for CH_4 are generally lower than for H_2 . However, the long-term price forecasts for H_2 indicate a potential for lower prices due to cost reductions through technological advances and economies of scale. Negative emissions are a marketable product already. In the future, ongoing standardization processes will increase their visibility, reliability, and, thus, market potentials in Europe and worldwide. This market potential correlates with the NEP and is thus about 2.7 times higher for process option 2.

In conclusion, producing biohydrogen via the SMR of biogas offers the possibility to capture between 2.2 and 5 times more of the carbon bound in biomass compared to biomethane production. Captured biogenic CO_2 has the potential to create negative emissions that can be marketed as a byproduct besides the resulting energy carrier. However, the

production of biohydrogen from biomethane results in an energy loss of approximately 37% in terms of the LHV of the product gases, but this loss can likely be partially compensated by efficient driving technologies such as FCEVs.

7. Outlook

Negative emissions will gain relevance in the coming decades, further highlighting the benefits of HyBECCS approaches such as process option 2 in this paper. Overall, they have the potential to make a significant contribution to lowering GHG emissions and establishing CO₂ sinks, thereby making the energy sector more sustainable. A deep economic analysis, considering several influencing factors such as investment and operational costs, product price forecasts, scale, location, and the time of a plant's operational initiation, must be carried out for a thorough comparison of the presented process options. Furthermore, the ecological impact of both process options has to be analyzed in a holistic assessment, including the GHG emissions caused throughout the entire lifecycle of each product couple. Therefore, developing an approach to determine the positive and negative impacts on the climate to identify the approach with the highest climate-change mitigation potential is required. This includes the development of an approach to evaluate different CCS and CCU options concerning their respective GHG mitigation potential. To allow for comparisons with other HyBECCS and NETs, the central Key Performance Indicators (KPIs) of leveled costs of carbon-negative H₂ (LCCNH) and leveled costs of negative emissions (LCNE) must be calculated for both process options [30]. Comparing the KPIs is essential for cost efficiency in NET and HyBECCS development and deployment.

Market integration of HyBECCS plants depends on establishing infrastructure for transporting both H_2 and CO_2 . The different options and their future perspectives must be analyzed considering their influence on the feasibility and economics of HyBECCS approaches. Furthermore, the feasibility of specific HyBECCS plants depends on the market for both products. Thus, analyzing the present and future market potential of H_2 and CO_2 in selected geographical areas is needed.

Author Contributions: Conceptualization, methodology, formal analysis: J.F.; writing (original draft), investigation, data curation, visualization: J.F., E.G., S.H., S.Z., T.S. and H.-P.S.; writing (review and editing): R.M. and A.S.; funding acquisition: J.F., R.M. and A.S. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the German Federal Ministry for Economic Affairs and Energy (grant number 03EI5407C); the Ministry of the Environment, Climate Protection and the Energy Sector Baden-Wuerttemberg, Germany and the European Regional Development Fund (ERDF) (grant number 2076391); and the German Federal Ministry of Education and Research (grant number 03SF0669B).

Data Availability Statement: Data is contained within the article.

Acknowledgments: The research shown in this paper was conducted within the projects "RhoTech", "SmartBioH2" and "H2Wood". The authors gratefully acknowledge the financial support of the German Federal Ministry for Economic Affairs and Energy, the Ministry of the Environment, Climate Protection and the Energy Sector Baden-Wuerttemberg, Germany, the European Regional Development Fund (ERDF) and the German Federal Ministry of Education and Research.

Conflicts of Interest: The authors declare no conflict of interest.

References

- IPCC. Global Warming of 1.5 °C: IPCC Special Report on Impacts of Global Warming of 1.5 °C above Pre-industrial Levels in Context of Strengthening Response to Climate Change, Sustainable Development, and Efforts to Eradicate Poverty; Cambridge University Press: Cambridge, UK, 2022; pp. 1–24. [CrossRef]
- 2. IPCC. *IPCC Special Report on Carbon Dioxide Capture and Storage*; Cambridge University Press: Cambridge, UK; New York, NY, USA, 2005; ISBN 13 9780521866439.
- 3. Weiland, P. Biogas production: Current state and perspectives. Appl. Microbiol. Biotechnol. 2010, 85, 849–860. [CrossRef]

111

- 4. Awe, O.W.; Zhao, Y.; Nzihou, A.; Minh, D.P.; Lyczko, N. A Review of Biogas Utilisation, Purification and Upgrading Technologies. *Waste Biomass Valor.* **2017**, *8*, 267–283. [CrossRef]
- 5. Ryckebosch, E.; Drouillon, M.; Vervaeren, H. Techniques for transformation of biogas to biomethane. *Biomass Bioenergy* **2011**, *35*, 1633–1645. [CrossRef]
- 6. Hajjaji, N.; Martinez, S.; Trably, E.; Steyer, J.-P.; Helias, A. Life cycle assessment of hydrogen production from biogas reforming. *Int. J. Hydrogen Energy* **2016**, *41*, 6064–6075. [CrossRef]
- 7. Kumar, R.; Kumar, A.; Pal, A. Overview of hydrogen production from biogas reforming: Technological advancement. *Int. J. Hydrogen Energy* **2022**, *47*, 34831–34855. [CrossRef]
- Shabani, B.; Andrews, J. Hydrogen and Fuel Cells. In *Energy Sustainability through Green Energy*; Springer: New Delhi, India, 2015; pp. 453–491. [CrossRef]
- 9. Sharma, A.; Kumar Kar, S. Energy Sustainability through Green Energy, 1st ed.; Springer: New Delhi, India, 2015; ISBN 9788132223375.
- 10. Full, J.; Merseburg, S.; Miehe, R.; Sauer, A. A New Perspective for Climate Change Mitigation—Introducing Carbon-Negative Hydrogen Production from Biomass with Carbon Capture and Storage (HyBECCS). *Sustainability* **2021**, *13*, 4026. [CrossRef]
- 11. Nahar, G.; Mote, D.; Dupont, V. Hydrogen production from reforming of biogas: Review of technological advances and an Indian perspective. *Renew. Sustain. Energy Rev.* **2017**, *76*, 1032–1052. [CrossRef]
- 12. Gao, Y.; Jiang, J.; Meng, Y.; Yan, F.; Aihemaiti, A. A review of recent developments in hydrogen production via biogas dry reforming. *Energy Convers. Manag.* **2018**, *171*, 133–155. [CrossRef]
- 13. Rosha, P.; Rosha, A.K.; Ibrahim, H.; Kumar, S. Recent advances in biogas upgrading to value added products: A review. *Int. J. Hydrogen Energy* **2021**, *46*, 21318–21337. [CrossRef]
- 14. Ivys Adsorption Inc. Xebec. Available online: https://xebecinc.com/ (accessed on 19 June 2023).
- 15. Air Liquide. A World Leader in Gases, Technologies and Services for Industry and Health. Available online: https://www.airliquide.com/ (accessed on 19 June 2023).
- 16. WS Reformer GmbH. X-to-Hydrogen. Available online: https://wsreformer.de (accessed on 19 June 2023).
- European Commission. Advanced Direct Biogas Fuel Processor for Robust and Cost-Effective Decentralised Hydrogen Production: Using Renewable Biogas to Produce Green Hydrogen. 2021. Available online: https://cordis.europa.eu/article/id/435357-using-renewable-biogas-to-produce-green-hydrogen (accessed on 19 June 2023).
- European Commission. BIOgas Membrane Reformer for Decentralized Hydrogen Production. Available online: https://doi.org/ 10.3030/671459 (accessed on 19 June 2023).
- Deutsches Biomasseforschungszentrum. BioH2Ref—Dezentrale Wasserstoffaufbereitung von Biogas durch Dampfreformierung. 2023. Available online: https://www.energetische-biomassenutzung.de/projekte-partner/details/project/show/Project/BioH2 Ref-706 (accessed on 19 June 2023).
- Mirza, N.; Kearns, D. State of the Art: CCS Technologies 2022. 2022. Available online: https://www.globalccsinstitute.com/wp-content/uploads/2022/05/State-of-the-Art-CCS-Technologies-2022.pdf (accessed on 21 April 2023).
- 21. Aresta, M. Carbon Dioxide as Chemical Feedstock; Wiley-VCH: Weinheim, Germany, 2010; ISBN 9783527324750.
- 22. Kemper, J. Biomass and carbon dioxide capture and storage: A review. Int. J. Greenh. Gas Control 2015, 40, 401–430. [CrossRef]
- 23. Full, J.; Trauner, M.; Miehe, R.; Sauer, A. Carbon-Negative Hydrogen Production (HyBECCS) from Organic Waste Materials in Germany: How to Estimate Bioenergy and Greenhouse Gas Mitigation Potential. *Energies* **2021**, *14*, 7741. [CrossRef]
- 24. Strevett, K.A.; Vieth, R.F.; Grasso, D. Chemo-autotrophic biogas purification for methane enrichment: Mechanism and kinetics. *Chem. Eng. J.* **1995**, *58*, 71–79. [CrossRef]
- 25. Abd, A.A.; Othman, M.R.; Majdi, H.S.; Helwani, Z. Green route for biomethane and hydrogen production via integration of biogas upgrading using pressure swing adsorption and steam-methane reforming process. *Renew. Energy* **2023**, *210*, 64–78. [CrossRef]
- 26. Chaemchuen, S.; Zhou, K.; Verpoort, F. From Biogas to Biofuel: Materials Used for Biogas Cleaning to Biomethane. *ChemBioEng Rev.* 2016, *3*, 250–265. [CrossRef]
- 27. Ardolino, F.; Cardamone, G.F.; Parrillo, F.; Arena, U. Biogas-to-biomethane upgrading: A comparative review and assessment in a life cycle perspective. *Renew. Sustain. Energy Rev.* 2021, 139, 110588. [CrossRef]
- 28. Rostrup-Nielsen, J.; Christiansen, L.J. *Concepts in Syngas Manufacture*, 10th ed.; Catalytic Science Series; Imperial College Press: London, UK, 2011; ISBN 9781908978004.
- LeValley, T.L.; Richard, A.R.; Fan, M. The progress in water gas shift and steam reforming hydrogen production technologies—A review. *Int. J. Hydrogen Energy* 2014, 39, 16983–17000. [CrossRef]
- Full, J.; Ziehn, S.; Geller, M.; Miehe, R.; Sauer, A. Carbon-negative hydrogen production: Fundamentals for a techno-economic and environmental assessment of HyBECCS approaches. *GCB Bioenergy* 2022, 14, 597–619. [CrossRef]
- 31. Boot-Handford, M.E.; Abanades, J.C.; Anthony, E.J.; Blunt, M.J.; Brandani, S.; Mac Dowell, N.; Fernández, J.R.; Ferrari, M.-C.; Gross, R.; Hallett, J.P.; et al. Carbon capture and storage update. *Energy Environ. Sci.* **2014**, *7*, 130–189. [CrossRef]
- 32. Verduyn, M.; Geerlings, H.; van Mossel, G.; Vijayakumari, S. Review of the various CO₂ mineralization product forms. *Energy Procedia* **2011**, *4*, 2885–2892. [CrossRef]
- 33. Da Pires Mata Costa, L.; Micheline Vaz de Miranda, D.; Couto de Oliveira, A.C.; Falcon, L.; Stella Silva Pimenta, M.; Guilherme Bessa, I.; Juarez Wouters, S.; Andrade, M.H.S.; Pinto, J.C. Capture and Reuse of Carbon Dioxide (CO₂) for a Plastics Circular Economy: A Review. *Processes* 2021, *9*, 759. [CrossRef]

- 34. Ravikumar, D.; Keoleian, G.A.; Miller, S.A.; Sick, V. Assessing the Relative Climate Impact of Carbon Utilization for Concrete, Chemical, and Mineral Production. *Environ. Sci. Technol.* **2021**, *55*, 12019–12031. [CrossRef]
- Full, J.; Baumgarten, Y.; Dokur, Y.; Miehe, R.; Sauer, A. Biogas Plants as Hydrogen Production Facilities and Greenhouse Gas Sinks: Technology Comparison, Challenges and Potentials for Carbon Negative Hydrogen Production (HyBECCS). *Procedia CIRP* 2022, 107, 185–190. [CrossRef]
- 36. Tomita, E.; Kawahara, N.; Azimov, U. *Biogas Combustion Engines for Green Energy Generation*; Springer International Publishing: Cham, Swizterland, 2022; ISBN 9783030945381.
- 37. Mishra, A.; Kumar, M.; Bolan, N.S.; Kapley, A.; Kumar, R.; Singh, L. Multidimensional approaches of biogas production and up-gradation: Opportunities and challenges. *Bioresour. Technol.* **2021**, *338*, 125514. [CrossRef] [PubMed]
- 38. Sun, Q.; Li, H.; Yan, J.; Liu, L.; Yu, Z.; Yu, X. Selection of appropriate biogas upgrading technology-a review of biogas cleaning, upgrading and utilisation. *Renew. Sustain. Energy Rev.* **2015**, *51*, 521–532. [CrossRef]
- Mallon, W.; Buit, L.; van Wingerden, J.; Lemmens, H.; Eldrup, N.H. Costs of CO₂ Transportation Infrastructures. *Energy Procedia* 2013, 37, 2969–2980. [CrossRef]
- Becattini, V.; Gabrielli, P.; Antonini, C.; Campos, J.; Acquilino, A.; Sansavini, G.; Mazzotti, M. Carbon dioxide capture, transport and storage supply chains: Optimal economic and environmental performance of infrastructure rollout. *Int. J. Greenh. Gas Control* 2022, 117, 103635. [CrossRef]
- Schaffelhofer, K. Neuer Gasmotor MAN E3872 mit 44.0% Wirkungsgrad und 735 kW Leistung aus nur 12 Zylindern, München. 2021. Available online: https://press.mantruckandbus.com/corporate/de/neuer-gasmotor-man-e3872-mit-44 0{-}{-}wirkungsgrad-und-735-kw-leistung-aus-nur-12-zylindern/ (accessed on 2 May 2023).
- 42. Mayr, K.; Hofer, F.; Ragowsky, G.; Gruber, W.; Arnberge, A.; Kabza, A.; Wolf, P.; Schmidt, M.; Jörissen, L. Systemvergleich Zwischen Wasserstoffverbrennungsmotor und Brennstoffzelle im Schweren Nutzfahrzeug: Eine Technische und Ökonomische Analyse Zweier Antriebskonzepte, Stuttgart. 2021. Available online: https://www.e-mobilbw.de/fileadmin/media/e-mobilbw/ Publikationen/Studien/e-mobilBW-Studie_H2-Systemvergleich.pdf (accessed on 2 May 2023).
- 43. Cullen, D.A.; Neyerlin, K.C.; Ahluwalia, R.K.; Mukundan, R.; More, K.L.; Borup, R.L.; Weber, A.Z.; Myers, D.J.; Kusoglu, A. New roads and challenges for fuel cells in heavy-duty transportation. *Nat. Energy* **2021**, *6*, 462–474. [CrossRef]
- 44. Schmidt, T. Wasserstofftechnik: Grundlagen, Systeme, Anwendung, Wirtschaft, 2nd ed.; Carl Hanser Verlag: München, Germany, 2022; ISBN 9783446473539.
- 45. Berne, E.K.; Berner, R.A. *Global Environment: Water, Air, and Geochemical Cycles*; Princeton University Press: Princeton, NJ, USA, 2012; ISBN 9780691136783.
- 46. Deutsches Institut für Normung e.V. DIN EN ISO, 14040:2021-02. Environmental Management—Life Cycle Assessment—Principles and Framework (ISO 14040:2006 + Amd 1:2020). German Version EN ISO 14040:2006 + A1:2020; Beuth Verlag: Berlin, Germany, 2021. [CrossRef]
- 47. Klöpffer, W.; Grahl, B. Ökobilanz (LCA): Ein Leitfaden für Ausbildung und Beruf; Wiley-VCH: Weinheim, Germany, 2009; ISBN 9783527320431.
- 48. Gupta, R.B. Hydrogen Fuel: Production, Transport, and Storage, 1st ed.; CRC Press: Boca Raton, FL, USA, 2008; ISBN 9781420045758.
- 49. Acar, C.; Dincer, I. Analysis and assessment of a continuous-type hybrid photoelectrochemical system for hydrogen production. *Int. J. Hydrogen Energy* **2014**, *39*, 15362–15372. [CrossRef]
- 50. Green, M.A. Hydrogen Safety Issues Compared to Safety Issues with Methane and Propane. *AIP Conf. Proc.* 2006, 823, 319–326. [CrossRef]
- 51. Moradi, R.; Groth, K.M. Hydrogen storage and delivery: Review of the state of the art technologies and risk and reliability analysis. *Int. J. Hydrogen Energy* **2019**, *44*, 12254–12269. [CrossRef]
- Ullah Khan, I.; Hafiz Dzarfan Othman, M.; Hashim, H.; Matsuura, T.; Ismail, A.F.; Rezaei-DashtArzhandi, M.; Wan Azelee, I. Biogas as a renewable energy fuel—A review of biogas upgrading, utilisation and storage. *Energy Convers. Manag.* 2017, 150, 277–294. [CrossRef]
- 53. Mertins, A.; Heiker, M.; Rosenberger, S.; Wawer, T. Competition in the conversion of the gas grid: Is the future of biogas biomethane or hydrogen? *Int. J. Hydrogen Energy* 2023, in press. [CrossRef]
- 54. Kumar, S.; Kwon, H.-T.; Choi, K.-H.; Lim, W.; Cho, J.H.; Tak, K.; Moon, I. LNG: An eco-friendly cryogenic fuel for sustainable development. *Appl. Energy* **2011**, *88*, 4264–4273. [CrossRef]
- 55. Economides, M.J.; Sun, K.; Subero, G. Compressed Natural Gas (CNG): An Alternative to Liquefied Natural Gas (LNG). SPE Prod. Oper. 2006, 21, 318–324. [CrossRef]
- 56. Smajla, I.; Karasalihović Sedlar, D.; Drljača, B.; Jukić, L. Fuel Switch to LNG in Heavy Truck Traffic. *Energies* 2019, 12, 515. [CrossRef]
- 57. Gustafsson, M.; Svensson, N. Cleaner heavy transports—Environmental and economic analysis of liquefied natural gas and biomethane. *J. Clean. Prod.* 2021, 278, 123535. [CrossRef]
- Ogden, J.; Jaffe, A.M.; Scheitrum, D.; McDonald, Z.; Miller, M. Natural gas as a bridge to hydrogen transportation fuel: Insights from the literature. *Energy Policy* 2018, 115, 317–329. [CrossRef]
- Johansson, J.; Leirnes, J.; Walton, K.; Yilmaz, E.; Zindel, E. Hydrogen Power and Heat with Siemens Energy Gas Turbines: Reliable and Flexible Carbon-Free Energy, Erlangen. 2022. Available online: https://www.siemens-energy.com/global/en/offerings/ technical-papers/download-hydrogen-gas-turbine-readiness-white-paper.html (accessed on 3 May 2023).

113

- 60. Stephan, P.; Stephan, K. Dubbel Taschenbuch für den Maschinenbau 1: Grundlagen und Tabellen; Springer: Berlin/Heidelberg, Germany, 2020; ISBN 9783662597118.
- 61. Hermeling, W. Handbuch für den LNG- und CNG-Praktiker; Springer Fachmedien Wiesbaden: Wiesbaden, Germany, 2020; ISBN 9783658285517.
- 62. Bossel, U. Wasserstoff löst keine Energieprobleme. TATuP 2006, 15, 27–33. [CrossRef]
- 63. Trávníček, P.; Vítěz, T.; Koutný, T. The Equation of State of Biogas. Acta Univ. Agric. Silvic. Mendel. Brun. 2017, 65, 537–543. [CrossRef]
- 64. Kapoor, R.; Ghosh, P.; Kumar, M.; Vijay, V.K. Evaluation of biogas upgrading technologies and future perspectives: A review. *Environ. Sci. Pollut. Res. Int.* **2019**, *26*, 11631–11661. [CrossRef]
- 65. Liu, K.; Song, C.; Subramani, V. *Hydrogen and Syngas Production and Purification Technologies*; John Wiley & Sons Inc.: Hoboken, NJ, USA, 2010; ISBN 9780470561256.
- 66. Barton, J.W.; Bowers, S.L.; Butcher, T.G.; Harshman, R.A.; Lee, G.D.; Warfield, L. Appendix. In *Specifications, Tolerances, and Other Technical Requirements for Weighing and Measuring Devices*; National Institute of Standards and Technology: Gaithersburg, MD, USA, 2021. [CrossRef]
- 67. Apostolou, D.; Xydis, G. A literature review on hydrogen refuelling stations and infrastructure. Current status and future prospects. *Renew. Sustain. Energy Rev.* 2019, 113, 109292. [CrossRef]
- 68. European Commission. Fuel Price Comparison. 2022. Available online: https://alternative-fuels-observatory.ec.europa.eu/ consumer-portal/fuel-price-comparison (accessed on 1 June 2023).
- 69. Sandhu, G.S.; Frey, H.C.; Bartelt-Hunt, S.; Jones, E. Real-world activity, fuel use, and emissions of heavy-duty compressed natural gas refuse trucks. *Sci. Total Environ.* **2021**, *761*, 143323. [CrossRef] [PubMed]
- 70. European Commission. Quarterly Report: On European Gas Markets. 2022. Available online: https://energy.ec.europa.eu/ document/download/0e69ba99-07e9-4aed-8c34-dbfa3ae0add7_en?filename=Quarterly%20report%20on%20European%20 gas%20markets%20Q1_2022.pdf (accessed on 19 June 2023).
- 71. Duan, Z.; Mei, N.; Feng, L.; Yu, S.; Jiang, Z.; Chen, D.; Xu, X.; Hong, J. Research on Hydrogen Consumption and Driving Range of Hydrogen Fuel Cell Vehicle under the CLTC-P Condition. *WEVJ* **2022**, *13*, 9. [CrossRef]
- 72. Abbas, H.F.; Wan Daud, W. Hydrogen production by methane decomposition: A review. *Int. J. Hydrogen Energy* **2010**, *35*, 1160–1190. [CrossRef]
- 73. Miyagawa, T.; Goto, M. Hydrogen Production Cost Forecasts since the 1970s and Implications for Technological Development. *Energies* **2022**, *15*, 4375. [CrossRef]
- 74. Poralla, M.; Honegger, M.; Ahonen, H.-M.; Michaelowa, A.; Weber, A.-K. *Sewage Treatment for the Skies: Mobilising Carbon Dioxide Removal through Public Policies and Private Financing*; NET-Rapido Consortium and Perspectives Climate Research: London, UK; Freiburg im Breisgau, Germany, 2021.
- 75. Puro.Earth. CORC Supplier Listing. Available online: https://puro.earth/CORC-CO2-removal-certificate (accessed on 15 May 2023).
- 76. European Commission. Certification of Carbon Removals—EU Rules. Available online: https://ec.europa.eu/info/law/better-regulation/have-your-say/initiatives/13172-Certification-of-carbon-removals-EU-rules_en (accessed on 1 June 2023).

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.

Appendix C: Paper C

Contributions to the work

Title: Carbon-negative hydrogen production: Fundamentals for a techno-economic and environmental assessment of HyBECCS approaches

Published in: GCB Bioenergy 14.5 (2022): 597-619.

DOI: https://doi.org/10.1111/gcbb.12932

CRediT (Contributor Role Taxonomy) author statement:

Authors	Johannes Full	Sonja Ziehn	Marcel Geller	Robert Miehe	Alexander Sauer
Conceptualization	\checkmark				
Methodology	\checkmark				
Formal analysis	\checkmark				
Investigation	\checkmark	\checkmark	\checkmark		
Data Curation	\checkmark	\checkmark	\checkmark		
Writing (Original Draft)	\checkmark	\checkmark	\checkmark		
Writing (Review and Editing)				\checkmark	\checkmark
Visualization	\checkmark	\checkmark	\checkmark		
Project administration	\checkmark				
Funding acquisition	\checkmark			\checkmark	\checkmark

The co-authors hereby agree to the contributions summarized in the table avove.

<u>Johannes Full</u> Name <u>Sonja Ziehn</u> Name <u>Marcel Geller</u> Name <u>Robert Miehe</u> Name

<u>Alexander Sauer</u> Name

Signature s. Ziehn Signature i Wes

Signature

Signature

Laur

Signaturé

25.11.2022 Date 25.11.2022 Date 25.11.2022 Date 3.5.2023 Date 03.05.2023 Date DOI: 10.1111/gcbb.12932

RESEARCH REPORT

Carbon-negative hydrogen production: Fundamentals for a techno-economic and environmental assessment of HyBECCS approaches

Accepted: 16 February 2022

Johannes Full^{1,2} | Sonja Ziehn¹ | Marcel Geller¹ | Robert Miehe¹ | Alexander Sauer^{1,2}

¹Fraunhofer Institute for Manufacturing Engineering and Automation IPA, Stuttgart, Germany

²Institute for Energy Efficiency in Production (EEP), University of Stuttgart, Stuttgart, Germany

Correspondence

Johannes Full, Fraunhofer Institute for Manufacturing Engineering and Automation IPA, 70569 Stuttgart, Germany.

Email: johannes.full@ipa.fraunhofer.de

Funding information

This research was funded by the German Federal Ministry for Economic Affairs and Energy and the Ministry of Rural Affairs and Consumer Protection of Baden-Wuerttemberg, Germany.

Abstract

In order to achieve greenhouse gas neutrality, hydrogen generated from renewable sources will play an important role. Additionally, as underlined in the Sixth Assessment Report of the Intergovernmental Panel on Climate Change (IPCC), new technologies to remove greenhouse gases from the atmosphere are required on a large scale. A novel concept for hydrogen production with net negative emissions referred to as HyBECCS (Hydrogen Bioenergy with Carbon Capture and Storage) combines these two purposes in one technological approach. The HyBECCS concept combines biohydrogen production from biomass with the capture and storage of biogenic carbon dioxide. Various technology combinations of HyBECCS processes are possible, whose ecological effects and economic viability need to be analyzed in order to provide a basis for comparison and decision-making. This paper presents fundamentals for the techno-economic and environmental evaluation of HyBECCS approaches. Transferable frameworks on system boundaries as well as emission, cost, and revenue streams are defined and specifics for the application of existing assessment methods are elaborated. In addition, peculiarities concerning the HyBECCS approach with respect to political regulatory measures and interrelationships between economics and ecology are outlined. Based on these considerations, two key performance indicators (KPIs) are established, referred to as levelized cost of carbon-negative hydrogen (LCCNH) and of negative emissions (LCNE). Both KPIs allow deciding whether a specific HyBECCS project is economically viable and allows its comparison with different hydrogen, energy provision, or negative emission technologies (NETs).

KEYWORDS

HyBECCS, carbon capture and storage, CCS, carbon capture and utilization, CCU, hydrogen, bioenergy, BECCS, DACCS, PCC, biohydrogen, carbon-negative hydrogen, green hydrogen, negative emission technologies, NET, carbon dioxide removal, CDR, bioenergy, waste to energy, waste to hydrogen, biogenic hydrogen, biogenic carbon dioxide, climate protection, greenhouse gas mitigation, GHG mitigation, climate change mitigation, renewable hydrogen, green technologies, carbon removal certificates, carbon removal certification, CRC

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

© 2022 The Authors. GCB Bioenergy published by John Wiley & Sons Ltd.

GCB-BIOENERGY WILEY



1 | INTRODUCTION

On a global level, only with the combination of greenhouse gas (GHG) reductions and the fast deployment of negative emission technologies (NETs), humanity has a chance to limit global warming to 2°C or even 1.5°C, as strikingly alerted by the Intergovernmental Panel on Climate Change (IPCC) in its latest report (Masson-Delmotte, et al., 2021). The IPCC clearly states that a successful limitation of global warming to 1.5°C builds upon two pillars. One is the fast and deep reduction of anthropogenic GHG emissions from fossil fuel combustion and the destruction of natural carbon storages such as forests and wetlands, referred to as GHG mitigation or reduction. The second pillar is the active removal of carbon dioxide (CO_2) from the atmosphere through NETs (IPCC). Only with the combination of these two approaches, humanity can possibly stay on an emission pathway toward 1.5°C. This pathway reaches net-zero GHG emissions around 2050 and enters into a net-negative GHG emission zone afterward (Masson-Delmotte, et al., 2021; IPCC). All scenarios calculate with gross negative GHG emissions, implying the implementation of large-scale NETs (Honegger & Reiner, 2018a, p. 307; Poralla et al., 2021, p. 19; Rogelj et al., 2015; UNEP, 2017, p. 59; UNFCCC). Based on this global target, over 100 countries, together emitting over 50% of global GHG, have set themselves the goal of GHG neutrality or negativity to be achieved at different points in time before 2060 (Poralla et al., 2021, p. 26).

Most nations, as well as international associations, consider technological innovations as key factors to avoid GHG emissions in order to achieve their respective climate goals. In this context, hydrogen (H₂) production from renewable resources as well as the generation of negative emissions by means of bioenergy with carbon capture and storage (BECCS) are frequently listed as key technologies for this transition (Full, Trauner, et al., 2021; FCH JU, 2019; IPCC; Miehe et al., 2019; Wietschel et al., 2021). One innovative approach that combines hydrogen production with BECCS is the production of biohydrogen from biomass with capture and storage of the co-produced biogenic carbon dioxide, abbreviated to HyBECCS (Hydrogen Bioenergy with Carbon Capture and Storage) (Full, Merseberg, et al., 2021; Full, Trauner, et al., 2021). A selection from multiple technology combination options is the next step in the development of the HyBECCS approach. These options have to be comparable and evaluable in terms of ecological and economic aspects. Therefore, the research objective of this paper is to introduce fundamentals for a techno-economic and ecological assessment of HyBECCS approaches or individual HyBECCS projects based on standardized methods and indicators. It shall provide a basis for the optimization



FULL ET AL.

FIGURE 1 Schematic overview of the contents of the paper

toward a more sustainable HyBECCS process design by maximizing the positive effects on the climate and economic returns. This might serve as a reference for private investors, as well as political decision makers, to evaluate HyBECCS approaches and enable targeted public subsidies and private investments. The assessment fundamentals are elaborated as summarized in Figure 1.

First of all, a classification and delimitation of HyBECCS are essential to describe the approach in the context of negative emission technologies. Furthermore, the classification serves to ensure a consistent use of the term "HyBECCS" in further research, thus ensuring its ecological integrity. Uniform system boundaries and framework conditions are elaborated to provide a basis for standardized economic and environmental evaluation of different HyBECCS projects. Within the regulatory status quo of negative emission technologies, existing literature on BECCS and direct air carbon capture and storage (DACCS) is reviewed to identify and highlight the urgent next steps for NET implementation such as HyBECCS. Additionally, the need to develop and implement internalization mechanisms for negative emission technologies is examined. Once these steps are completed, the fundamentals for conducting an environmental and techno-economic assessment of a HyBECCS project are outlined. Based on the environmental and techno-economic assessment, two key performance indicators (KPI) are derived, namely the levelized cost of carbon-negative hydrogen (LCCNH) and the levelized cost of negative emissions (LCNE). These KPIs allow for a comparison of HyBECCS projects with different hydrogen or energy provision technologies and NETs.

2 | BASICS

2.1 | Classification and Delimitation of the HyBECCS Approach

As elaborated by Full et al. 2021, the generation of biogenic CO_2 as a by-product in most biomass to hydrogen production processes is the basis for the HyBECCS approach. This applies to both thermochemical and biotechnological technologies of biohydrogen production (Full, Merseburg, et al., 2021; Full, Trauner, et al., 2021). A simplified illustration of the approach is shown in Figure 2: On the left side, atmospheric CO_2 and water are stored via photosynthesis in biomass. In the biomass conversion plant, the biomass is converted mainly into hydrogen and biogenic CO₂. Hydrogen can be used, for example, in fuel cells to generate electricity or heat. With O₂ as educt and water as a product of the reaction, mass balance equilibrium is established for oxygen and water. Biogenic CO₂ can be stored or used in the long term, leading to negative emissions (Full, Merseburg, et al., 2021; Full, Trauner, et al., 2021). The biomass conversion plant and the carbon storage together constitute the HyBECCS project.

According to the work of Full et al. 2021, the HyBECCS approach can be split into four basic process steps: 1. substrate biomass pretreatment, 2. production of biohydrogen and biogenic CO_2 , 3. separation of the product gases CO_2 and H₂, and 4. their processing for use of hydrogen and long-term storage or use of biogenic CO₂. For each of the four steps, there are different technology options to choose. The main technological advantage of the HyBECCS approach is the energy-efficient capture of biogenic CO₂ as a point source (Full, Merseburg, et al., 2021). It, furthermore, provides a double effect on climate mitigation through the provision of the emission-free energy carrier hydrogen and its potential as a CO₂ sink. The comparative advantage of HyBECCS over many other bioenergy production approaches, such as the production of hydrocarbon-based biofuels like biogas or biomethane, is that the energy carrier distributed to the end users is CO_2 -free and the CO_2 emissions already occur at the plant, where they can be centrally captured (Full, Merseburg, et al., 2021). Another advantage of HyBECCS over the BECCS approach of combined heat and power (CHP) plants burning biomass is the higher flexibility and usability of hydrogen compared to electricity and heat. Hydrogen can be stored over long periods of time, transported over long distances without

Photosynthesis / biomass generation Production of hydrogen and biogenic carbon dioxide Biogenic carbon dioxide

FIGURE 2 Simplified illustration of the HyBECCS approach (arrow = mass flow, dashed arrow = mass balance equilibrium)

significant losses, and has multiple applications beyond the generation of electricity and heat (Wietschel et al., 2021) (S. 20ff). Examples are its use as auxiliary in many industrial applications and as basic material, e.g., in the urea and fertilizer production (IEA) (S. 32); (Hydrogen Council). Based on these advantages, HyBECCS is considered to be a promising and highly effective climate protection technology that is expected to play a major role in future energy systems by making a significant contribution to meeting the world's rapidly growing demand for new renewable energy sources, such as biohydrogen, and negative emissions at the same time (Coalition for Negative Emissions; FCH JU, 2019; IPCC; Wietschel et al., 2021). The approach can also be seen as part of the biological transformation process, aiming to build the economy upon sustainable, nature-based processes instead of fossil resource depletion (Full et al., 2019; Full, Baumgarten, et al., 2021; Miehe et al., 2018, 2019, 2020).

-WILEY

HyBECCS can be placed in the context of existing technology clusters and definitions some of which are not yet clearly distinguished. Experts criticize the continued lack of clarity, for example, on the exact definition of NETs, carbon dioxide removal (CDR), and carbon capture and storage (CCS) (Poralla et al., 2021; Tanzer & Ramírez, 2019). To establish a consistent use of terms in the interest of the ecological integrity of HyBECCS, the following clarifications are stated. Carbon capture and storage (CCS) embraces all anthropogenic activities to capture and store carbon dioxide in the-yet to be defined-long term (IPCC). CO_2 to be stored in CCS approaches can be of any type: fossil, atmospheric or biogenic. It is, however, of utmost importance to keep in mind that CCS of fossil carbon dioxide emissions is not a NET technology and can never generate net-negative emissions. Fossil CO₂ and other GHG are emitted when burning fossil fuels or when their derivatives decompose after use. Even in the yet utopic case of a 100% carbon dioxide capture rate, the process always remains GHG positive, or carbon-neutral when not considering process emissions and other GHG than CO₂. Fossil CCS can, in the best case, only reduce the total GHG footprint of the overall process, qualifying thus merely as a so-called GHG mitigation activity or GHG reduction activity (Poralla et al., 2021). In contrast, carbon dioxide removal (CDR) and NET refer to anthropogenic activities that remove CO₂ from the atmosphere and thus lower its overall atmospheric concentration. Hence, only the CCS of atmospheric or biogenic CO_2 can qualify as CDR or NET. NETs and CDR can be grouped into nature-based approaches such as afforestation or reforestation, wetland restoration, enhanced weathering, soil carbon sequestration and accelerated mineralization, on the one hand, and technological or hybrid approaches such as DACCS and BECCS, on the other hand (Poralla

et al., 2021, p. 18). Biohydrogen production technologies that enable the storage of biogenic carbon in solid form, such as biomethane pyrolysis, can also be considered as HyBECCS concepts. However, the focus of consideration in this paper is on the capture and storage of gaseous CO₂ from biohydrogen processes. According to the definitions above, all considered concepts (CCS, NET, CDR) can apply to the HyBECCS concept. It is able to remove carbon dioxide from the atmosphere and can, therefore, be considered as CDR via CCS of biogenic CO₂. This implicit condition is specified in more detail in 3.2. In addition to storage options, the biogenic carbon dioxide can also be combined with carbon utilization (CU). Carbon utilization comprises all approaches where carbon dioxide is used as feedstock or auxiliary (IPCC). In case the utilized carbon dioxide is stored over a very long period for such use, the respective CU approach can be considered as carbon storage (CS). However, the definition of how long CO_2 has to be stored in order to count as "stored" with a positive effect on global climate is still pending for CS, in general, and for long-term CU, in special. The term "CS" within "HyBECCS" comprises the long-term carbon storage as well as the long-term extraction of CO_2 from the atmosphere via carbon utilization. The following Table 1 summarizes the characteristics of the technologies and technology clusters under consideration.

Also for hydrogen, clarification is needed due to inconsistencies and imprecisions in literature. Hydrogen is often categorized in colors. In the "theory of colors" of hydrogen, grey hydrogen is extracted from fossil resources, thus releasing fossil GHG emissions and accelerating global warming. Blue hydrogen is grey hydrogen combined with CCS. Fossil CO₂ emissions from hydrogen production by steam reforming of natural gas are captured and stored in this process. Even though blue hydrogen is partly considered as low-carbon or even carbon-neutral (Bundesministerium für Bildung und Forschung - BMBF; Minli et al., 2021; Noussan et al., 2021; Yan et al., 2020), recent lifecycle assessments show that it causes as little as 9%–12% less GHG than grey hydrogen, especially due to high fugitive methane emissions (Howarth & Jacobson,

2021). Turquoise hydrogen, derived from the pyrolysis of methane and producing solid carbon, can theoretically be GHG neutral when the used electricity is entirely renewable (Newborough & Cooley, 2020). Green hydrogen is either only defined as hydrogen from water electrolysis using electricity generated with renewable energy or additionally includes biomass-based hydrogen production methods, the product being called biogenic hydrogen (Newborough & Cooley, 2020) or biohydrogen (Full, Merseburg, et al., 2021). Hydrogen from electrolysis and biohydrogen is mostly not GHG-neutral due to process GHG emissions, but can mostly be considered as low-carbon hydrogen. Despite on-going initiatives, there is no established uniform threshold of GHG intensity for hydrogen to qualify as "green" or "low-carbon" (certifhy; Hinicio; Newborough & Cooley, 2020). Therefore, hydrogen is categorized in this paper as GHG-neutral, -positive, or -negative according to its GHG balance. The prefix "GHG-" can also be replaced by "carbon-" in this context, resulting in the terms carbon-neutral, -positive, or -negative, which are, respectively, used as synonymous alternatives.

2.2 System Boundaries and Framework Conditions

To evaluate the economic efficiency and the environmental performance of HyBECCS approaches, a uniform framework is required. Figure 3 represents the system boundaries for techno-economic and environmental assessment of HyBECCS approaches. The system boundary of HyBECCS projects (B3) is divided into two system elements, each of which can be considered separately with its own system boundary: The generation of hydrogen and biogenic CO_2 from biomass is referred to as the system element HyBECC and as system boundary B1. The storage of carbon dioxide or its long-term utilization is referred to as system element CS/CU and as system boundary B2.

In addition to the actual step of hydrogen and carbon dioxide derivation from biomass substrates, the HyBECC

TABLE 1 Summarized classification of the considered technologies and technology clusters and their respective characteristics. (Own elaboration based on AACE International (2020), Couper (2003), Kunysz (2020), Matthes et al. (2020), Towler and Sinnott (2013), Weber (2016), Zimmermann et al. (2020))

Type of CO ₂	CCS	CDR/NET	GHG balance	Technology options
Fossil	\checkmark	×	Positive	
Biogenic from non-renewable biomass (e.g. slow-growing biomass)	\checkmark	×	Positive	
Biogenic from renewable biomass (e.g. fast- growing biomass)	\checkmark	1	Potentially negative	BECCS, HyBECCS
Atmospheric	1	1	Potentially negative	DACCS

system boundary B1 also includes the pretreatment steps of the biomass input (I) as well as gas processing of the resulting hydrogen and the captured CO₂. The CS/CU system boundary B2 includes the transport of captured CO_2 to the storage or utilization site, the storage itself and its maintenance (CS) as well as the processing of the CO₂ for utilization (CU). Both subsystem boundaries B1 and B2 can be considered individually or combined to evaluate the HyBECCS system (B3). Related costs (C), revenues (R), and emissions (E) are assigned to the two subsystem boundaries B1 and B2 and summed up in the HyBECCS system boundary B3. In general, the emission flows E1, E2, E2b, and E3 can be of different types (e.g., noise, pollutants, and greenhouse gases) and the system abstraction above can be used for the analysis of emission flows of any kind. For the present analysis, however, all emission streams only represent the respective greenhouse gas emissions. Their characteristics and interrelationships are described in more detail in the following.

The system element HyBECC (B1) causes GHG emissions (E1), for example, indirectly through the biomass generation, procurement, pretreatment, and process energy demand. Various types of costs (C1) are incurred, such as capital and operating expenditures. Revenues of the HyBECC system element are generated by the sales of hydrogen (R1a) and of biogenic carbon dioxide (R1b), summed up in R1. In addition to the production of hydrogen and biogenic CO_2 , fermentation digestates or residues (D) are produced as by-products. Depending on the composition of these residues, either further revenues can be generated by selling them, e.g., as fertilizer, or costs that can incur for their disposal. Therefore, they can either be included in the costs C1 or the revenues R1. The same applies for the purchase of the biomass input (I), which can either generate costs or an income, the latter in case of waste or residues for which disposal fees incur (Thrän & Pfeiffer, 2013). It should be noted that using residual and waste biomass for biohydrogen production, the long-term storage of the carbon bound in the digestates can lead to further negative emissions. This is the case if, for example, the residual is used as agricultural fertilizer and a substantial amount of the carbon in the digestate remains in the soil (Antonini et al., 2020). This effect would be taken into account within B1, where it would be included negatively in the total of E1.

System boundary B2 covers the CS/CU system element. The emissions in subsystem B2 contain the process emissions of the CS/CU process (E2b) and the amount of biogenic CO₂ permanently stored or used, being the gross negative GHG emissions (E2a). Since the amount of biogenic CO₂ permanently stored or used has been removed from the atmosphere during the biomass growth

B-BIOENERGY

120

via photosynthesis, negative emissions can be generated (Kemper, 2015). E2a enters the emissions balance as a negative value and is fundamental for HyBECCS approaches to achieve an overall negative GHG balance. It must be noted that the physical amount of negative GHG emissions (E2a) actually stored is not necessarily equal to the amount of biogenic CO₂, which is transferred from balance boundary B1 to B2. This difference is due to potential leakages (L) in the system B2 which can occur during the transport to the storage site or the storage itself. Revenues (R2) accrue when being recompensated through incentives for generating negative emissions. It is important to keep in mind that, in the case of CS, the subsystem B2 does not generate any other income than the compensation for negative emissions. The costs (E2) are composed of the capital and the operational expenditures including the purchase price for biogenic CO₂ produced within B1 (C2b) as well as all other costs for the carbon storage or utilization process (C2a). The material flow of biogenic CO₂ leaving the HyBECC system boundary B1 corresponds to the material flow of biogenic CO₂ entering the CS/CU system boundary B2 as feedstock for negative emission generation. From the perspective of system boundary B1, the biogenic CO₂ generates a revenue flow (R1b) from its sale, whereas, for B2, the same biogenic CO₂ material flow generates costs (C2b) for its purchase. For CU approaches, revenues depend on the respective product. Summarized, the total costs of the HyBECCS approach (C3) incurred within the system boundary B3 can then be calculated according to Equation 1.

$$C3 = C1 + C2 = C1 + C2a + C2b \tag{1}$$

The total profit P3 achieved through HyBECCS can be calculated according to Equation 2. It is summed up from the revenues generated by the sales of hydrogen R1a, of the biogenic carbon dioxide R1b, and of the negative emissions or long-term carbon dioxide utilization R2, which together give the overall revenues for biohydrogen production and negative emissions R3, minus the total costs C3. In the case of separate operation of the system elements HyBECC and CS/CU, the respective profit can also be calculated within the subordinate system boundary B1 and B2. The profit for the production of hydrogen and biogenic CO₂ (P1) is then considered independently of the profit from the storage or long-term use of the biogenic CO₂ (P2).

$$P3 = P1 + P2 = R1 - C1 + R2 - C2$$
(2)
= R1a + R1b - C1 + R2 - C2a - C2b = R3 - C3

To assess the environmental impact of HyBECCS, all emissions of the system boundary B3 have to be considered. As explained above, this paper focuses on the

FIGURE 3 System boundaries (B) for HyBECCS approaches with mass flows for biomass input (I), hydrogen (H2), carbon dioxide (CO2), stored carbon dioxide referred to as gross negative emissions (GNE), residues or digestate from biohydrogen production (D) and carbon dioxide leakages (L), monetary flows

divided into costs (C), revenues (R), and

emission flows (E)

N°	Framework condition (FC)	Consequence
FC1	E3 < 0	HyBECCS project = Negative Emission Technology (NET) project and Biohydrogen = Carbon-Negative Hydrogen (CNH) and permission to issue carbon removal certificates (CRC)
FC2	P3 > 0	HyBECCS project is economically viable

TABLE 2 Framework conditions (FC) for HyBECCS approaches and projects

impact on the global climate of HyBECCS approaches. For this purpose, only positive and negative (stored) GHG emissions are to be balanced. The sum of all GHG emissions within B3 can be considered as its greenhouse gas footprint, which can be calculated according to Equation 3.

$$E3 = E1 + E2 = E1 - E2a + E2b$$
(3)

In case, the greenhouse gas footprint of a specific HyBECCS project (E3) is negative, it actually generates net negative GHG emissions and thus qualifies as a NET project. The co-generated hydrogen can then be considered carbon-negative hydrogen (CNH).

The definitions and framework conditions (FC) summarized in Table 2 are established as described above and apply to all HyBECCS projects:

This means that only HyBECCS projects that fulfill the first framework condition (FC1) above (E3 < 0) are NET and CDR and its produced biohydrogen is allowed to be qualified as CNH. Furthermore, only in case the total profit is positive (P3 > 0), the HyBECCS project is economically viable as the second framework condition (FC2) states.

Regulatory Status Quo of NETs 2.3

All emission pathways to limit global warming to 2°C or even 1.5°C calculate with gross-negative GHG emissions, and thus CDR, from 2030 on (Honegger & Reiner, 2018a; Poralla et al., 2021; Rogelj et al., 2015; UNEP, 2017). The CDR amount needed to reach global climate goals is estimated to be 5–20 GtCO₂ per year, summing up to 444–1000 GtCO₂ until 2100 (Boysen; Honegger & Reiner, 2018b; Poralla et al., 2021). With each lost year that CDR is not implemented on a large scale, the challenge of sufficient



adoption of these technologies increases (Poralla et al., 2021, p. 19). However, the scientific evidence and urgency of NET implementation are not reflected in reality: Globally, government policies and private sector initiatives supporting NETs are very sparse to date (Jeffery et al., 2020, p. 9f). The few governmental or private ambitions focus on biological CDR approaches such as afforestation and reforestation. NETs like DACCS and BECCS are supported by single countries such as Sweden, some public funds, and very few private entities. Poralla et al. (2021) provide a good overview on the supportive instruments existing worldwide. Emerging voluntary markets for CDR credits also primarily focus on biological storages (Jeffery et al., 2020, p. 9f; Poralla et al., 2021, p. 31). Incentives to mobilize technological approaches linked with permanent underground storage, such as BECCS and DACCS, are widely missing. As the contribution to GHG mitigation, an external benefit of NETs, does not generate revenues by itself and thus does not translate into an economic advantage for the operator, the externalities of GHG mitigation and emission must be internalized (Mankiw, 2001, p. 172). One option is the internalization of the external costs caused by GHG emissions by carbon pricing mechanisms. However, the internalization approach with resultbased crediting of the actual negative emissions achieved is only one option of financial NET support and primarily adequate for mature NETs (Poralla et al., 2021, p. 23). Even though the technology readiness level of BECCS is relatively high with 6-9 for bioenergy processes and 4-7 for CCS, there are only single large-scale BECCS plants in operation (Poralla et al., 2021, p. 17). At the current stage of BECCS implementation, technology selection, learning curves, scale-up, and cost reduction promise significant efficiency gains. For the current development stage of NETs, subsidies for research, design, development, and demonstration (RDD&D) are thus more appropriate and urgently needed (Honegger, 2020). Besides direct subsidies, further instruments to speed up the NET implementation range from tax credits, concessional loans, grants to feed-in tariffs, all instruments successfully applied for the deployment of renewable energies around the world (Poralla et al., 2021, p. 23). Furthermore, existing barriers that hinder the implementation of CDR, such as the regulatory uncertainty on the export of CO₂ for geological storage, or on the geological storage itself, have to be abolished (Poralla et al., 2021, p. 31). There will be no NET activities without external incentives due to the special economics of these approaches. This is especially true-and unfortunate from a climate protection perspective-for HyBECCS as it has multiple benefits for the global climate, not only actively removing CO₂ from the atmosphere, but at the same time mitigating fossil GHG emissions by generating biohydrogen, potentially pushing fossil alternatives from

the market. The special economics of HyBECCS and how to solve the dependency between the political framework, its economics, and climatic impact, will be discussed in the following subsection.

₩ILEY

2.4 | Interrelationships between Ecology and Economics

-BIOENERGY

The economic and ecological, especially climatic benefits of HyBECCS approaches can be best explained by looking at the two main products of HyBECCS: biohydrogen and negative emissions. On the one hand, the generation of hydrogen from biomass provides the valuable energy carrier hydrogen at a low GHG footprint and is carbon neutral when only considering the biomass conversion. The reason for this is that the CO₂ emitted during the biomass conversion equals the amount of CO₂ removed from the atmosphere during the growth of the biomass (Singh & Rathore, 2017). GHG emissions of HyBECCS processes are thus limited to process emissions from the hydrogen production process (incl. biomass generation) (E1) and CCS (E2b). Compared to fossil energy carriers, the use of biohydrogen avoids fossil GHG emissions from fossil fuel generation and its combustion or the decomposition of its derivatives after utilization. This contribution to GHG mitigation is the first external benefit of HyBECCS, as depicted as white box in Figure 4. It does not generate revenues by itself and thus does not translate into an economic advantage for the operator. Without external intervention, the GHG reduction would not entail any financial added value. Ideally following the so-called "polluter pays principle," the external costs caused by GHG emissions are internalized. Common approaches are cap-and-trade systems such as the EU-ETS, where GHG-emitting installations have to purchase emission allowances, or carbon taxes, potentially complemented by baseline-and-credit mechanisms such as the Clean Development Mechanism (CDM) (European Commission). If applied sufficiently, this translates into a comparative price advantage for biohydrogen in case it causes less GHG emissions and thus has lower GHG internalization costs to bear than fossil alternatives, as depicted as grey boxes in Figure 4. It is important to mention that the GHG emissions of all technologies, even if considered as low- or zero-emission technologies, should be included in such GHG internalization mechanisms to allow the comparison of the social costs of HyBECCS products with alternatives for energy or hydrogen provision. The negative CO₂ emissions depicted as green box in Figure 4 constitute the second external benefit of HyBECCS. The generation of negative emissions is linked with costs for CO₂ capture, transportation, and storage (cf. Equation 1). CCS cost estimations in the

603



FIGURE 4 External benefits of HyBECCS compared to a (fossil) reference. green bar = negative emissions; grey bar = process emissions; dashed white bar = difference of process emissions resulting in external benefit (Not to scale)

literature vary broadly and depend, beyond others, on the CO_2 concentration at source, plant scale, transportation mode, the distance to the storage site, and the type of storage (Agora Energiewende und Wuppertal Institut, 2019; Poralla et al., 2021; Trippe, 2013). These costs are from the operator's perspective—not (over)compensated by any income. The external benefit of the negative emissions for global climate must thus be internalized to make them happen.

High concentration and purity of the CO₂ to be captured using HyBECCS approaches compared to direct air capture (DAC) or post-combustion capture (PCC) as well as the emission-free use of hydrogen resulting in a higher possible CO₂ capture rate compared to hydrocarbonbased biofuels generated with other BECCS technologies are further advantages of the HyBECCS approach. They reduce the total energy demand and increase the CO₂ yield (Full, Merseburg, et al., 2021, p. 15; Singh & Rathore, 2017). However, these advantages translate into lower production costs-and comparative advantages of HyBECCS over other NETs. They thus do not have to be internalized. The production of hydrogen as a marketable product likewise constitutes a direct economic advantage of HyBECCS over other NETs, who often lack any revenue other than from negative emissions. This reduces the external financial need of HyBECCS approaches or increases the possible profit margin, making it a highly valuable model for both efficient GHG mitigation and removal in the pursuit of the 1.5°C goal. The example of HyBECCS and its externalities-a typical case of market failure in political economics (Mankiw, 2001, p. 172)-proves the urgency of developing and implementing internalization

mechanisms for negative emissions and of sharpening existing instruments for GHG reduction.

3 | FUNDAMENTALS AND GUIDELINES FOR ENVIRONMENTAL ASSESSMENT

Over the last decades, the increasing awareness of the negative impact of human activities on the environment and the regulatory adoption of environmental awareness gave birth to many ecological guidelines and standards. They range from macro-level approaches such as national GHG inventories over meso-level analysis like environmental assessments to micro-level assessments addressing products or services (DIN e.V. DIN EN ISO 14040, 2021, p. 8; Frischknecht, p. 17; Hauschild et al., 2018; Kaltschmitt & Schebek, 2015; Klöpffer & Grahl, 2009). HyBECCS projects can have multiple impacts on the environment, such as acidification and eutrophication or the depletion of the ozone layer and abiotic resources. They depend on the feedstock used, the distances and mode of transportation of biomass and products as well as the chosen conversion technology (Fajardy et al., 2019). All those impacts have to be evaluated in an environmental assessment (EA). The EA has to follow the local legislation or, if not existent or insufficient, best international practice, always aiming to minimize negative impacts. However, the key of HyBECCS in climate protection lies in its ability to deliver biohydrogen and negative emissions. Any HyBECCS approach or project can only contribute as NET in case it actually has a negative GHG balance, which means that it actually removes more CO_2 from the atmosphere than it emits GHG. In such a case, the co-generated hydrogen can be considered as CNH as per framework condition FC1 above. The focus herein is to develop a guideline to assess the ability of a HyBECCS project to deliver negative emissions and to produce CNH. Even though all environmental impacts are equally important, they are not included in the further assessment, as established evaluation methods already exist. The following fundamentals for environmental assessment of HyBECCS are limited to considerations on climate change mitigation. A distinction is made between technology level (3.1.) assessing the technology choice as a first step and project level (3.2.) evaluating the actual GHG balance of the specific HyBECCS project.

3.1 | Eligibility on a Technology Level

There are two conditions that need to be fulfilled on a technology level for a HyBECCS project in order to be able to deliver net negative emissions: The theoretical ability for CO₂ removal and the theoretical permanence of the CO_2 storage. The theoretical ability for CO_2 removal is given for all types of BECCS approaches, as the CO₂ in the product gas is removed from the atmosphere during biomass growth. Accordingly, this has also to be verified for HyBECCS technology options. Controversy exists over biomass substrates that are not clearly assigned. For example, very slow-growing biomass, such as peat, does not qualify as renewable (Kaltschmitt & Schebek, 2015, p. 246). Consequently, for HyBECCS or BECCS approaches in general, a positive list of eligible technologies and raw material has to be elaborated and adapted to the state of the art and technology over time. Only in case technology and raw material from that list are applied, the HyBECCS project has the theoretical ability to generate negative emissions. The second condition for a HyBECCS approach in order to deliver negative emissions is that the captured CO_2 is permanently removed from the atmosphere. Different storage options are characterized by different time horizons of storage as well as different risk profiles concerning reversibility and leakage. Considering the current state of the art, HyBECCS approaches can be combined with three different storage options: (1) mineral carbonation, also known as mineralization or enhanced weathering, (2) geological underground storage, for instance in depleted oil/gas reservoirs or saline aquifers, and (3) in long-term CCU application, such as in the built environment in low-carbon concrete (Azapagic et al., 2018, p. 21; IPCC, p. 39). Mineralized carbon is safely stored in the long term, whereas geological underground storage is estimated to be safe for over 1000 years when best practice is applied (IPCC, 2005b; Möllersten et al., 2020; Poralla

et al., 2021, p. 21). The third storage option for biogenic CO₂ from HyBECCS, long-term CCU applications, has a higher risk of reversals, insecurity concerning the time horizon of storage, and further challenges such as monitoring the permanence due to the dispersed places of use (Otto et al., 2017). Moreover, there is no clear definition of how long CO₂ has to be safely stored in order to count as negative emission with an effective impact on global climate. For the CO₂ storage technologies, a positive list of eligible approaches has to be elaborated and updated over time to reflect the state of the art. Only in case technology from that list is applied, the HyBECCS project has the theoretical ability to generate negative emissions. In case a HyBECCS approach or project fulfills both conditions on the technology eligibility, it has the theoretical ability to generate negative emissions and CNH, which means that it has the theoretical ability to fulfill the framework condition FC1 above.

WILEY

3.2 | Eligibility and Assessment on a Project Level

Even though a HyBECCS approach has the theoretical ability on the technological level to deliver net negative emissions according to 3.1., this does not mean that every HyBECCS project applying such a technological approach is able to do so, which means that not every HyBECCS project is able to fulfill framework condition FC1 above. The actual GHG footprint highly depends, for instance, on the raw material production, the efficiency of the bioenergy conversion, and the distance to the CO₂ storage site. For each project, the expected climatic performance thus has to be assessed ex-ante before its implementation and monitored ex-post during its operation. The relevant category to evaluate the contribution of HyBECCS to a global negative emission budget are the GHG emissions, internationally convened to be quantified in CO₂ equivalents (CO_2eq) and with regard to the global warming potential over 100 years (GWP₁₀₀) (DIN e.V. DIN EN ISO 14044), p. 31; IPCC, 2015c; p. 87; Lozanovski, 2013, p. 60). A proven practice to quantify the GHG emissions of projects and services is a life cycle assessment (LCA) in line with the regulations of the international standards ISO 14040 and 14044 (DIN e.V. DIN EN ISO 14040: 2021; Klöpffer & Grahl, 2009; Kaltschmitt & Schebek, 2015; DIN e.V. DIN EN ISO 14044: Lozanovski, 2013). The application of these LCA standards guarantees the scientific veracity of the assessment and is the basis for the comparability of the results with competing approaches. Furthermore, the holistic perspective of an LCA is especially suitable for new approaches like HyBECCS. In addition to GHG emissions, potential sustainability issues other than the contribution

605

to global warming can be revealed during the inventory phase of the LCA. Specifics for the determination of the GHG footprint of a HyBECCS project are elaborated in the following.

First, as the LCA allows climatic assessments relative to a specific product or service, a so-called functional unit has to be defined. To answer the question whether a HyBECCS project generates negative emissions, the functional unit shall be one ton of CO_2 permanently stored. For the climatic appraisal of the HyBECCS approach, negative emissions are thus considered to be the main product of the process. Hydrogen is considered as co-product. For different valuation approaches like the economic and the holistic assessment, however, other perspectives can be chosen. Equation 3 above can thus be converted as follows, now expressing the GHG emission factor (EF) of the negative emissions, in t CO_2 eq per t CO_2 stored.

$$EF = \frac{E1 + E2b}{|E2a|} \tag{4}$$

In case the emission factor is smaller than one (EF < 1), the HyBECCS project generates net negative emissions and the eligibility to qualify as NET is proven; the framework condition FC1 above is fulfilled. The hydrogen produced can thus be referred to as CNH. In case of an emission factor bigger than one (EF > 1), more process GHG emissions are generated than CO₂ is permanently stored, the project generates net positive emissions and contributes to global warming. The comparison of the emission factors of different HyBECCS approaches or NETs, in general, allows conclusions on the efficiency of the different approaches. The GHG emissions of HyBECCS to be considered are the following, according to Figure 3 above: The process GHG emissions emitted during the HyBECC process of producing hydrogen and biogenic CO₂ (E1), the process GHG emissions emitted for the storage (CS) or long-term use (CU) of the biogenic CO₂ (E2b) and the amount of CO_2 stored in the long term, being the negative CO_2 emissions (E2a). The GHG footprint of the HyBECCS project can be calculated according to Equation 3 and allows the determination of the emission factor EF. It shows whether-in the case of ex-ante estimations-the project is expected to generate net negative emissions and-in case of measurements during operation-whether it actually generates net negative emissions. In case the total emissions (E3) are negative, which means that EF is smaller than one, the HyBECCS project actually generates net negative emissions and the generated hydrogen can thus be considered as CNH, as defined in the framework condition FC1 above. The specifics to determine E1, E2a, and E2b are discussed in detail in the following. The calculation of the process GHG emissions E1 and E2b shall, in general, follow the rules of the ISO 14040 and 14044 standards.

However, the following specifications and choices given by the ISO framework should be followed in order to safeguard the ecological integrity of HyBECCS approaches. For the sake of comparability and fairness, the same rules must be applied to competing approaches. The process GHG emissions of HyBECC (E1) are generated during all steps necessary to provide hydrogen and biogenic carbon dioxide. The provision of the biomass used, which can have an important share in the total GHG emissions, must be considered (IE, p. 26). This is especially important for international traded biomass in order to avoid GHG accounting leakages in global supply chains. All direct emissions for machines, transportation, etc., and indirect emissions generated, for example, during the production of fuels or fertilizers, the generation of purchased electricity, or due to land-use changes shall be included (Hauschild et al., 2018, p. 494). However, double accounting of GHG has to be avoided. No direct and indirect GHG emissions are allocated on waste and residues. They "leave" the upstream product system without being burdened with any GHG footprint (DIN e.V. DIN EN ISO 14044, p. 26; Thrän & Pfeiffer, 2013, p. 87). This means that for bioenergy approaches using waste or residues as raw material for the energy conversion, no GHG emissions must be taken into account for the provision of such biomass except for its transportation to the plant and its processing. This rule honors the ecological advantages of approaches growing or obtaining raw material in an efficient and ecological way, whereas dedicated, land- or resource-intensive cultivation of biomass is disadvantaged. A recent study shows that optimization of the hydrogen production process can lead to negative emissions even without CCS: In case the digestates from the hydrogen production are used as fertilizer and a major part of its carbon content is stored in the soil, the GHG balance of the hydrogen production over its entire life cycle can be negative (Antonini et al., 2020). Consequently, when CCS is additionally applied, the GHG negativity of this optimized HyBECCS approach and its contribution to climate change mitigation can increase.

The process emissions E1 should furthermore ideally include the making, construction, and deconstruction of the infrastructure needed for the biohydrogen generation process in order to realistically calculate the total GHG emissions. For the main process, the biomass conversion to hydrogen, all relevant material and energy streams have to be taken into account. For the present impact category, the global warming potential, minor inputs, and outputs can be cut-off to a maximum of 5%, the impact of such cut-off however being scrutinized in a sensitivity analysis (Lozanovski, 2013, p. 56). For hydrogen, all steps of further processing and transportation until its intended end use should be included in the LCA. Long transportation routes to the site of usage might spoil the ecological advantages of hydrogen. The inclusion of all transportation emissions thus favors decentralized projects using local biomass over large-scale projects located far from any site of consumption. This might alleviate the disadvantages of missing economies of scale of decentralized plants. The required conditions (e.g., pureness and pressure) of the produced hydrogen depend on the envisaged use. It also affects the energy need, thus the GHG footprint and needs to be defined beforehand. The same applies for the required characteristics of the biogenic CO_2 , to be defined as so-called reference flow in the LCA (Klöpffer & Grahl, 2009, p. 37). One major advantage of hydrogen is that it oxidizes to water; its use is thus CO₂-emission-free. Fossil fuels, on the contrary, emit GHG when burned. A "cradle-to-grave" approach of the LCA is thus advisable for a realistic comparison of fossil and hydrogen-based applications in order to reward this benefit. The HyBECC system element simultaneously delivers hydrogen and biogenic CO₂. Potentially, further co-products like digestates usable as fertilizer or other usable components in the product gas can be generated. An allocation of the total GHG emission over the different products would be possible (DIN e.V. DIN EN ISO 14044, p. 26; Lozanovski, 2013, p. 43). However, the complete accounting of all GHG is essential for the present assessment and thus no allocation is undertaken in order to avoid an accounting leakage through co-products not covered by GHG internalization mechanisms. Concerning the biogenic carbon dioxide, only its generation, capture, compression and, if needed, conditioning for transportation (IPCC, p. 5) are to be included in the LCA of the HyBECC system boundary B1. Assumptions for the project lifetime can be taken from proven comparable technologies. Dividing the total GHG emissions (E1) resulting from the HyBECC system element by the product quantity, results in the GHG intensity of such product, for example, in tCO₂eq/kWh H₂. The comparison of the results from several HyBECC approaches allows a prioritization according to the climatic efficiency of biomass conversion.

The further lifecycle of the biogenic CO_2 is covered by the system boundary B2. For process, GHG emissions of CS/CU (E2b), multiple factors like the CO_2 transportation mode, distance and storage or long-term utilization technology (Honegger & Reiner, 2018b, p. 308; Poralla et al., 2021, p. 21) influence the GHG footprint of CS/CU and have thus to be considered. The calculation of the GHG emissions emitted for the storage of the biogenic carbon dioxide shall, in general, follow the same rules elaborated above for the HyBECC process within the framework of the ISO 14040 and 14044. The system boundary encompasses all processes needed for the transportation of the CO_2 from its site of generation to the storage site, the storage itself, for example, the injection into the underground, as well as all provisions necessary to assure the permanence of the storage. Further compression might be needed at the storage site in case the pressure of the arriving CO_2 is not sufficient for the storage (IPCC, p. 11). All direct and indirect process GHG emissions are to be included in the LCA. In the case of biogenic CO₂, losses of this product through equipment leaks or accidental releases are GHG-neutral and only result in a loss of negative emissions. Contrary to fugitive emissions of fossil CO₂ (IPCC, p. 7), they do not have to be included in the LCA. They should, however, be avoided. The first result of the LCA of boundary B2 is the total GHG emission accruing for the transportation and storage of the biogenic carbon dioxide. The second result is, when dividing the total GHG by the product amount, the GHG intensity of the product, in tCO₂eq/ tCO₂stored. The comparison of different CS/CU approaches allows selecting efficient technologies to minimize their impact on global warming.

B-BIOENERGY

Negative emissions (E2a) are the total amount of biogenic CO₂ stored in the long term. The stored CO₂ amount has to be measured continuously by proven measurement equipment at the injection or storage site (IPCC, p. 11). CO₂ leakage from transportation and compression is already taken into account at this point, as they occur before measurement. Other types of leakages, such as CO₂ migration and blowouts, have to be avoided by a diligent selection of technology and site as well as the use of adequate equipment, materials, and techniques (IPCC). However, as they cannot be excluded to happen, risk deductions depending on the technology applied are likely to be necessary for a realistic estimation of negative emissions over the long term. Such deductions may be applied as default values depending on the specifics of the CS plant and need to be elaborated and introduced on a global level. Pre-operational estimations of the negative emissions are possible by stoichiometric approaches and experimental results on the product gas composition, giving the amount of biogenic CO₂. After the deduction of transportation and storage losses, as well as risk deductions for leakages at the storage site, the amount of CO₂ actually stored in the long term can be estimated.

After the determination of E1, E2a, and E2b, the emission factor EF of the specific HyBECCS project under evaluation can be calculated according to Equation 4 above. In case the emission factor is smaller than one, the HyBECCS project generates net-negative GHG emissions and, consequently, carbon-negative hydrogen. This means that the framework condition FC1 above is fulfilled. Such HyBECCS project thus contributes to the deceleration of global warming.

It is important that E1, E2a, and E2b are stated separately in order to allow evaluations on the absolute impact of the HyBECCS approach or project on the environment.

607

WILEY

608 WILEY- GCB-BIOENERGY

Two projects with the same amount of net-negative emissions could show significant differences in the absolute amount of positive (E1 + E2b) and negative (E2a) emissions. The separate statement is thus especially important to save potentially scarce CO₂ storage capacities (Azapagic et al., 2018) and to safeguard resources. Furthermore, a separate statement is required in case different incentive schemes (or prices) apply to the different types of emissions. Another reason for the separate statement is that the climate goals for emission reduction, on the one hand, and negative emissions, on the other, might change over time, reflecting the progressing defossilization and the increasing need of global net negative emissions. Moreover, different emission caps for each process type (BECC, CCS) might be introduced to enforce the technological progress and would only be verifiable in case of separate statements.

The following Table 3 shows an overview of the most relevant aspects for the calculation of the GHG balance of a HyBECCS project.

4 | FUNDAMENTALS AND GUIDELINES FOR TECHNO-ECONOMIC ASSESSMENT

HyBECCS approaches have to be economically viable in order to bring its products-hydrogen and negative emissions-to the market. Techno-economic assessments of production processes are of great importance in order to examine their profitability, to identify potential barriers for a successful market integration like cost drivers, and give an important basis for technical development. Furthermore, they help to identify business cases and are valuable tools for decision making and monitoring (Zimmermann et al., 2020). Generally, to estimate profitability, the capital and operational expenditures must be determined. At the example of the HyBECCS processes, these must be derived for the biohydrogen production as well as for the capture, transport, and storage or the use of biogenic carbon dioxide, according to Equation 1. Fundamentals for the cost estimation of HyBECCS processes and proposals for the selection of suitable estimation methods are outlined in 4.1. Once both the capital and operating costs of the HyBECC and the CS/CU plant are determined, the total profit P3 of the HyBECCS plant can be calculated by summing up the costs (C3) and revenues (R3), according to Equation 2. In case of higher revenues than costs (R3 > C3), the HyBECCS project is profitable (P3 > 0), which represents the second framework condition (FC2). If the costs outweigh the revenues (C3 > R3), the project should not be implemented from an economic point of view.

In system boundary B1, the main product is biohydrogen, which generates revenues (R1a) as shown in 2.2. The additional revenues from the sale of co-products, here coming from the sales of biogenic CO₂ (R1b), are deducted from the total production costs. Dividing the annualized production costs by the annual product quantity results in the unit production costs, in the case of hydrogen the levelized cost of energy (LCOE) (IE). The LCOE can be related either to mass (e.g., in EUR/ kg H_2) or to net calorific value (e.g., in EUR/kWh H_2) (Lozanovski, 2013; Matthes et al., 2020). In case biohydrogen from HyBECCS is only compared to other hydrogen generation systems, mass-based production costs can be chosen. A more general comparison with other types of energy conversion technologies can be envisaged by choosing the energy-based LCOE. In case only the carbon storage system (B2) is considered, where negative emissions are the main product, the unit production costs are expressed in EUR/tCO2stored. In case the HyBECCS system boundary (B3) is considered, the main product may either be biohydrogen or negative emissions. In such a case, the revenues from the sales of the co-product must be deducted from the total costs when determining the unit production costs of the main product. In case the negative emissions are chosen as the main product, the comparison of the HyBECCS approach with other NETs is possible. The outcome is the unit production costs of the main product in EUR per unit of the main product, for example, the LCOE of biohydrogen. In the following, the unit production costs before the internalization of external costs and benefits are defined as "internal LCOE" of the product. The GHG internalization costs are stated separately from the internal LCOE. This differentiation applies independently from whether the external costs and benefits are partly or entirely internalized, which is partly the case in some countries and/or sectors with established carbon pricing mechanisms. The separate statement of the GHG internalization costs shall raise the transparency and comparability of different alternatives in terms of their climatic impact. The internal LCOE for hydrogen with biogenic CO₂ as co-product (system boundary B1) is defined, for example, in EUR/ kWh H₂, according to Equation 5.

$$internal \ LCOE \ (B1) = \frac{C1 - R1b}{annual \ H2 \ production}$$
(5)

The internal levelized costs of stored carbon (LCSC) for the system boundary B2 can be defined as follows, in EUR/tCO_2 stored.

internal LCSC (B2) =
$$\frac{C2}{E2a}$$
 (6)

FULL ET AL. GCB-BIOENERGY WILL EX 609
Image: Constrained a substantiable Biology During For a
E1: GHG emissions from HyBECC (Hydrogen Bioenergy & Carbon Capture)
Life Cycle Assessment according to ISO 14040/14044
Impact category: Global warming potential over 100 years (GWP ₁₀₀)
Functional unit: one ton of CO ₂ permanently stored
 "Cradle-to-grave"-approach, including GHG emissions from: Biomass cultivation/provision All direct emissions (transportation, etc.) All indirect emissions (incl. land use changes, infrastructure, deconstruction)
No allocation (side effect: co-product biohydrogen is "zero-emission")
No GHG allocation on waste and residues \rightarrow "zero-emission" biomass input
Double accounting of GHG has to be avoided
Cut-off of material & energy flows at maximum of 5% \rightarrow effect to be checked in sensitivity analysis
+ Potential sustainability issues of new HyBECCS approaches can be revealed during LCA inventory phase
E2b: GHG emissions from CS (Carbon Transport & Storage)
Life Cycle Assessment according to ISO 14040/14044
Impact category: Global warming potential over 100 years (GWP ₁₀₀)
Functional unit: one ton of CO ₂ permanently stored
 "Cradle-to-grave"-approach, including GHG emissions from: CO₂ transportation from HyBECC plant to the storage site Storage (for example, injection into the underground) O&M (Operation and Maintenance) (all provisions necessary to assure the permanence of the storage)
E2a: Negative Emissions
Unit: tons of CO ₂ permanently stored
Ex-ante estimations through stoichiometric approaches & experimental results on the product gas composition
 Measurements during operation must be: continuously with proven measurement equipment at the injection or storage site Lack of uniform standards for measurement, reporting and verification (MRV)
Risk deductions (for leakage, accidental release) to be applied — Lack of uniform definition of default values for risks

- Scientific & regulatory uncertainty on necessary duration of storage, responsibilities, liabilities, etc.

! Separate statement of absolute amount of E1, E2a and E2b necessary.

! Uniform LCA system boundaries for comparative assessments

The internal levelized costs of negative emissions (LCNE) for the HyBECCS system boundary B3 can be defined according to the following equation, in EUR/ tCO₂stored.

internal LCNE (B3) =
$$\frac{C3 - R1}{E2a}$$
 (7)

The internal LCOE for carbon-negative hydrogen (system boundary B3) is defined as follows, where no revenues from the co-product, the negative emissions, are included. This is because they come from the internalization of the external benefit from negative emissions, not considered in the internal cost calculation (expressed, for example, in EUR/kWh H_2).

internal
$$LCOE = \frac{C3}{annual H2 production}$$
 (8)

However, an important aspect for the economic viability of HyBECCS concepts is monetary incentives for GHG reduction as well as for storing biogenic CO_2 . After presenting the most relevant aspects for the capital and operational cost estimation in 4.1., the internalization of those external benefits will be discussed in 4.2, resulting in the derivation of two key performance indicators (KPIs) for the evaluation of HyBECCS projects: The levelized cost of carbon-negative hydrogen (LCCNH) and negative emissions (LCNE). The chapter closes by presenting an exemplary basic model for HyBECCS regulatory systems in net-zero or net-negative emission economies.

4.1 | Capital and Operational Cost Estimation

Suitable estimating approaches for capital costs of HyBECCS projects like capacity, factor, module, and detail methods are to be chosen in terms of estimation accuracy and effort and must be selected depending on the project planning degree of the HyBECCS project, the data availability, and the objective of the estimation (Weber, 2016). In order to select an adequate method for estimating capital cost for HyBECCS approaches, the current degree of the project definition and some key indicators or information about the cost of the equipment must be known (Kunysz, 2020). The degree of the project definition depends solely on the available information about the project. Certain project states are associated with a certain degree of the project definition. In preliminary, approval and detailed planning it ranges from 10% to 15%, 25% to 35%, and 85% to 95%, respectively (Weber, 2016). Figure 5 illustrates a schematic guide to select one of the mentioned methods described below.

Capacity methods usually are applied for projects with planning degrees below 2% (Kunysz, 2020) for giving order-of-magnitude estimates or to carry out process comparisons. The capital expenditures can be estimated using key indicators such as investment costs and capacity of similar existing plants or through gross annual sales and capital turnover ratio. If the technical sizing has not yet been carried out, but key indicators are known, the capacity methods are suitable (Kunysz, 2020; Towler & Sinnott, 2013). Factorial methods are based on rough technical sizing of the plant and an estimate of the major equipment's purchase cost. They usually are applied at project definition degrees up to about 30% (Weber, 2016). Depending on the applied factorial method, purchase equipment costs are estimated by multiplication with specific factors, for instance, location or material peculiarities (Towler & Sinnott, 2013). Due to its higher project planning degree, factorial methods often deliver higher estimation accuracies than capacity methods, which is why they commonly are used for more detailed estimations like feasibility studies, concept assessments, or preliminary budget approvals (AACE International, 2020). Cost estimations according to the module methods require a functional classification of the plant into modules and are typically deployed for budget approvals. Each module is designed based on a P&I diagram by cost-relevant equipment data, such as material, process temperatures, or pressures. Therefore, higher project planning degrees than 2%, as well as information

about module costs are required (Weber, 2016). Due to the modularity of HyBECCS technologies, the module method is especially suitable. If the project definition lies in between 2 and 30%, either factorial or module methods are applicable (AACE International, 2020). Module methods are more commonly used at higher project definition degrees. Otherwise, in the case of known equipment cost factors, such as Lang- or Hand-factors, the factorial methods can be applied (Couper, 2003; Kunysz, 2020). The detailed estimation methods are based on binding offers or offers from completed similar projects and are applied for cost control or budget compilation. They require the highest project planning degrees and provide the most accurate estimation of these four methods (Kunysz, 2020). If the project definition is above 30% (AACE International, 2020) and binding offers or offers of already completed, similar projects are available, detailed methods are commonly used (Kunysz, 2020; Towler & Sinnott, 2013). If there are no offers, but it is possible to form functional units and information about modular costs is available, the module method can be applied. If equipment cost factors are known, the factorial methods are suitable. Access to reliable data for completely new developed technologies is often not available. As recent data on actual prices for similar process equipment is a highly reliable source, economic data from existing plants can be used as a basis for the techno-economic assessment (Towler & Sinnott, 2013). Therefore, as HyBECCS is a combination of biohydrogen production and CCS/CCU, data from existing BECCS- or CCS-plants and biohydrogen processes can be used for HyBECCS plant cost estimations.

Operational expenditures can be divided into variable and fixed production costs. Variable costs are proportional to plant output, meaning for HyBECCS, increased hydrogen and carbon dioxide production or a greater amount of carbon dioxide stored is generally associated with increasing variable costs. Typical variable costs for HyBECCS plants comprise costs for raw materials like biomass, auxiliaries such as process water or electricity, consumables, waste disposal for biomass residues as well as the costs for hydrogen and carbon dioxide processing and its transport to the CS/CU plant site. Variable costs can usually be decreased by operating the plant more efficiently, for instance by increasing the biomass to hydrogen conversion rate or energy efficiency. Estimating the raw material or consumable costs of a new plant requires an estimate of the amounts of raw materials or consumables needed as well as corresponding prices. The prices can either be forecasted using existing pricing methods or determined from current market prices. Estimating the utility costs of the HyBECCS system requires mass and energy balances as well as a preliminary design of the heat recovery system. Costs for waste



FIGURE 5 Guideline for the selection of a capital cost estimation method (Own elaboration based on AACE International (2020), Couper (2003), Kunysz (2020), Peters et al. (2003), Towler and Sinnott (2013), Weber (2016), Zimmermann et al. (2020))

disposal in HyBECC (B1) may occur, for instance, from biomass residues or wastewater that cannot be recycled or sold as a by-product. As HyBECCS plants can be configurated on a variety due to their modularity regarding the biomass pretreatment, hydrogen production pathway, product gas separation, and its processing as well as in terms of the carbon storage option or carbon utilization case, specific raw materials, consumables, auxiliaries, and additional waste treatments can be required. Fixed costs incur irrespective of the plant output, meaning that a decrease in production does not lead to a reduction in fixed costs. Fixed costs for HyBECCS generally include operating labor, supervision, maintenance, property taxes and insurance, general plant overhead, sales and marketing costs as well as license fees and capital charges. Labor costs, property taxes, and insurance or capital charges depend on the location of a plant (Towler & Sinnott, 2013). Due to the high variety of possible HyBECCS configurations and the dependence of some fixed costs on the plant location, operational expenditures need to be estimated for each HyBECCS plant individually. Guidelines and general reference values for estimating variable and fixed costs are outlined, for example, in Towler and Sinnott, (2013) (p. 373ff) or Peters et al. (2003) (pp. 259ff).

4.2 Internalization of External Costs

The economics of HyBECCS approaches are characterized by a very high dependency on the regulatory framework it is embedded in and, in the ideal case, its ecological benefits. In order to evaluate HyBECCS concepts with regard to their economic viability, these aspects must be considered. In 4.2.1., two single-value KPI are developed taking into account monetary incentive schemes for environmental benefits. These KPIs allow the comparison of different HyBECCS approaches or projects, but also the comparison with other NETs like BECCS approaches or even conventional, potentially fossil-based alternatives. The target KPIs presented merge the outcome of the economic analysis with the outcome of the climatic assessment (cf. 3.). It represents the production costs of a product, including the internalization costs and benefits of the external effects on the global climate. For CNH, it can be considered as the levelized cost of carbon-negative hydrogen (LCCNH), and, for negative emissions, as the levelized cost of negative emissions (LCNE). Uniform incentive schemes for external cost savings through greenhouse gas reduction as well as the creation of negative emission have to be established in order to reach and maintain greenhouse gas neutrality. A basic model for

612 WILEY- GCB-BIOENERG

such a regulatory, result-based system building on carbon removal certificates (CRCs), also known as negative emission certificates (NECs), is shown in 4.2.2.

4.2.1 | Determination of the Levelized Cost of Carbon-Negative Hydrogen (LCCNH) and Negative Emissions (LCNE)

One result of the techno-economic assessment, to be carried out in accordance with the provisions elaborated in 4.1., are the total internal production costs of the main product. To understand the economic interrelationships of the HyBECCS approach in a more holistic view, the internalization of the external costs and benefits of the HyBECCS project must be considered. As elaborated in 2.4., external costs are caused by the process GHG emissions of the HyBECC system element (E1) and the CS/CU system element (E2b). The total amount of E1 and E2b is an outcome of the LCA (cf. 3.2.). For the internalization, the external costs of the GHG emissions have to be monetarized. An option is the offsetting of process GHG emissions with emission reduction certificates from renewable energy projects. Such offsetting renders all HyBECCS products GHG neutral. An alternative to be discussed is the offset of unavoidable process GHG emission against CO₂ permanently stored. After this first internalization step, the biohydrogen can thus be considered carbon-neutral hydrogen. The offsetting means that only the net-negative emissions, meaning those negative emissions that exceed the process emissions of the HyBECCS process, are fully rewarded on the market. The internalization costs have to be added to the total production costs. When the annualized internalization costs are divided by the annual product quantity, they can be added to the internal LCOE. In case the internal LCOE and internalization costs per unit are stated separately, different approaches can be compared in terms of their relative GHG footprint. For such a comparison, the same carbon price has to be applied to all alternatives.

Additionally, the external, GHG-related benefit of the HyBECCS project, which is according to 3.4., the negative emissions, need to be internalized. The determination of the total amount of biogenic carbon dioxide that is permanently stored, referred to as gross negative emissions (E2a), is an outcome of the LCA (cf. 4.2.). In case the project generates net-negative emissions (EF < 1) as per framework condition FC1, the project qualifies to receive a revenue stream (R2) based on a uniform negative emission. The price to be received per unit of gross negative emissions has to be the same for all alternative NETs in order to render the different approaches comparable. In

reality, however, different prices may be negotiated in direct carbon removal certificates (CRCs) purchase contracts. A possible approach for incentives based on CRCs is described in 4.2.2. As the process GHG emissions have been internalized, only the net-negativity is fully rewarded through additional income. In case the hydrogen to be sold can prove that the co-produced biogenic CO_2 is actually used for long-term storage or use and the project generates net-negative emissions (EF < 1), the biohydrogen, already carbon-neutral after the first internalization step, can be considered as CNH. For the internalization of the external profit of the negative emissions, the revenue from the sales of the co-product, the negative emissions, is to be deducted from the total production costs. As a quality feature for mere marketing purposes, either the "net negative intensity" (in tCO2stored/kWh H2) could be indicated. Alternatively, only the share of hydrogen equal to the total amount of net negative emissions can be declared as CNH, whereas the remainder is to be declared carbon-neutral hydrogen. The production costs of one unit of carbon-negative hydrogen produced, including (i) the internalization costs for all process GHG emissions as well as (ii) the revenue for the negative emissions, for example in EUR/kWh H₂, is referred to as levelized cost of carbon-negative hydrogen (LCCNH). It has to be underlined that only in case the biogenic CO_2 is actually used for long-term storage or use and the project generates netnegative emissions (EF < 1), the product is CNH and the KPI can be called LCCNH, as defined in the framework condition FC1 above. This means that the designation as LCCNH already includes the validation of the hydrogen being carbon-negative. In all other cases, the resulting KPI would merely represent and be called the external, GHGrelated LCOE of hydrogen. The target KPI for NET comparison is the levelized cost of negative emissions (LCNE), expressed in EUR/tCO2stored. Also, LCNE can only be used in case the biogenic CO₂ is actually used for longterm storage or use and the project generates net-negative emissions (EF < 1). For a better understanding, the stepwise internalization procedure is exemplarily shown for the HyBECCS system (B3) in the following and expressed as Equation 9.

First, the internal LCOE of produced biohydrogen, in EUR/kWh H₂, is determined from the total production costs. If the project generates net-negative emissions, as per framework condition FC1, all process GHG emissions (E1 and E2b) are monetized via carbon pricing, leading to an increase in the LCOE. Afterward, the revenues from the sales of gross negative emissions (R2) are deducted, leading to a decrease in the LCOE. The sum of the internalization costs is divided by the annual H₂ production amount (in kWh H₂). The result is the LCCNH (in EUR/kWh H₂), which can be easily compared to the LCCNH of

0,25

0,20

0,15

LCOE reduction

from CS profit

other specific HyBECCS projects or the LCOE of any other energy carrier such as fossil fuels.

LCCNH = Internal LCOE

+ E1 internalization costs + E2 binternalization costs - R2 annual H2 production (9)

Figure 6 illustrates the functionality of one of the developed KPI. The LCCNH from an exemplary HyBECCS process is compared to the LCOEs of several other energy carriers. Due to missing data from real HyBECCS plants, the LCCNH is only an assumption for illustration purposes and not based on real values. The values of the alternatives are based on actual data from Howarth and Jacobson, (2021), IPCC and Couper, (2003). The blue bars represent the respective internal LCOE (before GHG internalization). The red bars represent the internalized costs for the specific GHG emissions depending on the assumed carbon price. The GHG internalization costs are shown separately in this example (in red), independently from whether the GHG emissions are partly already internalized through carbon pricing mechanisms for some alternatives and in some countries (World Bank, 2021) and would thus be part of the internal costs. For HyBECCS, the green bars represent the LCOE reduction due to the application of CCS. This means that, in this example, the capture and storage of the biogenic CO₂ generate more income from sales of carbon removal certificates than costs (including GHG internalization costs for CCS process emissions). In this example, the costs can be reduced from a total of 0,20 EURO/kWh to a final LCCNH of 0,13 EURO/kWh (dotted line) at an assumed carbon price of 300 EUR/tCO₂eq. The LCCNH as the final KPI shows the competitiveness of biohydrogen from HyBECCS in comparison to alternatives. Here, the LCCNH of biohydrogen from an exemplary HyBECCS plant is compared to the

LCOE of hydrogen from other sources as well as to the wholesale prices of fossil energy carriers (IRENA, 2019; Matthes et al., 2020, p. 29).

Figure 7 compares the levelized costs of negative emissions (LCNE) from HyBECCS with the LCNE of other negative emission technologies. Due to missing data from real HyBECCS plants, the LCNE for HyBECCS is only an assumption for illustration purposes and not based on real values. The values of the alternatives are based on actual data from Poralla et al. (2021) and Möllersten et al. (2020). For HyBECCS, the purple bar represents the internal LCNE before GHG internalization. The red bars represent the internalization costs for the specific GHG emissions depending on the assumed carbon price. For HyBECCS, the green bar represents the cost reduction due to the generation and sale of hydrogen. This means that, in this example, the total unit costs can be reduced from a total of 360 EUR/tCO2stored to a final LCNE of 110 EUR/tCO2stored (dotted line) at a carbon price of 300 EUR/tCO₂eq. The LCNE as the final KPI shows the competitiveness of negative emissions from HyBECCS in comparison to alternative NETs. Here, the LCNE of negative emissions from an exemplary HyBECCS plant is compared to the costs of BECCS and DACCS. For both alternatives, the striped bars represent cost ranges gathered from studies (Möllersten et al., 2020; Poralla et al., 2021). As long as no GHG internalization for process emissions is undertaken for DACCS and BECCS, those cost ranges represent the internal LCNE.

4.2.2 Exemplary Basic Model for HyBECCS Regulatory Systems in Net-zero or Net-negative **Emission Economies**

As the CS/CU subsystem of HyBECCS, in most cases, does not generate any other marketable product than the

Additional LCOE reduction from CS @ 300 EUR/tCO2eq

LCOE reduction from CS @ 200 EUR/tCO2eq



Levelized Costs of Carbon-

with hydrogen from other sources and wholesale prices of fossil energy carriers. Assumptions: Net calorific value for hydrogen. Specific GHG emission for ammonia corresponds to those of grey hydrogen. Ammonia is calculated as H2 equivalents. (Own elaboration based on Fischedick and Adolf (2017), Hinicio, Matthes et al. (2020))

132

WILEY-GCB-BIOENERGY

negative emissions, but generate costs for the carbon storage, the CS/CU subsystem would not be implemented without external incentives to (over)compensate such CO₂ capture and storage costs. In order to achieve a global economy with first net-zero and then net-negative GHG emissions, according to the emission pathways of the IPCC (Masson-Delmotte, et al., 2021), uniform incentive schemes have to be established. Various approaches have been developed and rules for such certification systems have been set forth (Honegger & Reiner, 2018a; Tanzer & Ramírez, 2019). The variety of incentives schemes can be divided into result-based and non-result-based approaches. In the early implementation phase of HyBECCS, non-result-based incentives schemes are appropriate and urgently needed. Examples are subsidies for research, design, development, and demonstration (RDD&D), tax credits, concessional loans, and grants (Poralla et al., 2021, p. 23). At a later stage with HyBECCS and other NETs being broadly operational, a result-based approach, based on carbon removal certificates (CRCs), can be established to strive for cost efficiency (Rickels et al.). The European Commission aims to develop such a regulatory framework for the certification of carbon removals in 2022 (European Commission). A basic model for such a regulatory, resultbased system building on CRCs from HyBECCS could work as described in the following. It shall give an example of how negative emissions from HyBECCS plants can be economically rewarded in order to convert HyBECCS into a working business case. This basic model respects the general rules elaborated and broadly accepted in scientific

discussion on carbon removal certification (Honegger, 2020; Tanzer & Ramírez, 2019).

In HyBECCS plants, CRCs can be generated when verified negative emissions are produced, for example, through measurement at CS sites that store certified and GHG-neutral (after internalization) biogenic CO_2 . The precondition for the issuance of CRCs is that net-negative emissions are generated as per framework condition FC1 (E3 < 0). CRCs can be issued on a quantity basis per ton of biogenic CO₂ permanently stored. These certificates can be sold on a free market to offset emissions of certain, predefined types. Figure 8 below depicts the mechanism at the example of HyBECCS. As shown in the figure, the process emissions of the HyBECCS process chain (E1 and E2b) could be compensated through CRC purchase and offsetting. For an operator with internal access to negative emissions, e.g. when considering the whole HyBECCS process (B3) or only the CS boundary (B2), the offset could be done with the cancellation of CRCs in the amount of process emissions. At the same time, a revenue stream is generated by the sale of CRCs issued for the negative emissions (E2a) and sold on the CRC market. This would create a self-regulating market mechanism ensuring, in a first step, net-zero emissions, in case all process emissions are obliged to be offset against negative emissions. Risk deductions for uncertainties on the permanence of CO₂ storage would have to be considered within such an offsetting mechanism.

Regulatory intervention is also possible and enables the targeted pursuit of predefined emission paths: By removing



FIGURE 7 Exemplary comparison of the Levelized Costs of Negative Emissions (LCNE) of negative emissions from HyBECCS with other NETs. (Own elaboration based on Möllersten et al. (2020), Poralla et al. (2021))



FIGURE 8 Basic model for regulatory systems based on carbon removal certificates (CRCs) at the example of HyBECCS processes (Abbreviations cf. Figure 3)

certificates from the market, negative emission paths can also be ensured. Potential buyers of CRCs are thus government or private institutions aiming to build up a negative emission budget through the reduction of CRCs. The second group who could be obliged to purchase CRCs are emitters of unavoidable emissions. It is, however, important that the unavoidability of GHG emissions is regularly questioned in the light of technological progress. It must be avoided that the CRC offsetting hinders or decelerates innovation for GHG mitigation. Furthermore, the priority of avoiding GHG emissions should be secured by excluding the offsetting of avoidable (fossil) emissions with CRCs, as it would prolong the lifetime of fossil utilities. Moreover, the offsetting of CRCs with fossil GHG emissions would have the same GHG balance as the CCS of fossil carbon dioxide: Both are, at a maximum, GHG-neutral, but would occupy potentially scarce CO₂ storage capacities (Azapagic et al., 2018). Such GHGneutral or -positive uses of storage capacities would increase the costs of NETs: Storage sites would be scarce, potentially less accessible, and thus more expensive to access when the economically best locations are already used for the storage of avoidable, GHG-neutral, or -positive CO2. This would limit the total potential of HyBECCS and NETs in general. CRC offsetting with avoidable GHG emission thus has to be ruled out, such as CCS of avoidable fossil CO₂, both ideally by the avoidance of such avoidable CO2 emissions in the first place. In the proposed offsetting mechanism for HyBECCS, potentially avoidable GHG emissions (of the HyBECCS process) are offset against CRCs. It should be discussed if such offsetting is allowed, in the interest of the NET implementation, whether the GHG emission internalization should happen via offsetting of process emissions against emission reduction certificates from renewable energy projects, or if a ratio for emission reduction certificates and CRCs should be introduced for NETs (Geden, 2021). The approach presented above gives an example of how an incentive scheme for the generation of negative emissions might look like aiming to

SCB-BIOENERGY

present a basis for discussion of how to solve the externalities of NETs in a positive way.

5 DISCUSSION AND LIMITATIONS

It must be clearly stated that there is no claim to the completeness of the described methods for a final techno-economic and ecological evaluation of HyBECCS projects. However, the evaluation basics and specifics are provided in order to enable a uniform evaluation of these approaches. Using those basics and discussing the elaborated KPIs LCCNH or LCNE, a review of the TEA and EA results should be carried out in order to check the completeness, reliability, and consistency of the model, assumptions, data quality, and outputs. Therefore, an uncertainty and sensitivity analysis should be conducted to identify the most influential input indicators. (Zimmermann et al., 2020). Based on the technical evaluation and the uncertainty and sensitivity analyses, targeted technical optimization can be carried out to improve efficiency in terms of energy consumption and biomass conversion as well as the economics and ecological aspects of the HyBECCS plant (Zimmermann et al., 2020). The LCCNH (and LCNE) can only serve as a benchmark to price scenarios and projections for external costs and benefits of HyBECCS processes into cost calculations as long as they are not internalized and reflected as actual internal costs in the calculations.

Further, for the environmental assessment specifics elaborated, a special focus is on the impact category of global warming potential. This is due to the main objective of the considered HyBECCS technologies for climate protection. However, many important impact categories remain unconsidered. These include, for example, acidification and eutrophication potential or photochemical oxidant formation as well as ozone layer and abiotic depletion according to ISO 14040/44. As HyBECCS represents a BECCS technology, this novel approach is subject to the sustainability criteria of BECCS technologies. Therefore, it should secure food supply, avoid land degradation and land use conflicts, conserve water resources, as well as preserve biodiversity in order to be socially and ethically tenable (Fajardy et al., 2019). The implementation of HyBECCS projects also depends on social acceptance and social compatibility (Gough et al., 2018), since social values are crucial in decision-making (Wainger et al., 2010) and technological development (Buck, 2016). Hence, HyBECCS projects must take place in a socio-ethical context, which is why those impacts need to be assessed in order to determine the full implementation potential of HyBECCS projects (Gough et al., 2018). Therefore, a social life cycle

615

-WILEY

616 WILEY- GCB-BIOENERGY

analysis (S-LCA) should be carried out. Furthermore, the deployment of each HyBECCS plant should be aligned with the Sustainable Development Goals (Fajardy et al., 2019). In addition, any other category that influences a social-ethical friendly environment, such as health and safety conditions, local work and employment conditions, fair competition, respect of indigenous rights, infrastructure development, etc. must be investigated within the S-LCA (Benoît Norris et al., 2013). These aspects were not considered in the present work. Thus, guidelines for an S-LCA for HyBECCS approaches need to be elaborated.

6 | SUMMARY AND OUTLOOK

In this paper, fundamentals for the holistic evaluation of HyBECCS process combinations in terms of technoeconomic and climatic performance are presented for the first time. Since there is no HyBECCS plant so far, this work provides the theoretical basis for comparing future HyBECCS plants. For this purpose, first, a classification and delimitation of the HyBECCS approach are elaborated. Respectively, the concepts CCS, NET, and CDR apply to the HyBECCS approach as it removes biogenic carbon dioxide from the atmosphere. However, further requirements must be met: Only if the GHG footprint of a specific HyBECCS project is negative, it actually generates net negative GHG emissions and thus qualifies as NET and the generated hydrogen is carbon-negative hydrogen (CNH). System boundaries to be applied and emission streams to be considered are described in order to provide a comparable frame of reference. Thereby, the HyBECCS approach is divided into two system elements, each with an associated subsystem boundary: First, the biohydrogen production with CO₂ co-production and capturing (HyBECC) and, second, the storage or long-term use of biogenic CO₂ (CS/CU). In addition to the emission flows, monetary flows of the system boundaries are defined as a basis for a better understanding of the economic relationships. A special focus was put on the description of interrelationships between climatic and economic aspects by elaborating the double climate protection effect through GHG emission reduction and removal and their resulting possibilities to receive incentives.

In the main part, this basis was used to explain how the economic viability as well as ecological effects, specifically on the climate, can be determined. On the ecological side, firstly, the conditions for positive climate impacts of the HyBECCS technology are discussed. This is done within an eligibility check related to the selection of the basic technology as well as project-specific life cycle impacts. The eligibility check on technology level ensures that the approach considered is suitable for the production of CNH. The theoretical ability for

CO₂ removal and the theoretical permanence of the CO₂ storage is evaluated by matching the substrate biomass and the selected CO₂ storage technology via positive lists. These lists should be drawn up and continuously maintained by authorities. Second, if this step is successful, a life cycle assessment is carried out for the specific project proposal and the suitability for generating net negative emissions is verified via the greenhouse gas balance. Special features for the greenhouse gas balancing that apply to the HyBECCS subsystems are elaborated and described. As a key figure for the climatic impact, the emission factor EF was established and defined as the most important indicator for the classification of the HyBECCS approach as NET and the designation of the produced hydrogen as CNH. Eligibility on project level to qualify as NET is proven if the calculated emission factor EF is less than 1. The EF can be estimated ex-ante based on the project planning, but has to be verified expost through on-site measurements during the operation of the HyBECCS plant(s).

On the economic side, particularities for the determination of profitability were explained. Special focus was placed on determining internal capital and operating costs, initially without taking into account external effects and incentive systems. The selection of appropriate methods for estimating capital cost with respect to the current degree of the project definition and data availability is elaborated within this part. Afterward, peculiarities concerning the HyBECCS approach with respect to political regulatory measures and interrelationships between economics and ecology are outlined. Based on this, two key performance indicators (KPIs) are determined: The levelized costs of carbon-negative hydrogen (LCCNH) and negative emission (LCNE). Both KPIs allow deciding whether a specific HyBECCS project is economically viable when compared to the LCOE of competing technologies for the provision of negative emissions, hydrogen, or energy, in general. It furthermore allows the comparison of HyBECCS projects with different hydrogen or energy provision technologies and NETs. This is especially important to ensure that the biological transformation process of modern economies is efficient, minimizing the depletion of scarce resources such as energy, non-waste biomass, land, potentially CO2 storage sites, and, ultimately, money. The guidelines for the KPI determination furthermore give first indications of how to optimize the GHG balance of a HyBECCS project. For instance, the design of the HyBECCS project as a "waste to hydrogen" plant, using waste or residuals as biomass feedstock, may significantly reduce the process emissions, allowing for a higher output of net negative emissions footprint (DIN e.V. DIN EN ISO 14044: Umweltmanagement - Ökobilanz - Anforderungen und

Anleitungen (ISO 14044:2006 + Amd 1:2017 + Amd. 2:2020); IE; Thrän & Pfeiffer, 2013). These holistic target KPIs can be calculated based on the total production costs and GHG balance. Both the negative emissions and the co-generated biohydrogen contribute to limiting global warming. However, only the internalization of these external benefits and the external costs of GHG emissions can turn innovative technologies such as HyBECCS into viable business cases. This particularity is reflected by the LCCNH. The LCCNH should represent the value to be optimized in order to achieve higher sustainability. However, this KPI depends strongly on political framework conditions, which must be taken into account.

In order to reach the GHG reduction targets set by governments worldwide, existing carbon pricing schemes have to be expanded and sharpened, as well as incentive schemes for negative emissions need to be disseminated. As shown in this paper, both of them affect the economic viability of HyBECCS. It is important to underline that the complete internalization of all GHG emissions and negative emissions - as proposed here is important to give a realistic picture of the real costs of energy provision and climate protection. The still missing or incomplete internalization of external costs of conventional technologies hinders the implementation of novel, climate-friendly approaches such as HyBECCS, giving the presented assessment approaches a speculative component. Only realistic carbon price levels will have the needed steering effect on the individual decision process of operators and investors to direct investment into eco-friendly and efficient options like HyBECCS (OECD, 2021). The implementation of regulations to account for all GHG emissions of all processes should thus be the aim on a global level and the basis for future negative emission incentive schemes. This paper increased the understanding of these regulation's impact on the economics of HyBECCS systems, and evaluability, as well as comparability, have been initiated via suggestions for basic frameworks and metrics. The work can be understood as an initial standard that enables meaningful comparison of HyBECCS systems. However, it needs to be confirmed by implementing the assessment approaches on real plants.

Furthermore, the innovative concept of HyBECCS with its novel product carbon-negative hydrogen calls for a more detailed categorization of hydrogen. On-going initiatives work on the establishment of a uniform threshold of GHG intensity for hydrogen to qualify as "green" or "low-carbon" (certifhy; Hinicio; Newborough & Cooley, 2020). However, as the GHG footprint of "green" hydrogen from different sources and projects varies significantly,

GCB-BIOENERGY

the "theory of colors" always falls short of introducing real transparency. Imposing a product declaration stating the specific GHG intensity of hydrogen (e.g., in kg-CO₂eq/kgH₂) would allow such transparency. Besides the economic steering effect of the monetary internalization schemes introduced in this paper, such transparency could give an important impulse for the dissemination of CNH. Moreover, in order to safeguard the economic integrity of HyBECCS, efforts need to be directed toward the integration of scientific evidence on the risks of specific CO₂ storage technologies into a regulatory framework. The same applies to the definition of a minimum time horizon of storage for carbon utilization applications. Further needs for regulations will appear during the future development of the HyBECCS concept. Thorough monitoring, reporting, and verification (MRV) standards need to be elaborated and generally adopted so that market mechanisms, such as a CRC market exemplary described above, can lead the world on emission pathways toward the 1.5°C target.

ACKNOWLEDGMENTS

The research presented in this paper was conducted within the projects "RhoTech" and "HyBECCS-BW". The authors gratefully acknowledge the financial support of the German Federal Ministry for Economic Affairs and Energy and the Ministry of Rural Affairs and Consumer Protection of Baden-Wuerttemberg, Germany. Open Access funding enabled and organized by Projekt DEAL.

CONFLICTS OF INTEREST

The authors declare no conflict of interest.

AUTHOR CONTRIBUTIONS

Writing, J.F., S.Z., and M.G.; supervision, R.M. and A.S. All authors have read and agreed to the published version of the manuscript.

ORCID

Johannes Full D https://orcid.org/0000-0002-0748-7689

REFERENCES

- Masson-Delmotte, V., Zhai, P., Pirani, A., Connors, S. L., Péan, C., Berger, S., Caud, N., Chen, Y., Goldfarb, L., Gomis, M. I., Huang, M., Leitzell, K., Lonnoy, E., Matthews, J. B. R., Maycock, T. K., Waterfield, T., Yelekçi, O., Yu, R., & Zhou, B. (Eds.) (2021). *IPCC. Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change.* Cambridge University Press.
- AACE International. (2020). Cost Estimate Classification system as applied in engineering, procurement, and construction for the

WILEY 617

WILEY-GCB-BIOENER

process industries'. Association for the Advancement of Cost Engineering International Recommended Practice No. 18R-97.

- Agora Energiewende und Wuppertal Institut. (2019). Klimaneutrale Industrie: Schlüsseltechnologien und Politikoptionen für Stahl.
- Antonini, C., Treyer, K., Streb, A., van der Spek, M., Bauer, C., & Marco Mazzotti, M. (2020). Hydrogen production from natural gas and biomethane with carbon capture and storage—A techno-environmental analysis. *Sustainable Energy & Fuels*, 4(6), 2967–2986. https://doi.org/10.1039/D0SE00222D
- Azapagic, A., Beerling, D., Cheeseman, C., Henderson, G., Hepburn, C., House, J., Le Quéré, C., Markusson, N., Shah, N., Shepherd, J., & Smith, P. (2018). *Greenhouse gas removal*. The Royal Society & Royal Academy of Engineering. ISBN 978-1-78252-349-9.
- Benoît Norris, C. et al (2013). The methodological sheets for subcategories in social life cycle assessment (S-LCA).
- Boysen, L. R. Potentials, consequences and trade-offs of terrestrial carbon dioxide removal: Strategies for climate engineering and their limitations 2017.
- Buck, H. J. (2016). Rapid scale-up of negative emissions technologies: Social barriers and social implications. *Climate Change*, 139(2), 155–167. https://doi.org/10.1007/s10584-016-1770-6
- BMBF. Eine kleine Wasserstoff-Farbenlehre. Retrieved August 23, 2021, from: https://www.bmbf.de/bmbf/shareddocs/kurzm eldungen/de/eine-kleine-wasserstoff-farbenlehre.html
- certifhy CertifHy enters into phase 3 to build a H2 GO market as well as a H2 certification scheme for REDII.
- Coalition for Negative Emissions. The case for Negative Emissions 2021. https://doi.org/10.1080/15374417609532700
- Couper, J. R. (2003). Process Engineering Economics.
- DIN e.V. DIN EN ISO 14044: Umweltmanagement Ökobilanz - Anforderungen und Anleitungen (ISO 14044:2006 + Amd 1:2017 + Amd. 2:2020) Deutsche Fassung EN ISO 14044:2006 + A1:2018 + A2:2020 2021.
- DIN e.V. DIN EN ISO 14040: Umweltmanagement Ökobilanz -Grundsätze und Rahmenbedingungen (ISO 14040:2006 + Amd. 1:2020). (2021). Deutsche Fassung EN ISO 14040:2006 + A1:2020.
- European Commission. Sustainable Carbon Cycles: Communication from the Commission to the European Parliament and the Council.
- European Commission. EU Emissions Trading System (EU ETS) -Climate Action - European Commission. Available online: https://ec.europa.eu/clima/policies/ets_en (accessed on 22 August 2021)
- Fajardy, M., Koeberle, A., MacDowell, N. & Fantuzzi, A. BECCS deployment: a reality check 2019.
- Fischedick, M., & Adolf, J. (2017). Energie der Zukunft?: Nachhaltige Mobilität durch Brennstoffzelle und H2.
- Frischknecht, R. Umweltverträgliche Technologien: Analyse und Beurteilung: Teil 2: Ökobilanzen (Life cycle assessment, LCA).
- Full, J., Baumgarten, Y., Delbrück, L., Sauer, A., & Miehe, R. (2021). Market perspectives and future fields of application of odor detection biosensors within the biological transformation—A systematic analysis. *Biosensors*, 11(3), 93. https://doi.org/10.3390/ bios11030093
- Full, J., Merseburg, S., Miehe, R., & Sauer, A. (2021). A new perspective for climate change mitigation—Introducing carbonnegative hydrogen production from biomass with carbon capture and storage (HyBECCS). *Sustainability*, 13, 4026. https:// doi.org/10.3390/su13074026

- Full, J., Miehe, R., Kiemel, S., Bauernhansl, T., & Sauer, A. (2019). The biological transformation of energy supply and storage— Technologies and scenarios for biointelligent value creation. *Procedia Manufacturing*, 1204–1214. https://doi.org/10.1016/j. promfg.2020.01.349
- Full, J., Trauner, M., Miehe, R., & Sauer, A. (2021). Carbon-negative hydrogen production (HyBECCS) from organic waste materials in Germany: How to estimate bioenergy and greenhouse gas mitigation potential. *Energies*, 14(22), 7741. https://doi. org/10.3390/en14227741
- Geden, O. (2021). Mit CO2-Rückholung gewinnt man keine Wahlen. Neue Energie, 26–30.
- Gough, C., L. Mabon, & S. Mander (Eds.) (2018). Social and ethical dimension of BECCS: Bioenergy carbon capture and storage: Unlocking negative emissions.
- Hauschild, M. Z., Rosenbaum, R. K., & Olsen, S. I. (2018). *Life Cycle Assessment*. Springer International Publishing.
- Hinicio, F. B. CertifHy Developing a European guarantee of origin scheme for green hydrogen.
- Honegger, M. (2020). Carbon dioxide removal in climate change mitigation policy planning.
- Honegger, M., & Reiner, D. (2018). The political economy of negative emissions technologies: Consequences for international policy design. *Climate Policy*, 18, 306–321. https://doi. org/10.1080/14693062.2017.1413322
- Honegger, M., & Reiner, D. Global Policy Instruments to Mobilize Carbon Dioxide Removal 2018.
- Howarth, R. W., & Jacobson, M. Z. (2021). How green is blue hydrogen? Energy Science & Engineering, 9(10), 1676–1687. https:// doi.org/10.1002/ese3.956
- Hydrogen Council. Hydrogen scaling up: A sustainable pathway for the global energy transition 2017.
- IE. (Institutfür Energetik und Umwelt) Energie- und Klimaeffizienzausgewählter Biomassekonversionspfade zur Kraftstoffproduktion. (2008).
- IEA. The Future of Hydrogen.
- IPCC. Carbon Dioxide Transport, Injection and Geological Storage. Guidelines for National Greenhouse Gas Inventories 2006.
- IPCC. Carbon Dioxide Capture and Storage 2005.
- IPCC. (2005). Carbon Capture and Storage.
- IPCC. (2015). *Climate change 2014: Synthesis report*. Intergovernmental Panel on Climate Change.
- IPCC. Global Warming of 1.5°C.: An IPCC Special Report on the impacts of global warming of 1.5°C above pre-industrial levels and related global greenhouse gas emission pathways, in the context of strengthening the global response to the threat of climate change, sustainable development, and efforts to eradicate poverty. Retrieved June 17, 2021, from: https://www.IPCC.ch/sr15/
- IRENA. (2019). Hydrogen: A renewable energy perspective.
- Jeffery, L., Hoehne, N., Moisio, M., Day, T., & Lawless, B. (2020). Options for supporting carbon dioxide removal. Discussion Paper.
- FCH JU. (2019). Hydrogen Roadmap Europe: A sustainable pathway for the European Energy Transition.
- Kaltschmitt, M., & Schebek, L. (2015). Umweltbewertung für Ingenieure: Methoden und Verfahren. Springer Vieweg. ISBN 978-3-642-36989-6.
- Kemper, J. (2015). Biomass and carbon dioxide capture and storage: A review. International Journal of Greenhouse Gas Control, 40, 401–430. https://doi.org/10.1016/j.ijggc.2015.06.012

- Klöpffer, W., & Grahl, B. (2009). Ökobilanz (LCA): ein Leitfaden für Ausbildung und Beruf. Wiley-VCH. ISBN 978-3-527-32043-1.
- Kunysz, D. O. (2020). Kostenschätzung im chemischen Anlagenbau.
- Lozanovski, A. (2013). *Guidance document for performing LCA on hydrogen production systems*. Fraunhofer Verlag.

Mankiw, G. (2001). Grundzüge der Volkswirtschaftslehre.

- Matthes, F., Heinemann, C., Hesse, T., Kasten, P., Mendelevitch, R., Seebach, D., Timpe, C., & Cook, V. (2020). Wasserstoff sowie wasserstoffbasierte Energieträger und Rohstoffe - Eine Überblicksuntersuchung (German language). Öko-Institut e.V.
- Miehe, R., Bauernhansl, T., Beckett, M., Brecher, C., Demmer, A., Drossel, W.-G., Elfert, P., Full, J., Hellmich, A., Hinxlage, J., Horbelt, J., Jutz, G., Krieg, S., Maufroy, C., Noack, M., Sauer, A., Schließmann, U., Scholz, P., Schwarz, O., ... Wolperdinger, M. (2020). The biological transformation of industrial manufacturing—Technologies, status and scenarios for a sustainable future of the German manufacturing industry. *Journal* of Manufacturing Systems, 54, 50–61. https://doi.org/10.1016/j. jmsy.2019.11.006
- Miehe, R., Bauernhansl, T., Schwarz, O., Traube, A., Lorenzoni, A., Waltersmann, L., Full, J., Horbelt, J., & Sauer, A. (2018). The biological transformation of the manufacturing industry— Envisioning biointelligent value adding. *Procedia CIRP*, 72, 739–743. https://doi.org/10.1016/j.procir.2018.04.085
- Miehe, R., Full, J., Scholz, P., & Demmer, A. (2019). The biological transformation of industrial manufacturing—Future fields of action in bioinspired and bio-based production technologies and organization. *Procedia Manufacturing*, 39, 737–744. https://doi.org/10.1016/j.promfg.2020.01.437
- Minli, Y., Wang, K., & Vredenburg, H. (2021). Insights into lowcarbon hydrogen production methods: Green, blue and aqua hydrogen. *International Journal of Hydrogen Energy*, 46(41), 21261–21273. https://doi.org/10.1016/j.ijhydene.2021.04.016
- Möllersten, K., Yan, J., & Naqvi, R. (2020). Qualitative Assessment of Classes of. Negative Emission Technologies.
- Newborough, M., & Cooley, G. (2020). Developments in the global hydrogen market: The spectrum of hydrogen colours. *Fuel Cells Bulletin*, 2020, 16–22. https://doi.org/10.1016/S1464-2859(20)30546-0
- Noussan, M., Raimondi, P. P., Scita, R., & Hafner, M. (2021). The role of green and blue hydrogen in the energy transition—A technological and geopolitical perspective. *Sustainability*, 2021(13), 298. https://doi.org/10.3390/su13010298
- OECD. (2021). Effective Carbon Rates 2021. OECD.
- Otto, A., Markewitz, P., & Robinius, M. (2017). Technologiebericht 2.4 CO2-Nutzung: Technologien für die Energiewende. Teilbericht 2 an das Bundesministerium für Wirtschaft und Energie (BMWi). Wuppertal Institut, ISI, IZES (Hrsg.).
- Peters, M. S., Timmerhaus, K. D., & West, R. E. (2003). Plant Design and Economics for Chemical Engineers.
- Poralla, M., Honegger, M., Ahonen, H.-M., Michaelowa, A., & Weber, A.-K. Sewage Treatment for the Skies: Mobilising carbon dioxide removal through public policies and private financing 2021.

<u>GCB-BIOENERGY</u>

Rickels, W., Proelß, A., Geden, O., Burhenne, J., & Fridahl, M. The Future of (Negative) Emissions Trading in the European Union.

-WILEY

- Rogelj, J., Luderer, G., Pietzcker, R. C., Kriegler, E., Schaeffer, M., Krey, V., & Riahi, K. (2015). Energy system transformations for limiting end-of-century warming to below 1.5 °C. *Nature. Climatic Change*, 5, 519–527. https://doi.org/10.1038/nclimate2572
- Singh, A., & Rathore, D. (2017). Biohydrogen production: Sustainability of current technology and future perspective. Springer India.
- Tanzer, S. E., & Ramírez, A. (2019). When are negative emissions negative emissions? *Energy & Environmental Science*, 12, 1210– 1218. https://doi.org/10.1039/C8EE03338B
- Thrän, D., & Pfeiffer, D. (Eds.). (2013). Methodenhandbuch -Stoffstromorientierte Bilanzierung der Klimagaseffekte. Energetische Biomassenutzung (German language). Volume 4. ISBN 2192-1806.
- Towler, G. & Sinnott, R. (2013). Chemical engineering design: Principles, practice and economics of plant and process design.
- Trippe, F. (2013). Techno-ökonomische Bewertung alternativer Verfahrenskonfigurationen zur Herstellung von Biomass-to-Liquid (BtL) Kraftstoffen und Chemikalien. KIT Scientific Publishing.
- UNEP (2017). The emissions gap report 2017: A UN Environment synthesis report; United Nations Environment Programme (UNEP): Nairobi, Kenya.
- UNFCCC. Paris Agreement 2015.
- Wainger, L. A., King, D. M., Mack, R. N., Price, E. W., & Maslin, T. (2010). Can the concept of ecosystem services be practically applied to improve natural resource management decisions? *Ecological Economics*, 69(5), 978–987. https://doi.org/10.1016/j. ecolecon.2009.12.011
- Weber, K. H. (2016). Engineering verfahrenstechnischer Anlagen. Praxishandbuch Mit Checklisten Und Beispielen.
- Wietschel, M., Zheng, L., Arens, M., Hebling, C., Ranzmeyer, O., Schaadt, A., Hank, C., Sternberg, A., Herkel, S., Kost, C. et al (2021). *Metastudie Wasserstoff – Auswertung von Energiesystemstudien. Studie im Auftrag des Nationalen Wasserstoffrats*. Fraunhofer ISI, Fraunhofer ISE, Fraunhofer IEG (Hrsg.).

World Bank. (2021). State and trends of carbon pricing 2021. World Bank.

- Yan, Y., Manovic, V., Anthony, E. J., & Clough, P. T. (2020). Technoeconomic analysis of low-carbon hydrogen production by sorption enhanced steam methane reforming (SE-SMR) processes. *Energy Conversion and Management*, 226, 113530.
- Zimmermann, A., Wang, Y., & Wunderlich, J. (2020). Technoeconomic assessment & life cycle assessment guidelines for CO₂ utilization (version1.1).

How to cite this article: Full, J., Ziehn, S., Geller, M., Miehe, R., & Sauer, A. (2022). Carbon-negative hydrogen production: Fundamentals for a technoeconomic and environmental assessment of HyBECCS approaches. *GCB Bioenergy*, *14*, 597–619. https://doi.org/10.1111/gcbb.12932 619

Appendix D: Paper D

Contributions to the work

Title: Carbon-Negative Hydrogen Production (HyBECCS) from Organic Waste Materials in Germany: How to Estimate Bioenergy and Greenhouse Gas Mitigation Potential Published in: MDPI Energies 14.22 (2021): 7741 DOI: https://doi.org/10.3390/en14227741

CRediT (Contributor Role Taxonomy) author statement:

Authors	Johannes Full	Mathias Trauner	Robert Miehe	Alexander Sauer
Conceptualization	\checkmark			
Methodology	\checkmark	\checkmark		
Formal analysis	\checkmark	\checkmark		
Investigation	\checkmark	\checkmark		
Data Curation	\checkmark	\checkmark		
Writing (Original Draft)	\checkmark	\checkmark		
Writing (Review and Editing)			\checkmark	\checkmark
Visualization	\checkmark	\checkmark		
Project administration	\checkmark			
Funding acquisition	\checkmark		\checkmark	\checkmark

The co-authors hereby agree to the contributions summarized in the table avove.

<u>Johannes Full</u> Name <u>Mathias Trauner</u> Name <u>Robert Miehe</u> Name

<u>Alexander Sauer</u> Name

Signature muer Signature Signature

Alexandy Laur

Signature

 16.11.2022

 Date

 17.11.2022

 Date

 3.5.2023

 Date

 03.05.2023

 Date




142

Article Carbon-Negative Hydrogen Production (HyBECCS) from Organic Waste Materials in Germany: How to Estimate Bioenergy and Greenhouse Gas Mitigation Potential

Johannes Full ^{1,2,*}, Mathias Trauner ¹, Robert Miehe ¹, and Alexander Sauer ^{1,2}

- ¹ Fraunhofer Institute for Manufacturing Engineering and Automation IPA, 70569 Stuttgart, Germany; mathias.trauner@ipa.fraunhofer.de (M.T.); robert.miehe@ipa.fraunhofer.de (R.M.); alexander.sauer@ipa.fraunhofer.de (A.S.)
- ² Institute for Energy Efficiency in Production (EEP), University of Stuttgart, 70569 Stuttgart, Germany
- * Correspondence: johannes.full@ipa.fraunhofer.de; Tel.: +49-711-9701434

Abstract: Hydrogen derived from biomass feedstock (biohydrogen) can play a significant role in Germany's hydrogen economy. However, the bioenergy potential and environmental benefits of biohydrogen production are still largely unknown. Additionally, there are no uniform evaluation methods present for these emerging technologies. Therefore, this paper presents a methodological approach for the evaluation of bioenergy potentials and the attainable environmental impacts of these processes in terms of their carbon footprints. A procedure for determining bioenergy potentials is presented, which provides information on the amount of usable energy after conversion when applied. Therefore, it elaborates a four-step methodical conduct, dealing with available waste materials, uncertainties of early-stage processes, and calculation aspects. The bioenergy to be generated can result in carbon emission savings by substituting fossil energy carriers as well as in negative emissions by applying biohydrogen production with carbon capture and storage (HyBECCS). Hence, a procedure for determining the negative emissions potential is also presented. Moreover, the developed approach can also serve as a guideline for decision makers in research, industry, and politics and might also serve as a basis for further investigations such as implementation strategies or quantification of the benefits of biohydrogen production from organic waste material in Germany.

Keywords: HyBECCS; biohydrogen; bioenergy potential; hydrogen; waste-to-hydrogen

1. Introduction

The findings in the Sixth Assessment Report of the Intergovernmental Panel on Climate Change (IPCC) show that the 1.5 °C target is likely already being reached in the timeframe from 2021–2040, with each 1000 GtCO₂ estimated to increase the global surface temperature from 0.27 °C to 0.63 °C and approximately 900 to 2300 GtCO₂ remaining of the carbon budget to limit warming by 2 °C. A rapid and profound reduction in anthropogenic greenhouse gas emissions from fossil fuel combustion and, in addition, active removal of carbon dioxide from the atmosphere through negative emission technologies (NETs) are inevitably required in the immediate future [1].

In this regard, hydrogen is globally considered to be a key energy carrier of the future to be used as a substitute for fossil energy sources. Germany and many other countries have adopted a hydrogen strategy to bring it into widespread application. Hydrogen can not only be used for fuel or energy storage but is also a key substance of ammonia production or for refineries. In the future, it will be capable of decarbonizing decisive industries, such as steel or cement [2] (p. 2). What distinguishes hydrogen from conventional fuels is that the only combustion product—besides energy—is water. Further, hydrogen gas releases an astonishing amount of 142.35 kJ per g when reacting with oxygen. Natural gas, in comparison, releases energy of between 35 and 45 kJ per g during combustion [3]



Citation: Full, J.; Trauner, M.; Miehe, R.; Sauer, A. Carbon-Negative Hydrogen Production (HyBECCS) from Organic Waste Materials in Germany: How to Estimate Bioenergy and Greenhouse Gas Mitigation Potential. *Energies* 2021, 14, 7741. https://doi.org/10.3390/en14227741

Academic Editor: Viviana Cigolott

Received: 7 October 2021 Accepted: 10 November 2021 Published: 18 November 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). (p. 42). Hydrogen can be produced sustainably using renewable electrical energy for water electrolysis or from biomass through several processes, with the product being called biohydrogen. Utilizing waste materials for biohydrogen production offers several benefits, as it increases the economic and ecological potential, avoids competitive land use, and represents another step towards a circular economy. According to Full et al. the ecological benefits for climate change mitigation can be further increased by capturing the produced biogenic CO_2 during the process and thereby achieving biohydrogen production as a carbon-negative process [4]. This means a net effect of CO_2 being removed from—rather than emitted to—the atmosphere during energy generation. This NET approach of carbon-negative hydrogen production by capturing biogenic CO_2 during biohydrogen processes is referred to as hydrogen bioenergy with carbon capture and storage (HyBECCS) [4].

There are many different ways to produce renewable energy. Bioenergy is one option that is of great importance, as biomass accounted for about 54% of the total share of renewable energy in Germany's primary energy consumption in 2017 [5]. A common way of producing bioenergy is the production of biomethane [6]. Biohydrogen production with additional carbon capture and storage or long-term use (HyBECCS) presents an alternative to biogas and -methane production and allows the generation of a carbonnegative product with growing demand [4,7]. The development of HyBECCS processes can additionally contribute to the global climate goals by creating negative emissions. Such methods of carbon dioxide removal (CDR) are key measures in the pathways to achieve climate targets [1]. Moreover, in order to decarbonize the energy sector more quickly and sustainably, non-utilized resources such as residual and waste materials should also be harnessed. These resources should be used primarily for NETs, such as the HyBECCS technology, as they are scarce. However, in order to accelerate the development and rollout of such promising technologies, it must be made clear to potential investors that they are economically viable [8]. An analysis of how much bioenergy from waste materials is available and suitable for biohydrogen production provides a basis for making such a statement. Research dealing with the estimation of bioenergy potential already exists [9–16]. However, none of the work deals with bioenergy potentials for biohydrogen production. Hence, there are no uniform evaluation methods for biohydrogen production. Therefore, the overall objective of this paper is summarized by the central question:

'How can the feasible bioenergy potential of several biohydrogen production processes from organic residuals and waste materials in Germany be evaluated and how much reduction in greenhouse gas emissions can be attained as a result?'

Initially, the classification and legal framework of biomass and organic waste materials are analyzed in general. The suggested methodical conduct is presented by the means of an exemplary biotechnological biohydrogen process. These specifications lay the foundation for identifying available material flows suitable to be converted into biohydrogen. After ascertaining and quantifying the amount of utilizable substances, the bioenergy potential can be estimated. Subsequently, this potential can be adopted into a possible amount of non-emitted and captured GHGs using HyBECCS approaches.

In this paper, for the first time, a recommendation for estimating bioenergy potentials specifically tailored for biohydrogen production in the German energy and waste system is developed. Furthermore, it should help to establish a generally accepted method to determine bioenergy potentials in order to make their quantification more accessible and uniform, which will additionally contribute to the development of new emerging technologies and their market rollout.

2. Basics

This section lays the foundation for this work by providing essential information regarding the different topics touched upon in this work.

2.1. Organic Waste Classification in Germany

This section outlines the legal framework for renewable energy as well as biomass utilization and disposal in Germany. Moreover, common treatment processes for biomass are presented, briefly touching upon their (dis)advantages and products. Lastly, two significant challenges concerning biomass are outlined, which are also highly relevant to this work.

The general regulation concerning renewable energies is the Renewable Energy Act (orig. Erneuerbare-Energien-Gesetz, EEG), which serves the purpose of enabling sustainable development of energy supply, reducing the economic costs by including long-term external effects, saving fossil resources, and supporting the development of technologies which generate electricity from renewable sources [§1, EEG]. A key development in the 2021 amendment of the law is that carbon neutrality of German electricity by 2050 and a 65% share of renewable energy by 2030 are made statutory [17]. The application of the EEG regarding the biomass definition, utilization technologies, and environmental requirements is established in the Biomass Ordinance (orig. Biomasseverordnung, BiomasseV). At the same time, the Circular Economy Act (orig. Kreislaufwirtschaftsgesetz, KrWG) aims to support a circular economy in order to save natural resources and to protect people and the environment during waste generation and management [§1, KrWG]. More specifically, the treatment, utilization, and disposal of organic waste material are widely regulated in the Organic Waste Ordinance (orig. Bioabfallverordnung, BioAbfV). Material of animal origin falls within the scope of the Animal By-Products Disposal Ordinance (orig. Tierische Nebenprodukte-Beseitigungsverordnung, TierNebV), which sets additional requirements on how such material is handled and disposed of, due to its potential hazard for the health of humans and animals. Data collection, processing, and publication are regulated by the Environmental Statistics Act (orig. Umweltstatistikgesetz, UStatG), which delivers a foundation for (environmental) policy decision making and fulfills the reporting obligation to European and other institutions [18]. German waste statistics are published annually by the German Federal Statistical Office (orig. Statistisches Bundesamt) on its GENESIS-online database, which is open access and provides a list of all waste types and their respective annual amounts. The different types of waste and their respective hazardousness are defined in the German Waste Classification Ordinance (orig. Abfallverzeichnisverordnung, AVV), which transposed the European Waste List into German national law. This work draws upon the definition of biomass from the BiomasseV. It defines biomass as energy sources consisting of phytomass and zoomass, including secondary products, by-products, residues, and wastes of which the energy content originates from phytomass and zoomass [(1),§2, BiomasseV]. In accordance with the KrWG, waste means any substance or object which the holder discards or intends or is required to discard [(1),§3, KrWG]. By-products are a substance or object resulting from a production process, the primary aim of which is not the production of that substance or object. They are different from waste if they meet the following criteria:

- 1. Further use of the substance or object is certain;
- 2. The substance or object can be used directly without any further processing other than normal industrial practice;
- 3. The substance or object is produced as an integral part of a production process;
- 4. Further use is lawful, i.e., the substance or object fulfills all relevant product, environmental, and health protection requirements for the specific use and will not lead to overall adverse environmental or human health impacts [(1),§4, KrWG].

The term residue is not defined specifically but used in various works and legal documents. As a consequence, the term cannot be clearly distinguished from waste or by-products. In this work, all biomass material, which is not produced as a primary product, falls within the definition of residues. That includes by-products and waste [19] (p. 21). In general, the sectors of agriculture, forestry, fisheries, food production, animal feed production, and municipalities are mainly accountable for organic residues and wastes in Germany [20] (p. 39). All varieties of material such as wastes from livestock farming,

fish and animal processing, crop harvesting, fruit and vegetable agriculture and processing to organic household waste, green waste, sewage sludge, and secondary wastes need to be evaluated regarding their potential usability for biohydrogen production. Therefore, this work classifies the present biomass flows as listed in Annex 6 of Brosowski et al. [21], as their work provides estimations of potential material flows from the German Biomass Research Centre, DBFZ.

Organic wastes can be used via several pathways, depending on the origin and composition of the material. First, organic material can be used directly, such as animal excrements as fertilizer, bark as mulch, or food industry residues as animal feed. Harvest residues are often left on the field as well [22] (p. 81). Second, various composting processes are established, which treat the material in order to be consecutively applied to (agricultural) land. Advantages are that the stabilized organic substances in compost contribute to the formation of humus, which sustains the productivity of land and increases the water storing capacity of the surface soil [23] (pp. 11, 12). Third, energetic use can be realized by fermenting organic material under anaerobic conditions, incinerating material of suitable composition, or using gasification as a means of treatment. Anaerobic fermentation is mainly used for animal excrements, but also for household biowaste or industry residues to a certain extent [24] (p. 38). Sewage sludge is also treated anaerobically on the site of large water treatment plants, whereas the majority of smaller ones have no such treatment option [25] (p. 18). The desired product of anaerobic digestion is biogas, which can be used directly in a combined heat and power plant or treated further to obtain biomethane, which can be fed into the gas grid, for instance. Biohydrogen production from biomass is also a promising utilization pathway in the future. Incineration or gasification is possible for material with low water content that is not usable for fermentation. As a consequence, mainly wood, wood-like residues, or straw are used as input material for those processes [25] (pp. 10-13). Lastly, biodiesel production is a widely applied technology to utilize oils and fats via transesterification [25] (p. 17).

At present, a lot of organic waste material is not used efficiently. It should be noted that material use of biomass, which also includes use as animal feed, is generally seen as higher value than energetic use [22] (p. 20), cf. Figure 1. What is considered inefficient is material use of substrates, which could be used energetically prior to material use. Examples are biowaste or animal excrements, which could be utilized energetically first and subsequently be composted and/or applied to land. This multiple-use principle is suggested to be a more efficient way to utilize organic wastes [22,26] and is also part of the Circular Economy Act [§8, KrWG]. Currently, just 40% of organic household waste, 12% of manure, and 2% of dung are utilized energetically [24,27], even though it would not impede the subsequent material use. In the following, only material flows which do not compete with established, high-value utilization routes such as animal feed, for instance, will be considered. In addition, the material is only analyzed for biohydrogen production, if it is not directly competing with food production or other human use. Overall, this work is based on the principles that food production is prioritized over energy crop cultivation in all cases and focuses on the utilization of true waste products, avoiding material which can be treated through a utilization path of higher value, cf. Figure 1. In consequence, certain biomass flows, theoretically utilizable for biohydrogen production, will be discarded, as higher-value utilization is assumed. However, biomass flows with an established material utilization pathway in which their potential is not harnessed will also be considered available for this process. Animal excrements are an example of that. Further, usable municipal wastes and certain industrial residues can be considered available for biohydrogen production.



Figure 1. Hierarchy of use for biomass [22].

Another challenge is that, as of 2020, 39% of mixed municipal solid waste (MMSW) is still of native organic origin [28]. This fact is problematic for several reasons. First, the EU set the goal to recycle 60% of municipal wastes by 2030, which urges Germany to act [28] (p. 26). Redirecting organic material from mixed fractions to biowaste could be a crucial opportunity to do so, as 97% of separately collected material from the organic waste bin is already being recycled [29] (p. 30). Secondly, organic waste, with its relatively high water content and hence lower caloric value, is not a desired input material for thermal treatment, which is the major route of disposal for mixed municipal waste [29] (p. 30). Even though organic waste is not causing problems in incineration, the desired high-quality utilization of biomass for biogas or, in this case, biohydrogen with multiple uses, is again preferred and demanded by the Circular Economy Act [24] (p. 92). The separate collection of recyclables and organic waste is already mandatory, but the necessary collection systems are not yet implemented consistently [28] (p. 27).

2.2. Biohydrogen Production and the HyBECCS Approach

Hydrogen produced from renewable energy or biomass is called green hydrogen as it is CO₂ neutral [30]. Furthermore, hydrogen derived from biomass feedstock is called biohydrogen. Processes for producing such green biohydrogen comprise electrochemical, thermochemical, and biotechnological production routes. Microbial electrolysis represents the electrochemical pathway. Thermochemical biohydrogen production processes comprise pyrolysis and gasification, whereas biotechnological process options are biophotolysis, photo-fermentation, dark fermentation, and dark photosynthesis. The following summary is limited to a number of relevant technologies according to the material source. Each technology is briefly described in 2.2.1. Further, in 2.2.2., the HyBECCS approach is presented, which is, at present, the only known way to produce hydrogen with a negative carbon footprint [4].

2.2.1. Biohydrogen Technologies

Thermochemical biohydrogen production processes include, for example, pyrolysis and gasification. Pyrolysis is gasification under the absence of oxygen at temperatures between 500 and 900 °C and pressures between 0.1 and 0.5 MPa. As a result, no dioxins or carbon oxides are formed, given that the input material is dry and no air or oxygen is present. The products are synthesis gas and carbon residues, such as liquid oils and solid charcoal [31] (p. 14), see reaction (1.1) in Table 1. Advantages of the process are the wide variety of possible input materials, a simple underlying process, compact plants, a clean carbon by-product (fuel oil or char), and reduced CO or CO₂ emissions compared to gasification [32] (p. 5). That is, the hydrogen produced from biomass conversion can

be classified as green, due to the carbon-neutral nature of all biomass. In other words, the resulting GHGs during the processes were already removed from the atmosphere in the first place. Biomass is seen as the most feasible candidate to substitute fossil fuels in the near future [32] (p. 5) and thermochemical processes, especially gasification, are already mature. However, biomass conversion faces logistics issues since it requires enormous amounts of input material in large-scale plants. Another problem is tar fouling the plant and the produced gas, which impedes wider market implementation. Possible solutions for the logistics issue are efficient plant downscaling and the smart distribution of conversion plants [32] (p. 5), paving the way for thermochemical processes as the only non-biological means of hydrogen production from biomass [31] (p. 13).

Table 1. Overview of chemical reactions for H₂ production from biomass [32-34].

Number	Process	Reaction
(1.1)	Biomass pyrolysis	$C_nH_m \rightarrow n C + 0.5 m H_2$
(1.2)	Glucose formation	$6 \operatorname{H}_2\operatorname{O} + 6 \operatorname{CO}_2 \rightarrow \operatorname{C}_6\operatorname{H}_{12}\operatorname{O}_6 + 6 \operatorname{O}_2$
(1.3)	Enzymatic hydrogen generation	$C_6H_{12}O_6 + 6 H_2O \rightarrow 6 CO_2 + 12 H_2$
(1.4)	Microbial electrolysis (acetate)	$\rm CH_3COO^- + 4~H_2O \rightarrow 2~HCO_3^- + H^+ + 4~H_2$
(1.5)	Photo-fermentation (acetic acid)	$CH_3COOH + 2 H_2O \rightarrow 4 H_2 + 2 CO_2$
(1.6)	Acidogenesis (butyric acid)	$C_6H_{12}O_6 \rightarrow CH_3CH_2CH_2COOH + 2\ CO_2 + 2\ H_2$

One of the biotechnological process options is biophotolysis. In this process, lightsensitive microorganisms in illuminated reactors perform photosynthesis first to convert water and CO₂ to glucose and oxygen (1.2) before photo-activated enzymes split the sugar and release hydrogen gas (1.3). Microalgae are promising organisms for this process, as they can be cultured and allow better hydrogen capture. In general, the process happens under standard conditions in a water environment and does not require cost-intensive input material, although the sufficient illumination of the substrate is an arising upscaling issue [34] (p. 11101). Moreover, the hydrogenase is inhibited by O_2 , which presents an additional obstacle in the process [33] (p. 8).

Another process, microbial electrolysis, is based on the ability of certain microorganisms, called exoelectrogenic microbes, to produce electric current by oxidizing organic matters. To generate hydrogen, a small external power supply drives electrons—produced from organic matter in the anode chamber—through a wire to the cathode. The applied current is typically 0.2 V, which is very low compared to 1.6–1.8 V for water electrolysis [33] (p. 11). Together with protons, which permeate through the proton exchange membrane, the electrons form hydrogen gas in the cathode chamber. An exemplary reaction for acetate as an organic compound (1.4) is shown in Table 1. This system is called a microbial electrolysis cell, which presents high hydrogen yields and efficiencies. However, the performance depends on the microbes used and the exchange efficiency of the membrane, as well as on the electrical conductivity and chemical stability of anode and cathode. Additional tasks are the reduction in internal resistance and material cost, as well as the increase in biomass concentration in the anode chamber [33] (p. 11, 12).

Photo-fermentative purple and green bacteria are able to preferably convert organic acids, but also carbohydrates or industrial effluents, to H_2 and CO_2 , utilizing energy from light in anaerobic environments [33,35]. An exemplary reaction of this photo-fermentation process is executed by purple non-sulfur bacteria, as shown for acetic acid (1.5) in Table 1. The process is highly sensitive to the presence of oxygen or ammonia and depends on light intensity, carbon source, and type of microorganism [35] (p. 575). The utilization of solar energy and organic wastes promises cheap operation. However, as hydrogen production rates and efficiencies for light conversion are low, while reactor costs are high and illumination problems immense, the process is rather impractical [33] (p. 9).

Compared to photo-fermentation and bio-photolysis, the so-called dark fermentation provides the huge advantage of not being dependent on light as well as featuring simple reactor design and operation [33] (p. 10). The main idea in dark fermentation is to inhibit the hydrogen-consuming methanogenesis, which is the last step of an anaerobic biogas

process, and harvest the hydrogen. The environment can be mesophilic, thermophilic, or hyperthermophilic, depending on the microbial strain to be used for production [36] (pp. 24, 25). The acidogenic phase particularly degrades carbohydrates, fats, and proteins to volatile fatty acids (VFAs), alcohols, amino acids, H_2 , and CO_2 . For example, pyruvate— an intermediate in the metabolization of glucose—is converted to volatile fatty acids, e.g., acetic, butyric, propionic, and malic acid, and hydrogen [31] (p. 9). As an example, the reaction of glucose to butyric acid (1.6) is shown in Table 1. Due to the similarity to the classical biogas process, dark fermentation is seen as a feasible way to produce hydrogen from organic wastes. However, low hydrogen yields and substrate degradation need to be optimized. To achieve that, process operation is still a matter of development, with the suppression of methanogens being crucial to the final hydrogen yield [32] (p. 11).

The so-called dark photosynthesis process is defined by the possibility of accomplishing photosynthetic growth patterns for Rhodospirillum rubrum, a Gram-negative, nonsulfur purple bacterium and a facultative anaerobe, with a special culture medium under dark conditions. This growth medium, called M2SF, is capable of enabling mass cultivation under semi-aerobic dark conditions. Such behavior is usually limited to anaerobic, photoheterotrophic growth [37,38]. The result of this new means of growth—mimicking light signals to the photosynthetic genes in a dark environment—is that the versatile metabolism of R. rubrum can be utilized in the absence of light, which makes it perfectly scalable.

2.2.2. The HyBECCS Approach

According to the sixth assessment report of the IPCC, emissions pathways that limit global average warming to $1.5 \,^{\circ}$ C or $2 \,^{\circ}$ C by 2100 require the use of carbon dioxide removal (CDR) approaches in combination with emissions reductions to pursue a pathway with net negative CO₂ emissions from 2050 on. Hence, affordable as well as environmentally and socially responsible CDR options on a large scale well before 2050 are an important element for climate mitigation and play a central role in limiting global warming to $1.5 \,^{\circ}$ C or $2 \,^{\circ}$ C [1]. Negative emission technologies (NETs), of which bioenergy carbon capture and storage (BECCS) as well as ocean fertilization, mineral carbonation, afforestation, and direct air capture (DAC) are part, are important by means of reducing anthropogenic net emissions in the near future [39] (pp. 1124, 1125). An emerging subclass of BECCS technologies is the concept of hydrogen bioenergy coupled with carbon capture and storage (CCS) technology [4]. These process combinations, referred to as HyBECCS, are described briefly below.

Most of the biohydrogen technologies are able to split biomass substrates into H_2 and CO_2 , which is beneficial for an effective carbon capturing process, as this mixture can be separated. The hydrogen fraction can be used to substitute fossil energy carriers in industry or transport in order to decarbonize these sectors. The carbon dioxide can be put to long-term use or in storage, leading to a carbon sink since it was previously removed from the atmosphere through plant biomass growth in relatively short periods of time. Carbon dioxide storage possibilities include enhanced oil or gas recovery, storage in saline formations, depleted oil and gas fields, and others [39] (Figure 1). An alternative to storage is usage, where CO_2 is stored for a certain period of time in applications such as fuels, construction materials, chemicals, and plastics [39]. All process combinations referred to as HyBECCS can be split into four basic process steps: biomass pretreatment, biohydrogen production, product gas separation, and its processing. For each of the four steps, there are variable options to be selected and combined into an overall process.

The main technological advantages of HyBECCS over competing NETs are, on the one hand, the energy-efficient capture of CO_2 with high purity and concentration compared to DAC approaches. On the other hand, it offers high carbon dioxide removal potential compared to BECCS technologies based on hydrocarbon energy carriers, which emit CO_2 during their decentralized use. Based on these advantages, HyBECCS is considered to be a promising and highly effective BECCS approach that is expected to play a major role in future energy systems by making a significant contribution to meeting the world's

rapidly growing demand for green hydrogen and negative emissions [4]. The HyBECCS approach can also be seen as part of the biological transformation process, which aims to achieve a sustainable economy that is no longer dependent on the depletion of fossil fuels by linking classical production with bio- and information technologies, leading to a new and disruptive innovation space [40,41].

2.3. Terminology and System Boundaries

In the past, the term 'potential' was not clearly defined in various studies, decreasing comparability and complicating interpretation. System boundaries define the resulting potentials and need to be chosen to reduce the effort of elaboration, but they still have to support the purpose of a piece of work, so it is strictly necessary to define them precisely. As a result, the terminology and systematic approach for this sort of estimation of potential have been standardized to some extent. In this work, the definitions according to Thrän et al. are used and summarized in the following [19]. The different types of potentials are outlined and their differences, e.g., level of observation and system boundaries, are highlighted and placed in the present context.

Biomass potentials can be differentiated according to the level of consideration [19]. Land potential, feedstock potential, fuel potential, and bioenergy potential can be distinguished. An overview of these levels and the resulting potential terms can be seen in Figure 2. The land potential describes the area available for energy crop cultivation, which is not relevant to this work, as it deals with organic waste material, and the feedstock potential is not calculated from the underlying land potential but represents the amount of applicable material for biohydrogen production. From that point, the fuel potential can be obtained from the hydrogen yield per unit of substrate and the respective lower heating value of H_2 . Lastly, the bioenergy potential can be derived by considering the relevant conversion efficiency of hydrogen to heat and/or electric energy and the energy input for pretreatment and process operation.



Figure 2. Levels of utilization pathway and corresponding potential terms [19].

Each potential requires specific system boundaries and framework conditions, necessitating further explanations. The biomass potentials obtained in a piece of work depend on a variety of different factors, such as the biomass material considered, the geographic level, the temporal reference, the type of data collection, and the methodology of conduction. In addition, different perceptions of competitive usage, sustainability aspects, or nature conservation issues often cause confusion [19] (p. 40).

These aforementioned potentials need to be defined more precisely, as to whether they reflect a theoretical, technical, or other potential. First, a theoretical potential describes the overall energy amount for a specific region in a specified period of time. It resembles the maximum amount of energy supply contained in the considered biomass, as it is based on the physical utilization boundaries alone. The practical relevance is minor, because the theoretical potential provides no statement of the actually usable amount of energy [19] (p. 42).

Second, the technical potential considers certain restrictions, and hence gives a more realistic result of utilizable biomass in a specific region and time. Considered restrictions are of technical (e.g., recovery rates and conversion losses), legal, environmental (e.g., environmental protection regulations, cross-compliance regulation), and social (e.g., food production, material utilization) concern. The technical potential is widely used in studies, as it is less affected by temporal fluctuations than the economic potential, for instance [19] (p. 42).

The economic potential reflects a part of the technical potential, which can be utilized considering further restrictions related to the economic framework. Adding even more restrictions allows for distinguishing additional potential terms. If increased importance is laid on ecological and environmental factors, one is speaking of sustainable biomass potential. Going one step further in the implementation direction, realizable potential can be obtained by further considering sociopolitical and practical restrictions [19] (p. 43).

That is, a scientific piece of work needs to explicitly lay out its use and interpretation of the mentioned terms. Moreover, the boundary conditions such as region, timescale, considered biomass, process operation, material handling, output, and methodical conduct need to be mentioned. An overview of the relevant potential terms is given in Figure 3. The following restrictions are suggested to be included when deriving a technical potential by Thrän et al. [19] (p. 44):

- Societal variables (as general agreement whether certain feedstock should receive a generally preferred form of utilization).
- Technical variables (cultivation, harvest, recovery, and conversion technology).
- Demand for food and material utilization.
- Ecological/environmental variables (legal requirements to ensure a sustainable resource base).



Figure 3. Types of potential according to system boundaries [19].

As this work aims to present a methodological approach to derive technical potential for biohydrogen production, these factors are included as far as possible. Some aspects, however, have to be simplified or neglected. Relevant factors, which exceed the scope of this work, are of demographic background, such as population trend, of socio-economic background, such as consumption habits and environmental consciousness, or of policy-making, regulatory, or organizational background, such as improved circular economy, reduced food waste, or stricter EHS standards [22] (p. 84). Other specific topics relevant to this process have not been included, such as substrate logistics, and storage or process residue handling. The restrictions applied to theoretical amounts of waste materials are, together with chosen simplifications and unincluded aspects, laid out in Section 3.

3. Methodological Approach

The following presents a systematic four-step methodological approach to obtain bioenergy potential and negative emission potential of biohydrogen production. Each step is elaborated in the following subsections, as shown in Figure 4. In the first step, a procedure to select relevant biomass categories for hydrogen production is elaborated, which is divided into six sub-steps. Second, suggestions are made for assumptions and framework conditions regarding biomass pretreatment, process energy demand, and logistical specifics. In the third step, in order to derive a calculation equation for the determination of the bioenergy potential for biohydrogen production, relevant indicators such as the lower heating value (LHV) or the fuel potential (FP) are introduced. This was done in accordance with standard practice and in compliance with certain criteria intended to ensure comparability and transparency. Finally, the estimation of the negative emission potential of biohydrogen processes is discussed regarding their contributions to climate mitigation. Furthermore, a calculation to estimate the saved amount of CO₂ emissions as well as for the substitutable amount of (fossil) energy carriers is presented.



Figure 4. Overview of the presented methodological approach for bioenergy potential estimation and ecological assessment (**left**) and the respective results (**right**).

3.1. Feedstock Selection

The first and most important step is to identify and quantify biomass flows utilizable for biohydrogen production. These resource streams are the basis, as the hydrogen yield

and all further results directly depend on the amount of utilizable substrate. Brosowski et al. defined 93 different biomass categories (BCs), which are drawn upon as a foundation for substrate selection in this work. In this section, the different BCs are evaluated regarding their applicability to the considered biohydrogen process, taking into account the prerequisites of the process and the biomass utilization hierarchy. All of the categories can be seen in Annex 6 of Brosowski et al. [21]. The selection process, which is conducted according to the following steps exemplified for biotechnological, fermentative biohydrogen production, can be illustrated in a decision tree as seen in Figure 5:

- Step 1 Data validity, utilization pathways, and classification issues: According to Brosowski et al., data and information are inconsistent for 16 biomass categories (BCs) and no technical potential is presented for eight categories. Those classes are sorted out in this step. Moreover, many food industry wastes, such as coffee or tobacco residues, must be disposed of according to special regulations. Other biomass flows, such as residues from convenience food production, are highly heterogeneous and assumed not to contain large amounts of relevant contents [42] (Chapter 2). Thus, it is not possible to make reasonable assumptions about the composition and such material flows should not be considered for utilization. In this case, 32/93 BCs can be discarded in this step.
- Step 2 Lignocellulosic biomass: Biomass categories containing wood or wood-like waste, straw, stalk, or other green waste are currently not suitable for some biotechnological processes, as they need to be readily biodegradable by microorganisms. Wood essentially consists of cellulose, hemicellulose, and lignin, which are poorly biodegradable without extensive pretreatment. Hence, they are rather unsuitable for fermentative processes [20] (p. 53); 28/93 BCs match this criterion and should be sorted out.
- Step 3 Wastes of animal origin: Such wastes should not be considered for alternative use because of the potentially contagious material and an established industry for efficient utilization of animal origin wastes. Gaida et al. describe the utilization of this biomass as already optimized. For instance, animal skin is processed to leather, and protein-rich residues are processed into tankage or fertilizer. Other potential pathways are biodiesel production and utilization as substances in the oleochemical industry [42] (pp. 22, 24). However, if a special suitability can be derived for specific biohydrogen process options, a comparison with the respective reference processes might be of interest. If not, 2/93 BCs match this criterion and should be sorted out.
- Step 4 Oils and fats: In a comprehensive evaluation of utilization pathways for this category, Fehrenbach et al. found that biodiesel production might be superior to other means of utilization [22] (pp. 112, 179). A comparison of the biohydrogen process under consideration with this form of use is therefore to be initiated. If the comparison is decided in favor of biodiesel production, another 2/93 BCs are sorted out in this step.
- Step 5 Waste material untypical for fermentation: Some waste types can be considered as biomass, but are not readily biodegradable, such as textiles or packaging material. These must also be subjected to a special test to determine their suitability for biohydrogen production. If this test is negative, 2/93 BCs match this criterion and have to be sorted out.
- Step 6 Animal feed or low sugar content: According to Section 2.1., material use of biomass should be preferred over energetic use in the case of animal feed, for instance. Hence, plenty of different process residues from the food industry cannot be seen as waste and are consequently not considered in this estimation of potential. Usage competition can arise if such materials are utilized energetically. Residues from the sugar industry would be perfectly suitable for fermentative biohydrogen processes, due to their sugar content. However, this material is widely used as animal feed, in the yeast industry, or in distilleries. The work of Gaida et al. indicates high-value material use for most food industry residues at present [42] (Section 2). However,

153

dairy industry residues, which mostly consist of whey [42] (p. 40), are excluded from this step. The amount currently fed to pigs is found to be better used for human consumption, which mainly consists of the contained whey protein [43] (p. 271). The contained lactose, which is a substrate for homolactic fermentation, remains in the permeate after protein extraction, hence it can be considered. Other biomass categories, such as residues from alcohol production, must first be checked as to whether they contain significant amounts of utilizable substances for biohydrogen production. Finally, 17/93 BCs could be excluded in this step.



Figure 5. Decision tree to select suitable biomass categories for an exemplary biotechnological biohydrogen process.

In summary, the majority of waste types can be sorted out for this exemplary biotechnological biohydrogen production, as illustrated in Figure 5. Technical feedstock potentials with maximum and minimum values are available in Brosowski et al. [21] for the respective material flows. Since those estimations of potential are of different sources, going as far back as 2000, certain values can be updated where possible. For instance, technical figures of potential are in some cases directly related to official waste statistics, or new values for 2015 can be accessed in the online database of the DBFZ. In the next sections, the waste flows are developed into a technical bioenergy potential by making assumptions according to Sections 3.2 and 3.3.

3.2. Definition of Assumptions and Framework Conditions

This section briefly presents proposed assumptions required by certain steps in the process of developing energy potentials.

3.2.1. Substrate Pretreatment

To increase process efficiency and hydrogen output, it is necessary to prevent disruptions and clogging and ensure sufficient homogenization for stable process operation. Additionally, the pretreatment of substrate is essential in biochemical conversion processes [44] (p. 34). The considered pretreatment methods for this exemplary process are grinding, hygienization, and thermochemical hydrolysis. Homolactic fermentation and anaerobic storage are more of a recommended storage option, rather than a pretreatment measure in that sense. Filtration or other more elaborate pretreatments (e.g., fructose extraction) could be required. All of the mentioned processes are shortly described and reasoned in the following. After that, each type of waste is assigned a probably suitable pretreatment method or combination of methods to ensure ideal process operation. This allocation is carried out based on subjective estimations. Certain pretreatments are especially common in biogas processes but are assumed necessary or advantageous for biohydrogen production, especially for biotechnological process routes. The derived energy demands are related to tons of fresh mass (FM).

Grinding (G): Grinding is a common pretreatment of substrates in biogas processes for several reasons. It increases the surface area, which supports sufficient and fast microbial degradation. In addition, it reduces the risk of issues such as clogging, reduced pumping ability, and agitator stagnation [44] (p. 35). Water or liquids, such as dairy industry residues, can be added during grinding to transform the substrate into the right condition. For the energy demand of grinding, a shredder (P = 37 kW with a throughput of 15 t/h) was adduced as an example. This results in an electric energy demand of $W_{el} = 2.46 \text{ kWh/t} \stackrel{\circ}{=} 8.9 \text{ MJ/t}$.

Hygienization (H): As the selected material flows can include smaller amounts of material from animal origin, e.g., municipal waste, hygienization could be necessary. If no material from category 1 or 2 (except animal feces, which are subject to exceptions) is assumed to be in the selected material, pasteurization at 70 °C for 60 min (Cat. 3: ground to <12 mm) is a sufficient treatment, according to Regulation (EC) No 1774/2002 Annex VI, Chapter II, C. In this work, pasteurization is considered sufficient to ensure hygienic input substrate. Besides that, pasteurization might trigger certain hydrolysis processes in the material and consequently increase process output. An advanced sterilization method would be pressure sterilization (133 °C, 3 bar for at least 20 min), which could be used if pasteurization does not achieve satisfactory bacterial reduction. The energy demand for pasteurization can be considered as follows:

- electric: $W_{el} = 2.5 \text{ kWh/t} \doteq 9 \text{ MJ/t} (W_{el} = 2.1-2:3 \text{ kWh/m}^3 [45] (p. 105); assumed substrate density 1000 kg/m^3);$
- thermal: $W_{th} = 12 \text{ kWh}/t = 43.2 \text{ MJ}/t (W_{th} = 11.6-11.9 \text{ kWh}/m^3 [45] (p. 105).$

Thermochemical hydrolysis (TCH): Vavouraki, Angelis, and Kornaros showed that a lot of glucose can be mobilized in kitchen wastes through thermochemical hydrolysis (TCH) by relatively simple hydrochloric acid pretreatment. As all material needs to be hygienized, especially for biotechnological biohydrogen production, this process can be used for additional sugar mobilization. Tests showed that glucose content can be increased by almost 600% by treating kitchen wastes at 100 °C with 1.12% added hydrochloric acid for a period of 108 min [46] (p. 744). It is assumed that the thermal energy demand is higher by 50% than for hygienization while the electrical one remains equal. It should be noted that hydrolyzed material requires subsequent neutralization to not harm the culture. The energy demand for TCH can be considered as follows:

- electric: $W_{el} = 2.5 \text{ kWh/t} \doteq 9 \text{ MJ/t} (W_{el} = 2.1-2.3 \text{ kWh/m}^3 [45] (p. 105);$ assumed substrate density 1000 kg/m³);
- thermal: $W_{th} = 18 \text{ kWh/t} \stackrel{\circ}{=} 64.8 \text{ MJ/t}$.

155

The controlled anaerobic storage (AS) of animal excrements enables an initial hydrolysis process, in which VFAs are produced [47] (p. 160). In consequence, anaerobically stored slurry and manure present a significantly higher content of VFAs, which are considered utilizable for certain biohydrogen processes. The temperatures observed in [47] ranged between close to 0 and almost 25 °C, therefore no thermal energy input is assumed. The electrical energy input is neglected.

Enhanced homolactic fermentation (HLF) is another suitable pretreatment step for wastes from the dairy industry. In this step, lactic acid is produced, which is utilizable for some biohydrogen processes. Dairy products, especially whey, are suitable raw materials as they already contain lactic acid-producing bacteria and important nutrients such as lactose [48] (p. 2). As the process temperatures range between 5 and 45 °C, no energy input must be considered for this pretreatment. Further details, such as reactor type, retention times, or potential additives, are not further discussed in this work.

All pretreatment methods, including the respective energy demands, are listed in Table 2. In addition, a suitable pretreatment or a combination of pretreatments is assigned to each type of biomass, which can be seen in Table 3.

Ν	lethod	Energy 1	Demand
Symbol	Name	W _{el} in MJ/t	W _{th} in MJ/t
(G)	Grinding	8.9	0 *
(H)	Hygienization	9	43.2
(TCH)	Thermochemical hydrolysis	9	64.8
(AS)	Anaerobic storage	0 *	0 *
(HLF)	Homolactic fermentation	0 *	0 *

Table 2. Overview of energy demands for pretreatment methods.

* No energy input allocated or negligible as described in Section 3.2.1.

Table 3. Overview of selected pretreatment for each type of waste.

Biomaco Catagora	Pretreatment						
biomass Category	(G)	(H)	(TCH)	(AS)	(HLF)		
Cattle manure and slurry		х		х			
Pig manure and slurry		х		x			
Cattle dung	х	х		x			
Pig dung	х	х		х			
Organic household waste	х		х				
Kitchen and canteen waste	х		х				
Waste from markets	х		х				
Dairy industry residues		х			х		
Organic fraction of MSW	х		х				
Commercial food waste	х		x				

3.2.2. Process Energy Demand

As the process requires similar technical equipment and reactor design compared to a biogas process (e.g., agitators, pumps), the overall process energy demand for biotechnological biohydrogen processes can be estimated by average figures for biogas production from the Agency for Renewable Resources (orig. Fachagentur Nachwachsende Rohstoffe e. V., FNR). Energy-demanding features for biohydrogen-specific subprocesses have to be estimated separately or can be neglected in some cases. The FNR publishes key figures from several sources and its own calculations on its webpage, where the estimated figures represent percentages of the plant's total energy output. Pasteurization and thermochemical hydrolysis already preheat the material, therefore the actual thermal energy demand

156

(heating) of the bioreactors can be neglected in this case. Preheated substrate from pasteurization is assumed to maintain process temperatures between 25 and 30 °C. The process energy demand is estimated to be 7.6% electric (f_{el}) [49] and, due to the aforementioned reasons, 0% instead of 28% thermal (f_{th}) of the overall energy generation capacity.

3.2.3. Logistics, Substrate Storage, and Handling of Residues

A comprehensive assessment of technical energy potentials is a multi-faceted challenge, which also requires examination of the material throughout its life cycle. For biohydrogen processes, real sustainability also includes avoiding long transport routes, efficient use of existing infrastructure, and material sources, among other aspects. This paper does not cover such aspects, due to their complex nature, regional variation, and the lack of traceability for waste materials on a large-scale basis. Such elaborate factors are found to be difficult to generalize due to the wide variety of possible hydrogen production options, and hence are to be assessed on a case-by-case basis. In the scope of this work, the process impact is evaluated from material pretreatment until hydrogen generation. The residues of the process can contain additional potential, which could further enhance the efficiency of the overall utilization. In any case, the material needs to be disposed of or treated by other means consecutively.

3.2.4. Produce Gas Separation

Depending on the biohydrogen production process, the initial produce or synthesis gas may be composed of various compounds. For biotechnological biohydrogen production, occurring produce gas compounds include CO₂ and H₂ and may contain N₂, water vapor, or trace compounds such as NH₃, CO, and H₂S, for instance [50,51]. As fuel cell technology demands high hydrogen purities (99.99%), effective separation is a key process in biohydrogen production [51] (p. 1277). Common separation systems include scrubbing, pressure swing adsorption, and cryogenic separation, all of which are energy intensive, and membrane technologies show great potential, but many are still in development [31,51]. Membrane technologies allow for simple separation at lower energy requirements, hence further development of these technologies is essential for biohydrogen production as a whole. The detailed consideration of energy inputs during this part of a production process is not part of this work. Nevertheless, this process step and its related energy inputs demand consideration in bioenergy potential estimations and should be included in future research and case studies.

3.3. Calculation of Energy Potential

In this step, the obtained feedstock potential and producible hydrogen can be converted into fuel and bioenergy potentials, respectively. With an established amount of moles H₂ (n_{H_2}), the annual mass of hydrogen gas output (m_{H_2}) can be calculated. The total energy content of an energy carrier is described either by the higher heating value (*HHV*) or the lower heating value (*LHV*), with combustion being the underlying process. The difference between *HHV* and *LHV* is that *HHV* assumes the produced water to be in a liquid state. Combustions usually include fuel of organic origin or, in this case, hydrogen, which reacts with oxygen. The reaction products are CO₂ and H₂O for carbon-containing sources and H₂O for hydrogen. In practice, the heat generation causes water to evaporate, hence the *HHV* can only be achieved if the flue gas energy is also fully harnessed, which is not always possible [48] (p. 23). As a result, the *LHV* is used to determine the utilizable energy content in this work. For hydrogen gas, the *LHV* is 120 MJ per kg [36] (p. 239). The resulting fuel potential (*FP*) can be calculated according to Equation (1).

$$FP = m_{H_2} \cdot LHV_{H_2} \tag{1}$$

The overall energy conversion efficiency of modern fuel cells can be as high as 90%, with 45–60% electrical. Hydrogen-only operation of such cells, which currently mostly use natural gas as fuel, can bring several benefits and even higher efficiencies [52] (pp.

374, 377). This work considers an overall conversion efficiency of 90% total (η_{tot}) and 50% electrical efficiency. In addition, the energy demands for substrate pretreatment (W_{PT}) and process operation (W_{PO}), which reduce the overall energy output, are considered in this final step. The respective energy amount of pretreatment is calculated by summing up the electric and thermal energy consumption for each method, see Equation (2). To include the process energy demand, the bioenergy potential can be reduced by the factors of Section 3.2.2 (f_{el} and f_{th}). The resulting bioenergy potential (*BEP*) is the final, utilizable amount of energy after conversion [19] (p. 41), reduced by the amount of energy consumed by pretreatment and process operation, and is calculated according to Equation (3). Additional energy inputs from logistics, storage, residue treatment, or gas separation can be included here for more detailed case studies.

$$W_{PT} = W_G + W_H + W_{TCH} + W_{AS} + W_{HLF}$$
⁽²⁾

$$BEP = FP \cdot \eta_{tot} \cdot (1 - \text{fel} - \text{fth}) - W_{PT}$$
(3)

3.4. Negative Emission Potential Estimation

In the last step, the environmental impact of the biohydrogen process is estimated. Aspects such as the achievable substitution of other energy carriers and the resulting emission reduction, as well as the extent of the possible negative carbon footprint by adding CCS to the process leading to HyBECCS approaches, are to be considered in this assessment step. It should be mentioned that HyBECCS projects can have multiple impacts on the environment, such as acidification and eutrophication or the depletion of the ozone layer and abiotic resources [53]. Even though all environmental impacts are equally important, they are not included in the following estimation, as it is limited to considerations on climate mitigation.

First, the carbon-neutral energy output of the biohydrogen process can be expressed by an equivalent mass of other energy carriers, which could be substituted ($m_{subs,x}$). This step can be realized by dividing the bioenergy potential (*BEP*) by the *LHV* of other energy carriers (*LHV_x*), see Equation (4). At the same time, the saved amount of CO₂ emissions ($m_{CO_2,x}$) can be calculated by dividing the energy output of the biohydrogen process by respective CO₂ emission factors from other energy carriers (*ef*_x), see Equation (5).

$$m_{subs,x} = \frac{BEP}{LHV_x} \tag{4}$$

$$n_{\mathrm{CO}_{2},x} = \frac{BEP}{ef_{x}} \tag{5}$$

The considered fossil energy carriers to be substituted are natural gas, coke, hard coal, and heavy fuel oil and hydrogen from natural gas steam reforming and water electrolysis powered by the German electricity mix. Further, green hydrogen from renewable electrical power can be considered. The fossil energy sources were chosen for comparison, as they are currently used in potential future application fields for hydrogen (e.g., steel industry, (heavy-duty) traffic, heat generation). The alternative hydrogen generation options highlight the direct benefits of biohydrogen (with and without CCS) as compared to other energy sources. Reference values for LHV_x and ef_x can be seen in Table 4.

1

158

Table 4. Reference values for various energy carriers.

Energy Carrier	LHV_x in MJ/kg	ef_x in t_{CO_2}/TJ	References
Natural gas (NG)	47.4 ³	56	[54] (p. 138)
Coke (C)	27.6	105	[54] (p. 138)
Hard coal (HC)	28.3	93	[54] (p. 138)
Heavy fuel oil (HFO)	39.5	81	[54] (p. 138)
H_2 steam reforming (SR) ¹	120	67	[26] (n, 220)
H_2 electrolysis electricity mix (EM) ²	120	83	[50] (p. 239) [55] (Table 1)
H_2 electrolysis renewables (REN) ²	120	0	[55] (Table 1)
H_2 electrolysis renewables (REN) ²	120	0	

 $\overline{1}$ η = 84%; 1190 GJ_{NG}/GJ_{H₂}; 2 η = 70%; 1428 GJ_{el}/GJ_{H₂}, EM 2018 has 30% lower emission factor compared to 2012 [56]. Hence, the value from [57] (Table 1) is reduced by the same percentage. ³ *LHV*_{NG} = 36.0 MJ/Nm³ and assumed density ρ_{NG} = 0.76 kg/Nm³ (Nm³ = norm cubic meter).

HyBECCS was introduced in Section 2.2. as an innovative way to produce carbonnegative hydrogen and is considered as an evaluation step in this work. The methodical conduct to obtain a theoretically storable amount of CO₂ through HyBECCS is defined as follows: The mass of the product gas mixture (m_{gas}) in the respective biohydrogen process and its CO₂ content (x_{CO_2}) must be known or estimated. It should be noted that the real CO₂ production rate often is a matter of uncertainty and depends largely on the composition of the input substrate. Real data are therefore to be preferred for the specific process. The resulting amount of CO₂ produced during the biohydrogen process can be calculated according to Equation (6). This can be seen as the theoretical amount of storable biogenic carbon dioxide leading to the same value of theoretically achievable negative emission potential (*NEP*) in CO_{2,eq} through the specific HyBECCS approach.

$$NEP = x_{CO_2} \cdot m_{gas} \tag{6}$$

4. Discussion

This section discusses the conducted work. First, the considered material flows are briefly touched upon. Next, limiting factors and assumptions are taken up and evaluated in light of associated uncertainties, restrictions, and alternative points of view.

Initially, a range of biomasses applicable to the process can be determined using the methodological approach presented. The decision to exclude by-products and biomass, which can be used as material resources, is a very important first restriction. Utilizing industrial residues, for instance, which are commonly used as animal feed, could create shortages in the supply for animal feed production. Consequently, this shortage would have to be filled by other resources. In terms of sustainable impact, those resources could possibly be of higher value, such as maize or soy, resulting in a shortage for human consumption so that in the further course these resources would have to be imported from other countries. This is just one example of emerging competition for usage, especially of relevance with industrial residues. In fact, the efficient use of by-products saves a lot of resources and energy as they fulfill demand, which otherwise would have to be satisfied elsewhere, potentially in a more energy-intensive manner. For those reasons, the material considered in this work is chosen to not present competition in material usage. Theoretically, industrial residues, animal-processing wastes, or oils and fats, amongst others, could still be used in biohydrogen production processes. However, the consideration of such material would have to be conducted in greater detail for specific regions or industries, as several sources indicate that nationwide data for industry by-products and other smaller biomass fractions are not available or based on vague assumptions. Hence, the respective data embody broad spreads and limited reliability [20] (p. 17), [22] (p. 103), [42] (p. 108).

An initial and expected finding based on the methodological approach shown in this paper is that pretreatment—especially hygienization—has an extensive impact on the bioenergy potential. The additional step of including other means of treatment, though not being necessary in the same sense as hygienization, can be reasoned by the expected increased degree of mobilization and the resulting enhanced process outputs. Altogether, the reliability of the obtained energy values and substrate properties can be considered as high, as they are taken from scientific papers. Associated risk factors are, on the one hand, the requirement of pressure sterilization in case of insufficient sterilization performance of pasteurization. On the other hand, the necessity to extract utilizable substances from the substrate could be required. Both of these factors could significantly increase pretreatment energy demand. Next, the process energy demand is an essential factor as well. The uncertainties concerning the process energy demand are associated with inaccuracies. Additional energy demand might result from further heat demand for process operation, for instance. Overall, a conservative approach has to be chosen to evaluate the processes.

The derivation of environmental impacts can provide an initial basis for decisions on the sustainability of the processes. Other relevant aspects, however, are substrate logistics and storage, as well as the handling of process residues, which have not been considered in the methodological approach presented. These topics were beyond the scope of this work but might pose certain risks or chances. For example, lengthy substrate transportation or improper storage could decrease the overall net potential of the biohydrogen processes, depending on the hauling distance or the loss of utilizable substances during storage. On the contrary, process residues might contain substances to be metabolized in further energetic treatment or biomass with high-value components. As a result, extraction of such high-value components, further energy generation from the substrate, or just compost producible from it could improve the economic and ecological value of the process.

The methodological approach shows weaknesses regarding process-specific developments. Some of those factors affecting the amount of available waste materials could be increased awareness of food waste, changing consumption habits, and the population trend. Regulatory measures, policy goals, and environmental concerns have been considered to some extent, but not in greater detail. It is considered acceptable to assume a status quo situation for these factors in the short term but has to be adjusted for the evaluation of long-term potentials. This work provides a starting point for evaluating bioenergy and climate mitigation potentials by providing a quickly implementable guide for estimating the bioenergy potential for hydrogen production from biomass and its potential greenhouse gas savings in Germany. However, many technology-specific factors that influence these potentials remain unconsidered. In addition to substrate suitability and conditions like wetness and hydrogen yield, these include, for example, technology maturity, energy efficiency, conversion rates, and plant complexity, in order to obtain a conclusive picture for the assessment. In addition, other environmental impact categories according to DIN EN ISO 14040/44, besides global warming potential, such as acidification and eutrophication potential or photochemical oxidant formation, as well as ozone layer and abiotic depletion, should be considered within holistic assessments.

5. Summary and Prospect

It is common ground that the future energy market requires efficient carbon-neutral or negative solutions in green energy production. Biohydrogen production technologies are expected to be capable of producing large amounts of hydrogen from organic biomass in the near future. Further, co-produced biogenic carbon dioxide can be stored or put to longterm use by applying hydrogen bioenergy with carbon capture and storage (HyBECCS) approaches to generate greenhouse gas sinks. This work aims to provide a guideline for deriving the bioenergy potential of organic waste material and the connected ecological impacts of biohydrogen and HyBECCS processes in Germany.

A four-step methodical conduct, including material selection, assumptions, calculation of energy potential, and environmental impact evaluation, is elaborated to achieve the desired outcome. First, biomass categories are evaluated regarding their applicability to an exemplary biohydrogen process. Therefore, a six-step selection process to exclude material is conducted, resulting in specific biomass categories found to be utilizable and fulfilling the criteria of waste material. Second, a number of assumptions regarding the pretreatment of the substrate and the required process energy demand are made to include the energy input required by the process. Third, the derivation of energy potentials is conducted following common practice and adhering to certain criteria for comparability and transparency. Lastly, the substitutable amount of (fossil) energy carriers and related emission savings, on the one hand, and the negative emissions by applying HyBECCS approaches, on the other hand, are evaluated to obtain a statement about the environmental benefits. The scientific contribution of this article was to develop a methodological approach for a bioenergy potential and a negative emission potential analysis illustrated using a fictitious fermentative hydrogen production process, rather than research based on experimental studies.

In conclusion, rising demand and a functioning hydrogen market are not only expected in Germany, but throughout Europe, which is key to triggering scaling effects and research progress for biohydrogen processes and HyBECCS approaches [2] (p. 5). Therefore, the use of hydrogen as an energy source in the transport sector and decisive industries (steel, cement, and chemical) is desired. These industries play a key role in the development of a German hydrogen market. In addition, the often decentralized produced hydrogen needs to be distributed all over the continent, demanding a modern network of storage and pipelines, suitable for safe H₂ transport. Safety concerns and the applicability of existing infrastructure are to be evaluated and resolved on the part of the German government. Lastly, research on and development of hydrogen production, storage, distribution, and utilization are still in their early stages. This means that technologies are often not advanced enough for market rollout. More applied research and development are needed on biohydrogen processes and HyBECCS approaches as they can contribute to climate mitigation targets to a large extent [5]. Additionally, the new technologies need to be fully analyzable and evaluable during the innovation and upscaling processes in order to be successful on the market and prove their ecological potential. Therefore, the ecological and economic potential including all life cycle costs and emissions should be further investigated and a common framework be created, since the competitive performance is the main short-term indicator of success for biohydrogen processes in Germany. Additionally, the processes need to be compared to the competitive utilization technologies for various biomass flows in order to see whether biohydrogen production can be a better option in terms of energy output, carbon emissions, and life cycle impact. In addition, other impact categories besides global warming potential, such as acidification and eutrophication potential or photochemical oxidant formation, as well as ozone layer and abiotic depletion, should be considered using the LCA method according to DIN EN ISO 14040/44. This work, however, provides a starting point for investigating bioenergy and climate mitigation potentials.

Author Contributions: Writing, J.F. and M.T.; supervision, R.M. and A.S. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the German Federal Ministry for Economic Affairs and Energy.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Acknowledgments: The research shown in this paper was conducted within the project "RhoTech" (FKZ 03EI5407C). The authors gratefully acknowledge the financial support of the German Federal Ministry for Economic Affairs and Energy.

Conflicts of Interest: The authors declare no conflict of interest.

References

- IPCC. Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change; Masson-Delmotte, V., Zhai, P., Pirani, A., Connors, S.L., Péan, C., Berger, S., Caud, N., Chen, Y., Goldfarb, L., Gomis, M.I., et al., Eds.; Cambridge University Press: Cambridge, UK, 2021; In Press.
- 2. BMWi. The National Hydrogen Strategy, Federal Ministry for Economic Affairs and Energy, Berlin. 2020. Available online: https://www.bmwi.de/Redaktion/DE/Publikationen/Energie/die-nationale-wasserstoffstrategie.html (accessed on 4 March 2021).

- 3. Juhrich, K. CO₂ Emission Factors for Fossil Fuels. 2016. Available online: https://www.umweltbundesamt.de/sites/default/fil es/medien/1968/publikationen/co2_emission_factors_for_fossil_fuels_correction.pdf (accessed on 26 March 2021).
- 4. Full, J.; Merseburg, S.; Miehe, R.; Sauer, A. A new perspective for climate change mitigation–introducing carbon-negative hydrogen production from biomass with carbon capture and storage (HyBECCS). *Sustainability* **2021**, *13*, 4026. [CrossRef]
- 5. Agentur für Erneuerbare Energien. Bundesländer-Übersicht zu Erneuerbaren Energien (Overview Renewable Energies). Renewable Energies Agency (AEE). 2019. Available online: https://bit.ly/3s3ws6s (accessed on 4 March 2021).
- 6. Schmid, C.; Horschig, T.; Pfeiffer, A.; Szarka, N.; Thrän, D. Biogas upgrading: A review of national biomethane strategies and support policies in selected countries. *Energies* **2019**, *12*, 3803. [CrossRef]
- Weide, T.; Feil, F.; Kamphus, S.; Peitzmeier, J.; Brügging, E. Bio-H₂ aus Organischen Reststoffen Mittels Dunkler Fermentation in Deutschland und den Niederlanden. 2020. Available online: http://www.biohydrogen.eu/wp-content/uploads/2020/09/2020 -09-18-Wasserstoff-Potential-DE-NL_final.pdf (accessed on 4 March 2021).
- 8. Thomassen, G.; van Dael, M.; van Passel, S.; You, F. How to assess the potential of emerging green technologies? Towards a prospective environmental and techno-economic assessment framework. *Green Chem.* **2019**, *21*, 4868–4886. [CrossRef]
- 9. Naqvi, S.; Ali, I.; Nasir, S.; Taqvi, S.A.A.; Atabani, A.E.; Chen, W. Assessment of agro-industrial residues for bioenergy potential by investigating thermo-kinetic behavior in a slow pyrolysis process. *Fuel* **2020**, 278. [CrossRef]
- Thrän, D.; Bauschmann, M.; Dahmen, N.; Erlach, B.; Heinbach, K.; Hirschl, B.; Hildebrand, J.; Rau, I.; Majer, S.; Oehmichen, K.; et al. Bioenergy beyond the German "Energiewende"–Assessment framework for integrated bioenergy strategies. *Biomass Bioenergy* 2020, 142, 105769. [CrossRef]
- 11. Long, H.; Li, X.; Wang, H.; Jia, J. Biomass resources and their bioenergy potential estimation: A review. *Renew. Sustain. Energy Rev.* **2013**, *26*, 344–352. [CrossRef]
- 12. Fu, T.; Ke, J.H.; Zhou, S.; Xie, G.H. Estimation of the quantity and availability of forestry residue for bioenergy production in China. *Resour. Conserv. Recycl.* **2020**, *162*, 104993. [CrossRef]
- 13. Batidzirai, B.; Smeets, E.M.W.; Faaij, A.P.C. Harmonising bioenergy resource potentials—Methodological lessons from review of state of the art bioenergy potential assessments. *Renew. Sustain. Energy Rev.* **2012**, *16*, 6598–6630. [CrossRef]
- 14. Bharti, A.; Paritosh, K.; Mandla, V.R.; Chawade, A.; Vivekanand, V. GIS Application for the Estimation of Bioenergy Potential from Agriculture Residues: An Overview. *Energies* **2021**, *14*, 898. [CrossRef]
- 15. European Environment Agency (EEA). *EU Bioenergy Potential from a Resource-Efficiency Perspective*; EEA: Brussels, Belgium, 2013; Volume 6, ISBN 9789-2921-3397-9.
- 16. Winkler, B.; Lemke, S.; Lewandowski, I. IREPA-Biomass for Bioenergy Potential Assessment by a Holistic and Participatory Approach. In Proceedings of the European Biomass Conference and Exhibition Proceedings, Vienna, Austria, 4 June 2015; pp. 273–281. Available online: https://doi.org/10.5071/23rdEUBCE2015-1CV.3.21 (accessed on 17 November 2021).
- 17. BMWi. Gesetz zur Änderung des Erneuerbare-Energien-Gesetzes und Weiterer Energierechtlicher Vorschriften. 2021. Available online: https://www.bmwi.de/Redaktion/DE/Artikel/Service/gesetz-zur-aenderung-des-eeg-und-weiterer-energierechtlic her-vorschriften.html (accessed on 1 April 2021).
- 18. Umweltbundesamt. Waste Statistics. 2016. Available online: https://www.umweltbundesamt.de/en/topics/waste-resources/waste-management/waste-statistics (accessed on 1 April 2021).
- Thrän, D.; Adler, P.; Brosowski, A.; Fischer, E.; Herrmann, A.; Majer, S.; Oehmichen, K.; Pfeiffer, D.; Schmersahl, R.; Schröder, T.; et al. Method Handbook-Material Flow-Oriented Assessment of Greenhouse Gas Effects. 2015. Available online: https://www.energetische-biomassenutzung.de/fileadmin/media/6_Publikationen/04_MHB_en_web.pdf (accessed on 25 March 2021).
- 20. Köppel, W.; Erler, R.; Senner, J.; Kiefer, J. Einsatz Von Industriellen Und Kommunalen Abfallstoffen für die Biogaseinspeisung. 2014. Available online: https://www.dvgw.de/medien/dvgw/forschung/berichte/g1_08_10.pdf (accessed on 9 March 2021).
- 21. Brosowski, P.; Adler, G.; Erdmann, W.; Stinner, D.; Thrän, U.; Mantau, C.; Blanke, B.; Mahro, T.; Hering, G. Reinholdt, Biomassepotenziale von Rest- und Abfallstoffen- Status quo in Deutschland. 2015. Available online: https://mediathek.fnr.de/b and-36-biomassepotenziale-von-rest-und-abfallstoffen.html (accessed on 9 March 2021).
- Fehrenbach, H.; Giegrich, J.; Köppen, S.; Wern, B.; Pertagnol, J.; Baur, F.; Hünecke, K.; Dehoust, G.; Bulach, W.; Wiegmann, K. BioRest: Verfügbarkeit und Nutzungsoptionen biogener Abfall- und Reststoffe im Energiesystem (Strom-, Wärme- und Verkehrssektor).
 2019. Available online: https://www.umweltbundesamt.de/sites/default/files/medien/1410/publikationen/2019-09-24_texte_115-2019_biorest.pdf (accessed on 9 March 2021).
- Knappe, F.; Vogt, R.; Turk, T.; Hüttner, A.; Dehoust, G.; Schneider, T. Hochwertige Verwertung von Bioabfällen-Ein Leitfaden. 2015. Available online: https://pudi.lubw.de/detailseite/-/publication/94776 (accessed on 2 April 2021).
- Scholwin, F.; Grope, J.; Clinkscales, A.; Daniel-Gromke, J.; Rensberg, N.; Denysenko, V.; Stinner, W.; Richter, F.; Raussen, T.; Kern, M.; et al. Thomas and Reinhold, Aktuelle Entwicklung und Perspektiven der Biogasproduktion aus Bioabfall und Gülle. 2019. Available online: https://www.umweltbundesamt.de/sites/default/files/medien/1410/publikationen/2019-04-15_te xte_41-2019_biogasproduktion.pdf (accessed on 11 March 2021).
- 25. Mühlenhoff, J.; Dannemann, B. Biogene Rest- und Abfallstoffe-Flexibler Baustein der Energiewende. *Renews Spez.* 2017, 81, 18.
- 26. Knappe, F.; Vogt, R.; Lazar, S.; Höke, S. Optimierung der Verwertung Organischer Abfälle. 2012. Available online: https://www.umweltbundesamt.de/sites/default/files/medien/461/publikationen/4310.pdf (accessed on 12 March 2021).

- 27. Zeller, V.; Gröngröft, A.; Kirsten, C.; Weller, N.; Schenker, M.; Wedwitschka, H.; Wagner, B.; Deumelandt, P.; Reinicke, F.; Vetter, A.; et al. Basisinformationen Für Eine Nachhaltige NUTZUNG von Landwirtschaftlichen Reststoffen zur Bioenergiebereitstellung. 2012. Available online: https://www.energetische-biomassenutzung.de/fileadmin/user_upload/Downloads/Ver %C3%B6ffentlichungen/02_Basisinformationen_Reststoffe_web.pdf (accessed on 12 March 2021).
- 28. BMU. Abfallwirtschaft in Deutschland 2020-Fakten, Daten, Grafiken. 2020. Available online: https://www.bmu.de/fileadmin/D aten_BMU/Pools/Broschueren/abfallwirtschaft_2020_bf.pdf (accessed on 18 March 2021).
- 29. Destatis. Waste Balance 2018. 2020. Available online: https://www.destatis.de/EN/Themes/Society-Environment/Environment/Waste-Management/Tables/liste-brief-overview-waste-balance.html (accessed on 10 March 2021).
- 30. BMBF. Eine Kleine Wasserstoff-Farbenlehre. 2020. Available online: https://bit.ly/3mJz07R (accessed on 12 April 2021).
- 31. Pandey, A.; Chang, J.-S.; Hallenbeck, P.C.; Larroche, C. (Eds.) *Biohydrogen-Biomass, Biofuels, Biochemicals*, 2nd ed.; Elsevier: Amsterdam, The Netherlands, 2013; ISBN 9780-4446-4203-5.
- 32. Poullikkas, A.; Al-Assaf, Y.; Kalamaras, C.M.; Efstathiou, A.M. Hydrogen production technologies: Current state and future developments. *Conf. Pap. Energy* 2013, 690627, 1–9. [CrossRef]
- 33. Wang, J.; Yin, Y. Biohydrogen Production from Organic Wastes; Springer: Berlin/Heidelberg, Germany, 2017; ISBN 9789-8110-4675-9.
- 34. Dincer, I.; Acar, C. Review and evaluation of hydrogen production methods for better sustainability. *Int. J. Hydrog. Energy* **2015**, 40, 11094–11111. [CrossRef]
- 35. Kapdan, I.K.; Kargi, F. Bio-hydrogen production from waste materials. Enzym. Microb. Technol. 2006, 38, 569–582. [CrossRef]
- 36. Singh, A.; Rathore, D. (Eds.) *Biohydrogen Production: Sustainability of Current Technology and Future Perspective*, 1st ed.; Springer: Berlin/Heidelberg, Germany, 2017; ISBN 9788-1322-3577-4.
- Ghosh, R.; Grammel, H.; Sauer, A. Neue Strategie zur Wasserstoff-Produktion aus Frucht- und Molkereiabfällen mit Hilfe von Purpurbakterien. Available online: https://www.energetische-biomassenutzung.de/projekte-partner/details/project/show/ Project/rhotech-637/ (accessed on 3 April 2021).
- Zeiger, L.; Grammel, H. Model-based high cell density cultivation of Rhodospirillum rubrum under respiratory dark conditions. *Biotechnol. Bioeng.* 2010, 105, 729–739. [CrossRef]
- 39. Bui, M.; Adjiman, C.S.; Bardow, A.; Anthony, E.J.; Boston, A.; Brown, S.; Fennell, P.S.; Fuss, S.; Galindo, A.; Hackett, L.A.; et al. Carbon capture and storage (CCS): The way forward. *Energy Environ. Sci.* **2018**, *11*, 1062–1176. [CrossRef]
- 40. Full, J.; Miehe, R.; Kiemel, S.; Bauernhansl, T.; Sauer, A. The Biological Transformation of Energy Supply and Storage–Technologies and Scenarios for Biointelligent Value Creation. *Procedia Manuf.* **2019**, *39*, 1204–1214. [CrossRef]
- 41. Miehe, R.; Bauernhansl, T.; Beckett, M.; Brecher, C.; Demmer, A.; Drossel, W.-G.; Elfert, P.; Full, J.; Hellmich, A.; Hinxlage, J.; et al. The biological transformation of industrial manufacturing–Technologies, status and scenarios for a sustainable future of the German manufacturing industry. *J. Manuf. Syst.* **2020**, *54*, 50–61. [CrossRef]
- 42. Gaida, B.; Schüttmann, I.; Zorn, H.; Mahro, B. Bestandsaufnahme zum biogenen Reststoffpotential der deutschen Lebensmittel und Biotechnik-Industrie. 2013. Available online: https://www.fnr-server.de/ftp/pdf/berichte/22001910.pdf (accessed on 9 March 2021).
- 43. Kopf, K.A.; Bisig, W.; Jungbluth, N.; Denkel, C. Quantitatives Potenzial zur Verwertung von Molke in Lebensmitteln in der Schweiz. *Agrar. Schweiz* **2015**, *6*, 270–277. [CrossRef]
- 44. Aschmann, V.; Effenberger, M.; Gronauer, A.; Kaiser, F.; Kissel, R.; Mitterleitner, H.; Neser, S.; Schlattmann, M.; Speckmaier, M.; Ziehfreund, G. Biogashandbuch Bayern-Grundlagen und Technik. 2007. Available online: https://www.lfu.bayern.de/energie/biogashandbuch/index.htm (accessed on 15 March 2021).
- 45. Pospiech, J.; Ullrich, M.; Göttling, S.; Truyen, U.; Speck, S. Hygienisierung von Wirtschaftsdünger und Gärresten. 2014. Available online: https://nbn-resolving.org/urn:nbn:de:bsz:14-qucosa-161847 (accessed on 17 November 2021).
- Vavouraki, A.I.; Angelis, E.M.; Kornaros, M. Optimization of thermo-chemical hydrolysis of kitchen wastes. *Waste Manag.* 2013, 33, 740–745. [CrossRef]
- 47. Patni, N.; Jui, P.Y. Volatile fatty acids in stored dairy-cattle slurry. Agric. Wastes 1985, 13, 159–178. [CrossRef]
- 48. Panesar, P.S.; Kennedy, J.F.; Gandhi, D.N.; Bunko, K. Bioutilisation of whey for lactic acid production. *Food Chem.* **2007**, 105, 1–14. [CrossRef]
- 49. Fachagentur Nachwachsende Rohstoffe e. V. (n.d.). Faustzahlen. Available online: https://biogas.fnr.de/daten-und-fakten/fau stzahlen (accessed on 18 March 2021).
- 50. Bakonyi, P.; Nemestóthy, N.; Bélafi-Bakó, K. Biohydrogen purification by membranes: An overview on the operational conditions affecting the performance of non-porous, polymeric and ionic liquid based gas separation membranes. *Int. J. Hydrog. Energy* **2013**, *68*, 9673–9687. [CrossRef]
- 51. Ramírez-Morales, J.E.; Tapia-Venegas, E.; Toledo-Alarcón, J.; Ruiz-Filippi, G. Simultaneous production and separation of biohydrogen in mixed culture systems by continuous dark fermentation. *Water Sci. Technol.* **2015**, *71*, 1271–1285. [CrossRef]
- 52. Staffell, I. Zero carbon infinite COP heat from fuel cell CHP. *Appl. Energy* **2015**, *147*, 373–385. [CrossRef]
- 53. Fajardy, M.; Koeberle, A.; MacDowell, N.; Fantuzzi, A. BECCS deployment: A reality check. *Grantham Inst. Brief. Pap.* **2019**, *28*, 1–13.
- 54. DEHSt. Leitfaden zur Erstellung von Überwachungsplänen für Stationäre Anlagen-4. Handelsperiode (2021–2030). 2020. Available online: https://bit.ly/31nG9Rz (accessed on 26 March 2021).

56. Umweltbundesamt. CO₂ Emissions Per Kilowatt Hour of Electricity in Further Decline in 2019. 2020. Available online: https://www.umweltbundesamt.de/presse/pressemitteilungen/bilanz-2019-co2-emissionen-pro-kilowattstunde-strom (accessed on 25 April 2021).

^{57.} Pehnt, M. *Energieeffizienz;* Springer: Berlin/Heidelberg, Germany, 2010; ISBN 9783-6421-4250-5.

Appendix E: Paper E



Available online at www.sciencedirect.com



journal homepage: www.elsevier.com/locate/he



Carbon-negative hydrogen production (HyBECCS): An exemplary techno-economic and environmental assessment



Johannes Full ^{a,b,*}, Marcel Geller ^a, Sonja Ziehn ^a, Tobias Schließ ^a, Robert Miehe ^{a,c}, Alexander Sauer ^{a,b}

^a Fraunhofer Institute for Manufacturing Engineering and Automation IPA, 70569 Stuttgart, Germany

^b Institute for Energy Efficiency in Production (EEP), University of Stuttgart, 70569 Stuttgart, Germany

^c Biointelligence Competence Center, Nobelstr. 12, 70569 Stuttgart, Germany

HIGHLIGHTS

• A techno-economic and ecological assessment for an exemplary HyBECCS approach is conducted.

• Effects resulting from the greenhouse gas (GHG) balance are considered in the assessment.

• Compared to other negative emission technologies, the approach shows high GHG removal efficiency.

• Compared to other hydrogen production processes, it shows environmental-economic advantages.

ARTICLE INFO

Article history: Received 16 April 2023 Received in revised form 20 September 2023 Accepted 22 September 2023 Available online 8 October 2023

Keywords: Biohydrogen BECCS CCS Negative emission technologies Carbon dioxide removal Life cycle assessment

ABSTRACT

An exemplary techno-economic and environmental assessment of carbon-negative hydrogen (H₂) production is carried out in this work. It is based on the so-called "dark photosynthesis" with carbon dioxide (CO₂) capture and geological storage. As a special feature of the assessment, the economic consequences due to the impact on the global climate are taken into account. The results indicate that the example project would be capable of generating negative GHG emissions under the assumptions made. The amount is estimated to be 17.72 kgCO₂ to be removed from the atmosphere per kilogram of H₂ produced. The levelized costs of carbon-negative hydrogen are obtained, considering the economic impact of greenhouse gas emissions and removals. They are estimated to be 0.013 EUR/kWhH₂. Compared to grey hydrogen from natural gas (0.12 EUR/kWhH₂) and green hydrogen from electrolysis using renewable electricity (0.18 EUR/kWhH₂), this shows a potential environmental-economic advantage of the considered example. Even without internalization of GHG impacts, an economic advantage of the project (0.12 EUR/kWhH₂) over green hydrogen (0.17 EUR/kWhH₂) is indicated. Compared to other NETs, the GHG removal efficiency is at the lower end of both BECCS and DACCS approaches.

© 2023 The Author(s). Published by Elsevier Ltd on behalf of Hydrogen Energy Publications LLC. This is an open access article under the CC BY license (http://creativecommons.org/ licenses/by/4.0/).

https://doi.org/10.1016/j.ijhydene.2023.09.252

^{*} Corresponding author. Fraunhofer Institute for Manufacturing Engineering and Automation IPA, 70569 Stuttgart, Germany.

E-mail addresses: johannes.full@ipa.fraunhofer.de (J. Full), marcel.geller@ipa.fraunhofer.de (M. Geller), sonja.ziehn@ipa.fraunhofer. de (S. Ziehn), tobias.schliess@ipa.fraunhofer.de (T. Schließ), robert.miehe@ipa.fraunhofer.de (R. Miehe), alexander.sauer@ipa. fraunhofer.de (A. Sauer).

^{0360-3199/© 2023} The Author(s). Published by Elsevier Ltd on behalf of Hydrogen Energy Publications LLC. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

1. Introduction

To limit global warming below 2 °C, a combination of two climate change mitigation measures is required. First, a fast and deep reduction of anthropogenic greenhouse gas (GHG) emissions is essential. Within GHG reduction, substituting fossil fuels with sustainably produced bioenergy can play an important role. For 2050, bioenergy with a total of 465 TWh per year is predicted to contribute with 23% to the total energy demand of the Federal Republic of Germany [1]. Thereof, 279 TWh can be obtained from biogenic residual and waste materials [1]. Second, in addition to GHG reductions, a fast deployment of negative emission technologies (NETs) is required to enable removal of carbon dioxide (CO₂) from the atmosphere [2,3]. The permanent storage or long-term use of biogenic carbon dioxide in so-called BECCS (Bioenergy with Carbon Capture and Storage) approaches can lead to a reduction of greenhouse gas concentration in the atmosphere [4,5]. This permanent net removal is labeled negative emissions [6]. BECCS can play a major role in the area of NETs. Therefore, carbon dioxide captured during the process must be of biogenic and renewable origin, i.e. derived from fastgrowing renewable biomass, biogenic waste or residues [7,8]. In Europe, the potential for negative emissions through BECCS is estimated to be 200 MtCO2 per year [9]. In Germany, the potential for negative emissions through BECCS is estimated to be about 62 MtCO₂ per year [10]. The Federal Climate Change Act stipulates that GHG emissions in Germany must be reduced by 88% compared to 1990 until 2040 and GHG neutrality must be achieved in 2045. This means that from 2040 to 2045, approximately 130 MtCO2eq of remaining GHG emissions would need to be reduced or offset annually [11]. With a potential of approx. 62 MtCO₂ per year, BECCS could contribute almost the half of this amount through the generation of negative emissions [10].

HyBECCS (Hydrogen Bioenergy with Carbon Capture and Storage) is a BECCS approach based on the production of biogenic hydrogen or biohydrogen. The concept combines biohydrogen production from biomass with capture and storage of carbon dioxide, aiming to produce carbon-negative hydrogen (CNH) [12]. An important basis for the HyBECCS approach is the generation of biogenic carbon dioxide as a byproduct in most biohydrogen processes. Each HyBECCS implementation project (short: HyBECCS project) consists of four modular basic process steps: 1. biomass selection and pretreatment, 2. biohydrogen production, 3. gas processing, and 4. H_2 utilization and CO_2 storage [13]. The selection of biomass for the process has a major impact on the economic efficiency and environmental impact of the processes. It is advisable to use residual and waste materials [12]. Different technology options can be applied for the four basic process steps. The variety of technology options contributes to the capability of using a wide range of residual and waste materials. A general technological advantage of HyBECCS over other NETs is e.g. the energy-efficient capture of carbon dioxide at a point source compared to Direct Air Capture (DAC) approaches [13]. In addition, the relative carbon dioxide removal potential of HyBECCS is higher compared to other BECCS technologies that produce hydrocarbon-based energy

carriers. For example, in biogas or biomethane production with carbon capture and storage (CCS), a stoichiometric fraction of the carbon from the biomass remains in the methane (CH₄) energy carrier. This carbon is usually distributed decentrally in the energy system (e.g., as fuels for transport applications or for building heating) and emitted as carbon dioxide during the biogas or biomethane combustion [13]. In HyBECCS projects, the carbon-free energy carrier biohydrogen is produced and carbon dioxide is already emitted centrally as a point source at the plant, where it is captured energy efficiently and stored. Generally, HyBECCS projects can be designed from the ground up or existing plants, e.g. biogas plants, can be upgraded to HyBECCS plants [14]. The HyBECCS production process with a negative carbon footprint and the ability to substitute fossil energy carriers with biohydrogen result in contributing to both necessary climate change mitigation measures: the reduction of GHG emissions and carbon dioxide removal.

For HyBECCS technologies, techno-economic assessment fundamentals and guidelines have been elaborated by Full et al. (2022) [15]. The abstractions, standardizations, uniform system boundaries and key performance indicators (KPIs) can serve as an important basis for their comparison with other hydrogen production technologies or NETs. In addition to classical techno-economic evaluation approaches, the economic consequences due to the impact on the global climate are taken into account. In this paper, a techno-economic and environmental analysis of an exemplary HyBECCS project is carried out according to these guidelines. The objective of the work is to validate the suitability of the methodology and KPIs developed in Ref. [15].

2. Description of the evaluated example project

The example project assessed in this work is based on the production of biohydrogen via so-called "dark photosynthesis" process. This specific biohydrogen production process has been established by Ghosh and Autenrieth at the University of Stuttgart, Germany [16] and is introduced briefly in the following. The "dark photosynthesis" is a biotechnological process that is based on the purple bacterium Rhodospirillum rubrum (R. rubrum) to produce biohydrogen from different biomass sources containing fructose or lactate. For example, waste residues from the milk or fruit processing industry can be used. For the evaluation in this paper, fruit waste and residues are assumed as a substrate for the biological process. The material selection meets the criteria for appropriate residual and waste streams as it can be assigned to the category of "commercial food waste" according to Ref. [12]. The uniqueness of the process stems from the ability of the bacteria to induce - in the absence of light - the "photosynthetic" growth regime, during which all components of the photosynthetic membranes (i.e. the photosynthetic units consisting of protein subunits, bacteriochlorophylls and carotenoids), as well as components for electron transfer (i.e. quinones, ferredoxins), and the enzymes for biohydrogen production, are expressed at photosynthetic levels [17]. The "light"-signal during microaerophilic growth in the dark in M2SF medium is the establishment of a negative redox state of the quinone pool in the membranes as a result of the breakdown of fructose via the Embden-Meyerhof-Parnas pathway [18,19]. The independency from light facilitates an easier scale-up to industrial size using basic steel fermentation technology instead of photo-bioreactors. Fig. 1 represents a simplified overview of the "dark photosynthesis" process. The gaseous products of this process are biohydrogen and biogenic carbon dioxide, which constitutes a prerequisite for being extended to a HyBECCS project by capturing and storing the carbon dioxide [13,20].

For carbon storage (CS), the approach of geological underground storage in saline formations is selected for the example project under evaluation. This appears a suitable choice, as the amount of carbon dioxide that is injected into saline formations can be measured, and the storage can be considered permanent. Risks like short storage periods before releasing the carbon dioxide back into the atmosphere, reversals, insecurity concerning the time horizon of storage, and challenges such as monitoring the permanence due to the dispersed places of use, as they are often associated with carbon utilization (CU) approaches, can be kept as low as possible with the selected process [21]. Saline formations that are CS-suitable are porous and permeable salt water-filled rock formations that are sealed by a clay-rich cap rock. Together, the saline formations and the cap rock form an upward closed structure that keeps the carbon dioxide safely

in place. Thus preventing it from reentering the atmosphere. Storage periods of over 1000 years are possible and the actual amount of stored carbon dioxide can be monitored by continuous pressure and regular seismic measurements [22]. Several commercial and demonstration plants are already operating successfully and technological maturity is given [23–26].

Fig. 2 illustrates the overall process chain of the HyBECCS example project evaluated within the scope of this work. The production system is divided into the system elements "HyBECC" (Hydrogen Bioenergy with Carbon Capture) with the associated subsystem boundary B1, and "CS" (Carbon Storage) with the associated subsystem boundary B2, according to Ref. [15]. HyBECC includes the production of biohydrogen and the capturing of biogenic carbon dioxide from biomass through "dark photosynthesis". CS includes the transport of carbon dioxide and its geological storage. The cultivation, harvesting, transportation, and processing of fruits are not considered within system boundary B3 as they are part of fruit juice production. Within the system element HyBECC, fruit waste or residues must be grinded and hygienized before being used as a substrate for "dark photosynthesis". The inoculum (R. rubrum) and processed substrate are then fed into the bioreactor where biohydrogen and biogenic carbon dioxide are produced under micro-aerobic conditions. After that, the product gas is separated and the relevant product fractions (H₂, CO₂) are purified and compressed by a gas



Fig. 1 - General overview of the "dark photosynthesis" process (own representation).



Fig. 2 — Illustration of the exemplary HyBECCS project, divided into the subystem boundaries B1 (system element HyBECC), B2 (system element CS), and overall system boundary B3. Illustration of CS process based on [22].

processing system. Digestates are processed and separated into liquid and solid components. They can be used as further valuable goods such as fertilizers, which however, is not part of the present assessment. The biogenic carbon dioxide is transported over an assumed average distance of 1500 km to the CS plant, first via trucks to the North Sea 500 km and then by ship another 1000 km to the CS plant site. There, it is injected into saline formations located about 1 km below sea level [22].

3. Assumptions and framework conditions

The example project considered in this paper is a compilation of sub-processes, each of which has been implemented at different locations and different scales, combining to a fictitious HyBECCS project. All assumptions regarding these subprocesses are explained in the following and are summarized in Appendix A, Table A1. Fig. 3 illustrates the plant diagram of biohydrogen and biogenic carbon dioxide production (HyBECC facility) from fruit waste or residues based on the described "dark photosynthesis" process in section 2.

It is assumed to be located close to the fruit processing facility generating the waste or residue used as substrate. Therefore, no costs and GHG emissions are assumed to occur for transporting the substrate to the HyBECC facility. The gross and net working volumes of the bioreactor are assumed to be 7200 m³ and 6000 m³, respectively. The net working volume refers to the maximum filling volume of the bioreactor with the substrate. An additional volume of about 20% is considered for the product gas above the substrate, summing up to the gross working volume. The net working volume is filled 80% with substrate and 20% with process water. Based

on averages for biogas plants in Germany, annual full load hours for "dark photosynthesis" are assumed to be 8000 h/a, resulting in an annual plant availability of 91% [27]. Also, based on average biogas plants in Germany, the lifetime of the HyBECC facility, showing similar plant configurations as biogas plants, is conservatively assumed to be 20 years [27,28]. Based on unpublished data from experiments, the gross biohydrogen yield and the gross amount of biogenic carbon dioxide are assumed to be approximately 896 and 19,521 tons per year, respectively. These assumptions are based on the usage of waste or residues from fruit juice production with a fructose content of 6% as feedstock and considering the plant availability of 91%. It results in a plant capacity of approximately 3409 kW_{el}.

The gas processing system to separate, purify and compress the product gas includes a low-pressure gas tank, a carbon dioxide compressor, an activated carbon filter, an absorbing type desiccator, a compression refrigerator, and a storage tank. Since "dark photosynthesis" and dark fermentation processes are similar in terms of the produced product gas, it is assumed that the gross biohydrogen yield is reduced by 8% due to losses in the gas processing system [29]. The product gas reaches a purity level of 99% hydrogen, as the practice of dark fermentation processes shows [29,30]. It is assumed to be compressed to 700 bar. Further losses or leakages for hydrogen are not taken into account.

The amount of biogenic carbon dioxide produced within system boundary B1 represents the theoretical maximum potential for the generation of negative emissions, also referred to as negative emission potential (NEP) [12]. Losses and leakages are considered throughout the process chain. During the carbon dioxide capture process using pressure swing adsorption, 10% of losses are assumed due to a limited



Fig. 3 - Plant diagram of the HyBECC system element (own representation).

capture rate of 90% [31,32]. Transportation by truck and ship is assumed to cause boil-off losses of carbon dioxide from the storage vessel into the atmosphere within a range of 3–4% per 1000 km [22]. Estimated conservatively, this accounts for another 6% of leakage over the assumed distance of 1500 km from the HyBECC facility to the CS plant site. Summing up, a total deduction of 15.4% and is assumed. Thus a NEP of about 16,515 tons of carbon dioxide per year can be assumed.

As no project-specific data on the purchase price for fruit waste or residues and the disposal of its digestates is available, it is assumed that the costs associated with the disposal of digestates are equal to the profit realized from the acceptance of fruit waste or residues, or vice versa. The bulk density of the substrate is assumed to be 0.35 t/m³ [33]. Based on the reactor volume, plant availability, and feedstock bulk density, 186,667 tons of waste or residues are assumed to be processed annually. The power demand of the HyBECC plant is set to account for 7.25% of the total amount of generated electricity, based on typical values for biogas plants [27]. This covers the electricity needed, including substrate pre-treatment (grinding, pumping, measurement, and control technology) and product gas processing. Electricity is received from the German grid at an average price for the industry of 0.176 EUR/ kWhel [34]. Heat is used for substrate hygienization. Its demand is estimated based on the amount of substrate and process water. Heating costs are assumed to be 0.09 EUR/ kWh_{th} [35]. District heating supply temperatures are sufficient to heat up the fruit waste or residues to the needed temperature level of approx. 70 °C [36]. Recovering heat from the fruit processing plant lowers the annual heat consumption,

assuming that the substrate can be preheated to 50 °C using waste heat. The amount of bacteria used as inoculum as well as the respective purchase and transport costs are neglected, since cultivation of the inoculum is carried out at the HyBECC facility by its operators.

Based on commercial CS plants, the CS plant lifetime of the example project is assumed to be 40 years, in which carbon dioxide can be injected consistently [26]. Capital expenditures of 11.33 EUR/tCO₂stored and operational expenditures of 14.90 EUR/tCO₂stored are assumed for the transport of biogenic carbon dioxide from the HyBECC facility to the CS plant site, its storage and monitoring [26, 28 + 34]. Market prices for hydrogen of 9.50 EUR/kgH₂ and for biogenic carbon dioxide of 19.30 EUR/tCO₂ are assumed [37,38]. Prices for negative emissions and GHG internalization are set at 195 EUR/tCO₂eq [39].

4. Methodology

The methodological approach for techno-economic and environmental assessment applied in this study can be subdivided into eight basic steps, as summarized in Fig. 4 and briefly explained in the following.

First, the theoretical ability for carbon dioxide removal and the permanence of carbon dioxide storage is checked. This is obtained according to the guidelines for eligibility check on technology level as described by Full et al. (2022) [15]. Second, GHG emission flows are calculated following ISO 14040/44 [40,41]. It focuses on the impact category of global warming potential over 100 years (GWP₁₀₀) and thus aims at quantifying



the GHG footprint of the example project and its subsystems. A "cradle-to-grave" system boundary is applied. The functional unit is chosen depending on the product under consideration. It is defined as one ton of CO2eq permanently (>1000a) removed or one kWh (net calorific value) of generated biohydrogen. Carbon removal certificates (CRC) are assumed to be paid for negative emissions, thus for the net-amount of carbon dioxide permanently removed from the atmosphere. Here, further risk deductions for long-term carbon storage leakages of 1% are considered [22]. The emissions for the construction of the infrastructure required are drawn from the ecoinvent database for the construction of a factory for synthetic gas with facilities for gas production and processing. The dataset includes materials for construction and operation of the building machines, as well as waste materials, such as waste concrete and reinforced steel. The output capacity of this factory of 5 MW is scaled down to the output capacity of the biohydrogen production plant of 3.409 MW by factor 1.47 and referenced from the factory lifetime of 50 years to the evaluations' scope of one year. The substrates assumed to be used in the HyBECC facility are waste or residue streams from fruit processing, i.e., a waste stream from the upstream product system of fruit juice production. Therefore, they enter the system without a CO₂eq burden [28,41]. If the amount of permanently stored biogenic carbon dioxide is higher than the GHG emissions caused by the example project, the eligibility on the project level is confirmed [15]. Costs for GHG process emissions and revenues for negative emissions are estimated based on the GHG balance applying the simplified basic model for regulatory systems in net-zero emission economies [15]. External costs and benefits on the climate are assumed to be entirely internalized.

Factorial cost estimation is applied for the calculation of capital expenditures (CapEx). Operational expenditures

(OpEx) are calculated based on plant dimensions, experimental data, experience values from biogas plants in Germany and surcharge factors [27,29]. Both internal and internalized costs and revenues are included in the calculation of three central key performance indicators (KPIs): the levelized costs of carbon negative hydrogen (LCCNH), the levelized costs of stored carbon dioxide (LCSC) and the levelized costs of negative emissions (LCNE). Those KPIs allow the comparison of different HyBECCS approaches with other hydrogen production technologies and NETs concerning their economic competitiveness and environmental performance in terms of GHG emissions.

The LCCNH (in EUR/kWhH₂) constitute the costs incurred to produce 1 kWh (net calorific value) of CNH in a climateneutralized way. They are determined according to Equation (1) by summing up all costs (C3) incurred within system boundary B3, the estimated internalization costs (C_{int}) of all process GHG emissions, and the assumed revenues (R2) from the internalization of the external benefits of generating negative emissions, divided by the hydrogen production in kWh (yH₂). C_{int} can be divided into $C_{int,E1}$ and $C_{int,E2b}$, referring to the internalized process emissions occurring within HyBECCS system boundary B1 and B2, respectively. R2 are obtained by internalizing the external benefits of negative emissions, for example through carbon removal certificates. The LCSC (in EUR/tCO₂stored) are defined as the costs that occur to permanently store one ton of carbon dioxide using a carbon-neutralized process. They are determined according to Equation (2) by the sum of C3, C_{int} , revenues (R1) obtained from the sales of hydrogen, divided by the absolute value of stored carbon dioxide E2a. The LCNE (in EUR/tCO2removed) are the costs that occur to achieve one ton (CO₂eq) negative emissions, i.e., to permanently remove one ton of carbon dioxide from the atmosphere, in a carbon-neutralized way. As shown in Equation (3) they are calculated by summing up the costs of C3 and C_{int} and deducting the revenues obtained from the sales of hydrogen (R1), divided by the absolute value of negative emissions E3, which is the absolute value of E2a minus process GHG emissions within system boundaries B1 (E1) and B2 (E2b). Capital expenditures, cash flows of revenues and expenses over the plant lifetime as well as the product yields for hydrogen, biogenic carbon dioxide and negative emissions (yH₂, E2a, E3) in Equations (1)–(3) are all discounted to a common reference point in time for the calculation of the levelized costs, i.e., annualized values.

$$LCCNH = (C3 + C_{int,E1} + C_{int,E2b} - R2) / yH_2$$
(1)

 $LCSC = (C3 + C_{int,E1} + C_{int,E2b} - R1) / |E2a|$ (2)

 $LCNE = (C3 + C_{int,E1} + C_{int,E2b} - R1) / |E3|$ (3)

Only in case of negative emissions (E3<0), the hydrogen qualifies as carbon negative and its LCOE can be referred to as LCCNH. Also, only in the case of negative emissions, the term LCNE can be used. According to Equation (4), this prerequisite can be examined by determining the emission factor (EF), which has to be smaller than one. The GHG emissions of HyBECCS can be divided, according to the system boundaries, into process GHG emissions (E1) emitted within system boundary B1, process GHG emissions (E2b) emitted during the carbon storage or carbon utilization process, and the amount of CO₂stored in the long term (E2a).

$$EF = (E1 + E2b) / |E2a|$$
 (4)

To investigate the impacts on the main KPIs LCCNH and LCNE, they are subjected to a sensitivity analysis. This analysis is conductetd in order to determine how uncertainties and risks affect economic and environmental performance factors. Therefore, the following parameters are varied in an interval of \pm 50% of their value and examined for their influences on the KPIs: H₂/CO₂ yield, GHG internalization costs/ revenues, process emissions of HyBECCS, total CapEx, total OpEx, process leakages, and (in the case of LCNE) the hydrogen sales price. Concluding the evaluations, the example process is compared with competing hydrogen production and negative emission technologies to evaluate its economic and environmental competitiveness.

5. Results

5.1. Environmental assessment

5.1.1. Eligibility on technology level

The first assessment step evaluates the eligibility on a technology level given under two conditions: (a) The theoretical ability for carbon dioxide removal and (b) the theoretical permanence of carbon dioxide storage [15]. For the HyBECCS project under evaluation, waste or residues from the processing of fruit, a renewable, fast-growing biomass, are used as substrate. Therefore, the carbon dioxide in the product gas is considered to be of biogenic origin and the theoretical ability for carbon dioxide removal is given. The theoretical permanance of carbon dioxide storage is given due to the technology choice made for CS. The selected carbon dioxide storage in saline formations can be considered to be permanent for over 1000 years in the case of a careful site selection [22]. Thus, the projects' eligibility on a technology level is given.

5.1.2. Carbon footprint

The process scheme for the calculation of the global warming potential over 100 years (GWP_{100}) in tCO₂eq per functional unit (carbon footprint) is depicted in Fig. 5. It shows the scope of the assessment. Following the "cradle-to-grave" approach, the system boundary of the LCA embraces all essential process steps described in section 2.

Table 1 summarizes the GHG emissions for the example project. The total carbon footprint for the example project is derived by subtracting the amount of CO_2 stored per functional unit from the process emissions, resulting in a negative carbon footprint of -0.53 kgCO₂eq/kWhH₂. The main share of 74.72% in the total GHG emissions of the example project (system boundary B3) is due to the provision of heat for hygienization. Electricity from the German grid accounts for 15.66%, while emissions from the CS process chain contribute another 9.38%. The shares for the construction and carbon dioxide transport remain at 0.24% and 0.01%, respectively.

The GHG emissions resulting from CS are drawn partially from an LCA conducted by Gassnova. Process emissions along the CS process chain, including storage, maintenance and monitoring of injected CO₂, as well as infrastructure construction emissions are assumed to be 10.81 kg CO₂eq/tCO₂stored, according to their work [24]. As GHG emissions from carbon dioxide capture are not included in that value, they are taken into account within the evaluation of system boundary B1 and its corresponding emission flow E1. GHG emissions from carbon dioxide transportation to the storage site are associated with system boundary B2 (cf. Fig. 2) and are calculated at 0.009 kgCO₂eq/tCO₂stored. The GHG emissions generated from the construction and operation of the CS plant drawn from Ref. [24], together with the carbon footprint for carbon dioxide transportation, sum up to a total carbon footprint of 10.82 kgCO₂eq/tCO₂stored.

5.1.3. Eligibility on project level

Even though the example project has the theoretical ability for carbon removal on a technology level (cf. 5.1.1.), this does not mean that it can actually deliver negative emissions. The actual ability on a project level is estimated ex-ante considering the carbon footprint of the process (cf. 5.1.2) and has to be monitored during its operation [15]. As the example project does not exist in reality, the ex-post evaluation of the eligibility on the project level is not part of this work. According to equation (4), the EF expresses the ratio between the GHG process emissions and the amount of stored carbon dioxide in $tCO_2eq/tCO_2stored$. For the example project, the EF is calculated at 0.12 $tCO_2eq/tCO_2stored$. This means that for each ton of biogenic carbon dioxide permanently stored, 0.12 tCO_2eq



Fig. 5 - Process scheme for life cycle assessment (own representation).

5.2.

are emitted throughout the whole life cycle of the HyBECCS project, including the process emissions from CS. In reverse, this means that 0.88 tCO₂ are actively removed from the atmosphere for each ton of biogenic carbon dioxide permanently stored. Considering biohydrogen production, 20.03 kg of carbon dioxide are removed for each kg (0.60 kgCO₂ for each kWhH₂) produced in the example project. As the EF is smaller than 1 (EF < 1), the analyzed example project produces CNH and qualifies as a NET on a project level, according to Ref. [15].

Techno-economic assessment

Capital expenditures (CapEx) 5.2.1.

Table 2 summarizes the composition of the CapEx. The total plant costs (TPC) for the HyBECC system element contain the total equipment costs, including the substrate management, bacteria management, bioreactor system, sterile compressed air system, digestate processing, and gas processing. They are estimated at 2,763,450 EUR. Direct and indirect costs (DIC)

Table 1 – GHG footprints of the respective system elements. (HyBECC = hydrogen bioenergy with carbon capture, CS = carbon storage, HyBECCS = hydrogen bioenergy with carbon capture and storage, own calculation according to ISO 14040/44).

System	Functional	Process	Carbon	Unit
Element (boundary)	Unit	Emissions	Footprint	
HyBECC (B1)	1 tCO ₂ stored	104.57	_	kgCO2eq/tCO2stored
CS (B2)	1 tCO ₂ stored	10.82	-	kgCO2eq/tCO2stored
HyBECCS (B3)	1 tCO stored	115 39	-884 62	kgCO_eq/tCO_stored
11,000 (00)	1 loo2stored	115.55	001.02	ngoo jeq/ too jotoreu
HyBECC (B1)	1 kWhH ₂	0.0628	_	kgCO ₂ eq/kWhH ₂
HyBECC (B1) HyBECC (B2)	1 kWhH ₂ 1 kWhH ₂	0.0628 0.0065		kgCO ₂ eq/kWhH ₂ kgCO ₂ eq/kWhH ₂

601

Table 2 – Absolute values and percentage of the different CapEx components. (HyBECC = hydrogen bioenergy with carbon capture, CS = carbon storage, Own Calculation based on factorial cost estimation, all sources are listed in Appendix A, Table A1).

Cost position (system element)	Value	Unit	Percentage of total CapEx
Substrate management	1,175,200	EUR	12%
Bioreactor system	706,300	EUR	7%
Digestate processing	506,500	EUR	5%
Bacteria management	145,450	EUR	1%
Sterile compressed air	140,850	EUR	1%
system			
Gas processing system	89,150	EUR	1%
Sum total plant costs (HvBECC)	2,763,450	EUR	28%
Direct and indirect costs (HyBECC)	2,686,050	EUR	27%
CapEx Carbon Storage (CS)	4,422,000	EUR	45%
Sum total CapEx (HyBECCS)	9,871,500	EUR	100%

include costs for equipment installation, piping, engineering, supervision, and legal expenses. They are estimated at 2,686,050 EUR in dependence of the TPC (cf. assumptions made in section 3). The CapEx of the system element CS are estimated at 4,422,000 EUR, resulting into a total CapEx of the example project of 9,871,500 EUR.

The CapEx for the HyBECCS project can be split into 55% on the system element HyBECC and 45% on the system element CS. TPC and DIC of system element HyBECC account for 28% and 27% of the total CapEx respectively. The substrate management has the largest share of the TPC with 43%, and third largest share of total CapEx with 12%.

5.2.2. Operational expenditures (OpEx)

OpEx estimations for the example project are summarized in Table 3. The variable costs, estimated at 1,412,800 EUR per

Table 3 — Absolute values and percentage of annualized OpEx (CS = carbon storage). (Own Calculation based on plant dimensions, experimental data, experience values from biogas plants in Germany and surcharge factors, all sources are listed in Appendix A, Table A1).

Cost position	Value	Unit	Percentage of total OpEx
Electricity	101,700	EUR/a	5%
Heat	696,100	EUR/a	32%
Process water	307,000	EUR/a	14%
Activated carbon	46,100	EUR/a	2%
CS (transport, storage,	261,900	EUR/a	12%
monitoring)			
Sum variable costs	1,412,800	EUR/a	65%
Taxes and insurance	98,700	EUR/a	5%
Maintenance	197,400	EUR/a	9%
Personnel expenses	480,100	EUR/a	22%
Sum fixed costs	776,200	EUR/a	35%
Sum OpEx	2,189,000	EUR/a	100%

year, include expenses for electricity, heat, process water, activated carbon, as well as for the transportation, storage, and monitoring of biogenic CO₂. Fixed production costs comprise expenses for taxes and insurance, maintenance, and personnel costs. They are estimated at 776,200 EUR per year, thus resulting into total OpEx of 2,189,000 EUR per year.

Heat accounts for the largest share of OpEx with 32%, followed by personal expenses with 22%, process water with 14% and CS with 12%.

5.2.3. Levelized costs

Table 4 summarizes the levelized costs. They can be divided into the internal levelized costs and the key performance indicators LCCNH, LCNE and LCSC (cf. section 4). The three KPIs are derived according to equations (1)-(3) taking into account the external costs and benefits of GHG emissions and removals.

The internal LCOE for system boundary B3 are higher than for system boundary B1. This is due to the additional costs for CS infrastructure and operations as well as the revenues generated from selling the co-product biogenic carbon dioxide. These revenues lower the internal LCOE in B1. However, they do not influence the internal LCOE in B3, as they occur in the same magnitude as a cost position in B2 and are thus offset against the revenues generated in B1. Internal LCSC in B2 are 385.02 EUR/tCO₂ higher compared to LCNE in B3. This is due to the revenues obtained from the hydrogen sales as part of the internal LCNE calculation. Considering LCCNH, the internalization of external effects leads to a reduction of about 0.11 EUR/kWhH₂ compared to the internal LCOE in B3. The LCNE is around 25.69 EUR/tCO₂ higher compared to the internal LCNE in B3. LCSC are reduced by about 322.40 EUR/tCO₂ compared to internal LCSC (B2).

5.3. Sensitivity analysis

Fig. 6 illustrates the results of the sensitivity analysis for the KPIs LCCNH and LCNE, as described in section 4.

The GHG internalization costs and revenues, H_2/CO_2 yield and the total OpEx can be identified as the parameters with the greatest impact on the LCCNH, as shown in Fig. 6 (a). The H_2/CO_2 yields have the greatest effect on the LCCNH, when being reduced. A reduction of the yields causes the LCCNH to converge to infinity. Increasing total OpEx and decreasing GHG internalization costs and revenues also have a comparatively high impact driving up the LCCNH. The highest value

Table 4 — Levelized costs, divided into internal levelized costs and key performance indicators (KPIs).				
Internal levelized costs	Value	Unit		
Internal LCOE (B1) Internal LCOE (B3) Internal LCSC (B2) Internal LCNE (B3)	0.08 2.59 0.12 3.85 62.64 322.38	EUR/kWh EUR/kgH ₂ EUR/kWh EUR/kgH ₂ EUR/tCO ₂ stored EUR/tCO ₂ removed		
KPIs				
LCCNH LCNE LCSC	0.013 0.43 296.69 259.76	EUR/kWh EUR/kgH ₂ EUR/tCO ₂ removed EUR/tCO ₂ stored		







Fig. 6 - Sensitivity analysis of LCCNH (a) and LCNE (b) (own representation).

for LCCNH is calculated at 14.17 EURcent/kWh for reducing H₂/ CO₂ yield by 50%. The lowest value for LCCNH is calculated at -3.84 EURcent/kWh for increasing GHG internalization costs and revenues by 50%. Fig. 6 (b) shows the influences of the parameters on the LCNE. The graph shows that the hydrogen sales price has the greatest impact on the LCNE. While a 50% reduction would lead to a cost increase of 91% to -25.89 EUR/ tCO₂removed, a 50% increase would reduce the LCNE by 91% to -567.48 EUR/tCO₂removed. Further, the H₂/CO₂ yield and the total OpEx have a considerably high impact on the LCNE. Corresponding to the influence on the LCCNH, decreasing H₂/ CO₂ yields leads to a disproportionate increase in the LCNE, converging to infinity. The impact of process emissions is not included in Fig. 6 (b) as the values correspond to the curve for GHG internalization costs and revenues.

6. Discussion of the exemplary results

6.1. Comparison with competing hydrogen generation technologies

In the following, the GHG emissions of the example project are compared to competing hydrogen production technologies. For each technology, the GHG emissions that occur throughout the production processes are indicated in Fig. 7 without CCS or, if applicable, with CCS. As shown in section 5.1.2., the example project causes GHG emissions of 0.0628 kgCO₂eq/kWhH₂ in system boundary B1, which is within the same magnitude as the GHG emissions of green hydrogen from electrolysis using renewable energy (0.05 kgCO₂eq/kWhH₂). This value is considerably lower than the GHG emissions of hydrogen generated via electrolysis using the EU electricity mix (0.72 kgCO₂eq/kWhH₂), as well as steam methane reforming (SMR) from both biogas (0.23 kgCO₂eq/kWhH₂) [42,43].

The GHG balance of the example project, including carbon storage (B3) is negative and amounts to -0.53 kgCO₂eq/ kWhH₂.

Hydrogen production through biogas SMR can also be extended to HyBECCS systems [14]. If the biogenic carbon dioxide occurring in the biogas as well as reformed gas is permanently stored (see e.g. Ref. [43]), the GHG intensity of a HyBECCS project based on biogas SMR (-0.45 kgCO₂eq/ kWhH₂) is within the same magnitude as the example project considered in this work. Application of CCS in natural gas SMR processes could reduce the carbon footprint from 0.32 to 0.17 kgCO₂eq/kWhH₂.

Fig. 8 shows the LCCNH of the example project compared to the levelized costs of energy (LCOE) of other hydrogen generation technologies. The internal LCOE constitute the levelized costs without internalization of the external effects of the process due to GHG emissions and removals. The internal LCOE of biohydrogen from the example project (0.12 EUR/kWhH₂ or 4.00 EUR/kgH₂) are lower compared to green hydrogen from electrolysis using renewable electrical energy (0.17 EUR/kWhH₂ or 5.67 EUR/kgH₂). This indicates that the example project could already be competitive to green hydrogen production from electrolysis under the assumptions made. The second bar from the left shows the assumed internalization costs and revenues for GHG emissions and carbon dioxide removal of the example project (cf. 5.1.2.).

For the example project, the LCCNH is 0.013 EUR/kWhH₂ (0.43 EUR/kgH₂). The comparison of the LCCNH with the total LCOE (including GHG internalization costs) of the alternative technologies underlines the high ecological benefit of CNH. Including all positive and negative effects in the calculations, the fossil alternative grey hydrogen from natural gas (0.12 EUR/kWhH₂ or 4.00 EUR/kgH₂) loses its competitiveness [42,44]. A direct comparison to the conventional fossil energy carrier natural gas (not part of Fig. 8) comes to the same conclusion: The internal LCOE of 0.02 EUR/kWh would rise to 0.06 EUR/kWh when including the external costs of GHG emissions [42,44]. In comparison to hydrogen from electrolysis using renewable electricity, the positive impact of the negative emissions also provide a considerable advantage to the example project. Overcompensating its GHG emissions with carbon dioxide removals leads to a net income of 0.107

(a)



Fig. 7 – GHG emissions from biohydrogen generated via the exemplary HyBECCS project and from other hydrogen production technologies. Own elaboration including data from Refs. [42,43].

EUR/kWh (3.57 EUR/kgH₂), whereas green hydrogen from electrolysis has to account for 0.01 EUR/kWhH₂ (0.33 EUR/ kgH₂) of GHG internalization costs. As elaborated by Full et al. (2021), the production costs for CNH in 2030 would have to be below 4.30 per kg in a worst-case scenario to be competitive in Germany, taking into account hydrogen market forecasts [13]. All estimated levelized costs of the example project are below this value.

6.2. Comparison with competing negative emission technologies

Changing the point of view from biohydrogen to the negative emissions as main product of the example project, its comparison with other NETs is possible. The EF of the example project is estimated to be $0.12 \text{ tCO}_2\text{eq/tCO}_2\text{stored}$ (cf. 5.1.3.). According to Fajardy et al. (2019) an average emission factor of $0.5 \text{ tCO}_2\text{eq/tCO}_2\text{stored}$ for BECCS can be obtained [45].

Borchers et al. (2022) indicate an emission factor between 0.1 and 0.9 tCO₂eq/tCO₂stored for BECCS, depending on the technology chosen [10]. The example project is, thus, at the very low EF range of BECCS approaches, as shown in Fig. 9. Considering DACCS approaches, the emission factors vary significantly depending on the type of energy provision and their location. According to an analysis of Terlouw et al. (2021), the EF of DACCS lies between 0.03 tCO₂eq/tCO₂stored for DACCS plants in Norway using waste heat and grid electricity and 0.91 tCO₂eq/tCO₂stored for DACCS plants in Greece using high-temperature heat pumps and grid electricity [46]. Borchers et al. (2022) indicate an EF of 0.6 tCO₂eq/tCO₂stored for both decentralized and centralized DACCS options [10]. The EF of the example project is, thus, comparable to DACCS plants at the lower end of their EF range.

The internal LCNE for the example project are estimated at -322 EUR/tCO₂removed (cf. 5.2.3.). Thus, even without internalizing any externalities, the removal of one ton of



Fig. 8 – Comparison of the levelized costs of carbon-negative hydrogen (LCCNH) from the exemplary HyBECCS project with the LCOE of other hydrogen production technologies. B1: system boundary B1 (HyBECC system element), B2: system boundary B2 (CS system element). Own elaboration including data from Refs. [42,44].





carbon dioxide from the atmosphere generates an income of 322 EUR due to the sales of the co-product biohydrogen. Other BECCS approaches generally show a wide range of internal LCNE due to the variety of the underlying technologies. Möllersten et al. (2020) indicate approx. 19.40 to over 97 EUR/tCO2removed, whereas the IPCC (2018) indicates a minimum of approx. 194 EUR/tCO₂removed [23]. DACCS approaches also show a wide cost range. For DACCS, Möllersten et al. (2020) indicate internal LCNE of approx. 39-581 EUR/ tCO₂removed, whereas the IPCC (2018) indicate approx. 19.40–969 EUR/tCO2removed [23]. The LCNE of the example project is estimated at -297 EUR/tCO2removed. This states that removing one ton of carbon dioxide from the atmosphere generates an income of 297 EUR, even after internalizing all externalities from GHG emissions. When applying internalization, the LCNE cost range is approx. 39-367 EUR/ tCO₂eq for BECCS. For DACCS, the LCNE cost range is 25–1127 EUR/tCO₂eq. Fig. 10 summarizes the comparison of the LCNE

of the example project to the Internal LCNE cost range as well as to the LCNE cost ranges of BECCS and DACCS. It can be seen that the Internal LCNE as well as the LCNE of the example project is within the BECCS cost range.

7. Limitations

As the example project considered in this work is not fully implemented, the results rely on the assumptions made. However, all assumptions are associated with uncertainties. The sensitivity analysis (cf. 5.3.) revealed that the performance of the KPIs is especially susceptible to the parameters H_2/CO_2 yield, hydrogen sales price, and OpEx with heating costs and personnel expenses showing the highest impacts. These parameters should be given particular attention when verifying the assumptions made. However, some assumptions and their uncertainties should be noted additionally.



Internal LCNE range (without GHG internalization)

Fig. 10 – LCNE (with GHG internalization) from the exemplary HyBECCS project based on "dark photosynthesis" compared to the (Internal) LCNE range of BECCS and the (Internal) LCNE range of DACCS. Source: Own elaboration including data from Refs. [23,47].
An important point of discussion from the authors' perspective is the waste or residual material used as substrate material. On the one hand, the large quantity considered must be available. On the other hand, its processing may involve environmental impacts or costs, which have been neglected in this work. Direct damages to ecosystems or indirect damages through increased competition for land use, with substantial risks to biodiversity, can occur. In this work, a one-sided focus is placed on GHG emissions. However, all impact categories of LCAs, especially conflicts with food production and biodiversity must be analyzed. Further, considering the economic impacts of the substrate material used, revenues or costs can arise for their acceptance or disposal. Thus, the price for the substrate used can be a significant factor in terms of costs. In Germany, for instance, the waste receipt of sewage sludge can generate revenues of about 30 EUR/t, whereas the use of sugar beet leaf silage can lead to costs of around 28 EUR/t [48]. In this work, the substrate residues from "dark photosynthesis" are considered a product of equal value to the originating residues from fruit juice production. They can be used e.g. as fertilizer. However, it must be highlighted, that whether the substrate used is associated with costs or even revenues. This has to be determined carefully for each specific substrate used. Another limitation concerns the bulk density of the material. Fruit waste consists of materials of different compositions and densities that in turn depend on the moisture content of the substance. For apple pomace as an example, the bulk density ranges from 251 kg/m³ for dry matter to about 1017 kg/m³ for fresh matter [49]. Since the product yields are linked to the amount of the used substrate, a higher bulk density results in higher yields assuming constant fructose content. At the same time, the amount of substrate to be processed increases. Besides moisture, pre-treatment of biomass is an important aspect of biohydrogen production [50]. Pre-treatment may lead to the formation of toxins and inhibitors, affecting bacterial conversion of the substrate to biohydrogen. Therefore physical, chemical, physiochemical and biological pretreatment methods as well as subsequent detoxification have to be implemented carefully for optimal feedstock quality. The neglected feedstock price (cf. section 3) and the annually needed amount may be bottlenecks for HyBECCS projects and must be carefully determined for each HyBECCS project.

Additionally, it should be noted that the assumed GHG internalization price of 195 EUR/tCO₂eq does not comply with reality but represents an idealized scenario representing an appropriate GHG internalization price for the ecological damage from climate change [39]. It also has to be underlined that the exemplary "dark photosynthesis" process analyzed in this paper is one technological approach amongst many HyBECCS approaches. Therefore, the example project and the results of this analysis are not representative for HyBECCS technologies in general. They only apply to the technology combination considered under the assumptions made. Generally, biotechnological process routes for biohydrogen production like "dark photosynthesis", face major challenges in their industrial implementation and scale up due to slow conversion rates [51]. In this regard, "dark photosynthesis" is no exception, existing yet only at a laboratory scale. Measures for scale up, such as continuous production, are highly

encouraged and will pave the way for faster development [52,53].

8. Summary and conclusion

In this paper, a techno-economic and environmental analysis is performed for an exemplary HyBECCS project. The evaluation is based on the methodological assessment approach presented by Full et al. (2022) and conducted primarily to validate this approach [15]. The methodology allows to compare HyBECCS technologies with other hydrogen production technologies and NETs. Therefore, in addition to conventional economic criteria, the specific key performance indicators integrate macroeconomic effects resulting from the GHG emissions and reductions as a special feature. The example project chosen combines the production of biohydrogen and biogenic carbon dioxide using fructosecontaining waste or residues from fruit processing with storage of the carbon dioxide in saline formations.

First, an estimation of the carbon footprint is conducted. It indicates that the overall process chain of the example project is a carbon sink under the assumptions made. For every kg of biohydrogen produced, 17.72 kg of carbon dioxide (0.53 kgCO₂/ kWhH₂) could be removed from the atmosphere. With this information, the eligibility check on the project level is fulfilled and the example project qualifies as HyBECCS technology being capable of delivering carbon-negative hydrogen. For the techno-economic analysis, the capital and operational expenditures are estimated. Considering CapEx, the estimations result in a total of 9,871,500 EUR. 55% of these costs are related to the production of biohydrogen and carbon dioxide capturing. Carbon storage accounts for the other 45% of the costs. Total OpEx sum up to 2,189,000 EUR per year with the highest shares of 32% for heating costs, 22% for personnel expenses, 14% for process water, and 12% for carbon dioxide transport, storage, and monitoring of the CS plant. The levelized costs of carbon-negative hydrogen are estimated at 0.013 EUR/kWhH₂ (0.43 EUR/kgH₂). They represent the costs incurred to produce hydrogen with negative emissions in a climate-neutralized way. The comparison of the LCCNH with the LCOE including GHG neutralization costs of alternative hydrogen production technologies, shows that the example project can provide hydrogen with the lowest costs under the assumptions made. The value for grey hydrogen from natural gas is estimated at 0.12 EUR/kWhH₂, for instance. Also, in comparison to green hydrogen from electrolysis using renewable electricity (0.18 EUR/kWhH₂), the positive climatic impact of the negative emissions provides a considerable advantage. Even without the assumed internalization of GHG impacts, the calculations show an economic advantage of the example project (0.12 EUR/kWhH₂) over green hydrogen (0.17 EUR/kWhH₂) [42,44]. To make a comparison with other NETs, the GHG removal efficiency represented by the emission factor EF is considered first. The EF of the example project is estimated at 0.12 tCO₂eq/tCO₂stored. This value is at the lower end of the EF ranges both from other BECCS and DACCS approaches, as literature shows. Additionally, the levelized costs of stored carbon dioxide LCSC are obtained. They represent the costs that occur to permanently store one ton of carbon

dioxide with a carbon-neutralized process. The LCSC of the example project are estimated at about -260 EUR/tCO₂stored. To further compare the costs of the example project with other NETs, the levelized costs of negative emissions LCNE are examined. They represent the costs that occur to achieve one ton (CO2eq) of negative emissions in a carbon-neutralized way. The LCNE of the example project are estimated at -297 EUR/tCO₂removed. This means that an income of 279 EUR due to the sales of the co-product biohydrogen could be generated. For BECCS, the LCNE range is approx. 39-367 EUR/tCO2removed, for DACCS, 25–1127 EUR/tCO2removed. The internal LCOE of the example project is estimated at -322 EUR/ tCO2removed. This indicates that even without internalizing any externalities, the removal of one ton of carbon dioxide from the atmosphere would generate an income of 322 EUR through biohydrogen sales.

A sensitivity analysis is carried out in order to determine uncertainties and risks affecting economic and environmental performance. It is shown that the GHG internalization costs and revenues, H_2/CO_2 yield, and the total OpEx have the greatest impact on the LCCNH of the example project. Further, the hydrogen sales price has the greatest impact on the LCNE of the example project. A 50% reduction would lead to a cost increase of 91% to -25.89 EUR/tCO₂removed. Further, the $H_2/$ CO₂ yield and the total OpEx have a considerably higher impact on the LCNE than the total CapEx, GHG internalization costs and revenues, and leakages.

Finally, it has been shown that the theoretical fundamentals elaborated by Full et al. (2022) are applicable in practice. A comparability of the example project with other NETs and hydrogen production technologies has been achieved. It has also been shown that it is possible to specifically evaluate the economic influences of the technologies considered for society coming from GHG emissions or removals.

9. Outlook

Further investigations should be conducted based on the results. For example, CS technologies and NETs are associated with different time horizons of storage and risks of reversal [23,47]. In the present evaluations, these two factors are not considered. However, to raise the comparability of different NETs and CS technologies, they have to be taken into account. Additionally, the ability of HyBECCS to deliver negative emissions and its potential economic competitiveness indicated by the results of the analysis in this paper are based on assumptions and are thus insecure. Further investigations on the potential of HyBECCS on a technology-specific, national and global level are necessary. Therefore, different biohydrogen production technologies should be subjected to the method and extended to HyBECCS facilities. Suitable options could include, for example, dark fermentation [54-57], biomass gasification [58-61] or biogas steam reforming [43,62,63].

Author contributions

Conceptualization, methodology, formal analysis: J.F.; writing (original draft), investigation, data curation, visualization: J.F.,

M.G., S.Z. and T.S.; writing (review and editing): R.M. and A.S.; funding acquisition: J.F., R.M. and A.S. All authors have read and agreed to the published version of the manuscript.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgments

The research shown in this paper was conducted within the projects "RhoTech", "SmartBioH2" and "H2Wood". The authors gratefully acknowledge the financial support of the German Federal Ministry for Economic Affairs and Energy, the Ministry of the Environment, Climate Protection and the Energy Sector Baden-Wuerttemberg, Germany, the European Regional Development Fund (ERDF) and the German Federal Ministry of Education and Research.

Appendix A. Supplementary data

Supplementary data to this article can be found online at https://doi.org/10.1016/j.ijhydene.2023.09.252.

REFERENCES

- Fachagentur Nachwachsende Rohstoffe e.V. (FNR). Basisdaten bioenergie deutschland 2022. Jul. 2022 [Online]. Available: https://mediathek.fnr.de/basisdaten-bioenergie. html. [Accessed 18 November 2022].
- [2] IPCC. In: Masson-Delmotte V, Zhai P, Pirani A, Connors SL, Péan C, Berger S, Caud N, Chen Y, Goldfarb L, Gomis MI, Huang M, Leitzell K, Lonnoy E, Matthews JBR, Maycock TK, Waterfield T, Yelekci O, Yu R, Zhou B, editors. Climate change 2021: the physical science basis. Contribution of working group I to the sixth assessment report of the intergovernmental panel on climate change. Cambridge University Press; 2021 [Online]. Available: https://www.ipcc. ch/report/ar6/wg1/downloads/report/IPCC_AR6_WGI_Full_ Report.pdf.
- [3] Azar C, et al. The feasibility of low CO2 concentration targets and the role of bio-energy with carbon capture and storage (BECCS). Climatic Change 2010;100(1):195–202. https:// doi.org/10.1007/s10584-010-9832-7.
- [4] Gough C, Upham P. Biomass energy with carbon capture and storage (BECCS or Bio-CCS). Greenhouse Gas Sci Technol 2011;1(4):324-34. https://doi.org/10.1002/ghg.34.
- [5] Kemper J. Biomass and carbon dioxide capture and storage: a review. Int J Greenh Gas Control 2015;40:401–30. https:// doi.org/10.1016/j.ijggc.2015.06.012.
- [6] IEA. Combining bioenergy with CCS: reporting and accounting for negative emissions under UNFCCC and the kyoto protocol. IEA Energy Papers 2011;16. https://doi.org/ 10.1787/5k9h0wd6407k-en.
- [7] Kraxner F, Nilsson S, Obersteiner M. Negative emissions from BioEnergy use, carbon capture and sequestration (BECS)—the case of biomass production by sustainable forest management from semi-natural temperate forests. Biomass

Bioenergy 2003;24(4-5):285-96. https://doi.org/10.1016/ S0961-9534(02)00172-1.

- [8] Fajardy M, Mac Dowell N. Can BECCS deliver sustainable and resource efficient negative emissions? Energy Environ Sci 2017;10(6):1389–426. https://doi.org/10.1039/C7EE00465F.
- [9] Rosa L, Sanchez DL, Mazzotti M. Assessment of carbon dioxide removal potential via BECCS in a carbon-neutral Europe. Energy Environ Sci 2021;14(5):3086–97. https:// doi.org/10.1039/D1EE00642H.
- [10] Borchers M, et al. Scoping carbon dioxide removal options for Germany–What is their potential contribution to Net-Zero CO2? Front. Clim. 2022;4. https://doi.org/10.3389/ fclim.2022.810343. Art. no. 810343.
- [11] Federal Office of Justice. Federal climate change Act of 12 december 2019 (federal law gazette I, p. 2513), as last amended by article 1 of the Act of 18 august 2021 (federal law gazette I, p. 3905). https://www.gesetze-im-internet.de/englisch_ksg/ englisch_ksg.html. [Accessed 25 November 2022].
- [12] Full J, Trauner M, Miehe R, Sauer A. Carbon-negative hydrogen production (HyBECCS) from organic waste materials in Germany: how to estimate bioenergy and greenhouse gas mitigation potential. Energies 2021;14(22):7741. https://doi.org/10.3390/en14227741. 2021.
- [13] Full J, Merseburg S, Miehe R, Sauer A. A new perspective for climate change mitigation—introducing carbon-negative hydrogen production from biomass with carbon capture and storage (HyBECCS). Sustainability 2021;13(7):4026. https:// doi.org/10.3390/su13074026.
- [14] Full J, Baumgarten Y, Dokur Y, Miehe R, Sauer A. Biogas plants as hydrogen production facilities and greenhouse gas sinks: technology comparison, challenges and potentials for carbon negative hydrogen production (HyBECCS). Procedia CIRP 2022;107:185–90. https://doi.org/10.1016/ j.procir.2022.04.031.
- [15] Full J, Ziehn S, Geller M, Miehe R, Sauer A. Carbon-negative hydrogen production: fundamentals for a techno-economic and environmental assessment of HyBECCS approaches. GCB Bioenergy 2022;14(5):597–619. https://doi.org/10.1111/ gcbb.12932.
- [16] Autenrieth C, Shaw S, Ghosh R. New approach for the construction and calibration of gas-tight setups for biohydrogen production at the small laboratory scale. Metabolites 2021;11(667). https://doi.org/10.3390/ metabo11100667.
- [17] Ghosh R, Hardmeyer A, Thoenen I, Bachofen R. Optimization of Sistrom medium for the semi-aerobic growth of Rhodospirillum rubrum with the maximal yield of photosynthetic membranes. Appl Environ Microbiol 1994;60(5):1698–700. https://doi.org/10.1128/aem.60.5.1698-1700.1994.
- [18] Grammel H, Gilles E-D, Ghosh R. Microaerophilic cooperation of reductive and oxidative pathways allows maximal photosynthetic membrane biosynthesis in Rhodospirillum rubrum. Appl Environ Microbiol 2003;69(11):6577–86. https:// doi.org/10.1128/AEM.69.11.6577–6586.2003.
- [19] Grammel H, Ghosh R. Redox state dynamics of ubiquinone-10 imply cooperative regulation of photosynthetic membrane expression in Rhodospirillum rubrum. J Bacteriol 2008;190(14):4912–21. https://doi.org/10.1128/JB.00423-08.
- [20] Bui M, et al. Carbon capture and storage (CCS): the way forward. Energy Environ Sci 2018;11(5):1062–176. https:// doi.org/10.1039/C7EE02342A.
- [21] Otto A, Markewitz P, Robinius M. Technologiebericht 2.4 CO2-Nutzung: technologien für die Energiewende. Teilbericht 2 an das Bundesministerium für Wirtschaft und Energie (BMWi). Wuppertal Institut, ISI, IZES; 2017.
- [22] IPCC. IPCC special report on carbon dioxide capture and storage. In: Metz B, Davidson O, de Coninck HC, Loos M,

Meyer LA, editors. Prepared by working group III of the intergovernmental panel on climate change. Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press; 2005. ISBN 13 978-0-521-86643-9.

- [23] Poralla M, Honegger M, Ahonen H-M, Michaelowa A, Weber A-K. Sewage Treatment for the Skies.: mobilising carbon dioxide removal through public policies and private financing. London, UK and Freiburg i.B. Germany: NET-Rapido Consortium and Perspectives Climate Research; 2021.
- [24] Gassnova. Report: full chain CO2 footprint. Nov. 2019 [Online]. Available: https://ccsnorway.com/carbon-footprint/. [Accessed 18 November 2022] (accessed Nov. 4, 2022).
- [25] Kearns D, Liu H, Consoli C. Technology readiness and costs for CCS. Global CCS Institute; Mar. 2021 [Online]. Available: https://www.globalccsinstitute.com/resources/publicationsreports-research/technology-readiness-and-costs-of-ccs/. [Accessed 18 November 2022].
- [26] Zero emissions platform, "the costs of CO2 storage: postdemonstration CCS in the EU," IEAGHG. Accessed: November.
 18, 2022. [Online]. Available: https://zeroemissionsplatform. eu/document/the-costs-of-co2-storage/.
- [27] Fachagentur Nachwachsende Rohstoffe e.V. (FNR), "Leitfaden Biogas - von der Gewinnung zur Nutzung," Accessed: November. 18, 2022. [Online]. Available: https:// mediathek.fnr.de/leitfaden-biogas.html.
- [28] Thrän D, Pfeiffer D. Leipzig. In: Methodenhandbuch: stoffstromorientierte Bilanzierung der Klimagaseffekte.: methoden zur Bestimmung von Technologiekennwerten, Gestehungskosten und Klimagaseffekten von Vorhaben im Rahmen des BMWi-Forschungsnetzwerkes Bioenergie/ BMWi-Förderbereich "Energetische Biomassenutzung. 4th ed. 2013. ISSN 2192-1806.
- [29] Han W, Liu Z, Fang J, Huang J, Zhao H, Li Y. Techno-economic analysis of dark fermentative hydrogen production from molasses in a continuous mixed immobilized sludge reactor. J Clean Prod 2016;127:567–72. https://doi.org/10.1016/ j.jclepro.2016.04.055.
- [30] Han W, Yan Y, Gu J, Shi Y, Tang J, Li Y. Techno-economic analysis of a novel bioprocess combining solid state fermentation and dark fermentation for H2 production from food waste. Int J Hydrogen Energy 2016;41(48):22619–25. https://doi.org/10.1016/j.ijhydene.2016.09.047.
- [31] Fischedick M, Görner K, Thomeczek M. CO2: abtrennung, Speicherung, Nutzung: ganzheitliche Bewertung im Bereich von Energiewirtschaft und Industrie. Berlin: Springer Vieweg; 2015. ISBN 978-3-642-19528-0.
- [32] George N. Nikolaidis, eustathios S. Kikkinides*, and michael C. Georgiadis*, "model-based approach for the evaluation of materials and processes for post-combustion carbon dioxide capture from flue gas by PSA/VSA processes,". Ind Eng Chem Res 2016;55. Art. no. 3.
- [33] Inaco Services GmbH, "Testbericht apfeltrester," Accessed: November. 18, 2022. [Online]. Available: https://inaco.ch/ testberichte-analysen/testbericht-apfeltrester (accessed Nov. 4, 2022).
- [34] Statista. Industriestrompreise (inklusive stromsteuer) in deutschland in den jahren 1998 bis 2022. https://de.statista. com/statistik/daten/studie/252029/umfrage/ industriestrompreise-inkl-stromsteuer-in-deutschland/. [Accessed 31 October 2022].
- [35] Heizspiegel. Durchschnittliche Heizkosten 2021 für ein Einfamilienhaus. https://www.heizspiegel.de/heizkostenpruefen/heizkosten-pro-m2-vergleich/. [Accessed 14 October 2022].
- [36] Pelda J, Holler S, Persson U. District heating atlas analysis of the German district heating sector. Energy 2021;233:121018. https://doi.org/10.1016/j.energy.2021.121018.

- [37] IEA. Energy technology perspectives: special report on carbon capture utilisation and storage - CCUS in clean energy transitions. Sep. 2020 [Online]. Available: https:// www.iea.org/reports/ccus-in-clean-energy-transitions. [Accessed 15 August 2022] (accessed Nov. 4, 2022).
- [38] EnBW Energie Baden-Württemberg AG. 5 Fragen zur Wasserstoffherstellung. https://www.enbw.com/ unternehmen/eco-journal/5-fragen-zurwasserstoffherstellung.html. [Accessed 4 November 2022].
- [39] Bünger B, Matthey A. Methodenkonvention 3.1 zur Ermittlung von Umweltkosten: kostensätze. Umweltbundesamt. Dessau-Roßlau; 2020. ISBN 1862-4804.
- [40] DIN e.V.. DIN EN ISO 14040: umweltmanagement Ökobilanz
 grundsätze und Rahmenbedingungen (ISO 14040:2006 + Amd. 1:2020); Deutsche Fassung EN ISO 14040:2006 + A1:2020. 2021.
- [41] DIN e.V.. DIN EN ISO 14044: umweltmanagement Ökobilanz
 anforderungen und Anleitungen (ISO 14044:2006 + Amd 1:2017 + Amd. 2:2020); Deutsche Fassung EN ISO 14044:2006
 + A1:2018 + A2:2020. 2021.
- [42] Antonini C, Treyer K, Streb A, van der Spek M, Bauer C, Marco Mazzotti M. Hydrogen production from natural gas and biomethane with carbon capture and storage – a technoenvironmental analysis. Sustain Energy Fuels 2020;6/2020. https://doi.org/10.1039/D0SE00222D.
- [43] Hinicio FB. CertifHy developing a European guarantee of origin scheme for green hydrogen. Oct. 2016 [Online].
 Available: https://fsr.eui.eu/wp-content/uploads/CertifHy_Presentation_19_10_2016_final_Definition_of_Premium_Hydrogen.pdf. [Accessed 18 November 2022] (accessed Nov. 4, 2022).
- [44] Fajardy M, Koeberle A, MacDowell N, Fantuzzi A. BECCS deployment: a reality check," London. Jan. 2019 [Online]. Available: https://www.imperial.ac.uk/media/imperialcollege/grantham-institute/public/publications/briefingpapers/BECCS-deployment—a-reality-check.pdf. [Accessed 18 November 2022] (accessed Nov. 4, 2022).
- [45] Terlouw T, Treyer K, Bauer C, Mazzotti M. Life cycle assessment of direct air carbon capture and storage with low-carbon energy sources, vol. 55. Environmental science & technology; 2021. p. 11397–411. https://doi.org/10.1021/ acs.est.1c03263.
- [46] Möllersten K, Yan J, Naqvi R. Qualitative assessment of classes of negative emission technologies," the 12th international conference on applied energy. Dec. 2020 [Online]. Available: https://www.researchgate.net/ publication/347309043_Qualitative_comparative_ assessment_of_Negative_Emission_Technologies. [Accessed 25 November 2022] (accessed Nov. 4, 2022).
- [47] Thrän D, Pfeiffer D. Methodenhandbuch: stoffstromorientierte Bilanzierung der Klimagaseffekte.: methoden zur Bestimmung von Technologiekennwerten, Gestehungskosten und Klimagaseffekten von Vorhaben im Rahmen des BMWi-Forschungsnetzwerkes Bioenergie/ BMWi-Förderbereich "Energetische Biomassenutzung". Leipzig. 5th ed. 2021. ISBN 978-3-946629-58-0.
- [48] Ueli W. Silierung von Apfel- und Birnentrester. Agrarforschung 2003;10(3):104–9 [Online]. Available: https:// www.agrarforschungschweiz.ch/2003/03/silierung-vonapfel-und-birnentrester/. [Accessed 4 November 2022].
- [49] Raj T, et al. Critical challenges and technological breakthroughs in food waste hydrolysis and detoxification for fuels and chemicals production. Bioresour Technol

2022;360:127512. https://doi.org/10.1016/ j.biortech.2022.127512.

- [50] Karapinar Kapdan I, Kargi F. Bio-hydrogen production from waste materials. Enzym Microb Technol 2006;38(Issue 5). https://doi.org/10.1016/j.enzmictec.2005.09.015.
- [51] Sivagurunathan P, et al. A critical review on issues and overcoming strategies for the enhancement of dark fermentative hydrogen production in continuous systems. Int J Hydrogen Energy 2016;41(Issue 6). https://doi.org/ 10.1016/j.ijhydene.2015.12.081.
- [52] Gorgec Firuze Karaosmanoglu, Karapinar Ilgi. Biohydrogen production from hydrolyzed waste wheat by dark fermentation in a continuously operated packed bed reactor: the effect of hydraulic retention time. Int J Hydrogen Energy 2019;44(1):136–43. https://doi.org/10.1016/ j.ijhydene.2018.08.155.
- [53] Ilgi Karapinar, Onur Balcan. Biohydrogen production from acid hydrolyzed wastewater treatment sludge by dark fermentation. Int J Hydrogen Energy 2020;45(5):3499–508. https://doi.org/10.1016/j.ijhydene.2019.03.230.
- [54] Guo Xin Mei, et al. Hydrogen production from agricultural waste by dark fermentation: a review. Int J Hydrogen Energy 2010;35(19). https://doi.org/10.1016/j.ijhydene.2010.03.008.
- [55] Krupp M, Widmann R. Biohydrogen production by dark fermentation: experiences of continuous operation in large lab scale. Int J Hydrogen Energy 2009;34(10):4509–16. https:// doi.org/10.1016/j.ijhydene.2008.10.043.
- [56] Karapinar Ilgi, et al. The effect of hydraulic retention time on thermophilic dark fermentative biohydrogen production in the continuously operated packed bed bioreactor. Int J Hydrogen Energy 2020;45(5):3524–31.
- [57] Cohce MK, Dincer I, Rosen MA. Thermodynamic analysis of hydrogen production from biomass gasification. Int J Hydrogen Energy 2010;35(Issue 10). https://doi.org/10.1016/ j.ijhydene.2009.08.066.
- [58] El-Emam Rami, Dincer Ibrahim, El-Emam, Salah. Gasification of biomass for hydrogen and power production: efficiency and environmental assessment." progress in clean energy, vol. II: novel systems and applications. Basel, Switzerland: Publisher: Springer International Publishing; 2015. https:// doi.org/10.1007/978-3-319-17031-2_12.
- [59] Kubilay Karayel G, Javani Nader, Dincer Ibrahim. Utilization of biomass and waste resources for renewable hydrogen production: a comprehensive study. Energy Sources, Part A Recovery, Util Environ Eff 2023;45(3):8553–67. https://doi.org/ 10.1080/15567036.2023.2225444 [Titel anhand dieser DOI in Citavi-Projekt übernehmen].
- [60] Tezer Özgün, et al. Biomass gasification for sustainable energy production: a review. Int J Hydrogen Energy 2022;47(34):15419–33.
- [61] Full Johannes, et al. Perspectives of biogas plants as BECCS facilities: a comparative analysis of biomethane vs. biohydrogen production with carbon capture and storage or use (CCS/CCU). Energies 2023;16(13):5066.
- [62] Park Min-Ju, et al. System optimization for effective hydrogen production via anaerobic digestion and biogas steam reforming. Int J Hydrogen Energy 2020;45(55):30188–200.
- [63] IEA and Nuclear Energy Agency. Projected costs of generating electricity: 2020 edition. OECD; 2020 [Online]. Available: https://www.oecd.org/iea/projected-costs-of-generatingelectricity-20798393.htm. [Accessed 20 November 2022] (accessed Nov. 4, 2022).

